University of Southern Queensland



Identifying strategies to improve the water productivity of permanent raised beds

A dissertation submitted by

Ghani Akbar

FOR THE AWARD OF DOCTOR OF PHILOSOPHY

2013

ABSTRACT

Permanent raised bed (PRB) farming systems combine several elements (e.g. reduced tillage, controlled traffic, stubble retention) of conservation agriculture. PRB systems have recently been introduced into irrigated areas but there is a lack of information on the agronomic and irrigation performance of these systems under these conditions. Three years of field research into the performance of irrigated PRBs was conducted in south east Queensland, Australia and north west Pakistan. These studies benchmarked the irrigation performance of variously sized PRB systems and explored the impact of bed renovation method (no tillage (NT), shallow cultivation (SC) and blade ploughing (BP)) on soil hydro-physical properties (bulk density, soil moisture storage, infiltration), irrigation performance, crop yield and input water productivity (*WP*). Irrigation management strategies to improve lateral infiltration and irrigation performance were also investigated.

The initial benchmarking study was conducted on two farms with clayey Vertisol (Australia) and three farms with sandy clay loam, Alfisol, (Pakistan) soils. The results showed low irrigation performance with excessive deep drainage potential. The irrigation application efficiency (*Ea*) on the clayey Vertisol and sandy clay loam was as low as 68% and 50%, requirement efficiency (*Er*) 96% and 77% and distribution uniformity (*DU*) 86% and 66%, respectively. However, the majority of the narrow beds (66 cm furrow spacing) were over-irrigated on the sandy clay loam while the wide beds (132 cm furrow spacing) were under-irrigated. Inappropriate renovation and bed furrow dimensions, sub-optimal irrigation management and poor lateral infiltration were the main factors likely to affect *WP* of irrigated PRBs.

The evaluation of bed renovation methods found that BP on the clayey Vertisol reduced (~6%) the average seasonal bulk density of the surface 0-30 cm compared to the NT treatment. BP was found to produce higher lateral infiltration into the beds than either NT or SC. Freshly applied BP and SC produced a higher (~23%) cumulative infiltration than NT but the effect was transient and was not significant in the following season when the PRB renovation treatments were not freshly applied. The SC beds slump more than the BP and NT beds. The soil water content in the beds indicated that there was increased water storage associated with BP.

PRB renovation (i.e. SC or BP) was found to reduce irrigation performance on the clayey Vertisol when the irrigation was farmer managed. The volume of irrigation water applied to the fresh SC and BP treatments increased by up 13% and 55%, and *Ea* was reduced by up to 9% and 29% respectively, compared with NT during the 2010 wheat and 2011 corn seasons. NT produced a higher wheat yield and *WP*, but the lower wheat yield in the cultivated treatments was associated with poor crop establishment from a rough soil surface and inadequate seeder performance in this season. In the subsequent seasons, there was no significant difference in either the crop yields or *WP* between the bed renovation treatments.

Field trials investigating the effect of bed renovation on infiltration were conducted on both soils. Significant differences in the amount of water infiltrated and stored in the bed shoulder and bed middle were found for periods of wetting consistent with normal irrigation practices. This suggests that the general assumption of uniform soil moisture distribution across the beds is not valid for wide PRB systems. The three different renovation methods also significantly affected lateral infiltration, suggesting tillage may be used as a management tool to improve bed wetting in low infiltration soils and subsided beds.

Infiltration and soil-water movement were simulated using Hydrus 2D. The simulations were found to be well correlated with the measured field data and an evaluation for the three PRB renovation methods confirmed that NT had the slowest, and BP the fastest, potential to wet the bed middle of the wide PRBs. Increasing the furrow water head from 4 cm to furrow full of water was also shown to reduce wetting time by more than 30%.

Lateral infiltration was poor in the sandy clay loam. The shortest wetting time (~15 hrs) required to wet the bed middle of a 132 cm wide bed to field capacity occurred when furrow full water head was applied to the BP treated bed. However, even this period of wetting may be difficult to achieve using current irrigation practices in Pakistan. The graphical model outputs developed can be used in developing guidelines to optimise bed width and irrigation management to ensure adequate lateral infiltration with different PRB renovation methods.

Strategies to optimise irrigation management and field design while ensuring adequate lateral infiltration into the bed middle were evaluated using the surface irrigation model SIRMOD. Optimising the inflow rate (Q) and the time to cut-off (Tco) for the particular field length was found to produce up to a 38% irrigation water saving compared with existing practices. However, the majority of the wide beds on the sandy clay loam required higher inflow volumes than currently applied, which increased Er but reduced Ea. Optimising furrow length indicated a further improvement in irrigation performance. Similarly, Q and Tco optimisation together showed up to 35% irrigation water saving and up to 33% improved Ea for the most sub-optimally managed BP treatment. Decreased furrow length improved irrigation performance for freshly renovated PRB, while irrigation performance of settled bed furrows tended to increase with increased furrow length. Relating optimum Tco to water advance to furrow tail end (Ta) was sensitive to furrow length, Q and soil infiltration functions. The decision support guidelines developed were helpful in improving existing irrigation performance under both soil conditions.

This research has shown that the existing irrigation performance of PRBs is often low and highly variable. A key constraint is the potential for poor lateral infiltration into the beds and inadequate wetting to the centre of wide beds. However, this work has also highlighted that the adoption of appropriate bed renovation methods (particularly BP) and irrigation management practices (e.g. Q and Tco) can substantially improve irrigation performance and input WP. Although the applicability of the specific decision support tools developed is restricted to the soil types and field conditions encountered, the general understanding and insight into the basic principles of performance optimisation and for agronomic and irrigation interactions is expected to be beneficial in refining understanding and the promotion of sustainable crop production under a wider range of environmental conditions.

CERTIFICATION OF DISSERTATION

I certify that the ideas, designs, experimental work, software, results, analyses and conclusions presented in this dissertation are entirely my own effort, except where otherwise indicated and acknowledged.

I further certify that the work is original and has not been previously submitted for assessment in any other course or institution, except where specifically stated.

Date

Endorsement:

Professor Steven Raine (Principal supervisor)

Date

Dr Allen David McHugh (Associate supervisor)

Date

ACKNOWLEDGEMENTS

I would like to express my sincere gratitude and appreciations to a number of people who have helped me during this research. I am grateful to Professor Steven Raine for his timely supervision, guidance, academic, technical and personal support. I greatly admire his capabilities of quick assessment, which greatly benefited me in solving multiple technical, writing and field related issues. I particularly acknowledge the major assistance provided to me by Dr Allen (Jack) McHugh in technical support and in improving my presentation and writing capabilities. Never once did he turn me away in an unscheduled meeting when I needed help, and I know this was at the expense of his own workload. I am extremely thankful to Mr Greg Hamilton, who not only supported me in getting the PhD fellowship but continued to guide me throughout the duration of this work. I cannot forget his personal contribution and support in my career development.

I acknowledge the support of the Australian Centre for International Agriculture Research (ACIAR) for providing me a fellowship, the National Centre for Engineering and Agriculture (NCEA) at the University of Southern Queensland (USQ), and its friendly staff for providing me the necessary equipment, field conveyance and technical support. I acknowledge the Pakistan Agricultural Research Council (PARC) and my colleagues for their cooperation and all possible support.

I greatly appreciate my collaborative farmers, especially Mr David House (Cambooya, Australia) and Mr Feroz Shah (Mardan, north west Pakistan) for providing me field access and support in establishing the field trials. Many thanks to Dr Jochen Eberhard, Dr Malcolm Gillies, Dr Simon White, and Professor Shah Jahan Khan for their guidance, discussion, technical and statistical consultation support.

I wish to express my humble obligation to my mother, wife, two sons and one daughter for their love, prayers, support, motivation and patience. The physical support in crop sample handling, moral support and encouragement from my wife during the hard days of lengthy field work was always crucial in motivating me to accomplish this research study.

PUBLICATIONS ARISING FROM THIS RESEARCH

- Akbar, G, Hamilton, GJ & Raine SR 2010, 'Permanent raised bed configurations and renovation methods affect crop performance', In: *19th World Congress on Soil Science*, August 2010, Brisbane Australia.
- Akbar, G, McHugh, AD, Raine, SR & Hamilton GJ 2011, 'A preliminary evaluation of furrow inflow rate and cut-off time on the performance of smallholder raised bed farming systems', In: World Congress on Conservation Agriculture, September 2011, Brisbane Australia.
- Akbar, G, Raine, SR, McHugh, AD & Hamilton GJ 2011, 'A preliminary evaluation of irrigation performance and in season changes under permanent raised beds on Vertisol in Queensland, Australia', In: World Congress on Conservation Agriculture, September 2011, Brisbane Australia.
- Akbar, G, Raine, SR, McHugh, AD & Hamilton, GJ 2012, 'A soil management technique to enhance lateral infiltration (subbing) in permanent raised bed', In: *ICID Irrigation Australia Conference and Exhibition*, Adelaide Convention Centre, June 25-29, 2012, Adelaide Australia.
- Akbar, G, Raine, SR, McHugh, AD & Hamilton, GJ 2012, 'Impact of renovating permanent raised beds on water productivity under Vertisol', In: *ICID Irrigation Australia Conference and Exhibition*, Adelaide Convention Centre, June 25-29, 2012, Adelaide Australia.
- The following journal papers are in the final stages of review between the authors and are expected to be submitted in the near future:
- Akbar, G, Raine, S, McHugh, AD & Hamilton, GJ 'Strategies to improve the irrigation performance of raised beds on small farms in north west Pakistan', will be submitted soon to *Water Resources Management* Journal (Springer).
- Akbar, G, Raine, S, McHugh, AD & Hamilton GJ 'Impact on irrigation performance and productivity of permanent raised bed renovation methods on a Vertisol', will be submitted soon to *Crop and Pasture Science* Journal (CSIRO).
- Akbar, G, Raine, S, McHugh, AD & Hamilton GJ "Modelling lateral infiltration under different renovation methods of permanent raised beds and furrow water heads in a Vertisol and a sandy clay loam', will be submitted to *Irrigation Science* (Springer).

TABLE OF CONTENTS

ABSTRACT	••••••	i
CERTIFICAT	ION OF DISSERTATION	iv
ACKNOWLEI	DGEMENTS	v
PUBLICATIO	NS ARISING FROM THIS RESEARCH	vi
TABLE OF CO)NTENTS	vii
LIST OF FIGU	URES	xii
LIST OF TABI	LES	XX
LIST OF ABBI	REVIATIONS	xxiii
LIST OF SYM	BOLS	xxiv
CHAPTER 1:	Introduction	1
1.1 Agricu	iltural sustainability and food security	2
1.2 water	productivity	
1.2.1 Agr	onomic management	4
1.2.2 Irrig	gation management	5
1.3 Resear	rch opportunities	6
1.4 Resear	rch question and aim	6
1.5 Hypot	neses	····· / 7
1.0 Specif	ure of this dissertation	····· / 8
CHAPTER 2:	Review of permanent raised bed farming system perfor	mance 10
2.1 Introd	uction	10
2.2 Need	to improve water productivity of raised beds	10
2.2.1 Ext	ent of agronomic and irrigation management issues	12
2.2.1.1	Agronomic and irrigation issues in an Australian context	12
2.2.1.2	Agronomic and irrigation issues in a Pakistani context	14
2.3 Factor	s affecting water productivity of raised beds	16
2.3.1 Agr	ronomic management factors	
2.3.1.1	Machinery traffic and tillage	17
2.3.1.2	Bed furrow configurations	
2.3.1.3	Cropping management	
2.3.1.4	waterlogging and satinity	
2.3.2 Irrig	Jation management factors	
2.3.2.1	Time to cut-off	23 21
2.3.2.2	Application depth and infiltration opportunity time	24 25
2.3.2.4	Field design	
2.3.2.5	Irrigation scheduling	
2.4 Adapt	ation of conservation agriculture to raised beds	
2.4.1 Per	manent raised bed farming systems	29
2.1.1 101	Davian of norman and naised hode in Anatualia	·····2) 20
2.4.1.1	<i>Keview of permanent raisea beas in Australia</i>	

2.4.	1.2 Review of permanent raised beds in Pakistan	32
2.4.	1.3 Summary of research needs in both countries	33
2.5 P	ermanent raised beds performance and assessment	33
2.5.1	Agronomic management performance and assessment	34
2.5.	1.1 Soil texture and structure	34
2.5.	1.2 Soil moisture	34
2.5.	1.3 Bulk density	36
2.5.	1.4 Hydraulic conductivity	37
2.5.	1.5 Lateral infiltration into bed	38
2.5.	1.6 Surface drainage and stability during wetness	40
2.5.	1.7 Seasonal changes	41
2.5.	1.8 Crop yield	42
2.5.2	Irrigation management performance and assessment	43
2.5.2	2.1 Models to assess irrigation performance	43
2.6 C	onclusions	50
		F1
	3: General material and methods	51
3.1 II	uroducuon	51
3.2 K	escarintian of field sites	51
3.3 D	escription of field sneds	52
3.4 0	eneral field conditions encountered in Australia	55
3.4.1	Site Al	53
3.4.2	Site A2	57
3.5 G	eneral field conditions encountered in Pakistan	58
3.6 S	oil and irrigation measurement technique	61
3.6.1	Soil moisture, bulk density and soil moisture deficit	63
3.6.2	Lateral infiltration	63
3.6.3	Soil infiltration and irrigation performance	65
3.6.	3.1 IRRIMATE TM equipment for irrigation performance evaluation	66
3.7 St	tatistical methods	66
CHAPIER	4: Benchmarking irrigation performance of permanent raised	 67
4.1 Ir	itroduction	67
4.1 M	laterial and methods	67
1.2 1.		60
4 2 1	1000000000000000000000000000000000000	09
4.2.1	Data collection and instruments used	60
4.2.1 4.2.	<i>L Australia</i>	69 70
4.2.1 4.2. 4.2. 4.3 R	Data collection and instruments used 1.1 Australia 1.2 Pakistan esults	69 70 71
4.2.1 4.2. 4.3 R 4.3 1	Data collection and instruments used 1.1 Australia 1.2 Pakistan esults Eield conditions and soil moisture deficit before irrigation	69 70 71 71
4.2.1 4.2. 4.3 R 4.3.1 4.3.2	Data collection and instruments used 1.1 Australia 1.2 Pakistan esults Field conditions and soil moisture deficit before irrigation Bod furrow dimensions	 69 70 71 71 71 72
4.2.1 4.2. 4.3 R 4.3.1 4.3.2 4.2.2	Data collection and instruments used 1.1 Australia 1.2 Pakistan esults	 69 70 71 71 72 72
4.2.1 4.2. 4.3 R 4.3.1 4.3.2 4.3.3	Data collection and instruments used 1.1 Australia 1.2 Pakistan esults Field conditions and soil moisture deficit before irrigation Bed furrow dimensions Inflow, water advance and time to cut-off.	 69 70 71 71 72 73 75
4.2.1 4.2. 4.3 4.3 4.3.1 4.3.2 4.3.3 4.3.4	Data collection and instruments used 1.1 Australia 1.2 Pakistan esults	 69 70 71 71 72 73 77
4.2.1 4.2. 4.3 R 4.3.1 4.3.2 4.3.3 4.3.4 4.3.5	Data collection and instruments used 1.1 Australia 1.2 Pakistan esults	 69 70 71 71 72 73 77 80
4.2.1 4.2. 4.3 R 4.3.1 4.3.2 4.3.3 4.3.4 4.3.5 4.4 D	Data collection and instruments used 1.1 Australia 1.2 Pakistan esults	 69 70 71 71 72 73 77 80 82
4.2.1 4.2. 4.3 R 4.3.1 4.3.2 4.3.3 4.3.4 4.3.5 4.4 D 4.4.1	Data collection and instruments used 1.1 Australia 1.2 Pakistan esults	 69 70 71 71 72 73 77 80 82 82

4.4.3	Sites and irrigation performance interactions	. 85
4.4.4	Irrigation management and performance interactions	. 85
4.4.5	Performance improvement options of existing permanent raised beds	. 87
4.5 C	Conclusions	. 88
СНАРТЕК	8.5: Impact of PRB renovation on irrigation performance and	
	productivity	. 89
5.1 II	ntroduction	. 89
5.2 N	Iaterial and methods	. 89
5.2.1	Site description, treatments and experiment layout	. 89
5.2.2	Agronomic management practices	. 92
5.2.3	Sampling and instrumentation	. 93
5.2.4	Irrigation performance and water productivity	.95
5.3 R	esults	.97
5.3.1	Weather conditions	.97
5.3.2	Observation of machinery operations	.97
5.3.3	Effect of PRB renovation method on bed and soil properties	.98
5.3.	3.1 Furrow almensions 3.2 Bulk density	.90 100
5.3.	3.3 Soil moisture deficit and infiltration properties	101
5.3.4	Water applied and profile water content	103
5.3.5	Effect of PRB renovation on irrigation characteristics	106
5.3.	5.1 Inflow	106
5.3.	5.2 Irrigation performance	106
5.3.6	Dry biomass yield	107
5.3.7	Water productivity	109
5.4 L		111
5.4.1	Effect of PRB renovation methods on soil physical properties	111
5.4.2	I reatment, irrigation management and performance interactions	114
5.4.3	Treatment and crop yield interactions	116
5.4.4	Treatment and water productivity interaction	117
5.4.5	PRB renovation options for improved performance on a Vertisol	118
5.5 C	-onclusions	119
СНАРТЕН	R 6: Field evaluation of lateral infiltration in permanent raised be	eds
(1 T	·····	120
6.1 II	Itroduction	120
621	Site description treatment and experimental layout	121
6.2.1 6.2.2	Bulk density and soil moisture measurements	121
622	Blocked furrow infiltrometer measurements	124
621	L steral infiltration and sorptivity	124
6.3 R	esults	125
6.3.1	Site A1-Vertisol	125
6.3.2	Site P4-Sandy Clay Loam	133

6.3.3 Field sorp	tivity	138
6.4 Discussion		139
6.4.1 Lateral inf	iltration on Vertisol	139
6.4.2 Lateral inf	iltration on sandy clay loam	141
6.5 Conclusions		143
CHAPTER 7: Strat	egies to improve lateral infiltration in permanent ra	ised
beds	using Hydrus 2D	144
7.1 Introduction		144
7.2 Material and	methods	144
7.3 Results		146
7.3.1 Hydrus ca	libration	146
7.3.2 Effect of r	enovation on infiltration with 9 cm furrow water head	146
7.3.2.1 Later	al infiltration	146
7.3.2.2 Verti	cal infiltration	147
7.3.3 Effect of f	urrow water head on soil moisture distribution	150
7.3.3.1 Later	al wetting on bed surface	150
7.3.3.2 Vertic	cal wetting below furrow centre	151
7.5.5.5 Cumi	<i>Mative inflitration</i>	152
7335 Deen	drainage losses	155
7.3.3.6 Soil n	noisture distribution for normal irrigation applications	156
7.3.4 Implicatio	ns of lateral infiltration for irrigation management	157
, ior implicatio	in of factorial minimum for mingunon management minimum	
7.4 Discussion		160
7.4 Discussion 7.4.1 Effect of r	enovation methods and furrow head on infiltration	160 160
7.4 Discussion 7.4.1 Effect of r 7.4.2 Implicatio	enovation methods and furrow head on infiltration	160 160 162
7.4 Discussion 7.4.1 Effect of r 7.4.2 Implicatio 7.5 Conclusions	enovation methods and furrow head on infiltration ns for irrigation management	160 160 162 163
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implicatio 7.5 Conclusions 	enovation methods and furrow head on infiltration ns for irrigation management	160 160 162 163
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implicatio 7.5 Conclusions CHAPTER 8: Mana 	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform	160 160 162 163 mance
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implicatio 7.5 Conclusions CHAPTER 8: Mana 8.1 Introduction 	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform	160 160 162 163 aance 164
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implicatio 7.5 Conclusions CHAPTER 8: Mana 8.1 Introduction 8.2 Material and	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform methods	160 160 162 163 nance 164 164 165
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implicatio 7.5 Conclusions CHAPTER 8: Mana 8.1 Introduction 8.2 Material and 8.2.1 Irrigation r	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform methods	160 160 162 163 nance nance 164 164 165 166
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implicatio 7.5 Conclusions CHAPTER 8: Mana 8.1 Introduction 8.2 Material and 8.2.1 Irrigation r 8.3 Results 	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform methods management and field design optimisation protocol	160 160 162 163 nance 164 164 165 166 167
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implicatio 7.5 Conclusions CHAPTER 8: Mana 8.1 Introduction 8.2 Material and 8.2.1 Irrigation r 8.3 Results 8.3.1 Optimising 	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform methods management and field design optimisation protocol g irrigation management of PRB systems in Australia	160 160 162 163 nance nance 164 164 165 166 167
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implicatio 7.5 Conclusions CHAPTER 8: Mana 8.1 Introduction 8.2 Material and 8.2.1 Irrigation r 8.3 Results 8.3.1 Optimising 8.3.1.1 Optimising 	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform methods management and field design optimisation protocol g irrigation management of PRB systems in Australia nising time to cut-off	160 160 163 nance 163 nance 164 164 165 166 167 167 168
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implication 7.5 Conclusions CHAPTER 8: Mana 8.1 Introduction 8.2 Material and 8.2.1 Irrigation r 8.3 Results 8.3.1 Optimising 8.3.1.2 Optimising 	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform methods management and field design optimisation protocol g irrigation management of PRB systems in Australia nising time to cut-off nising inflow rate and time to cut-off	160 160 163 nance nance 164 164 165 166 167 168 168
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implicatio 7.5 Conclusions CHAPTER 8: Mana 8.1 Introduction 8.2 Material and 8.2.1 Irrigation r 8.3 Results 8.3.1 Optimising 8.3.1.1 Optim 8.3.1.2 Optim 8.3.1.3 Optim	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform methods management and field design optimisation protocol g irrigation management of PRB systems in Australia nising time to cut-off nising inflow rate and time to cut-off	160 160 163 nance 163 nance 164 164 165 166 167 167 168 168 168 169
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implication 7.5 Conclusions CHAPTER 8: Manage 8.1 Introduction 8.2 Material and 8.2.1 Irrigation r 8.3 Results 8.3.1 Optimising 8.3.1.2 Optimising 8.3.1.3 Optimising 8.3.1.4 Optimising 	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform methods management and field design optimisation protocol g irrigation management of PRB systems in Australia nising time to cut-off nising inflow rate and time to cut-off nising furrow length nising time to cut-off relative to water advance time	160 160 163 nance 163 nance 164 164 165 166 167 168 168 168 169 170
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implicatio 7.5 Conclusions CHAPTER 8: Mana 8.1 Introduction 8.2 Material and 8.2.1 Irrigation r 8.3 Results 8.3.1 Optimising 8.3.1.1 Optim 8.3.1.2 Optim 8.3.1.3 Optim 8.3.1.4 Optim 	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform methods management and field design optimisation protocol g irrigation management of PRB systems in Australia nising time to cut-off nising inflow rate and time to cut-off nising furrow length nising time to cut-off relative to water advance time g irrigation management of PRB farming systems in Pal	160 160 163 ance 163 ance 164 164 165 166 167 167 168 168 168 169 170 kistan
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implication 7.5 Conclusions CHAPTER 8: Mana 8.1 Introduction 8.2 Material and 8.2.1 Irrigation r 8.3 Results 8.3.1 Optimising 8.3.1.2 Optimising 8.3.1.3 Optimising 8.3.1.4 Optimising 8.3.2 Optimising 	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform methods management and field design optimisation protocol g irrigation management of PRB systems in Australia nising time to cut-off nising inflow rate and time to cut-off nising furrow length nising time to cut-off relative to water advance time g irrigation management of PRB farming systems in Pal	160 160 163 mance 163 mance 164 164 164 165 166 167 168 168 168 169 170 kistan 171
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implicatio 7.5 Conclusions CHAPTER 8: Mana 8.1 Introduction 8.2 Material and 8.2.1 Irrigation r 8.3 Results 8.3.1 Optimising 8.3.1.1 Optim 8.3.1.2 Optim 8.3.1.3 Optim 8.3.1.4 Optim 8.3.2 Optimising 8.3.2.1 Optim 	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform methods management and field design optimisation protocol g irrigation management of PRB systems in Australia nising time to cut-off nising furrow length nising time to cut-off relative to water advance time g irrigation management of PRB farming systems in Pal nising time to cut-off	160 160 163 nance 163 nance 164 164 164 165 165 166 167 168 168 168 168 169 170 kistan 171 172 173
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implication 7.5 Conclusions CHAPTER 8: Mana 8.1 Introduction 8.2 Material and 8.2.1 Irrigation r 8.3 Results 8.3.1 Optimising 8.3.1.1 Optimising 8.3.1.2 Optimising 8.3.1.3 Optimising 8.3.1.4 Optimising 8.3.2.1 Optimising 8.3.2.1 Optimising 8.3.2.2 Optimising 8.3.2.3 Optimising 	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform methods management and field design optimisation protocol g irrigation management of PRB systems in Australia nising time to cut-off nising inflow rate and time to cut-off nising furrow length g irrigation management of PRB farming systems in Pa nising time to cut-off relative to water advance time g irrigation management of PRB farming systems in Pa nising time to cut-off	160 160 163 mance 163 mance 164 164 164 165 166 166 167 168 168 168 169 170 kistan 177 173 173 175
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implicatio 7.5 Conclusions CHAPTER 8: Mana 8.1 Introduction 8.2 Material and 8.2.1 Irrigation r 8.3 Results 8.3.1 Optimising 8.3.1.1 Optim 8.3.1.2 Optim 8.3.1.3 Optim 8.3.1.4 Optim 8.3.2.1 Optim 8.3.2.1 Optim 8.3.2.1 Optim 8.3.2.2 Optim 8.3.2.3 Optim 8.3.2.4 Optim 	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform methods management and field design optimisation protocol g irrigation management of PRB systems in Australia nising time to cut-off nising furrow length nising time to cut-off relative to water advance time g irrigation management of PRB farming systems in Pal mising time to cut-off nising time to cut-off	160 160 163 nance 163 nance 164 164 164 165 166 167 167 168 168 168 168 169 170 kistan 171 172 175 176
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implication 7.5 Conclusions CHAPTER 8: Mana 8.1 Introduction 8.2 Material and 8.2.1 Irrigation r 8.3 Results 8.3.1 Optimising 8.3.1.1 Optimising 8.3.1.2 Optimising 8.3.1.3 Optimising 8.3.2.1 Optimising 8.3.2.1 Optimising 8.3.2.2 Optimising 8.3.2.3 Optimising 8.3.2.4 Optimising 8.3.3 Optimising 	enovation methods and furrow head on infiltration ns for irrigation management aging irrigation and field design to improve perform methods management and field design optimisation protocol g irrigation management of PRB systems in Australia nising time to cut-off nising furrow length nising time to cut-off relative to water advance time g irrigation management of PRB farming systems in Pal nising time to cut-off nising time to cut-off g irrigation management of PRB farming systems in Pal nising time to cut-off nising time to cut-off nising time to cut-off relative to water advance time g irrigation management in response to PRB renovation	160 160 163 mance 163 mance 164 164 164 165 166 166 167 168 168 168 168 169 170 kistan 170 kistan 171 173 175 176
 7.4 Discussion 7.4.1 Effect of r 7.4.2 Implicatio 7.5 Conclusions CHAPTER 8: Mana 8.1 Introduction 8.2 Material and 8.2.1 Irrigation r 8.3 Results 8.3.1 Optimising 8.3.1.1 Optim 8.3.1.2 Optim 8.3.1.3 Optim 8.3.1.4 Optim 8.3.2.1 Optim 8.3.2.1 Optim 8.3.2.1 Optim 8.3.2.2 Optim 8.3.2.3 Optim 8.3.2.4 Optim 8.3.3 Optimising methods on the Vertice 	enovation methods and furrow head on infiltration ns for irrigation management	160 160 160 162 163 nance 164 164 164 165 165 166 167 168 168 168 168 168 168 169 170 kistan 171 175 175 176

8.3.3	.2 Optimising inflow rate and time to cut-off
8.3.3	.3 Optimising furrow length
8.3.3	.4 <i>Optimising time to cut-off relative to water advance time</i> 182
8.4 D1	scussion
8.4.1	Irrigation management practices of PRB systems in Australia
8.4.2	Irrigation management practices of PRB systems in Pakistan
8.4.3 8.5 Co	Irrigation management practices in response to renovation methods 186 onclusions
CHAPTER	9: General discussion and recommendations
9.1 Int	roduction
9.2 Ma	ajor outcomes and key findings 190
9.2.1	Benchmarking irrigation performance of PRB farming systems 190
9.2.2	Impact of renovations on irrigation performance and productivity 191
9.2.3	Effect of renovation on lateral infiltration
9.2.4	Strategies to improve lateral infiltration and irrigation performance 194
9.3 Su	mmary
9.4 Pr	actical implications of this research
9.5 Re	commendations for further research 199
CHAPTER	10: Conclusions 200
References	
Appendix A	: Specific field measured data 223
Appendix B	: IPARM and SIRMOD calibration233
Appendix C	: EnviroSCAN and Micro-Gopher calibration validation 236
Appendix D	: Irrigation performance and optimisation for all sites and
Appendix E	: Hydrus 2D calibration244

LIST OF FIGURES

Figure 1.1:	Frame work of management factors affecting water productivity of raised beds
Figure 1.2:	Outline of dissertation structure
Figure 2.1:	Bed cross section with three soil zones affecting irrigation and crop performance
Figure 2.2:	Screen shot of IPARM version 2 main user interface
Figure 2.3:	Screen shot of SIRMOD version III main output screen
Figure 3.1:	General location of study sites in Australia and Pakistan on World map53
Figure 3.2:	Furrow spacing at (a) site A1 (b) site A2 and (c) Furrow dimension (TW: Top width, MW: Middle width, BW: Bottom width and D: furrow depth or bed height)
Figure 3.3:	Available raised bed machinery (a) bed former; note the smearing of bed surface soil and the shallow, rough furrows (insert show furrower) and (b) no till seeder with 8-rows of crop per bed at site A1 (Marinya farm, Cambooya, Australia)
Figure 3.4:	On farm head channel with siphons, flow meters, soil moisture data loggers and weather station at site A1 (Marinya farm, Cambooya, Australia)
Figure 3.5:	General field conditions and crop stage during measurement at site A2, (Bandawing farm, Dalby, Australia)
Figure 3.6:	General field conditions of site P2 under corn crop in Mardan, Pakistan (Note that the corn is sown on the edge of the beds to overcome poor penetration of the wetting front, as seen by the wet areas on the surface of the beds)
Figure 3.7:	Furrow spacing of (a) NB: Narrow bed (b) WB: Wide bed evaluated at three research sites during the benchmarking study in Mardan, Pakistan
Figure 3.8:	Layout of Sentek EnviroSCAN [®] probes placed across half bed width at given positions of sensors for logging soil moisture variations at 10 minutes interval during 2010 wheat season at site A1, Marinya farm, Cambooya, Australia
Figure 3.9:	IRRIMATE TM equipment for irrigation measurements (a) Siphon flow meter, (b) water advance meter, (c) flume with flow meter
Figure 4.1:	Soil moisture deficit (SMD) in the root zone (0-100cm depth) prior to two irrigation applications at both sites on Vertisol, Australia (vertical bars show SD)
Figure 4.2:	Average <i>SMD</i> at given depths on three sites, prior to two measured irrigation applications at Mardan, Pakistan (Vertical bars show SD)71

Figure 4.3:	Average field measured furrow dimensions (recorded prior to two irrigations at all three sites for narrow bed (NB: 66cm furrow spacing) and wide bed (WB: 132cm furrow spacing) systems at Mardan, Pakistan (vertical bars show SD)
Figure 4.4:	Average field measured water advance curves of two irrigations at (a) sites A1, and (b) site A2 in Australia (vertical bars show SD)74
Figure 4.5:	Crop and field conditions before (a) irrigation 1 with no cracks; (b) irrigation 2 with flow crossing the bed through cracks at site A2, Australia
Figure 4.6:	Impact of (a) bed size and (b) wheel trafficking on average water advance along furrow for all three sites in Pakistan (Vertical bars show SD)
Figure 4.7:	Field conditions of site P2 with (a) NB: Fast water advance on trafficked furrows and slow on non-trafficked alternate furrows; (b) WB: Uniform water advance on both trafficked furrows
Figure 4.8:	Comparison of SIRMOD simulated cumulative infiltration at given times during two irrigations at both sites (A1 and A2) on a Vertisol, in Australia
Figure 4.9:	Average cumulative infiltration and infiltration opportunity time along the furrow length during the two irrigations at (a) site A1 and (b) site A2 in Australia
Figure 4.10	2: Average volume infiltrated per unit furrow length on two bed sizes at three sites during two irrigations in a sandy clay loam, Pakistan (Vertical bars show SD)
Figure 4.11	: Average volume infiltrated per unit area on two bed sizes at three sites in sandy clay loam, Pakistan (Vertical bars show SD)
Figure 4.12	2: Spatial variation in average cumulative infiltration along the furrow length to replenish 60 mm soil moisture deficit (<i>SMD</i>) during two irrigations each at (a) site P1, (b) site P2 and (c) site P3 under two bed sizes in sandy clay loam, Pakistan
Figure 5.1:	Layout of three PRB renovation methods including (a) NT: no tillage, (b) SC: shallow cultivation and (c) BP: blade ploughing treatments
Figure 5.2:	Establishment of two renovation treatments for 2010 wheat crop after 2009 soybean harvest at site A1, Marinya farm, Cambooya, Australia90
Figure 5.3:	Layout of bed renovation treatment plots at site A1, Australia91
Figure 5.4:	Overbank systems and IRRIMATE TM meters for measuring inflow rate. Insert shows water advance sensors and soil moisture sensors layout during irrigation. Photo taken during 2010 wheat season at site A1, Cambooya, Australia
Figure 5.5:	In-season variations in furrow dimensions (TW = top width, MW = middle width, BW = bottom width and D = furrow depth) during 2010 wheat season (vertical bars show SD)
Figure 5.6:	Average seasonal bulk density of three renovation treatments during three cropping seasons at site A1, Australia (Vertical bars show SDs) 100

Figure 5.7:	Average s	soil me	oisture	deficit	in 100	cm	depth	prior to	irrigations	during
	the three s	seasons	s (verti	cal bars	show	SD))			101

- Figure 5.10: Volumetric soil moisture (100 cm depth) at given distance from furrow centre during the first irrigation to 2011 corn at site A1, Australia 103

- Figure 5.14: Plant density (a) and dry plant biomass (b) for 2011 hemp three rows (row 1 on bed edge and row 3 on bed middle) across a half bed width for three tillage treatments at site A1, Australia (vertical bars show SD) .. 109

- Figure 6.2: Average bulk density (BD) before two irrigations for three PRB renovation treatments at site A1, Vertisol (vertical bars show SD)...... 125

Figure 6.7: Cumulative infiltration across a 200 cm wide bed with three renovation treatments during the first irrigation at site A1-Vertisol
Figure 6.8: Cumulative infiltration across a 200 cm wide bed with three renovation treatments during the second irrigation at site A1-Vertisol
Figure 6.9: Soil moisture distribution for three PRB renovation treatments across a 200 cm wide bed during the first irrigation at site A1-Vertisol
Figure 6.10: Soil moisture distribution for three PRB renovation treatments across a 200 cm wide bed during the second irrigation at site A1-Vertisol132
Figure 6.11: Average bulk density (BD) before two irrigations for three PRB renovation treatments at site P4-sandy clay loam (vertical bars show SD)
Figure 6.12: Impact of renovation method, wetting time, and distance from furrow centre on average soil moisture in 100 cm profile depth during two irrigations at site P4-sandy clay loam
Figure 6.13: Change in soil moisture content $(\Delta \theta)$ across a 132 cm wide bed at given distances from furrow centre and given times with three renovation treatments during two irrigations at site P4-sandy clay loam
Figure 6.14: Cumulative infiltration across a 132 cm wide bed with three renovation treatments during two irrigations at site P4-sandy clay loam 136
Figure 6.15: Soil moisture distribution for three PRB renovation treatments across a 132 cm wide bed during two irrigations at site P4-sandy clay loam 137
Figure 6.16: Effect of three PRB renovation methods on average field sorptivity for (a) site A1-Vertisol during first irrigation and (b) site P4-sandy clay loam during two irrigations
Figure 7.1: Locations of observation nodes on surface and vertical boundaries of flow domain for Vertisol
Figure 7.2: Correlation between measured vs. predicted soil moisture during 9 hrs wetting for three renovation treatments with 9 cm furrow water head . 146
Figure 7.3: Simulated soil moisture variation on the bed surface (shown by observation node number: Figure 7.1) for three PRB renovation treatments with 9 cm furrow water head applied on a Vertisol
Figure 7.4: Simulated soil moisture variation on the bed surface at $(L1 = 36, L2 = 46, L3 = 51, L4 = 56, L5 = 61 and L6 = 66 cm distance from furrow centre) for three PRB renovation treatments with 9 cm furrow water head applied on a sandy clay loam$
Figure 7.5: Simulated soil moisture variations below furrow (shown by observation node number: Figure 7.1) for three PRB renovation treatments with 9 cm furrow water head applied on a Vertisol
Figure 7.6: Simulated soil moisture variations on the vertical boundary under the bed middle (shown by observation node number: Figure 7.1) for three PRB renovation treatments with 9 cm furrow water head applied on a Vertisol

Figure 7.7: Simulated soil moisture variations below furrow (shown by observation node number: Figure 7.1) for three PRB renovation treatments with 9 cm furrow water head applied on a sandy clay loam
Figure 7.8: Simulated soil moisture variations on the vertical boundary under the bed middle (shown by observation node number: Figure 7.1) for three PRB renovations with 9 cm furrow water head applied on a sandy clay loam 149
Figure 7.9: Effect of furrow water head on the wetting time required to reach field capacity at different lateral distances from furrow centre on a Vertisol 150
Figure 7.10: Effect of furrow water head on the wetting time required to reach field capacity at different lateral distances from furrow on a sandy clay loam 151
Figure 7.11: Effect of furrow water head and wetting time on the depth of wetting beneath the furrow on a Vertisol
Figure 7.12: Effect of furrow water head and wetting time on the depth of wetting beneath the furrow on a sandy clay loam
Figure 7.13: Effect of furrow water head on simulated cumulative infiltration for three PRB renovation methods on a Vertisol
Figure 7.14: Effect of furrow water head on simulated cumulative infiltration for three PRB renovation methods on a sandy clay loam
Figure 7.15: Effect of wetting time and furrow head on the water stored in 100 cm deep root zone across half bed width for three renovation treatments on a Vertisol
Figure 7.16: Effect of wetting time and furrow head on the water stored in 100 cm deep root zone across half bed for three renovation treatments on a sandy clay loam
Figure 7.17: Effect of wetting time and furrow head on deep drainage losses below 100 cm root zone depth (across half bed x-section) for three PRB renovation methods on a Vertisol
Figure 7.18: Effect of wetting time and furrow head on deep drainage losses below 100 cm root zone depth (across half bed x-section) for three PRB renovation methods on a sandy clay loam
Figure 7.19: Simulated soil moisture distribution across the flow domain after 100 mm water application for three PRB renovation treatments with 9 cm furrow water head on a Vertisol
Figure 7.20: Simulated soil moisture distribution across the flow domain after 60 mm water application for three PRB renovation treatments with full furrow water head on a sandy clay loam
Figure 7.21: Effect of inflow rate on furrow water head measured in the field head section during experiments on a Vertisol and a sandy clay loam
Figure 7.22: Effect of inflow rate on wetting time of bed middle (200 cm wide bed) to field capacity for three PRB renovation methods on a Vertisol 158

Figure 7.23	: Effect of inflow rate on wetting time of bed middle (66 cm furrow spacing-NB) to reach field capacity for three PRB renovation treatments on a sandy clay loam
Figure 7.24	: Effect of inflow rate on maximum bed width required to ensure wetting of bed middle to field capacity within 7 hrs of wetting time on a sandy clay loam
Figure 8.1:	Impact of optimizing Tco at $Er = 85\%$ on the distribution of infiltration opportunity time along the furrow length during two irrigations of 2009 soybean. The dotted box highlights the tail end area susceptible to poor lateral infiltration
Figure 8.2:	Example irrigation management tool for illustrating the optimised Tco vs. Q and resulting Ea , Er and DU evaluated for two irrigations at two sites
Figure 8.3:	Example irrigation management and field design tool illustrating the effect of furrow length and Q on Tco , inflow volume and $Tco Ta^{-1}$ (solid lines) and <i>Ea and DU</i> (dotted line) for achieving $Er \ge 85\%$, water reaching furrow tail end on (a) site A1 with blocked end furrows and (b) site A2 with open ended furrows requiring wetting to bed middle on site A1 on Vertisol (vertical bars show SD)
Figure 8.4:	SIRMOD screen shots for showing the impact of irrigation management strategies on infiltration along a blocked end furrow (86 m) for a high infiltration (site P1-NB during first irrigation) sandy clay loam soil171
Figure 8.5:	SIRMOD screen shots for showing the impact of irrigation management strategies on infiltration along a blocked end furrow (90 m) for a low infiltration (site P2- NB during first irrigation) sandy clay loam soil 172
Figure 8.6:	Example irrigation management tool showing the effect of inflow rate on optimum time to cut-off for (a) narrow and (b) wide beds with ~90 m furrow length at three sites on a sandy clay loam (vertical bars show SD)
Figure 8.7:	Impact of irrigation management strategies on performance of (a) over- irrigated events (NB & WB at site P1 and NB at site P3) and (b) under- irrigated events (NB & WB at site P2 and WB at site P3) in a sandy clay loam (vertical bars show SD)
Figure 8.8:	Example field design and irrigation management tool illustrating the effect of furrow length and inflow rate on optimised <i>Tco</i> and <i>Tco</i> Ta^{-1} (solid lines) and <i>Ea</i> and <i>DU</i> (dotted lines) for (a) narrow beds (66 cm furrow spacing) and (b) wide beds (132 cm furrow spacing) in a sandy clay loam. Optimisation required $Er \ge 85\%$ and water reaching to the tail end)
Figure 8.9:	Impact of renovation methods and Q on optimal <i>Tco</i> and inflow volume (solid lines) and <i>Ea</i> and <i>DU</i> (dotted line) on (a) one season old PRBs (2010 wheat) (b) three season old PRBs (2011 corn) at site A1, Australia. Optimisation requires wetting of bed middle and accounts for furrow water head effect on lateral infiltration

Figure 8.10: Example field design and irrigation management tool for	showing the
effect of furrow length and Q on optimum Tco & Inflow v	volume (solid
lines) and Ea & DU (dotted lines). Optimisation data	for a single
irrigation to 2010 wheat requiring lateral wetting of bed	middle on a
Vertisol at site A1, Australia	

- Figure 8.12: Effect of furrow length on optimum $Tco Ta^{-1}$ (solid line) and Ea (dotted line) for freshly renovated PRBs (2010 wheat season) for different Q and requiring wetting of bed middle at site A1, Australia.. 182

- Figure D1: Example irrigation management tool showing the effect of inflow rate on optimum time to cut-off on irrigation performance (*Ea* and *DU*) by confirming $Er \ge 85\%$ for two bed sizes (NB & WB) with 86-90 m furrow length at three sites on a sandy clay loam (vertical bars show SD) 242

- Figure E3: Outline of flow domain with applied boundary conditions.......248

LIST OF TABLES

Table 3.1:	Research plan for improving water productivity of raised beds
Table 3.2:	Summary of experimental sites, objectives, field conditions and measurements (SM = soil moisture, BD = bulk density, SCL= sandy clay loam and three treatments are described in Chapter 5)
Table 3.3:	Input and output data of IPARM and SIRMOD models65
Table 4.1:]	Experimental details of measurement of existing irrigation performance in Australia and Pakistan (SM-soil moisture, BD- bulk density and <i>SMD</i> -soil moisture deficit, FC- Field capacity, NB-narrow bed and WB-wide bed)
Table 4.2:	Average furrow dimensions (TW = top width, MW = middle width, BW = bottom width and D = depth) recorded prior to two irrigations at two sites (A1 and A2) in Australia and three sites (P1, P2 and P3) in Pakistan (SD in brackets)
Table 4.3:	Average irrigation performance of two irrigations measured on two sites (A1 and A2 with two bed sizes) in south east Queensland, Australia and three sites (P1, P2 and P3 with two bed sizes) in north west Pakistan (SD in brackets)
Table 4.4:	Volume of surface water storage in furrows at the time of water arrival to tail end on three sites with two bed sizes in sandy clay loam, Pakistan (SD in brackets)
Table 5.1:	Soil tillage applied to each treatment over three cropping seasons at site A1
Table 5.2:	Experimental details of measurement of soil properties, irrigation, crop yield and water productivity per each PRB renovation plot during three cropping seasons at site A1, Australia (SM = soil moisture, BD = bulk density)
Table 5.3:	Average furrow dimensions (according to layouts shown in Figure 3.2c) under three PRB renovation treatments during each cropping seasons (SD in brackets)
Table 5.4:	Total water applied (<i>TWA</i>) during the three cropping seasons in three PRB renovation methods at site A1, Cambooya Australia
Table 5.5:	Irrigation performance for freshly applied PRB renovation treatments during two cropping seasons (2010 wheat & 2011 corn) and on unrenovated PRB (2011 hemp) treatments at site A1, Australia (SD in brackets)
Table 5.6:	Average dry plant biomass for three crops and dry grain yield for 2011 corn with three PRB renovation methods at site A1, Australia (SD in brackets)
Table 5.7:	Dry biomass input water productivity of three crops under three PRB renovation methods at site A1, Australia
Table 5.8:	Dry grain input water productivity of 2011 corn under three PRB renovation methods at site A1, Australia

Table 6.1:	Average furrow dimensions during two irrigations on a Vertisol and a sandy clay loam (furrow dimensions explained in Figure 3.2c), (SD in brackets)
Table 6.2:	Wetting front penetration (in cm from furrow centre) at the bed surface after 9 hrs of wetting for three treatments during the first irrigation at site A1-Vertisol
Table 6.3:	Wetting front penetration (in cm from furrow centre) at the bed surface after 9 hrs of wetting for three renovation treatments during the two irrigation at site P4-sandy clay loam
Table 6.4:	Field sorptivity (mm hr ^{-0.5}) during 9 hrs wetting time for three PRB renovation treatments during the first irrigation at site A1-Vertisol and average of two irrigations at site P4-sandy clay loam (SD in brackets) 138
Table 7.1:	Time required for the wetting front to reach the given positions on the two soils with three PRB renovation methods at 9 cm furrow water head applied
Table 8.1:	Average irrigation performance parameters for strategies of optimising (i) time to cut-off (<i>Tco</i>) and (ii) inflow rate (<i>Q</i>) plus <i>Tco</i> compared with field measured farmer practice (FP) for two sites (site A1 = 465 m long furrows with blocked tail ends and site A2 = 455 m long furrows with open end furrows at tail end) evaluated during two irrigation events on a Vertisol (SD in brackets)
Table 8.2:	Average irrigation performance parameters for strategies of optimising (i) time to cut-off (Tco) and (ii) inflow rate (Q) plus Tco compared with field measured farmer practice (FP) for two bed sizes evaluated during two irrigation events (~90 m furrow length) on a sandy clay loam (SD in brackets)
Table 8.3:	Impact of three fresh PRB renovation methods and optimised (Q and Tco) on irrigation performance during two cropping seasons on Vertisol (SD in brackets)
Table 8.4:	Impact of one season old renovation methods and the optimised (Q and Tco) on irrigation performance across all experimental plots during the 2011 hemp season (SD in brackets)
Table A 1	: Soil properties at given depths at site A1, Australia (Soil analysis conducted by Topoclimate services Pty Ltd)
Table A 2:	Soil properties at given depths on two bed sizes measured at the harvest of wheat 2008-09 crop at site P1, Pakistan
Table A3:	Average monthly weather data at site A1 during four cropping seasons (SD in brackets)
Table A4:	Average bed furrow dimensions measured before irrigations during the benchmarking study in both soils and three renovation experiments in Australia (TW = top width, MW = middle width, BW = Bottom width and D = furrow depth)
Table A5:	Average water advance and variability during the benchmarking study 226

Table A6:	Temporal variation in average bulk density (BD in gm cm ⁻³) and soil moisture (SM in %) of 0-30 cm profile at 10 cm interval for three PRB renovation treatments during three cropping seasons in a Vertisol, Australia (SD in brackets)
Table D1:	Existing irrigation performance of raised beds and deep drainage losses (<i>DD</i>) during two irrigations on two sites in south east Queensland, Australia (SD in brackets)
Table D2:	Existing irrigation performance of raised beds and deep drainage losses (DD) during two irrigations on three sites with two bed sizes in north west Pakistan (SD in brackets)
Table D3:	Average irrigation performance for all irrigation events applied on three PRB renovation methods during three cropping seasons on a Vertisol at site A1, Australia (SD in brackets)
Table D4:	Average seasonal irrigation performance for three PRB renovation methods during three cropping seasons on a Vertisol at site A1, Australia (SD in brackets)
Table D5:	Effect of two irrigation management optimisation strategies on current irrigation performance of three sites with two bed sizes during two irrigations under sandy clay loam during benchmarking study in Mardan, Pakistan (SD in brackets)
Table D6:	Effect of two irrigation management optimisation strategies on current irrigation performance of three PRB renovation treatments during three cropping seasons on Vertisol at site A1, Australia (SD in brackets) 241
Table E1:	Soil hydraulic parameters for the analytical functions of Van Genuchten equation after calibrating the Hydrus 2D model according to the measured field conditions (Units for time $(t) = hr$ and length $(l) = cm) 245$
Table E2:	Field measured furrow dimensions (TW = top width, BW = bottom width and D = depth of furrow) during experiments conducted on Vertisol and sandy clay loam
Table E3:	Antecedent soil moisture and soil moisture deficit (SMD) in 100 cm root zone depth in Vertisol and 60 cm depth in sandy clay loam
Table E4:	Relationship between measured and predicted soil moisture at given times and depths across the bed on a Vertisol at 6 cm furrow water head $(R^2 \text{ values in }\%)$
Table E5:	Relationship between measured and predicted soil moisture at given times and depths on Vertisol at 9 cm furrow water head (R ² values in %).
Table E6:	Relationship between measured and predicted soil moisture at given times and depths across bed on sandy clay loam soil at 9 cm furrow water head (R^2 in %)

LIST OF ABBREVIATIONS

ACIAR	Australian Centre for International Agricultural Research
BD	bulk density
BP	blade ploughing
BW	bottom width of furrow
CA	conservation agriculture
CTF	control traffic farming
D	depth of furrow
DD	deep drainage losses
DU	distribution uniformity
Ea	application efficiency
Er	requirement efficiency
FC	field capacity
MW	middle width of furrow
NB	narrow bed (66 cm furrow spacing)
NCEA	National Centre for Engineering in Agriculture
NT	no tillage
PARC	Pakistan Agricultural Research Council
PRB	Permanent raised bed
Qld	Queensland
SC	shallow cultivation
SD	standard deviation
SM	soil moisture
SMD	soil moisture deficit
Та	time of irrigation water advance to furrow tail end
Тсо	time to cut-off
TW	top width of furrow
WB	wide bed (132 cm furrow spacing)
WP	wetted perimeter
WP	input water productivity

LIST OF SYMBOLS

А	cross sectional area of flow
а	Kostiakov infiltration exponent
D_{f}	diffusivity
f_o	steady state infiltration rate of the soil
g	acceleration due to gravity
Ι	infiltration rate
k	Kostiakov infiltration equation coefficient
Q	inflow rate
r	fitted parameter
S	sorptivity
S_{f}	frictional slope
S_o	furrow slope
v	flow velocity
у	furrow water head
Ζ	cumulative infiltration
Θ	volumetric soil moisture content
$ ho_b$	bulk density
$ ho_s$	particle density
σ_{y}	surface storage shape factor
σ_{z}	subsurface storage coefficient
F	scaling factor
φ	boltzman transform
Ψ	soil moisture potential

CHAPTER 1: Introduction

Raised beds may be defined as a farming practice where the beds and furrows are constructed from the cultivated soil that is excavated by furrowers aligned behind the wheels of tractors, and sometimes, as well, with the mid-point the tractor track width. They are raised above the normal level of the soil surface by the soil excavated to form the furrows. The crop is then planted on the surface of the beds and irrigation water is applied to the field via the furrows, with the water absorbed laterally by the soil into the beds (Roth et al. 2005). The furrows are oriented in the direction of the slope and are also used for traffic, and, if need be, bed drainage. Furrow spacing is generally determined by a combination of the track width of machinery, row spacing of the crops and the stability of the soil to rapid wetting, which control the lateral infiltration of water into the centre of the beds. Raised beds are also used under rainfed conditions to prevent crop waterlogging.

There is a wide variety of cultivation and bed construction methods used around the world, depending on available labour resources, available machinery, farming practice tradition and knowledge of soil and water management. Cultivation methods vary from animal drawn (bullock, horse or camel) ploughing, to tractor drawn chisel ploughing, mould-board ploughing and rotary hoeing. Raised bed construction methods range from animal-drawn single furrowers to tractor drawn, GPS-steered, multiple bed forming implements mounted on hydraulic 3-point linkage arms on the rear of tractors.

Bed and furrow dimensions and field, or furrow length, also vary widely around the world. These vary as a consequence of farm size, slope length, water inflow rates, infield water distribution infrastructure, proportion of field taken up by unutilised headlands, available machinery, soil type and farmer knowledge. For instance, furrow spacing, or bed width, ranges from 0.5 to 1.0 m in China, India, Pakistan, Bangladesh and Mexico to 1.5 to 3.0 m in Australia (Sayre & Hobbs 2004). Consistent with the variation in bed width/furrow spacing, furrow dimensions range in top width from 40 to 60 cm and furrow depth from 10 to 25 cm.

1.1 Agricultural sustainability and food security

The increasing world population requires greater productivity from existing land, because there is little or no land available to expand agriculture (Hobbs 2007). Simultaneously, climate change is reducing and making less reliable water resources for both agriculture and human consumption (Vörösmarty et al. 2000), and land productivity is declining as a consequence of excessive cultivation, degrading soil structure and the chemical and biological fertility of soils (Beddow et al. 2009). However, with the advent of herbicides to kill weeds the need for cultivation has diminished (Koskinen & McWhorter 1986). This has given rise to reduced tillage and reduced traffic for crop production, known as conservation agriculture (CA).

CA is based on practices that minimally disturb soil, maximise soil cover and rotate crops (Unger 2006). This facilitates the natural soil formation processes of active soil biology and root retention, which enhance the build-up of organic matter and organic nutrients, and combined to increase soil aggregation and the stability of aggregates to wetting (Dexter 1991; Limon-Ortega et al. 2006). In turn, water infiltration and water storage properties improve, and these lead to enhanced irrigation efficiency (Hulugalle et al. 2004). Hence the benefits of CA are twofold: increased irrigation efficiency, and increased productivity (Hassan et al. 2005).

The application of CA farming practices to raised bed irrigation systems occurred because of the operational efficiencies of chemical weed control. However, it brought with it expectations that the soil conditions and fertility improvements, which have been reported where it was practised in rain-fed agricultural areas (Tullberg et al. 2007; McHugh et al. 2009) would ensure that adequate lateral infiltration properties would be maintained. It also brought with it the expectation that all wheel traffic would be confined to the furrows, reducing the loss of water through the base of furrows and avoiding wheel compaction on crop seedbeds. The introduction of CA to raised beds thus enabled the beds to become permanent, and so the term "permanent raised bed" (PRB) was used to describe this way of farming. Hence PRB is a type of soil and irrigation management where the field is divided into two sections: (i) furrows for irrigation application, drainage, and machinery traffic; and (ii) raised beds for growing crops with reduced tillage.

However, notwithstanding the practice of CA on PRB, the expected improvement in irrigation performance failed to eventuate (Akbar et al. 2007; Jin et al. 2007). There are two main reasons for this: one is that some soils used for irrigated PRB are inherently structurally unstable because they have a high exchangeable sodium percentage, which causes soil aggregates to disperse and seal when wet (Smith et al. 2001); and the second is that soils subjected to rapid wetting and drying will disaggregate, subside and consolidate even when they have a stable structure (Utomo & Dexter 1982). Hence, the infiltration properties of soils in PRB, although in better condition than soils in cultivated raised beds (Pagliai et al. 2004), continue to have lateral infiltration rates that are less than desirable (Lucy 1993).

The failure of lateral infiltration to be adequate is a major reason behind conflicting research findings on PRB under variable environmental conditions. Poor lateral infiltration into the bed centre (subbing) has been reported in PRB in many countries e.g. Australia, China, Pakistan and India, where the soils used have poor infiltration properties (Lucy 1993; Akbar et al. 2007; Jin et al. 2007). Thus, additional measures for improving the stability of soil structure to wetting and maintaining a loose friable soil are essential if PRB farming is to gain widespread adoption golbally.

What is needed to achieve these conditions for the soil in PRB is machinery and practices that can renovate beds in a way that loosens the soil with near-zero disturbances, retains all the roots from previous crops in a near undisturbed state and cleans and reshapes furrows and bed shoulders. If achievable, such practices would ensure the benefits of bed-furrow irrigation to be magnified in terms of both irrigation efficiency (lateral infiltration) and productivity. Thus, measures for improving the water productivity (*WP*) of PRBs are central to gaining widespread adoption for the sake of sustainable agriculture.

1.2 Water productivity

The *WP* is a generic term, which can be defined as "the physical mass of production or the economic value of production measured against gross inflows, net inflow, depleted water, process depleted water, or available water" (Molden 1997). This research is focussed on input *WP* because it deals with the physical mass of production (grain or straw yield) measured against water input (irrigation + rainfall). There are many factors that can affect the *WP*, but this research deals with two main areas comprising agronomic and irrigation management. Improving soil structural stability, lateral infiltration, and general soil fertility with better agronomic management (Pagliai et al. 2004) and enhancing irrigation performance with reduced irrigation application losses (Smith et al. 2005) and improved application uniformity are central for enhancing *WP* of PRB farming systems on a sustainable basis.

1.2.1 Agronomic management

Furrow irrigation, no matter how water is applied, depends for its effectiveness on the hydraulic properties of the soil in beds, which for relatively fast lateral movement of soil-water depends on the stability to wetting of the structure of the soil in the beds (Singer & Munns 1999). Despite this dependence and the wide spread scientific knowledge of the deleterious effects of cultivation on the stability of soil structure (Hulugalle & Daniells 2005; McHugh et al. 2009), most irrigation practices in the world still use an unnecessarily large amount of cultivation (Roth et al. 2005). For instance, in most of the under-developed and developing world beds are used only for row crops. In these types of countries, basin irrigation is the preferred irrigation practice for field crops, such as wheat, barley, rice, canola etc., and beds are used for row crops, such as cotton, maize, and vegetables.

The alternate use of beds for every second crop requires an unnecessarily large number of cultivation operations for, firstly, the construction and then, the destruction of beds and levelling of the field for subsequent basin-irrigated field crops. In countries where these practices are the norm, farmers overcome the slow rate of infiltration of water to the centre of beds by overtopping the beds in the first and, possibly, second irrigations (to ensure good germination and establishment of their crops) and/or filling the furrows and maintaining a high water level in the furrows for long periods (Hamilton (pers. comm.)).

Even in countries where beds are used for successive or near-continuous cropping, cultivation exceeds the small number and minimal soil disturbance that is necessary for the creation and maintenance of stable soil structure (Roth et al. 2005). In these

countries, various forms of cultivation continue to be practised because of the need to reshape the furrows and beds and to loosen the soil in the beds. Some bed cultivation is considered necessary because of the combined needs of creating a good seedbed, ensuring adequate rates of infiltration into the beds, and for controlling weeds, diseases and pests. With the advent of herbicides to control weeds and a general recognition that soil cultivation is constraining productivity and increasing the cost of production, there is an emerging awareness and desire to find improved ways of managing soil to increase water infiltration, crop production, and farm profitability.

1.2.2 Irrigation management

The Oxford English dictionary (<u>www.oed.com</u>) defines irrigation as "The action of supplying land with water by means of channels or streams; the distribution of water over the surface of the ground, in order to promote the growth and productiveness of plants". This research deals with furrow irrigation, where irrigation application rate and duration are the main irrigation management factors that affect the efficiency of irrigation application (Raine & Bakker 1996).

Furrow irrigation is one of the commonly used form of surface irrigation throughout the world (Burt et al. 1995). It is generally considered to be a more water-efficient method in comparison with the basin irrigation method because of (i) the speed with which water is conveyed to the low end of a field (Gillies 2008); and (ii) " the relatively small proportion of the soil surface is in contact with the flowing water during irrigation than the flat basin, where whole surface is inundated (Walker 1989).

Although furrow irrigation is more efficient than basin irrigation, it still has challenges. One of the challenges is the knocking down of beds after row crops for the next field crop (wheat, canola, rice etc.) and then reconstruction for the next row crop. Consequently soil in the bed consolidates with wetting and drying and its hydraulic properties deteriorate. Thus, the resulting poor lateral infiltration detracts from the efficiency potentially available by adopting furrow irrigation.

1.3 Research opportunities

Adoption of PRB in developing countries is currently low and existing PRB management is generally inappropriate. Several factors contribute to low adoption rates including uncertainty over the magnitude of crop and water productivity benefits, operational and management factors including lack of availability of suitable machinery and guidelines on appropriate irrigation design and management practices. Anecdotal evidence also suggests that bed subsidence of structurally degraded soils may also limit lateral infiltration into the bed and raises questions of what is an ideal bed width. In these cases, there is a need to identify strategies to enhance lateral infiltration by loosening subsided beds consistent with CA.

The available decision support guidelines on irrigation and soil management practices are limited for improving the *WP* of irrigated PRBs in general and in Pakistan in particular. Soil properties under variable agronomic management of PRB were studied under rain-fed conditions. Similarly, previous research on irrigated PRB has only provided water saving comparisons with traditional flat basin or bay application systems. Hence, there are real research needs to (i) find a way of loosening subsided beds that is consistent with the principles of CA and creates and maintains a stable, loose soil structure with high rates of lateral infiltration; and (ii) develop appropriate irrigation application practices that capture these soil benefits of PRB in the most efficient manner. Combined agronomic and irrigation management strategies need to be explored for improving *WP* of irrigated PRBs.

1.4 Research question and aim

The underpinning research question for this dissertation is: "Can the productivity of current irrigated PRB farming practices be improved by improved soil management (bed renovation) and irrigation management?"

Hence, the aim of this research is to identify and evaluate strategies which improve the input *WP* of irrigated PRB farming systems.

The input WP of raised beds may be affected by many factors including plant genetics, external plant environmental characteristics, physical processes, their

management and their complex interactions. However, the research reported here deals with the agronomic and irrigation management factors shown in Figure 1.1. In terms of crop management, it deals only with the establishment and yield of crops.



Figure 1.1: Frame work of management factors affecting water productivity of raised beds

1.5 Hypotheses

The key hypotheses for this research are that:

- The input *WP* of irrigated PRB is sub-optimal due to inappropriate soil management, field design, and irrigation management practices.
- The input *WP* of current PRB farming systems can be increased with appropriate bed renovation methods, improved field design (including bed-furrow dimensions) and optimised irrigation management.

1.6 Specific objectives

The specific objectives of this research are to:

- Evaluate the irrigation performance and input *WP* of PRB under commercial Australian and Pakistani farm conditions.
- Evaluate the effect of PRB renovation methods on seasonal changes in soil hydro-physical properties (including lateral infiltration), irrigation performance, crop yield and input *WP*.
- Identify soil and irrigation management strategies to improve input *WP* by increasing lateral infiltration and optimising irrigation performance of PRB.

1.7 Structure of this dissertation

The dissertation structure is outlined in Figure 1.2. This introductory chapter has provided a brief background to the research area including the hypotheses and specific objectives of this study. Chapter 2 reviews existing literature and will provide a rationale for improving input *WP* through keeping in view the prevailing agronomic and irrigation management issues both in Australia and in Pakistan. It reviews the adaptation of conservation agriculture to traditional raised bed farming systems in Australia and Pakistan. Chapter 2 then describes the main agronomic and irrigation methods that are required to improve the input *WP* of raised bed system.



Figure 1.2: Outline of dissertation structure

Chapter 3 is a general material and methods and describes the research plan, study sites, their background information, previous crop and cultivation history and field conditions. It also describes the general assessment methods that were applied across all or most sites during the three field studies and a desktop study components.

Chapter 4 benchmarks the irrigation performance of commercial PRB farming systems in Australia and Pakistan. It provides an assessment of the measured

irrigation performance and identifies the main agronomic and irrigation management issues affecting the *WP*.

Chapter 5 evaluates the effect of PRB renovation methods on irrigation performance and productivity. It describes the experiments for the combined implementation of agronomic and irrigation management on a Vertisol. It provides an analysis of seasonal variations in soil hydro-physical properties e.g. bulk density, soil moisture, lateral infiltration and bed furrow configurations with particular reference to crop yield and *WP* during the three cropping seasons under three renovation methods.

Chapter 6 covers detailed field experimentation and analysis of lateral infiltration into the centre of 200 cm wide beds on a Vertisol in Australia and 132 cm wide beds on a sandy clay loam in Pakistan.

Chapter 7 describes strategies for improving lateral infiltration under 200 cm wide PRB on a Vertisol and 132 cm wide PRB on a sandy clay loam using data from Chapter 6 with Hydrus 2D simulation. It provides an assessment of three different renovation methods, variable furrow water head, inflow rate and bed width for improving lateral infiltration and developing management guidelines. It also describes implications of lateral infiltration in irrigation management.

Chapter 8 evaluates various irrigation management and field design strategies using field measured data from Chapters 4, 5 and information from Chapter 7 with SIRMOD simulation. It outlines guidelines for improving the existing irrigation performance in both countries. Chapter 8 also describes alternative solutions and decision support aids for further improvement of the irrigation performance under three PRB renovation techniques.

Chapter 9 is a general discussion and provides a comprehensive overview of the overall research outcomes and their implication in existing PRB farming systems. It also identifies several areas for future research. Chapter 10 concludes the overall key research findings and hypotheses.

CHAPTER 2: Review of permanent raised bed farming system performance

2.1 Introduction

This chapter review the rationale for improving the existing water productivity (*WP*) sustainably using PRB farming system with particular reference to Australia and Pakistan. This review divides the existing challenges and opportunities into two key areas including agronomic and irrigation management. Therefore different agronomic and irrigation factors affecting raised bed *WP* along with their assessment methods has been reviewed. The review concluded a road map to be pursued during this study.

2.2 Need to improve water productivity of raised beds

The world's future food security is uncertain because the population continues to grow at a fast rate, no or little new land is available for agriculture, and water resources are predominantly used inefficiently to produce food and are under threat in terms of the reliability of supply and the quantity available because of the impact of climate change on existing farming areas. The UN (2010) forecasted the world population to reach ~9 billion in 2050, thus water demand will increase tremendously, based on another UN (2005) estimate that ~2700 litres of water per day per single person is required to avoid water scarcity. Expansion in agricultural lands is not viable because of the limited available land and the associated environmental cost. Therefore, improving the production per unit area (Bruinsma 2003) and producing more with less water are the major options available to meet the increasing food demand on sustainable basis.

Production increase with artificial fertilisers may be not sustainable (Tilman et al. 2001). Nevertheless, the use of nitrogen was helpful in improving around 50% of yield levels and according to Gilland (2006) its use at the current level, at least, is forecast as an essential part of the global industrial farming system in the near future. However, the cost of these oil and gas based fertilisers is likely to rise in future as the oil itself will be very expensive due to rapid depletion that will limit its application

(Edgerton 2009). Moreover, soil contaminants in fertilisers like arsenic, cadmium, fluorine lead and mercury and their transfer from soil to food is a matter of concern (McLaughlin et al. 1996). Therefore, there is a need to identify practices that improve productivity without relying on artificial fertilisers.

Traditional methods of farming involve intensive tillage, which is not sustainable. Intensive tillage destroys soil stable bio pores, soil fauna, and biota and exacerbates soil carbon and organic matter loss. Production increase with intensive tillage has been proved as a temporary solution (Tilman et al. 2002; Hobbs et al. 2008). For instance, one of the main reasons for a twofold yield increase in the subcontinent since the 1960s and threefold yield increase in Europe between 1950 and 1990 were attributed to intensive cultivation; however a gradual damage to soil structure and fertility was noted. Consequently, after five decades, a decline in land production potential is obvious (Beddow et al. 2009). Thus, better agronomic management is essential for sustainable production.

Gravity fed surface irrigation systems represent ~90% of irrigation applied globally. According to a UN (2009) report, irrigated agriculture produces ~40% of world food by consuming ~70% of total global water withdrawal for irrigating 20% (~275 million ha) of global cultivated land. Hence, agriculture contributes proportionately more in global water resources losses. The major reasons are frequently attributed to the huge irrigation losses estimated to be ~30% during conveyance (Bos 1985), ~19% during field application (Postel 1993) while only ~10% to 30% of available water is used by plants as transpiration (Wallace 2000). Furrow irrigation is the most common gravity fed surface irrigation method (Burt et al. 1995), but cause higher water application losses under its current management. Water application losses under furrow irrigation have been reported to the extent of 40-70% in Australian cotton and sugarcane industries (Bakker et al. 1996; Dalton et al. 2001; Smith et al. 2005) and up to 40% in Pakistani farm conditions (World Bank 1997). Hence, producing more with less water seems achievable particularly in both countries by controlling the irrigation application losses through better irrigation management.
2.2.1 Extent of agronomic and irrigation management issues

Agronomic and irrigation management issues largely prevail in existing farming systems of Australia and Pakistan, and both countries are the focus of this study. Australia has better farming systems, better farming infrastructure and fertile land resources, but irrigation water is limited, and furrow irrigation management is well short of optimal in all states and across all soil types. Pakistan is a major irrigated country but both land and irrigation management is poor. Both major agricultural countries represent the majority of issues that are generally experienced in the developed and developing countries of the world. A review of agronomic and irrigation management systems in both countries is presented below.

2.2.1.1 Agronomic and irrigation issues in an Australian context

Australian Vertisols have been shown to be very sensitive to poor agronomic management, although it has high natural fertility. The reasons are generally attributed to the existence of low volume of permanent structural pores (Coughlan 1981), which is affected by the shrink swell nature of the soil. Swelling can close unstable macro pores making the soil less permeable (Cook et al. 1992) and more susceptible to runoff and erosion (Smith et al. 1983). In contrast, stable bio pores resist swelling and maintain permeability (Dexter 1991), which can be formed and protected by root channelling, surface residue management, mulching and biological processes (Smith et al. 1983). Existing poor agronomic management involving intensive cultivation, mismatched machinery operations, and random traffic with heavy machines negatively affects these processes.

Vertisol cracks due to wetting and drying processes. Surface cracks are formed due to evaporation and deeper cracks are formed due to soil moisture extraction by plant roots (Dexter 1991). Cracks are beneficial for enhancing lateral infiltration in widely spaced furrows and the wet/dry cycles can also repair soil structure of compacted Vertisol (Samrah et al. 1996). However, during the crop establishment period, surface cracks are more pronounced due to evaporation, while deeper cracks are negligible due to absence of water extraction by plant roots. Generally dry soil profile increase lateral infiltration due to deeper cracking but in case of shallow dry surface layer underlain with wet soil, which generally occur in crop establishment stage, can reduce seed germination in wide bed middle due to limited lateral wetting of irrigation water (Lucy 1993). Similarly, if deeper cracks are present, generally when crops are sufficiently grown, then initial infiltration is very high until the cracks close. The drop is sudden due to declining infiltration into dense soil blocks between cracks. The rapid sealing of cracks by slaking and swelling creates air trapping and zones of low porosity in deeper wetted subsoil, resulting reduced infiltration in Vertisol (Chan 1989).

Vertisol is very sensitive to subsidence, which is caused by slumping due to wetting and compaction due to random mechanical operations (trafficking), especially under moist soil with heavy machinery (McGarry 1993). According to Chan (1989), both these processes generally result in denser and stronger aggregates, which result in the reduction in soil moisture retention properties, sorptivity, and hydraulic conductivity. Thus, water storage of the dense profile could not be substantially improved with prolonged irrigation intervals. Consequently *WP* is adversely affected.

In Australia, surface irrigation (predominantly border and furrow) utilises more than 70% of irrigation water (ABS 2010) under a wide range of fields, pastures and row crops. Unfortunately, irrigation water availability is limited, while the demand for water resources has been constantly increasing due to the development of competing industries. Water is considered one of the key resources for agriculture and economic development in Australia. According to Dalton et al. (2001), improving on farm water use efficiency is part of the solution to the competing demand for limited water resources. Better management of land and water are considered the key factors for improving the existing *WP* sustainably.

Current irrigation performance is poor under most furrow irrigated raised beds and crops. For instance, application efficiencies of 30-60 % were reported by Raine and Bakker (1996) and Dalton et al. (2001) for cotton and sugarcane crops in Queensland and 61% for sugarcane in Ord river irrigation area in Western Australia (Bakker et al. 2006). The major reasons are excessive deep drainage potential and tail drain runoff losses, estimated to the extent of ~43 mm per irrigation in the Queensland cotton industry using modelling, which was estimated to equal ~US\$140 million a year at

that time (Smith et al. 2005). Poor irrigation management and improper field design were the most common reasons for this (Jensen et al. 1990; Raine & Bakker 1996).

2.2.1.2 Agronomic and irrigation issues in a Pakistani context

Around 80% of cultivated lands in Pakistan (Qureshi & Barrett-Lennard 1998) are irrigated that produce > 90% of its total agricultural production (Barral 1994). The Indus Basin of Pakistan, one of the largest contiguous canal irrigation systems in the world, operates with 25% to 40% water application losses (World Bank 1997). Moreover, pumping groundwater to supplement the canal water resulted up coning of brackish water (Akbar et al. 2001). Consequently, 35% of Indus Basin is affected by waterlogging and salinity causing ~25% loss in productivity (Qureshi et al. 2008).

The government is currently spending billions of rupees on lining the water courses to mitigate the conveyance losses and control the waterlogging and salinity issues. Subsurface drainage systems were installed during the 1980s through salinity control and reclamation programs (SCARP) in many parts of Indus Basin. However, the potential benefits of such curative programs can only be achieved once the preventive measures are strengthened. This is possible through the conjunctive improvement of reducing losses during conveyance and field application.

Irrigated raised bed farming is generally practised for the majority of row crops and vegetables in the Indus Basin. Raised bed farming in Pakistan is characterised by a narrow bed width (65-75 cm furrow spacing), short field length (30-100 m) and blocked furrows at tail ends. Selection of bed furrow configurations is generally based on the farm manager's understanding of its apparent relationship with land use efficiency, crop type, soil type, lateral infiltration, traditional practices and available machinery (Akbar et al. 2007). Wide beds (132 cm furrow spacing) were recently introduced and are mainly used for research purposes. The main reason for the use of these bed sizes is the 132 cm track width of the commonly available 50 hp tractor.

The non-swelling rigid sandy clay loam soils with a history of intensive cultivation tend to subside and structurally degrade during a cropping season. Thus, the unstable soil structure and reduced hydraulic conductivity adversely affect crop performance, especially in the bed middle (Akbar et al. 2007). The irrigation and crop performance of existing raised bed farming systems in the Indus Basin can be improved on a sustainable basis by adopting soil friendly agronomic management practices involving minimum soil disturbance that improve soil structure by accelerating the natural soil amelioration processes and organic matter build up.

Irrigation water in the Indus Basin is generally available on a weekly schedule from a network of perennial earthen canals and tributaries, which have very poor conveyance efficiency (50%) according to Gill (1994). Flow from the water course to a small temporary manifold type head ditch is generally through a manually controlled outlet called 'nucca'. The head ditch simultaneously supplies water to several (generally 5 to 10) furrows. Inflow to furrow depends on the number of furrows served, while irrigation application time often depends on flow arrival to tail end and/or furrow filling. Frequent irrigation water overtopping of the bed at the tail end and lack of wetting front penetration into the centre of wide beds (132 cm furrow spacing) are generally obvious during irrigation (Akbar et al. 2007). To achieve uniform application and to fulfil the deficit, the furrow length is generally divided into small (10-15 m long) sections with bunds and each section is irrigated separately in sequence. This practice is highly laborious, time consuming and requires careful monitoring during irrigation, while its impacts on deep drainage losses and contribution to groundwater are largely unknown but seem considerable.

Field design and irrigation management can significantly affect raised bed performance (Raine et al. 1998), but have received only limited considerations in Pakistan. Despite the fact that improved irrigation performance and increased production of raised beds versus flat basin has been reported by Hameed and Solangi (1993) and Berkout et al. (1997), irrigation management of furrow irrigation is still largely sub-optimal (Kalwij 1997; Akbar et al. 2009). The main reasons are uninformed selection of inflow rate and time to cut-off causing over irrigation and sub-optimal bed-furrow configurations causing poor lateral infiltration in wide beds (Akbar et al. 2007).

2.3 Factors affecting water productivity of raised beds

Leaving aside plant genetics and crop husbandry, which are outside the scope of this research, water productivity is affected by the type of soil being used and the manner in which it is managed (Lal 1985), plus the quality, quantity, application method, and frequency with which water is applied. These factors are of importance in all forms of irrigation, but for furrow irrigation they have an elevated importance because of the fact that: (i) water is applied to only a small proportion of the surface area of a field and its lateral movement into the root-zone of plants is dependent on the hydraulic properties of the soil, and (ii) the velocity of water flow in furrows and the rate of advance of water in the furrows needs to be matched to the soil's ability to absorb it for long enough for the infiltrating water to reach the centre of beds.

The bed furrow environment, which the irrigation water and plant encounter during a cropping season, can be illustrated by Figure 2.1 where the bed cross section has been divided into three zones. Zone 1 is the most compact and least permeable, which is a consequence of traffic necessary to farm. By controlling the traffic to defined tracks the compaction of zone 1 can be enhanced (McHugh et al. 2009), which can be beneficial in enhancing lateral infiltration to bed and controlling deep drainage losses. Soil beneath the bed in zone 2 has a density that is a consequence mostly of soil type and partly of prior history of traffic and cultivation. Therefore, better agronomic management can produce long term improvements in zone 2. Soil structure, density, hydraulic properties and bed-furrow shapes within zone 3 can be significantly influenced by farmer operations (Panettieri et al. 2013), and management factors in short term. Therefore, improved water and air movement, organic matter and fertility in zone 3 are the keys for improving *WP*.



Figure 2.1: Bed cross section with three soil zones affecting irrigation and crop performance

According to the foregoing text the major factors affecting *WP* of raised beds can be divided into two broad groups. They are related to (i) agronomic management and, (ii) irrigation management. Agronomic management is aimed to improve *WP* on sustainable basis by improving soil physical, chemical and biological properties, crop health and land use efficiency. This can largely be accomplished by optimising the soil management techniques, bed-furrow configurations and cropping management under the current raised bed systems. However, this research is focussed on soil management only. Irrigation management involves delivery of water to all parts of an irrigated field, which is an engineering challenge and can be efficiently accomplished by minimising the losses and maximising the uniformity through optimising inflow rate, time to cut-off, application depth and field design.

2.3.1 Agronomic management factors

Because hydraulic conductivity of soil is controlled by the air filled porosity, pore sizes and geometric distribution (Rezanezhad et al. 2010), thus the soil factors important for irrigation are those that affect the pore size distribution and the stability of that distribution to wetting and drying. Some of the differences in these properties between soil types are inherent, resulting from differing types of mineralogy and amounts of clay and the age of the soil profile, all of which are consequences of differing parent material and soil formation processes. Others factors are affected by the way in which the soil is managed. The main agronomic management factors affecting *WP* are listed below.

2.3.1.1 Machinery traffic and tillage

The use of large machinery is an essential part of modern agricultural production systems, because it reduces time, labour and fuel cost of operation. However, intensive tillage under high soil moisture without matching track widths and lack of defined tracks can cause soil compaction (Allen & Musick 1997). According to Soane (1976), around 90% field surface was compacted during seed bed preparation, 25% during combine harvesting and 60% during straw baling. However, by controlling the traffic paths only 15% of the field was compacted, which caused significant improvement in soil infiltration properties. Similarly, up to sixfold increase in infiltration was identified under controlled traffic farming than

conventional random traffic farming by Tullberg and Murray (1988), while McHugh et al. (2009) identified up to 45% increase in plant available water and 9% decrease in bulk density in the top 40 cm profile over a 3 year period of controlled traffic farming. Therefore, minimum soil disturbance is helpful in improving soil properties.

Controlling the machinery traffic to fixed tracks can reduce the vertical infiltration into the furrow bottom, which can be helpful in enhancing application uniformity by accelerating the water advance rate in lengthy furrows. Reduced steady state infiltration under compacted furrows was identified to the extent of 20% to 80% in several past studies (Elliott & Walker 1982; Trout & Kemper 1983; Trout & Mackey 1988a; Hunsaker et al. 1999; Li et al. 2001), which can be used to enhance irrigation water advance rate in furrows up to 45% in compacted wheeled furrows in comparison with non-wheeled furrows (Allen & Musick 1992). Using a narrow tyre width can avoid compaction of bed shoulders leading to enhanced lateral infiltration. Reduced vertical infiltration into furrow bottom and enhanced lateral infiltration into the wide bed middle can be helpful in improving *WP*, especially under low infiltration soils.

The severity of tillage impact depends on the environment and the duration of the particular tillage farming system application (Rhoton 2000). Most tillage practices have pronounced effects on soil hydraulic properties following tillage application, but these effects can diminish rapidly (Mark et al. 2008), while porosity and soil infiltration gradually enhances under the no till system (Azooz & Arshad 1996). Comparison of the main cotton pre-planting tillage options indicated a favourable root environment in the direct listed soil, excessive water entered the ripped soil thus caused waterlogging while chisel ploughing indicated intermediate behaviour (Hulme 1987). Similarly, intensive tillage leaves the soil exposed to structural decline, destroys stable bio pores and fauna (Smith et al. 1983) and exacerbates soil carbon and organic matter loss (Boulal & Gómez-Macpherson 2010). Hence, due to effects on soil physical, chemical and biological properties, tillage should be carefully applied.

Tillage is generally applied with the aim of loosening the compacted soil and enhancing soil porosity. Other reasons are control soil borne diseases, pests, pupae and slugs associated with certain crops genetic varieties (McBeath & McBeath 2010). Similarly, shallow rotary cultivation is generally practised in Pakistan to clear the weeds at harvest and to prepare a soft seed bed for facilitation in manual sowing of row crops (Akbar et al. 2007). Blade ploughing with no inversion has been shown to improve *WP* by enhancing lateral infiltration in wide beds (Hamilton et al. 2005; Jin et al. 2007; Akbar et al. 2010) in low infiltration soil. However, studies focussed on the impacts of different soil loosening methods under Vertisol for enhancing soil properties, irrigation efficiency, and crop production sustainably, are scarce.

2.3.1.2 Bed furrow configurations

Bed furrow configuration is variable throughout the world. Bed widths (furrow spacing) of 2 to 3 m are common in Australia, while 0.6 to 1.5 m width is common in China, India, Pakistan, Mexico and Bangladesh (Sayre et al. 2005). Currently selection of bed width depends on available machinery dimensions, soil type, crop type, crop rotation, and farmers' preferences. Soil infiltration properties, crop row spacing, and crop row distance from furrow centre generally governs the bed width. Bed height and furrow dimensions are generally affected by variable furrower tilt angle, speed of machinery operation, soil moisture at the time of bed construction, tractor power and depth of excavation (Hamilton et al. 2005). In season processes generally modify bed furrow configurations.

Variable furrow dimensions, water depth in furrow and furrow compaction can affect the soil moisture distribution between the wetted furrows (Tabuada et al. 1995). For instance, too deep furrows at early crop stage can reduce furrow water head and lateral infiltration into bed surface, which can be detrimental to densely grown crop establishment (Hassan et al. 2005). Similarly narrow furrows than the tractor tyre width can reduce infiltration due to compaction of bed shoulders (Hao & Yali 2012). Furrow shape has been shown to cause 45% water saving by using compacted V shaped furrows rather than uncompacted U shape furrows by Raine and Shannon (1996). Furrow dimensions change wetted perimeter and surface storage of water, which can affect irrigation performance of short furrows (Trout 1992). The management of bed furrow dimensions is important in the current raised bed performance. Wide beds have been shown to reduce deep drainage losses, increase the production of land under crop and reduce tractor power requirement per unit area (Jin et al. 2007). Wide beds have also been shown to save irrigation water and enhance yield. However, poor lateral infiltration under low infiltration soil limit adoption of wide beds due to deleterious effects on crops in the bed middle. Moreover, furrow dimensions can affect infiltration due to changes in wetted perimeter (Trout 1992; Oyonarte & Mateos 2003). Wetted perimeter has been shown to affect infiltration evidenced by findings of Izadi and Wallender (1985), who identified a 33% share of wetted perimeter in infiltration variation. However, adoption of optimum bed furrow dimensions is rare under actual field conditions due to lack of knowledge and absence of proper decision guidelines. Therefore, optimised management of bed furrow sizes has the potential to improve the current *WP* of raised beds by reducing the irrigation losses and increasing crop yield through improved lateral infiltration and increased land use efficiency.

2.3.1.3 Cropping management

Crop rotation is one of the important farming considerations for improving raised bed performance. It can affect soil carbon and nitrogen (Edwards et al. 1992; Govaerts et al. 2006) and can increase yield by increasing the root activities (Nickel et al. 1995; Shafiq et al. 2002). It can be used for controlling the microorganism detrimental for root growth (Johnson et al. 1992) and other soil-borne pathogens by adopting non-host crops or pasture species (Gardner et al. 1998) in rotation. However, the cropping frequency and rotation is currently largely driven by market conditions or other farmer preferences.

The soil cover and residue retention tends to reduce the water and wind erosion (Langdale et al. 1991; Ma et al. 2006) thus can also reduce furrow filling and maintenance. It prevents crusting or sealing formation on the surface of the soil (Baumhardt & Lascano 1996) and facilitates improvement in soil infiltration rate (Findeling et al. 2003). Soil cover absorbs excess rainfall energy, thus mitigating the rainfall splash effect leading to enhanced aggregate stability (Pinheiro et al. 2004). The crop residues also tend to enhance the soil organic matter (Yang & Wander

1999). However, retention of straw and crop residues is an economical decision, because of the associated loss in product income. This residue is normally utilised as fodder, fuel or raw material for paper products, especially in the developing countries. Therefore, Halvorson et al. (2002) recommended no tillage while Soon and Arshad (1996) recommended crop choice in rotation for improving organic matter to compensate for the potentially negative impacts of straw removal.

Plant population per unit area can affect crop yields. The land loss due to row spacing and furrow width can affect yield, whose severity depends largely on specific crop species and variety. Fischer et al. (2005) reported 10% yield loss with 45-55 cm furrow width for sensitive wheat cultivars and no loss for least sensitive cultivars, while yield loss of all cultivars of wheat above 60 cm furrow width was confirmed. They also termed light capture as one of the key factors that compensate for the yield loss due to furrow gap. Therefore, the current trend of getting more population per unit area or uncontrolled furrow width can reduce crop production potential levels. Thus, crop row spacing affected by furrow width can affect *WP* if not managed properly.

2.3.1.4 Waterlogging and salinity

Waterlogging is caused when there is insufficient air in the soil for the plant roots to access enough oxygen for them to respire. Waterlogging generally occurs due to upward movement of soil moisture from a high water table. However, excessive rainfall or over-irrigation can also produce a temporary or perched water table, especially under duplex soil conditions (Bakker et al. 2005) that can damage crops due to lack of aeration, while raised beds have been shown to control waterlogging.

Waterlogging and salinity are two different things but are closely related as interconnection has been identified in many regions of the world. Waterlogging can raise a saline water table close to the surface or can mobilise a zone of salinity in a soil where there is no shallow water table. Crops can tolerate a low level of salinity. However, salinity become visible once the concentration of ions in the soil solution exceeds crop threshold levels as reviewed by Tanji (1990). In both cases salt moves to the surface by a combination of convection (i.e. soil water flow) and, if there is a

concentration gradient within the soil-water, by diffusion. Convection is always the faster and more influential process. Therefore, waterlogging facilitates the transport of salt to the surface profile or root zone followed by evaporation leaving a concentrated layer of salts at the surface. High concentration of sodium salts in soil for longer period deflocculate soil colloids and degrade the soil structure (Levy & Torrento 1995). The high salinity and sodicity levels make the land surface less productive and in many cases, barren due to their negative impacts on soil structural stability, infiltration rate and drainage rates. Waterlogging and salinity have been identified as the major threat to many fertile lands, especially the Indus Basin (Qureshi et al. 2008). Poor irrigation management and use of poor quality water are the major reasons exacerbating waterlogging and salinity.

Raised beds are generally considered to exacerbate salinity due to large exposed surface areas that enhance evaporation. According to Oster et al. (1984), lateral infiltration is considered to increase the salt content of the centre portion of raised beds. However, a sandy clay loam soil with low salt content failed to show any accumulation in the centre of raised beds during nine years of monitoring salt levels (and nutrients), probably because limited rainfall redistributed any accumulation. Similarly, Bakker et al. (2010) could not identify any increase in salinity build up on raised beds in Western Australia. Therefore, minor salinity variations should not be a major concern in affecting *WP* of raised beds, while proper soil and irrigation management can reduce salinity by the leaching process.

2.3.2 Irrigation management factors

Irrigation performance is the ability of an irrigation system to apply water efficiently and uniformly according to plant and environmental requirements. Generally the term 'efficiency' is used as the ratio between input and output. In irrigation terminology it was further subdivided into different performance indicators. For instance, irrigation application efficiency (*Ea*) is the ratio of irrigation water available to the crop in the root zone and water received at field inlet (Purcell & Associates 1999) and requirement efficiency (*Er*) is the ratio of the volume of water stored in the root zone immediately after irrigation to the pre-irrigation root zone soil moisture deficit (Michael 1999; Raine & Smith 2007). The evenness of applied depth across the field is generally indicated by the distribution uniformity (DU), which is the average infiltrated depth of water in the lower one quarter of the field, divided by the average infiltrated depth over the whole field (Merriam & Keller 1978).

The irrigation factors important for irrigation performance are those that affect the amount and distribution of water in the field relative to the amount and availability of water and oxygen to plants, and the potential for water to be 'lost' to plants through deep drainage, surface runoff and evaporation. Hence, rate, volume and duration of flow need to be matched to the rate at which the soil is able to absorb water and the quantity that needs to be added to the soil to provide plants with adequate amounts and ease of extraction. The main irrigation factors are the field design factors of slope and length of furrow, as well as flow rates and duration of application that need to match the plant requirements and the soil's ability to absorb the water.

2.3.2.1 Inflow rate

Inflow rate (Q) refers to volume of water applied per unit time to a single furrow. Optimum Q is considered to be the one which enhances uniformity, minimises losses and suits the field conditions. Generally, Q is variable from farm to farm and even within the same farm, while optimum Q can hardly be found even in more advanced countries (Renault & Wallender 1992). The reasons for this are generally attributed to the variable delivery methods (e.g. siphons, gated pipes, or bank less channels) used for introducing water to the furrow inlet, which are largely subjected to variations due to variable outlet dimensions and pressure head in the supply channel (Smith 1990). The sensitivity of Q to these contributing factors can be illustrated by the investigations, which showed that decreasing the head by 50% resulted in 25% reduction in O, while a 1 to 5 mm change in the diameter of outlets changed O by 25% (Trout & Mackey 1988a). High variations in Q can suppress the seasonal trends in infiltration (Mailhol et al. 2005), while Schwankl et al. (2000) attributed up to 50% of variability in uniformity to Q variability between furrows. Thus, better management of these factors is essential for achieving optimum Q, which is generally constrained by field limitations.

Irrigation performance improvement using Q as a management tool can be applied in several ways. These methods are generally referred to as surge irrigation, cutback irrigation, increased inflow, and cablegation. However the majority of these traditional methods are labour intensive and are difficult to adopt under actual commercial farm conditions. Surface irrigation optimisation is another method currently based on using data of previous irrigation's events to calibrate a simulation model and to identify adjustments to Q and time to cut-off (*Tco*) for the next irrigation. Optimising Q and *Tco* has been successfully demonstrated for improved irrigation performance in many past researches (Raine et al. 1998; Smith et al. 2005; Smith et al. 2009). Nevertheless, it can also provide an opportunity of real time control of irrigation management as mentioned by Khatri (2007) and has raised the prospects of future furrow irrigation automation. This method still has limited field application due to equipment constraints, field limitations, and lack of knowledge due to limited training opportunities, but is suitable for testing and evaluation.

2.3.2.2 Time to cut-off

Time to cut-off (*Tco*) is the time period between switching on and switching off inflow to individual furrow. Optimum *Tco* is the time required to replenish the target soil moisture deficit (*SMD*) with increased efficiency and uniformity. It is generally affected by Q, furrow length, furrow slope, irrigation scheduling and soil infiltration characteristics. Lengthy *Tco* causes low *Ea* while reduced *Tco* may result in underirrigation and can cause plant water stress, especially in tail reaches of the field as pointed out by Khatri (2007) and thus demand more frequent irrigation applications.

The *Tco* decision is generally based on water arrival at the tail end of the field and/or furrow filling, especially under short furrows with blocked ends. However, for lengthy furrows, with open ends, *Tco* is generally affected by the seasonally variable infiltration properties of soil, which govern the water advance rate and consequently affect the *Tco*. Lack of optimum *Tco* under current furrow irrigation management for the majority of crops is affecting the overall raised bed performance. For instance, identification of optimum *Tco* under long Australian sugarcane and cotton fields has been identified as the key in controlling deep drainage and runoff losses (Raine & Bakker 1996; Smith et al. 2005). Similarly, the uncontrolled *Tco* under short

Pakistani blocked end furrows frequently cause over-irrigation and bed over-topping, which have exacerbated the waterlogging issues (Gill 1994). Currently, lack of decision support aids for optimised *Tco* management is affecting irrigation performance in many regions of the world.

2.3.2.3 Application depth and infiltration opportunity time

The depth of application depends on the irrigation deficit prior to irrigation determined by the irrigation schedule, which is generally assumed as fixed. Generally, applying less application depth will result in less opportunity time along the furrows during an irrigation event, thus demanding more frequent irrigation. However, applying less irrigation depth than required will enhance *Ea* (Sepaskhah & Ghahraman 2004) due to reduced potential for deep percolation and runoff losses. In contrast, decline in performance is more pronounced, when achievement of higher *Er* is attempted by applying more application depth (Bautista & Wallender 1993a). Therefore, application depth can affect irrigation performance.

Variability of irrigation application depth from head to tail end of the field largely depends on the Q, furrow length, slope and soil infiltration properties. All these factors control the distribution of infiltration opportunity time along the furrow length and the head of water in the furrows. The field head reaches is normally over irrigated to the extent of 30% to replenish *SMD* of major portion (80%) of furrow length (Trout 1990). In contrast, the short furrows, especially at high Q and/or with steeper slopes cause larger irrigation application at field tail than at field head reaches. Therefore, irrigation management should account for the spatial variability in infiltration depth for improved DU.

Variable infiltration opportunity time is generally held responsible for variable application depth along furrow length. However, variation of infiltration rate during a given infiltration opportunity time can significantly affect irrigation performance. For instance, Bautista and Wallender (1985) admitted the importance of the short infiltration opportunity time in irrigation performance before the steady infiltration rate is achieved while Tarboton and Wallender (1988) also recognised the effects of infiltration opportunity time in field variability of applied depth, but also mentioned

greater susceptibility of irrigation performance to variability in the infiltration rate, which is largely affected by the soil properties and soil antecedent moisture content at the time of irrigation.

2.3.2.4 Field design

Field length is one of the important field design components, which can markedly affect the irrigation performance. Theoretically, shorter field length will cause uniform water application and reduced losses provided all other variables remain constant. For instance, a reduction of 42% in irrigation water application was predicted by changing field length from 600 m to 300 m on a highly permeable alluvial soil (Raine & Shannon 1996). Longer furrow lengths, specifically under Australian conditions where the furrow lengths generally range from 600 to 1000 m, necessitate the need for intensive irrigation management to control the large deep drainage losses (Smith et al. 2005). Short furrows, particularly under the Indus Basin field conditions where the furrows generally range from 30-100 m length with blocked ends, are very sensitive to irrigation management factors.

The longitudinal slope of the furrow affects uniformity of water application due to its influence on advance and recession time periods. An increase in slope can slightly enhance the advance and recession rate provided Q and soil infiltration properties are fixed as evident from examining Manning's equation. Larger infield slope variations have been reported as one of the major contributing factors that negatively affected the existing irrigation performance and exacerbated the waterlogging and salinity issues in the Indus Basin (Gill 1994). Minor infield slope variations can be remedied by land levelling techniques. However, generally field slope is difficult to alter due to the huge cost involved with land forming operations. Therefore, field slope is often regarded as fixed in the design stage.

Furrow spacing can affect irrigation and crop performance. One reason is the rapid completion of the advance phase on wide spaced furrows in comparison with narrow spaced furrows for the same total available field Q. Similarly, total water applied to 132 cm wide spaced furrows was 17% less than 66 cm furrow spacing (Akbar et al. 2010) for wheat and maize crops under sandy clay loam. However, in cracking

Vertisol the effect of wide furrow spacing on water application would be limited due to high lateral infiltration. Experiments conducted under low infiltration soils showed up to 45 times greater root density of maize crop next to the watered furrows in comparison with unwatered furrows (Oliveira et al. 1998). Similarly, poor crop performance of centre rows was reported by Hassan et al. (2005) due to reduced lateral infiltration under sandy clay loam with 180 cm furrow spacing. Therefore, bed width needs to be carefully designed for maximising yield and *WP*.

2.3.2.5 Irrigation scheduling

Irrigation scheduling is a practice that periodically applies sufficient water to maintain the crop root zone near the field capacity for better crop growth (Wichelns & Oster 1990). The soil moisture generally vary between saturation and wilting point (McBride 2004) but it is generally not recommended for the soil moisture to drop closer to wilting point, which can damage crop. Therefore, the lower limit of soil moisture is kept higher than the wilting point, termed readily available water, which generally on crop variety and stage, soil type and water availability.

Irrigation scheduling is generally aimed to minimise irrigation costs, maximise crop yields, optimally distribute a limited water supply and optimise the production from a limited irrigation water (Martin et al. 1990). Irrigation scheduling can influence both irrigation and crop performance. For instance, larger *SMD* prior to irrigation under cracking Vertisols can reduce water advance rate. Similarly, Cavero et al. (2001) identified significant correlation between maize yield and application depth, while Wichelns and Oster (1990) highlighted impact of *SMD* prior to irrigation on cotton crop yield. Therefore, adequate irrigation scheduling can improve *WP*.

Notwithstanding that such advance techniques of irrigation scheduling are present (Jones 1990), implementation of a proper irrigation schedule is still a great challenge due to existing poor irrigation management techniques and field limitation related to unpredictable water availability constraints in many regions of the world, especially the developing countries. Consequently, improper irrigation schedules reduce crop production and exacerbate waterlogging and salinity problems globally.

2.4 Adaptation of conservation agriculture to raised beds

Efforts around the world are underway in exploring innovative ways for ensuring sustainable crop production. Intensive tillage negatively affects soil structure, fertility (McGarry 1993), crop residue distribution, surface roughness, soil temperature and roots distribution (Dwyer et al. 1996), thus enhancing soil erosion, organic matter loss and evaporation losses (Yang & Wander 1999; Six et al. 2000; Kibblewhite et al. 2008). In contrast, minimum/no tillage and crop residue retention enhance productivity on a sustainable basis (Hobbs 2007), because it can protect the soil structure, soil fauna and biota that accelerate biological processes for natural soil amelioration (Tisdall et al. 1978; Govaerts et al. 2008; McHugh et al. 2009). Therefore, agronomic management based on the principles of conservation agriculture (CA) is essential for improving *WP* on a sustainable basis.

CA involves minimum soil disturbance, maximum soil cover and crop rotation with the aim of protecting the soil and water from losses/degradation, to maintain the productivity on an economically, ecologically and socially sustainable basis (Unger 2006). It facilitates soil amelioration by natural processes of wetting and drying without damaging structure, biological processes of root channelling, crop residue reinforcing, and some time by minimum controlled mechanical loosening of soil (McKenzie & Chan 1990; Dexter 1991; Hamilton et al. 2005). It tends to enhance soil structural stability, water absorption, and availability of soil-water to plants. Minimum tillage is better facilitated under the controlled traffic farming (Tullberg et al. 2007), which can also efficiently manage surface drainage and ground cover.

Attempts have been made, particularly in Australia, to improve the management of soils by introducing conservation agriculture practices. These have substantially reduced the amount of cultivation and increased the likelihood of a build-up in soil organic matter and biological activity. The practice of cultivation for weed control has been marginally reduced (Werth et al. 2006) due to chemical weed control and the adoption of minimum tillage crop establishment has further reduced the amount of soil disturbance. Thus the practice of permanent raised bed (PRB) farming has come into vogue. PRB involves distinctly separating the field into cropping zone

(bed) and uncropped zone (furrow) and the furrows also serve as a machinery traffic path and water channel for irrigation and drainage.

2.4.1 Permanent raised bed farming systems

PRB farming systems consist of developing a bed-furrow system that is kept in place for several seasons with minor pre-sowing renovation operations (McGarry 1995; McKenzie 1998; Cooper 1999). Generally renovation is performed to restore bedfurrow configuration, to loosen the subsided bed and to control weeds, diseases, insects, slugs, pupae and pests (Wightman et al. 2005). Usually the frequency of remaking ranges from two to seven years (Cooper 1999) at commercial farms to as much as 18 years (Hulugalle et al. 2004) in experiments, which largely depends on soil and crop conditions. Minimum tillage, stable structure and soil cover under PRB conserves soil moisture for longer (Boulal et al. 2012), increase drainage (HulugalleA et al. 2010) and reduces losses, thus enhancing water saving (Jat et al. 2008). By combining several elements of conservation agriculture, PRB systems have been proved useful for controlling erosion, minimising soil structural decline (Verhulst et al. 2011) and reduction of evaporation losses (Hulugalle & Daniells 2005). However, irrigated PRB systems are still largely in their infancy and need to be more productive to achieve the anticipated benefits under wide environmental conditions, according to Roth et al. (2005).

2.4.1.1 Review of permanent raised beds in Australia

PRB system development in Australia gained momentum gradually since the 1970s. Martin Maynard, a pioneer PRB farmer in the Riverina of NSW, found he could increase his farm gross margin by up to 50% by adopting higher value rotation in conjunction with PRB farming system in the 1970s. Similarly, the preliminary work based on investigating the genetic and agronomic approaches for controlling the waterlogging together with the associated zinc deficiency symptoms in soybean in south east Queensland (Mungomery & Byth 1976) provides the initial background for the subsequent development of PRBs according to Borrell and Garside (2005), although this work was not aimed at developing the PRB system. Then, after exploring the benefits with soybeans, the research area on PRB was extended to rice and soybean in rotation (Borrell et al. 1991; Garside et al. 1992). Similarly, in an effort to develop a cropping system that will provide comparable economic return to sugarcane crops in the Burdekin river irrigation area, the double cropping system with PRB was preferred due to its short turnaround time requirement facility between crops in rotation (Borrell & Garside 2005). Studies conducted on double cropping of rice, maize and soybean on PRB in a rotation system from 1986 to 1991 resulted in recognisable advantages of improvement in saving water, nitrogen, phosphorus, timeliness of operation, energy and reduced soil compaction (Garside et al. 1992; Borrell et al. 1999; Ockerby et al. 1999), which acted as stimulus in dissemination of PRB. Similarly, in the Murrumbidgee/Murray valleys all summer non-rice grain cropping systems with an estimated area of 35,000 ha were undertaken using PRBs (Thompson 1995), while now the use of PRBs has been expanded to all irrigated maize, soybean, faba bean and chickpea planting, more than 30% of irrigated canola planting and 25 to 30% of all irrigated wheat plantings (Beecher et al. 2005). The results of a replicated trial on a range of crop sequences in a rice based farming system (Beecher et al. 2006) has shown improvement in ponding water, controlling weeds at crop establishment stage and enhancing WP by using lateral PRBs. Therefore, PRBs were adopted due to their inherent benefits, while field research related to improving its performance is scarce.

Cotton is one of the major crops in Australia that is grown largely in the irrigated areas of Queensland and New South Wales. Approximately 80% of the cotton crop is irrigated (Hulugalle & Daniells 2005), while furrow irrigation is the widely used method (> 94%), with a smaller area being irrigated with drip (4%) and sprinkler (< 2%) methods (Raine & Foley 2002). Cotton wheat rotation is followed by ~75-80% of cotton growers, where land preparation varies from intensive tillage (deep ripping, chiselling and bed reconstruction every year) to minimum till PRB systems (Schoenfi sch 1999). PRB farming systems with a cotton-wheat rotation under the irrigated grey Vertisol of NSW, Australia, showed improved soil structure, increased number of drainage and aeration pores, which resulted in better drainage and storage (Hulugalle et al. 2010) increased plant available water and oxygen than cultivated raised beds (Hulugalle et al. 2004). Similarly, after reviewing a range of field research, Hulugalle and Daniells (2005) concluded that PRB is a suitable option for cotton crops and the problems related to heliothis pupae control, fertiliser, weeds and

water management can be overcome by using appropriate irrigation and agronomic management practices. This study identified benefits of PRBs relative to cultivated beds but did not identify strategies for improved performance of irrigated PRBs.

Sugarcane crops are largely based on monoculture, while excessive tillage with heavy machinery is generally used, which causes severe soil compaction. The fluctuation in prices and yield compels the farmers to think of alternative production methods. Comparison of PRB with traditional methods has shown comparable sugarcane yields and reduced cost of production with improved earth worm activity, infiltration and yields in PRBs (Bell et al. 2003). However, according to Garside (2005) the development of PRBs in the sugarcane industry is still in its infancy but it seems that PRBs will soon become an integral part of the sugarcane cropping system.

The PRB and controlled traffic farming system developed for overcoming the waterlogging and improving the soil structure in high rainfall regions of South East Australia is a recent phenomenon. Around 70000 to 80000 ha of raised bed area was estimated by Wightman et al. (2005) in South Australia. In many parts of south-western Victoria with high water tables adopting raised beds have doubled the crop yields and considerably improved the farm profit (Wightman et al. 2005), while the use of organic amendments with PRB further increase the soil macro porosity and plant available water capacity (Peries & Gill 2011) leading to reduced runoff and greater root proliferation. However, adoption of PRBs is still in its early stage.

Winter waterlogging has been identified as affecting approximately one million hectares each of crop and pasture every year in Western Australia. This causes an average productivity loss of ~25% to 40% (McFarlane et al. 1992) due to perched water table in duplex soil, while PRBs successfully reduced winter water-logging and salinity (Bakker et al. 2005). The reasons were attributed to better PRB management practices that involved bed loosening using horizontal blades without inverting the soil, which resulted in porous and stable soil structure with greater resistance to wetting. These improvements were attributed to a deepened seed bed, increased root mass acting as reinforcing rods, and a significant proportion of larger pores (Hamilton et al. 2005).

The foregoing text showed the farming systems in Australia are still in transition from fresh bed (intensive cultivated) to PRB systems. The majority of the past studies on PRBs provided comparative performance to traditional practices. However, studies focussed on improving the agronomic and irrigation management of PRBs are sparse. Thus, it can be envisaged that performance of current PRB farming practices can be improved with better agronomic and irrigation management.

2.4.1.2 Review of permanent raised beds in Pakistan

The furrow bed irrigation method was introduced in the Indus Basin for growing vegetables and fruits (Berkout et al. 1997) in the late 1960's. The furrow bed system is generally considered a relatively efficient method of irrigation (Kahlown et al. 1998), which has now been adopted in over 40% of wheat-cotton rotation in the Punjab province of Pakistan. Some early work showed ~20% yield increase for wheat (Hameed & Solangi 1993) and ~48% yield increase for cotton (Berkout et al. 1997) just by changing from traditional flat basin to furrow bed system. Notwithstanding that furrow bed system is now being adopted for majority of row crops there are growing concerns about the cost of production (Alberts & Kalwij 1999), weed infestation, soil compaction, lateral infiltration and land degradation because the beds are generally regularly knocked down at the harvest of row crops. Greater interest has been shown to transform the current tillage intensive system.

The concept of PRB farming system is new in the Indus Basin of Pakistan. A joint research study of Pakistan Agricultural Research Council (PARC) and Australian Centre for International Agricultural Research (ACIAR) was conducted during 1999 to 2004 comparing traditional flat basin (intensive cultivated) with 180 cm wide PRB (minimum till: blade ploughing) using maize wheat rotation. Results showed considerable improvement in yield (26% for maize and 15% for wheat), water saving (32% for maize and 36% for wheat), *WP* (65% for maize and 50% for wheat) and reduced weed infestation (24% for maize and 31% for wheat) under PRBs compared with flat basin (Hassan et al. 2005). The reasons were attributed to better crop health on PRB due to improved soil physical properties (e.g. reduced bulk density, increased infiltration) and chemical properties. However, dry bed middle due to limited lateral water movement during normal irrigations was identified. Another

study 2004-2009 at the same site using 132 cm and 66 cm furrow spacing in comparison with traditional flat basin showed up to 13% increased yield of maize and ~5% of wheat on PRBs compared with flat basins. This study showed increase of ~126% for maize and 13% for wheat compared with first study. The *WP* was increased by up to 30% for 132 cm and up to 18% for 66 cm PRB than for the flat basins for both crops (Akbar et al. 2010). The subbing issue was partially reduced by smaller bed size but irrigation management issue was still reported.

2.4.1.3 Summary of research needs in both countries

This review of previous work on PRBs in both countries indicated that the majority of the past research compared PRBs with traditional methods and research focussed on improving the renovation methods and irrigation performance of PRB practices was largely ignored. Moreover, the majority of past research used traditional methods of point measurement for field evaluation, which were subjected to variations. Therefore, a comprehensive evaluation of the current performance of PRBs is required using large area measurement methods that should also be helpful in developing strategies for improving soil and irrigation management practices. Evaluation of different PRB renovation methods and their impact on lateral infiltration seems essential for enhancing performance of existing PRBs. Therefore, detailed assessment of the major factors affecting *WP* of PRBs, explored during this review, needs to be critically investigated for identifying strategies to improve the current performance of PRBs.

2.5 Permanent raised beds performance and assessment

Two variables play a major role in controlling the performance of PRBs for improved *WP*. They are the crop yield and the volume of water used during a cropping season. The crop yield can be improved with better soil hydro-physical, chemical and biological properties, a manifest of agronomic management. Irrigation application can be minimised by improving the irrigation performance by reducing the losses with optimised irrigation management. In reality, the complexity of two dimensional infiltrations under furrow irrigation and lack of decision support guidelines has made it hard to manage PRBs properly under actual field conditions. Consequently, irrigation performance is negatively affected.

2.5.1 Agronomic management performance and assessment

2.5.1.1 Soil texture and structure

Soil texture (the relative proportion of sand, silt and clay) strongly affects the infiltration capacity of the soil. The particle sizes of normal soil range from < 0.0002 mm for clay colloids to the larger sand grains that range from 0.5 to 2 mm and even some time large boulders (Singer & Munns 1999). The smaller sized clay particles play an important role by filling the voids between the sand and silt particles, while the charged surfaces bind the particles together. The hydraulic conductivity of a medium with single size particles has been identified proportional approximately to the square of the particle diameter (Iwata et al. 1995). Thus, due to the variable nature of soil type within the same field, regions of soil with sandy texture often have higher infiltration rates (Childs et al. 1993), while clay texture was correlated positively and silt negatively with the initial infiltration rate (Van Es et al. 1991). The soil texture cannot be changed by any agronomic management practices.

Structure is the arrangement of soil particles, generally attributed to soil properties of pore sizes, their distribution, continuity and inter connectivity (McGarry 1993). Soil pores consist of macro pores (> 0.075 mm) and micro pores (< 0.03 mm) (Singer & Munns 1999). The soil pores are variable which largely depend on the biological activities, shrinking swelling, temperature, soil moisture and cultivation effects (Lal 2004). According to Cook et al. (1992), the cohesive forces between particles and the frictional resistance met by the particles impart strength to the soil matrix. Structured soil is more stable, in agricultural terminology, due to flocculation of clay particles and micro aggregates stability, which is enhanced by roots, microbial, fungal and bacterial exudates (Dexter 1991; Tisdall 1994; Czarnes et al. 2000) that also tend to enhance soil infiltration (Flury et al. 1994). The soil structure has been shown to gradually improve with the imposition of PRB farming systems.

2.5.1.2 Soil moisture

Soil management plays a key role in conserving soil moisture. Intensive tillage has been shown to uncover the underlying moist soil thus increase soil moisture loss due to evaporation. Conversely, PRB farming system with good soil cover and minimum soil disturbance conserve soil moisture and thus reduce irrigation requirement. It also helps in protecting the crop from water stress at critical crop stages. Similarly, tillage at low soil moisture level can promote clodding and surface disturbances while moist soil promote smearing and reduced infiltration. Therefore, optimal soil moisture during seed bed preparation is essential for improved performance.

There are two main conventional methods of soil moisture measurement (Jones 2004). The first one is based on the direct measurement of soil moisture status in the root zone and the second one is based on soil water balance calculations, which require detailed methods of estimating evapo-transpiration and water requirements for different crops and environments as reviewed by Allen et al. (1998). Further methods have also been discussed for future development by Jones (2004) that are mainly based on the sensing of plant stress, which include both water status and plant response measurement. However, these methods are generally very sensitive to environmental conditions that can cause fluctuations sometime greater than treatment difference and often require sophisticated and costly equipment.

The thermo gravimetric method is the standard soil moisture measuring technique, which requires oven drying at 105°C for 24-48 hours and determining the weight loss (Walker et al. 2004). However, this method is time consuming, labour intensive, destructive to the soil sampled, and cannot be used for repetitive measurements at the same location, but is still indispensable for calibration and evaluation purposes. There are a wide range of automotive indirect soil moisture measuring techniques including neutron scattering, gamma ray attenuation, soil electrical conductivity (based on electrical resistance and electromagnetic induction), hygrometers (capacitance, infra-red absorption and transmission, psychometers), tensiometers (based on capillary tension or soil matric potential) and that based on soil dielectric constants (including capacitance meters and time domain reflectometer-TDR). Reviews on the use, comparison and suitability of these techniques is given in Topp (2003) and Walker et al. (2004).

There are a wide range of commercially available point soil moisture measurement techniques in Australia. However, the capacitance based technology of (Sentek EnviroSCAN[®]), diviner and micro-gopher are used at the National Centre for Engineering in Agriculture (NCEA). The sensors are fitted into a probe at specific desired depth (usually multiples of 10 cm) that is inserted into a specific size access tube installed in the soil profile to the target profile depth. The sensors work on the principle of producing a high frequency electric field around the sensors that extends through the access tube into the soil; the sensors detect the changes in the dielectric constant or permittivity of the soil over time, which varies inversely with the amount of water present in the soil. The scaled frequency readings obtained for a specific soil texture are converted by using a default calibration equation. The output comes in the form of volumetric moisture content in 10 cm profile depth. The sensors can be attached to an automatic solar powered logging facility, which can store a complete record of soil moisture over an extended period of time with the measurements taken at flexible time intervals. These sensors have been successfully tested and used (Mead et al. 1995a, 1995b; Alva & Fares 1998; Starr & Paltineanu 1998; Fares & Alva 2000) but were also identified sensitive to soil type, temperature and soil salinity (Baumhardt et al. 2000) thus require careful installation and calibration.

2.5.1.3 Bulk density

Bulk density is the ratio between mass of the solid material and the volume it occupies, which is generally expressed in (gm cm⁻³). Thus for soil particles with fixed particle density the bulk density is inversely proportional to the porosity. Mixed results were achieved during attempts to establish a distinct relationship between infiltration properties of soil either with pore sizes distribution (Baker 1979) or with bulk density (Jaynes & Hunsaker 1989; House et al. 2001) due to the complex interactions between other large numbers of soil properties. Increase in bulk density due to compaction caused by grazing animals, machinery traffic and unfavourable agronomic management were reported under wide environmental conditions by Shafique and Skogerboe (1983) and Allen and Musick (1992). Similarly, gradually reduced bulk density were reported (Hassan et al. 2005; Akbar et al. 2009; McHugh et al. 2009) by imposition of PRB farming systems in different soils.

Bulk density variations can affect irrigation performance. The compacted trafficked furrows with high bulk density and non-trafficked loose furrows with low bulk density affect the water advance rates sometimes to the extent of 45% (Allen & Musick 1992) in the same irrigation, which often complicates the traditional irrigation management practices. However, bulk density indicative of variations in organic matter (Gupta & Larson 1979) and soil looseness in the manageable crop root zone under variable tillage systems can be beneficial in illustrating their agronomic performance and in comparing various agronomic management methods. Bulk density results are often inferred to soil porosity, water holding capacity and sometimes plant growth (Arya & Paris 1981).

Bulk density generally varies between less than 1 gm cm⁻³ to 2 gm cm⁻³ for normal soils, which can be determined in the field by oven drying, at 105°C for 24-48 hours, a soil sample of known volume. The ratio between the dry soil weight and initial volume sampled is the bulk density. Volumetric soil moisture can be determined by multiplying the gravimetric moisture content with the bulk density. The relationship between bulk density and porosity can be expressed as follows:

$$Porosity = \left(1 - \frac{\rho_b}{\rho_s}\right) * 100 \qquad 2.1$$

where ρ_b is bulk density in gm cm⁻³ and ρ_s is particle density in gm cm⁻³.

2.5.1.4 Hydraulic conductivity

Hydraulic conductivity is the ratio of the flux to the hydraulic gradient, which is affected by soil properties such as texture, structure, porosity, pore size distribution, their tortuosities, pore geometry and initial moisture content (Hillel 1982). Intensive farming systems exacerbate the soil subsidence/densification process during a crop season, that tends to convert macro pores into micro pores and the inter-connectivity and/or continuity is considerably reduced, thus hydraulic conductivity also tends to reduce as mentioned by Shipitalo et al. (2000). Therefore, supply of soil moisture to root zone to replenish *SMD* during an irrigation event also reduces. Consequently, crop yields are negatively affected leading to low *WP*.

Minimum soil disturbance under PRB system reduce soil aggregates disintegration (Adem et al. 1984) thus enhance soil hydraulic conductivity. Conversely, intensive

tillage and no residue retention accelerates surface crusting/sealing, subsidence, slaking, erosion and silting that negatively affects hydraulic conductivity, infiltration and sorptivity. The hydraulic conductivity of surface seals can reduce up to ~1-8% of the underlying soil and can cause 46% reduced infiltration (Segeren & Trout 1991).

The hydraulic conductivity can be determined in the field, in laboratory or using mathematical models. The field method is more representative but costly and time consuming. Laboratory and mathematical techniques are more convenient but require extensive comparison to field results.

2.5.1.5 Lateral infiltration into bed

Lateral infiltration (subbing) is influenced by the available hydraulic gradient. Hydraulic gradient is a combination of gravity, pressure head, osmotic and matric potentials (Singer & Munns 1999). The soil matric potential/suction plays an important role in soil moisture retention and sorptivity, which is vital for plant growth. The suction causes the beds to wet slowly, thus the pores remain stable for a longer period, while better control is maintained over soil moisture characteristics and biological activities (Adem & Tisdall 1984), which facilitates better root environment and lateral infiltration. However, intensive tillage negatively affects the lateral infiltration process.

Intensive tillage and operation at sub-optimal soil moisture can promote subsidence and compaction leading to reduced lateral infiltration. Poor lateral infiltration problems were reported under loamy soils by Ma and Wang (2005), Jin et al. (2007) and Akbar et al. (2007), under sodic Vertisol by McKenzie and Chan (1990), under red brown soil in the Murray Valley by Thompson and North (1994), and in black Vertisol by Lucy (1993) always where soil is unstable to wetting. Tillage at variable soil moisture content has been shown to affect soil structure and fabric (McGarry 1989) evidenced by up to 17% area of striated clay in wet cultivated soil compared with < 1% area in dry cultivated or land prior to cultivation. Limited controlled mechanical loosening using different kind of cutters/blades, that loosen the surface soil in the bed (zone 3 Figure 2.1) without inversion, have been shown to improve lateral infiltration, surface drainage and *WP* (Hamilton et al. 2005; Jin et al. 2007; Akbar et al. 2010). Soil loosening using horizontal blade involves slicing, lifting and dropping of soil over a height difference of 1-2 cm, which were shown to opens soil with zero soil inversion and maximum root retention (Hamilton et al. 2005). However, studies on lateral infiltration from furrow to the centre of bed are scarce.

The soil moisture sensors with loggers placed on bed at different distances from the furrow centre and inserted to target depths at desired depth intervals can be used to log the wetting front movement into the bed profile across wide bed widths during an irrigation event (White 2007). Evaluation of the lateral infiltration under variably managed PRB can be instrumental in identifying its impact on PRBs performance.

Lateral infiltration is affected by the sorptive behaviour of soil, which is driven by the capillary forces. Sorptivity is a measure of the capacity of the medium to absorb and desorb liquid by capillarity (Philip 1957a). Philip (1969) defined sorptivity by the following relationship:

$$S = \int \varphi \, d\theta = -2 D_f d\theta / d\varphi$$
 2.2

where *S* is sorptivity (mm hr^{-0.5}), θ is volumetric moisture content (mm³ mm⁻³), D_f is diffusivity (mm² hr⁻¹), and $\varphi = x t^{0.5}$ is the Boltzman transform (mm hr^{-0.5}) for describing phenomena that progress in proportion to the square root of time t (hr); and D_f = the hydraulic conductivity at a given moisture content (θ) times the slope of the moisture characteristic (soil moisture potential(Ψ)/moisture content (θ)) at the same moisture content. This analysis of infiltration by Philip (1969) led to the derivation of the simple equation 2.3.

$$Z = S\sqrt{t}$$
 2.3

where Z is the cumulative infiltration, S is sorptivity and t is time.

The soil intake characteristics were shown as the main source of irrigation depth variability (Oyonarte et al. 2002), indicating increased lateral infiltration is possible through increased soil intake using improved furrow bed management.

Apart from field experiments on lateral infiltration, the Hydrus 2D model (Šimunek et al. 1999) can be used to reproduce the results, which can be used for bed-furrow design and irrigation management optimisation. Hydrus 2D is based on the solution of Richards equation (Richards 1931) and has been broadly used in salinity and water movement studies. The Hydrus 2D model satisfactorily quantified soil moisture and solute distribution in PRB of various dimensions in Australia, Pakistan, China, Indonesia and India (Cook et al. 2008), which was helpful in developing guidelines for optimising furrow bed management under different soils. White (2007) described Hydrus 2D is able to adequately predict soil conditions in furrow bed system for short period < 26 days but the prediction capability may decrease due to compounding errors in soil moisture prediction. The Hydrus 2D simulation can be helpful in evaluating the optimum bed furrow dimension, furrow water head and infiltration opportunity time for improved lateral infiltration. It can be helpful in assessing soil moisture distribution under variable PRB renovation methods.

2.5.1.6 Surface drainage and stability during wetness

Removal of waterlogging and improved aeration is necessary for plant growth and better yield. Agronomic management practices affect waterlogging and surface drainage due to its effects on soil structure and infiltration characteristics. Under poor soil structure, which is normally caused by excessive tillage and traffic (McGarry 2003), there are few large stable pores to supply enough oxygen for the plant roots to quickly mitigate the effects of waterlogging. The prolong waterlogging reduces soil aeration and structural stability, which enhance soil subsidence. The subsided and structurally degraded soils are not capable of hosting and flourishing the beneficial microbial, bacterial ecosystem of micro-organisms necessary for achieving improved performance on a sustainable basis. Therefore PRBs are beneficial, because it generally enhance surface drainage (Hamilton et al. 2005), as the furrows can be used to act as drainage channels for disposing of the excess water.

Agronomic practices that strengthen the cohesive forces between soil particles by accelerating soil fungal and bacterial activities and their exudates (Smith et al. 1983) can be helpful in improving the stability of soil structure. A stable soil structure can enhance soil porosity and infiltration for longer. Moreover, the wet/dry cycles can be

used for repairing soil structure of compacted Vertisol (Samrah et al. 1996). All these processes can enhance surface drainage and soil stability during wetting.

Soil bulk density variation can be used to indicate soil structure stability but the variable nature of some soil e.g. Vertisol, due to shrink swell properties; can impede the accuracy of measurements, which can be overcome by extensive sampling. Similarly, soil resistance to deformation can also be related to seasonal variations in bed furrow dimensions. The extent of temporal variability in bed furrow dimensions during a cropping season illustrates soil resistance to wetting and deformation, which can also be helpful for comparing different agronomic management methods.

2.5.1.7 Seasonal changes

During a cropping season the land surface is subjected to wetting, flowing water, rainfall hammering, slaking, erosion, silting, sealing and compressive traffic impacts, which affect soil infiltration, aeration and moisture retention properties leading to subsidence (Hillel 2004). Subsidence can largely affect soil hydro-physical characteristics, while large soil fauna entry into small pores is also negatively affected (Brussaard & Van Faassen 1994). The severity of variations in all these processes during a cropping season largely depends on the type of tillage system (Coutadeur et al. 2002) and extent of managing surface cover or crop residues. However, the existence of current poor agronomic management under the majority of raised beds, specifically in developing countries, generally negatively affects the soil hydro-physical and chemical characteristics. Consequently, the productivity potential of soils is at low levels. Therefore, seasonal changes in soil hydro-physical properties need careful consideration in the design and management of raised beds.

Soil infiltration characteristics can vary temporally between irrigations (Rahimi 2011) and spatially between furrows (Gillies et al. 2008). The soil intake characteristics were shown to cause greater variability in infiltrated volume than that caused by variable intake opportunity time from field head to tail sections (Childs et al. 1993); (Oyonarte et al. 2002). Therefore, furrows and beds needs to be carefully managed keeping in view soil infiltration variability in two dimensions and their possible detrimental effect on irrigation performance.

Various soil properties have contrasting temporal and spatial scales of variations. Some soil properties that are responsive to management can easily be measured while others are impractical because of the large spatial variability and cost of measurement. Thus, adopting proper tactics for measurement is crucial to soil property monitoring procedures. Large changes in soil properties within a short spatial boundary have been identified during several studies (Wilding & Drees 1983; Burrough 1993). Selection of a proper method and frequency of soil change measurement depends on the objective of the study, understanding of the system behaviour, pattern of variations, measurement technology and resources (McKenzie et al. 2002). The nature of these highly variable soil properties during a cropping season genuinely requires field measurements recorded during or close to irrigation events using a representative soil sample of the whole field area. The seasonal variability in soil hydro-physical properties can be monitored by employing some inexpensive technologies or can be indirectly predicted by using combination of field measurements and simulation modelling techniques.

2.5.1.8 Crop yield

The ultimate purpose of optimum agronomic management is to get higher crop yield on a sustainable basis. Crop yield is mainly affected by soil fertility, soil structure, organic matter, soil moisture, weather conditions, pests, and diseases. However, agronomic management factors like tillage, bed width, crop variety, row spacing, and amount of inputs can also affect crop yields. For instance, no tillage produced up to 20% enhanced canola seed yield in comparison with conventional intensive tillage (Malhi et al. 2006) while (Phillips et al. 1980; Sayre et al. 2005; Govaerts et al. 2007) identified stubble retaining on PRB to improve organic matter, aggregate stability and reduce salinity leading to increased crop yield. Similarly, evidence of variable impact of *SMD* prior to irrigation on different crop yields has been reported by Wichelns and Oster (1990), Cavero et al. (2001) and Paz et al. (2001), who showed higher crops yields for soil moisture levels between field capacity and the refill points. However, some crops, e.g. cotton, can also enhance yield under stressed conditions (Morgan & Lowery 2003). Therefore, optimum agronomic management can improve crop yield on a sustainable basis. Limited controlled tillage without causing significant disturbance in the bed can be beneficial in removing compaction, loosening the soil and enhancing the lateral infiltration, particularly in structurally unstable soils. Better crop yields (5%) and improved *WP* (38%) were reported by Jin et al. (2007) by loosening the soil with horizontal blades without inverting the surface soil in comparison with traditional methods of cultivating raised beds. Similarly, up to 13% enhanced wheat yield and 50% enhanced *WP* were reported by Hassan et al. (2005) by using blade plough with horizontal blades in comparison with traditional flat basin systems under sandy clay loam. The reasons were mainly attributed to enhanced lateral infiltration in these soils of low infiltration capacity. However, the impact of blade ploughing on crop yield and *WP* of PRBs under cracking Vertisol soils is unknown. Therefore, crop yield can also indicate the effectiveness of agronomic management, which is easily measurable under actual field conditions.

2.5.2 Irrigation management performance and assessment

Raised beds with furrow irrigations are considered a relatively efficient surface irrigation method, because only the furrow base plus a small portion of the bed shoulders are wet by free water, compared with the whole field surface in basin irrigation. Also, because the surface area of soil in contact with free water is less and the depth of flow is greater, the advance times are much reduced. Notwithstanding these inherently better attributes of raised beds, there have been reports of poor irrigation performance (Gill 1994; Raine & Bakker 1996; Smith et al. 2005). The major reasons were attributed to poor irrigation management. Simulation models have been successfully used to identify optimum irrigation management options.

2.5.2.1 Models to assess irrigation performance

There are several simulation models based on the Saint-Venant equation (Khatri 2007) used to evaluate irrigation performance. These models also help to optimise the design and management factors (section 2.3.2) for enhanced irrigation performance. The Saint-Venant equation is based on the principles of conservation of mass (continuity) and momentum. The continuity part of the Saint Venant equation can be expressed by equation as follows (Walker & Skogerboe 1987):

$$\frac{\partial Q}{\partial x} + \frac{\partial A}{\partial t} + I_x = 0 \qquad 2.4$$

where Q is the inflow rate (m³ s⁻¹), A is the cross sectional area of flow (m²), x is the distance (m) and I is the infiltration rate (m³ m⁻¹ s⁻¹).

The expression for the momentum/motion part of the Saint-Venant equation can be expressed by the following equation (Walker & Skogerboe 1987):

$$\frac{v}{g}\frac{\partial v}{\partial x} + \frac{v}{gA}\frac{\partial Q}{\partial x} + \frac{v}{gA}\frac{\partial A}{\partial t} + \frac{1}{g}\frac{\partial v}{\partial t} + \frac{\partial y}{\partial x} = S_0 - S_f \qquad 2.5$$

The term *v* is the flow velocity in (m s⁻¹), *g* is acceleration due to gravity in (m s⁻²), *y* is the depth, S_0 is the furrow slope, and S_f is the friction slope.

Different assumptions were applied to the Saint Venant equations to make it simple. For instance, the zero inertia model neglects the inertia and acceleration terms in the momentum equation 2.5, which reduces the computational requirements (Alazba 1999) and works well under free flow conditions (Schwankl & Wallender 1988). Kinematic wave models further simplify by assuming the change in energy with distance for a constant Q equal to the slope of the furrow. However, it may only work correctly under fields with considerable slope (Gharbi et al. 1993). The volume balance model utilises the continuity equation (2.4) only and neglects the momentum term completely. The full hydrodynamic model is the most accurate, and is based on both continuity and the momentum equation, thus involving rigorous computation. However, the recent development of processor speeds and software has made it a favourable choice for irrigation evaluation and optimisation. Hence, this model is generally preferred for research purposes.

Irrigation performance assessment under furrow bed system is generally complicated by the two dimensional infiltrations that consist of vertical and horizontal components. Gravity forces are dominant in the vertical infiltration, while suction/capillary forces dominate under horizontal/lateral infiltration. There are many physical and empirical equations for describing the amount and rate of infiltration. However, the Kostiakov-Lewis variant or modified Kostiakov infiltration equation is more widely used, because it can provide adequate prediction of the water advance and cumulative infiltration depth (Hanson et al. 1993). The modified Kostiakov infiltration equation (Walker & Skogerboe 1987) is given by the following equation:

$$Z = Kt^a + f_0 t 2.6$$

$$I = akt^{a-1} + f_0 2.7$$

where Z is the cumulative infiltration, I is the infiltration rate, a and k are empirical constants, and f_0 is the empirical parameter that shows a close relationship with final infiltration rate of the soil.

Infiltration characteristics/functions of soil are very important parameters in evaluation, design and management of furrow irrigation. The water advance, depth of infiltration and uniformity of application during irrigation can be identified using infiltration characteristics (Jobling & Turner 1973). There are numerous methods of point infiltration determination including blocked furrow infiltrometer (Bondurant 1957), furrow by-pass infiltrometer (Shull 1961), double ring infiltrometer described by Tricker (1978) and Bouwer (1986) and flow through furrow infiltrometer (Childs et al. 1993). However, due to the spatially and temporally variable nature of infiltration, the point measurement (small area) methods often fail to provide the true overall field infiltration characteristics (Elliott & Walker 1982). One simple method of cumulative infiltration determination can be through determining the difference in pre- and post-irrigation volumetric moisture content by using the gravimetric method or capacitance and neutron probes. One of the drawbacks of such methods can be the inability to determine the water infiltration below the measuring zone (Shepard et al. 1993). Similarly Elliott and Walker (1982) failed to simulate the water advance or runoff using several point infiltration methods. Therefore, point measurement methods are not the preferred options to estimate infiltration in surface irrigation. However, methods involving large areas are considered more representative of the actual field conditions (Esfandiari & Maheshwari 1997), which are largely based on the volume balance approach and were recommended by many (Christiansen et al. 1966; Reddell 1981; Bautista & Wallender 1993b; Gillies & Smith 2005).

Substantial work has been directed recently to determine the spatial average of infiltration characteristics, which has been considered essential for the accurate simulation of surface irrigation. Various methods/models mainly based on the solution of inverse problems for determining infiltration parameters from measuring the water advances during irrigation have been developed. The inverse problems generally use equation 2.6 to determine soil infiltration and full hydrodynamic or volume balance model to determine distribution of water along the furrows.

The empirical constants of the modified Kostiakov equation are generally estimated by matching field measured data obtained during irrigation with the results of a simulation model using the inverse solution approach. Various methods of collecting water advance data in the field have been identified. The two point method (Elliott & Walker 1982) is considered a standard in the irrigation industry, as it performed better for 40 cm to 60 cm wide furrows (Holzapfel et al. 2004). It assumes that the advance curve can be approximated by a simple power function. This method requires separate calculations for identifying f_0 . Simulation models like INFILT (McClymont & Smith 1996) based on same basic equations of the two point method, but does not require separate calculation for the steady intake rate term f_0 , which was successfully used in the Australian sugarcane (Raine et al. 1998) and cotton (Smith et al. 2005) industries. However, INFILT treats surface storage as a fourth parameter, which causes sometime unreliable results. The IPARM (Gillies & Smith 2005) model is based on modification to the volume balance model to include runoff data and account for Q variability.

IPARM

The IPARM (Gillies & Smith 2005) model utilises the inverse procedure with volume balance model. Equation 2.6 is modified by introducing a constant factor C to account for cracking. Thus the equation 2.6 is modified to equation 2.8:

$$Z = kt^a + f_0 t + C 2.8$$

The advance distance along the furrow is estimated by rearranging the volume balance continuity equation into the following form:

$$x = \frac{Q_0 t}{\sigma_y A_0 - \sigma_{Z1} k t^a + \sigma_{Z2} f_0 t + C}$$
 2.9

where σ_y is the surface storage shape factor, which was identified to be in the range of 0.7 to 0.8 by different researchers, σ_{Z1} and σ_{Z2} are subsurface storage coefficients for the advance phase (Walker & Skogerboe 1987), where σ_{Z1} was assumed to be equal to 0.8 by Reddell and Latimer (1987). Both coefficients can be calculated as:

$$\sigma_{z1} = \frac{(a+r(1-a)+1)}{(1+a)(1+r)} \qquad 2.10$$

$$\sigma_{z2} = \frac{1}{(1+r)}$$
 2.11

where r is a fitted parameter, which can be determined from two advance points by simple logarithmic transformation of the power curve equation.

IPARM replaces the $Q_0 t$ term in equation 2.9 with the integral of inflow hydrograph $(\sum Q(t)dt)$ for each advance and runoff time point to account for variable Q. The input data for IPARM includes Q, slope, furrow length, furrow geometry, Manning's n and advance data at multiple points. IPARM can process multiple advance data, which enhances the accuracy of simulation. However, it increases the field data collection requirements. The output consists of infiltration functions $(a, k \text{ and } f_0)$, advance trajectory (measured and simulated) and cumulative infiltration with time curve. The IPARM version 2 (Figure 2.2) can create automatic SIRMOD (Walker 2003) input files and infiltration parameters. The infiltration functions and other information can be directly exported as a SIRMOD file, which can be retrieved directly by a SIRMOD model for irrigation performance evaluation. The IPARM model was successfully used by Gillies (2008) and Khatri (2007) for Vertisol.
Chapter 2: Review of PRB farming system performance



Figure 2.2: Screen shot of IPARM version 2 main user interface

SIRMOD

There is a range of available irrigation performance evaluation models but SRFR (Clemmens & Strelkoff 1999) and SIRMOD (Walker 2003) have gained major recognition. The SRFR model employs the zero inertia or kinematic wave models. It is capable of accommodating variations in Q and furrow dimensions, field slope and soil infiltration. SIRMOD (Walker 2003) was developed at Utah State University, US, and utilises the full hydrodynamic model i.e. both Saint-Venant equations. In case of unstable solutions with full hydrodynamic model, it also has the additional facility of applying the zero inertia or kinematic wave models. It utilises the modified Kostiakov infiltration equation with an optional term C (cracking factor) for describing infiltration rate. Simulation and evaluation with SIRMOD is event based and its accuracy depends on input data mainly for the modified Kostiakov infiltration parameters. The input data include field length, slope, infiltration functions, *SMD*, Q, Manning's n, furrow geometry, and furrow spacing. The output includes advance-recession curves, water infiltration along the furrow, run-off hydrograph, *Ea*, *Er*, *DU*, deep percolation fraction and volume balance as illustrated in Figure 2.3.



Figure 2.3: Screen shot of SIRMOD version III main output screen

SIRMOD can simulate furrow, border, or basin irrigations with open or closed end boundary conditions. It can simulate continuous flow with cutback and surge flow. It can test different management scenarios. For instance, it is capable of evaluating variable Q and modifying the infiltration functions accordingly by using a scaling factor F based on changes in furrow wetted perimeter, as illustrated below

$$f = \left(\frac{WP_2}{WP_1}\right)^b \qquad 2.12$$

where WP1 and WP2 are the wetted perimeters (m) when inflow rates are Q_1 and Q_2 , respectively and *b* is an empirical constant, which depends on field conditions.

The SIRMOD was extensively used for improving irrigation management and design in the Australian irrigation industry (Raine et al. 1998; Dalton et al. 2001; Smith et al. 2005) and under wide environmental conditions. SIRMOD is commercially available in Australia through IRRIMATETM, and consists of software and hardware tools developed by the National Centre for Engineering in Agriculture (NCEA) at the University of Southern Queensland (USQ), Australia. However, SIRMOD assume uniform irrigation application between the wetted furrows thus did not account for the lateral infiltration.

2.6 Conclusions

The increasing global requirement for food and fibre demands to produce more with less water on a sustainable basis. PRB farming practices have the potential to enhance productivity but are still in their infancy, while their current management is poor leading to negative impacts on their performance. This review revealed the possibility of improved *WP* of existing PRB farming systems through better agronomic and irrigation management practices. These are the key conclusions:

- Available research on the management of irrigated PRB for improved irrigation performance and WP is scarce, especially under Pakistani field conditions. Therefore, evaluation of existing PRB under commercial farm conditions can be helpful to benchmark their current irrigation performance and to refine the key contributing factors to be pursued in this research.
- Improved soil management practices are required to optimise root zone conditions if production is to be increased and sustained. Renovation methods of PRB that offer the best prospects for loosening subsided beds in a way that is consistent with conservation agriculture needs to be investigated by assessing its impact on crop production in PRB.
- Impact of blade loosening without inversion on soil structure stability to wetting and lateral infiltration in comparison with existing methods needs to be investigated in soils of various textures to ascertain its impact on furrow irrigation efficiency and WP.
- Furrow irrigation performance models offers the prospect of identifying the optimal application practices for soils of different textures and structures under variable soil loosening practices.
- The quantification of the lateral infiltration into beds with different textures and structures provides an experimental data set that can be used to verify the predicting capability of furrow irrigation performance models.

CHAPTER 3: General material and methods

3.1 Introduction

This chapter provides the research plan, describes all the study sites and details the general field conditions encountered during the experiments. Additionally, the general assessment methods that were applied across all, or most, sites during the three field studies and a desktop study are also presented in this chapter. However, details of the work of each experiment were expanded, as necessary, in their specific chapters.

3.2 Research plan

As per the findings in the foregoing Chapters 1 and 2 this study was planned to evaluate two main factors (irrigation and soil management) for improving the water productivity of raised bed systems. A brief research plan to be pursued in this study is outlined in Table 3.1.

Deceenab plan	Improve irrigation	efficiency and productivity		
Research plan	Irrigation management	Soil management		
What to do	Optimise inflow rate (Q) , time to cut-off (Tco) and field length	Improve soil structure stability by practices consistent with conservation agriculture (No tillage and blade loosening and furrow reshaping)		
How to achieve it	Quantify existing practices and use modelling (IPARM & SIRMOD) to identify potential improvement possible in irrigation practices with existing soil management	Quantify improvement in irrigation performance, lateral infiltration and <i>WP</i> with improved soil structural stability with different renovation methods (no tillage, shallow cultivation and blade loosening)		
Analysis	Identify from field matched modelling (a) Best combination of <i>Q</i> <i>Tco</i> and furrow length & (b) Assess practical achievability of the optimum set of irrigation practices with existing practices	Identify from field matched modelling (a) Best combination of <i>Q</i> , <i>Tco</i> and furrow length with no tillage, shallow cultivation and blade loosening & (b) Assess practical achievability of the optimum set of irrigation practices with different renovation methods		

 Table 3.1: Research plan for improving water productivity of raised beds

3.3 Description of field sites

The three components of this research were conducted on farms in Australia and Pakistan. The first component was related to benchmarking of existing PRB farming systems (objective 1), which was conducted at five sites, two in Australia: sites A1 and A2, and three in Pakistan: sites P1, P2, and P3. The second component was related to evaluation of irrigation performance and *WP* of three PRB renovation techniques (objective 2), which was conducted over three cropping seasons (winter 2010 to summer 2012) at site A1, in Australia. The third component evaluated lateral infiltration under three PRB renovation methods (objective 3) conducted at sites A1 (Australia) and P4 (Pakistan). The different soil properties of site A1 and P1 are given in Appendix A: Table A1 and Table A2 respectively.

The research sites were located in south east Queensland, Australia and north west Pakistan. Both these locations were selected on the basis of resources availability, proximity to working stations, liaison with cooperative farmers and availability of sites where PRB farming systems were actually practised. All field trials were conducted at farmers' fields in both countries. All the trial sites had irrigated PRB farming systems and growers had most farm equipment required for establishing the different treatments for this study.

South east Queensland, Australia is one of the most productive parts of the country, where irrigated agriculture is practised due to available water resources and subtropical climatic conditions. The major source of irrigation water in south east Queensland is from rain water harvesting and quite often supplemented by ground water bores. However, irrigation water usage is generally restricted due to tough government rules that vary spatially and temporally according to changing ground water level and quality.

The study sites in north west Pakistan were in the Mardan district of Khyber Pakhtunkhwa Province. Irrigation water in Mardan is from the northern mountains snow melt, which is carried through the Swat River and diverted into canals and then distributed to farms through distribution channels and minor water courses. Part of Mardan district is also served by the Pehur high level canal, which diverts water from the Indus River through the Tarbela Dam. This location was selected because of a research station of Pakistan Agricultural Research Council (PARC), where field research has been conducted on PRB systems since 1999 in collaboration with the Australian Centre for International Agriculture Research (ACIAR), thus the majority of farm equipment needed and support staff to assist in data collection were available. Study locations are shown on world maps in Figure 3.1.



Figure 3.1: General location of study sites in Australia and Pakistan on World map

3.4 General field conditions encountered in Australia

3.4.1 Site A1

Site A1 was established on Marinya farm, Karara Road, Cambooya, which is located on the eastern Darling Downs (27.734900°S, 151.828062°E), 30 km south of Toowoomba, Queensland, Australia. Long historical data showed annual mean minimum temperature of ~10°C and the mean maximum temperature at ~25°C, total summer rainfall ~ 466 mm and winter rainfall ~253 mm (www.weatherzone.com.au). However, summer 2010 rainfall was above average (Appendix A: Table A3).

The soil at site A1 was Irving-black Vertisol which is a deep to very deep (100-180 cm), fine self-mulching, brownish black cracking clay with brown or reddish brown subsoils on basaltic colluviums. It generally occurs on mid or lower slopes of low basalt hills and rises (Harris et al. 1999). The soil at site A1 contained a subsoil horizon with about < 10% white calcareous segregations (Tim et al. 2008). Maize, wheat, soybean, barley and millet were the main crops grown on the farm over the last few years, prior to the start of the trial. The crops grown were 2009 soybean, 2010 wheat, 2011 hemp, and 2011 corn during the experimental period from summer 2009 to summer 2012.

The cultivation practice at this farm has been no-till PRB since 1969. The current practice consisted of renovating the beds by furrow cleaning only, either before or after crop sowing. Chisel ploughing was occasionally needed, generally at a frequency of every five years. The aim was to loosen the soil compaction caused due to mismatched machinery track widths and to control pests and diseases according to farmer preferences. Renovation and land preparation methods and their frequency vary from farm to farm in the surrounding locality depending on crop varieties, crop rotation, available machinery, farmer preferences, and weather.

The cultural practices prior to 2009 soybean crop sowing during the benchmarking study, commenced with chisel ploughing to a depth of around 30 cm. Chisel ploughing was followed by a slight (20°) realignment of the furrows as part of recent farm modification processes to improve the water harvesting and surface drainage on the farm. The aim of chisel ploughing was to remove the compaction of previous furrows and to achieve uniform subsurface soil conditions. Furrow spacing of site A1 was 200 cm (Figure 3.2a), which was double that of site A2 (Figure 3.2b). The furrows developed at site A1 were generally shallow (D < 10 cm deep), narrow (TW < 40 cm top width) (Figure 3.2c) and 465 m long with blocked ends. According to the farmer, this was primarily to construct PRB with maximum land use efficiency as well as to prepare a smooth and level surface for the following seeding operation.



Figure 3.2: Furrow spacing at (a) site A1 (b) site A2 and (c) Furrow dimension (TW: Top width, MW: Middle width, BW: Bottom width and D: furrow depth or bed height)

During the renovation process the farmer used a bed former which consisted of a frame fitted to assist smoothing the bed surfaces from clods with small furrowers to clean the furrows (Figure 3.3a). Stubble at < 5% of plant height were generally retained during the subsequent cropping seasons due to machinery and economic limitations. The available double disc planter (Figure 3.3b), was 600 cm wide, capable of planting seed on the relatively hard seed bed of no till PRB systems. Both the bed former and seed planter covered three beds in one pass.



Figure 3.3: Available raised bed machinery (a) bed former; note the smearing of bed surface soil and the shallow, rough furrows (insert show furrower) and (b) no till seeder with 8-rows of crop per bed at site A1 (Marinya farm, Cambooya, Australia)

Irrigation water source at site A1 was from two ground water bores (150 m deep) operated separately or in combination according to supply requirements. Subject to availability of water, additional water was available from the rain water storage dam in the vicinity of the experimental plot. However, water availability on the farm was limited due to water restriction rules for the locality. The farmer also shared water with a vegetable contractor, which further constrained the water availability.

Water supply on the farm was through a distribution channel constructed at the head reaches of the irrigated command area of the farm (Figure 3.4). The distribution channel also served a small dam for the vegetable farmer's irrigation supply. Water supply to the distribution channel was from two bores through a PVC pipe (~20cm diameter). Irrigation water application to furrows from the distribution channel was through siphon tubes (50 mm diameter) 6-7 m long (Figure 3.4). Irrigation Tco and Q were variable depending on the supply, number of siphons operated at one time and head in the distribution channel. The head height in the distribution channel was adjustable with water release to the small dam. Current farm practice Tco was between 18 to 24 hours. This site was used for all three field trials (Chapter 4, 5 and 6) conducted at farming fields in Australia. The data collected, equipment used and specific field conditions during the experiments are discussed in details in the respective chapters.



Figure 3.4: On farm head channel with siphons, flow meters, soil moisture data loggers and weather station at site A1 (Marinya farm, Cambooya, Australia)

The chisel ploughing conducted at site A1, was not the general practice at this farm but was representative of many farms in the vicinity. The furrow dimensions were shallow and narrow as per farmer practice at this farm, which affected water advance in a few furrows during first irrigation in the benchmarking study. Therefore, the furrow dimensions were enlarged before the second irrigation to control water crossing between wet and dry furrows. The available Q from the source and the height of distribution channel from fields at this farm was limited, thus offering limited flexibly for variation when required. The blade plough at farm was used during the first season' which consisted of a single horizontal blade (140 cm wide, vertically inclined at an angle of ~25 degrees to the direction of travel and backward rake angle of ~15 degrees). However, due to operational difficulties and soil disturbance during the first season, a new blade plough with two horizontal sharp blades (each 70 cm wide with zero degrees inclination to the direction of travel and a backward rake angle of ~35 degrees) was manufactured, but made available during the last corn season. It was tried to conduct the blade ploughing at optimum soil moisture levels to avoid smearing and soil disturbance.

3.4.2 Site A2

Site A2 was at Bandawing farm, near Dalby (27.150830S, 151.182563 E), 80 km west of Toowoomba, Queensland. The annual mean minimum temperature was 12° C and the mean maximum temperature was 27° C. Total summer season average rainfall was 454 mm and total average winter season rainfall was 172 mm (<u>www.weatherzone.com.au</u>).

This site was selected for the benchmarking study because the PRB management, wheat-cotton rotation and the black cracking clay (Vertisol) at this site truly represented the common farming practices and conditions of this locality. Wheat-cotton rotation was practised at this site for the last ten years. The current agronomic practices in wheat-cotton rotation include annual renovation of PRB by deep ripping involving inversion and bed reforming without relocating the furrows and without disturbing the bed for the next crop. Deep ripping with inversion was to control diseases, pests, slugs and pupae. No stubble were retained at this site. The beds were 100 cm wide with single cotton row planted to the centre of the bed. The bed former was 600 cm wide capable of establishing six beds at a time. The furrow position was kept permanent with tractor mounted GPS. The furrows were ~60 cm wide, alternately irrigated and free draining at tail ends. The benchmarking experiments were conducted in a cotton crop at flowering stage as shown in Figure 3.5.



Figure 3.5: General field conditions and crop stage during measurement at site A2, (Bandawing farm, Dalby, Australia)

The water source was rain water stored in an on-farm storage dam in the vicinity of the experimental field. Water availability was limited in the dam because of the continuous drought conditions over the last few years. Water supply was piped to the distribution channel by gravity. Siphon tubes (50 mm diameter), 4.5 m long were used for irrigating the furrows. The selected experimental site was 455 m long and 24 m wide. Two sets of irrigation data were taken at this site.

This site was used only for the benchmarking study (Chapter 4) due to water availability constraints caused by the drought conditions in 2009-10. The furrows were wide (~60 cm top width), the beds were narrow (~40 cm bed top width) and furrow spacing was 100 cm. Soil cracking was observed to affect irrigation performance of alternate furrow irrigation because water from the wet furrows was able to break through into the dry furrows within soil cracks.

3.5 General field conditions encountered in Pakistan

The study area was located in Mardan (Khyber Pakhtunkhwa Province), in north west Pakistan. All four study sites were located in a radius of 15 km therefore the climatic conditions and soil types were comparable. Mardan lies in the semi-arid zone, where mean seasonal rainfall of 250 mm occurs in summer (April-September) and around 300 mm during winter (October-March). The mean maximum temperature ranges from 27-30°C during June, while the mean minimum temperature ranges from 5-8°C during January. Sites P1 (34.115608 N, 71.595392 E) and P2 (34.115141N, 71.595343E) were located near the Mardan Research Farm (Pakistan Agriculture Research Council (PARC)), a short distance from Mardan-Charsadda Road. Sites P3 (34.054170N, 72.030761E) and P4 (34.081084N, 72.014867E) were located about 30 km and 15 km respectively to the east of site P1. The benchmarking study was conducted at sites (P1, P2, and P3) in summer 2010.

The soil at all experimental sites belonged to Alfisol soil order that have been developed on pleistocene surfaces in subhumid/humid regions of Peshawar valley (Ahmad et al. 1986). The soil texture was sandy clay loam, that belongs to the Mardan soil series, which was classified as fine Usteric Camborthid, a greyish brown, non to slightly calcareous alluvial material of the Holocene age (Shafiq et al. 2002). The soil at these sites has negligible shrink- swell qualities. The subsurface tile drainage system installed in this locality did not allow the water table to rise and generally remains at 350 to 500 cm depth. Sites P1, P2 and P3 were in a low salinity zone and site P4 was in a relatively high salinity zone. No stubbles were retained in this locality. The general field conditions of site P2 (132 cm wide bed) are shown in Figure 3.6.



Figure 3.6: General field conditions of site P2 under corn crop in Mardan, Pakistan (Note that the corn is sown on the edge of the beds to overcome poor penetration of the wetting front, as seen by the wet areas on the surface of the beds)

Experimental plots at site P1 and site P2 were part of a 10 ha farm where water was available on a weekly schedule. The farm manager was able to control Q by changing the number of furrows per set. At site P3, Q and *Tco* were easily manageable due to the lengthy irrigation turn period for the relatively larger farm size (> 40 ha). The total available maximum Q at all four sites was ~28 L s⁻¹. The farm managers had flexibility in using total maximum inflow in any part of their farms according to their needs during the allocated irrigation period at all experimental sites. For instance, during the benchmarking study at site P1 and site P3 the farm manager utilised larger Q due to expected slow water advance on high infiltrative soils. However, at site P2 two fields were simultaneously irrigated to reduce the Q per furrow due to fast water advance in furrows. The lateral infiltration study was conducted at site P4 situated half way between site P1 and P3. The PRB at this site were two seasons old but the soil was degraded and had mild salinity.

The data for the benchmarking study were collected in the summer season 2010 under cotton crop at site P1 and maize crop at sites P2 and P3. Two bed sizes (NB: 66 cm furrow spacing (Figure 3.7a) and WB: 132 cm furrow spacing (Figure 3.7b)) were tested at each site. The field length at site P1 was 86 m and at site P2 and P3 was 90 m each with a common slope of 0.002 m m⁻¹ at all sites. All furrows of WB were trafficked, while only alternate furrows of NB were trafficked.



Figure 3.7: Furrow spacing of (a) NB: Narrow bed (b) WB: Wide bed evaluated at three research sites during the benchmarking study in Mardan, Pakistan

Site P1 was under cotton-wheat rotation while site P2 and site P3 were under maizewheat rotation. The PRB systems at all three sites were renovated by shallow rotary hoeing prior to furrow cleaning, a common renovation method in this locality, while the furrow positions were kept permanent. According to the farmer, the major aim was to control weeds and to loosen the soil for preparing a soft seed bed that facilitates manual seeding of maize and cotton crops.

The observed field conditions at different sites were variable during the benchmarking study. Large weed infestation was observed in the cotton crop at site P1, with an increased number of weeds concentrated in NB furrows in comparison with WB. The crops at all three sites were at different growth stages at the time of data collection. The cotton and maize crops at sites P1 and P3 respectively were at flowering stage, while the maize crop at site P2 was at establishment stage. Third and fourth irrigations were measured at site P1, first and second at site P2, while at site P3 fifth and sixth irrigations were measured. Similarly, manual weeding, a traditional method in the locality, was recently practised at site P3 which increased the furrow surface roughness, but reduced the hindrance caused by weeds to the flow of water. All these field conditions had consequences on results during the analysis.

3.6 Soil and irrigation measurement technique

This research involved field data collection related to soil moisture, bulk density, lateral infiltration, Q, Tco, irrigation water advance time in furrows at multiple points, bed-furrow configurations, crop establishment, crop height and yield. In addition to field evaluation research components, this study also included a fourth component that involved simulation for evaluation of irrigation performance (*Ea*, *Er*, *DU*, inflow and deep drainage losses) and to identify strategies for improving lateral infiltration and for optimising irrigation management and field design of raised beds. As such, all four components were covered separately in four chapters. However, the common field measurement technique and simulation modelling protocols that were applied across all or most sites for the four research components are discussed below.

Description of all six sites, study objectives, field conditions, measurements, and simulation models used during the analysis are summarised in Table 3.2.

		•		Field (conditions		Measure	ments	
Site	¹ Obj.	Experiment	Soil type	Cultivation practices	Furrow spacing	Crop & growth stage	Soil	Water	used
	1	Summer 2009-10		Deep ripping	200 cm	Soybean, full season data	SM, BD &	Q, Tco, Water	IPARM &
A1	7	Win.2010- Sum 2012	Heavy clay Vertisol with shrink-swell	Three treatments	200 cm	Wheat, hemp and corn, full seasons	Furrow dimension	advance time,	SIRMOD
	c	Summer 2011	and self- mulching properties	Three treatments	200 cm	Corn, establishment	SM, BD & Lateral infiltration	Water added	Hydrus
A2	1	Summer 2009-10		Annual ripping	100 cm	Corn, flowering stage	SM, BD, furrow dimension	Q, Tco, Advance time,	IPARM & SIRMOD
P1	1		SCI with	Shallow	WB-132 cm	Cotton, flowering stage	SM RD &	O T_{CO}	
P2	1	Summer 2010	non-swelling properties	rotary harrowing	& NB: 66 cm at each site	Corn, early stage	furrow	Advance	IPARM & SIRMOD
P3	1	I		0		Corn, late stage			
P4	3	Summer 2011	SCL, mildly saline	Three treatments	132 cm	Corn, establishment stage	SM, BD & lateral infiltration	Water added	Hydrus

3.6.1 Soil moisture, bulk density and soil moisture deficit

The soil moisture and bulk density was recorded using thermo gravimetric methods according to the procedure described in Marshall and Holmes (1988). The soil cores were collected from 0-30 cm profile depth (rings of 5 cm x 5 cm sizes were used) at 10 cm depth interval. The samples were packed in air tight and insulated containers immediately after collection to avoid any soil moisture loss. The samples were kept in an oven at 105[°]C for 48 hours after determining the wet weight. The soil moisture was determined as the ratio of weight lost (water weight) and dry soil weight after 48 hours of drying and bulk density was calculated on dry mass basis as the ratio between dry weight of soil and internal ring volume. Volumetric soil moisture was determined by multiplying gravimetric soil moisture content by mean bulk density for each layer. The root zone soil moisture deficit (SMD) was determined as the difference between antecedent and field capacity moisture levels. Field capacity is the upper limit of plant available water or is the soil moisture content after drainage of gravitational water has become very slow and the moisture has become relatively stable, that generally occur after 2-3 days of irrigation (Michael 1999), which was determined at 43% for Vertisol during experiments that closely agreed with values reported by (Hansen et al. 1979; Payero & Harris 2010) and 23.5% for sandy clay loam that closely agreed with values reported by (Shafiq et al. 2002; Hassan et al. 2005).

3.6.2 Lateral infiltration

Lateral infiltration data during the four cropping seasons were collected at site A1, Australia. Automatic soil moisture meters (Sentek EnviroSCAN[®], discussed in section 2.5.1) 100 cm deep with sensor slots at 10 cm depth interval, were used. Three probes were placed at 33 cm, 67 cm and 100 cm distances from furrow centre across the bed as shown in Figure 3.8. The holes for the probes were made with a soil auger slightly smaller than the outer diameter of the PVC tube and slightly deeper (120 cm) than the maximum measuring depth of 100 cm. The tubes were then installed by pushing, twisting and hammering at the top of the tube. This procedure reamed the hole to the exact size of the tube while the soil reamed was settled in the extra depth of the hole drilled. This procedure ensured a perfect contact between the tube and the soil. All the sensors were standardised (normalised) by taking free air

and water measurements before installations. Initially the commonly used soil properties of Vertisol were used for calibrating the enviroSCAN probes which were latter on readjusted with minor modifications to match the gravimetric soil moisture values on these sites. The EnviroSCAN readings were validated by matching with the calculated volumetric soil moisture obtained from a 0-30 cm profile in a nearby location using the gravimetric method (section 3.6.1). The gravimetrically determined soil moisture values matched well with the calibrated EnviroSCAN readings as given in Appendix C: Figure C1. The soil moisture was logged at 10 minutes interval throughout the cropping season. The probes were removed shortly prior to harvest, and reinstalled shortly after sowing. However, replication of this setup, along with the number and location of sensors per probe, was variable during the experimental period due to equipment constraints. These data collections facilitated in irrigation scheduling and soil moisture deficit (SMD) calculations. The number and location of sensors used during the three cropping seasons (2010 wheat, 2011 hemp, and 2011 Corn) are discussed in Chapter 5. The lateral infiltration study during 2011 summer in both soils (Vertisol and sandy clay loam) is detailed in Chapter 6.



Figure 3.8: Layout of Sentek EnviroSCAN[®] probes placed across half bed width at given positions of sensors for logging soil moisture variations at 10 minutes interval during 2010 wheat season at site A1, Marinya farm, Cambooya, Australia

3.6.3 Soil infiltration and irrigation performance

The IPARM model (Gillies & Smith 2005) (section 2.5.2.1) was used for identification of Kostiakov infiltration functions (a, k and f_0 of equation 2.6). IRRIMATETM equipment were used for irrigation data collection in Australia and manual method in Pakistan. SIRMOD III (Walker, 2003) (section 2.5.2.1) was used for evaluation of irrigation performance. The field measured input data and outputs of IPARM and SIRMOD are summarised in Table 3.3. The SIRMOD model was calibrated by adjusting the furrow roughness coefficient (Manning "n") and f_0 in the model until a close match between measured and simulated advance time was achieved. The measured versus simulated advance times was correlated with trend line analysis (Appendix B). Potential for deep drainage due to irrigation excess (DD) was calculated by the equation (Smith et al. 2005);

$$DD = \frac{(1 - Ea)Vi - Vo}{LS}$$
 3.1

where *Ea* is application efficiency, *Vi* is volume of furrow inflow, *Vo* the volume of furrow tail-water outflow, *L* the length of the furrow, and *S* is furrow spacing.

Model	Field measured inputs	Output		
IPARM	 Inflow rate Furrow slope Furrow length Furrow top width Furrow middle width Furrow bottom width Furrow depth Manning's n Water advance at multiple points along furrow Runoff 	Infiltration functions of modified Kostiakov infiltration equation 1 - a 2 - k $3 - f_0$ 4 - C and 5-Advance curve 5-Runoff curve 6-Cumulative infiltration curve		
SIRMOD	 All IPARM inputs and outputs except water advance and runoff Time to cut-off Free draining or blocked Furrow spacing Root zone soil moisture deficit (<i>SMD</i>) 	Irrigation performance 1- Ea 2- Er 3- DU 4- Inflow volume 5- Outflow volume 6-Complete log of advance, recession time and cumulative infiltration along the furrow length		

Table 3.3: Input and output data of IPARM and SIRMOD models

3.6.3.1 IRRIMATETM equipment for irrigation performance evaluation

The IRRIMATETM suit of equipment developed at NCEA was used in irrigation data collection. It include siphon inflow meters (Figure 3.9a), which gives a log of Q. The advance sensors (Figure 3.9b) measure the water advance in furrows. The sensor in each furrow triggers when water touches the sensor. The flume flow meter (Figure 3.9c) is installed on flumes specially designed for measuring the runoff volume, which gives a complete log of flow passing through the flume. Data from all equipment is downloaded through a small pocket PC which can communicate using infrared communication with the equipment.



Figure 3.9: IRRIMATETM equipment for irrigation measurements (a) Siphon flow meter, (b) water advance meter, (c) flume with flow meter

3.7 Statistical methods

Data sets were analysed using Excel and IBM SPSS version 19. All the data sets were checked for compliance with the underlying ANOVA assumption. When normality of data sets was found to violate the ANOVA assumption then data were transformed using SPSS built in functions for improving the symmetry of the distribution prior to analysis. The equal variance assumption was evaluated using the Levene test. If the Levene test showed heterogeneity but the ratio between largest to smallest standard deviation was less than 2, then the data sets were considered fit for ANOVA at P=0.05 significance level, otherwise P=0.01 was recommended to control the type I error (White 2007). A general linear model (univariate) with Tukey (post hoc) analysis was used to identify significant differences between the treatments (main groups) and plant rows as subgroup for plant data and profile depth as subgroup for bulk density analysis.

CHAPTER 4: Benchmarking irrigation performance of permanent raised beds

4.1 Introduction

Literature reviewed in Chapter 2 illustrated that the existing research on irrigated PRB systems under actual farmer managed conditions is sparse, while the majority of available research largely covers rain fed conditions. The limited field studies focussed on irrigated PRB systems under experimental field conditions, generally used small area or point evaluation methods, which were subject to large spatial variations. Therefore, this dissertation involved preliminary evaluation of existing irrigated PRB farming practices to assess their impact on irrigation performance under both Australian and Pakistani farm conditions. It aimed to benchmark irrigation performance of farmer managed PRB farming systems and identify strategies to improve their *WP*.

Field trials were conducted in Australia and Pakistan on existing irrigated PRB farming systems in farmers' fields. Two sites (A1, A2) were selected in south-east Queensland, Australia. The soil type on both sites was a black Vertisol exhibiting substantial shrinking-swelling, cracking and self-mulching properties as described in section 3.4. Three field sites (P1, P2 and P3) were selected with contrasting environmental conditions in north west Pakistan. The soil type at each Pakistani site was a non-swelling, rigid sandy clay loam with a history of intensive cultivation as described in section 3.5. The measurements were conducted under actual field conditions with full farm manager control.

4.2 Material and methods

The experimental details of measurements of irrigation performance in both countries are summarised in Table 4.1. Irrigation water generally moves to 100 cm and 60 cm depths under Vertisol and sandy clay loam respectively during normal irrigation applications, thus these depths are generally analysed as active root zone depths.

Table 4.1: Experimen density and <i>SMD</i> - soil	tal details of measure moisture deficit, FC-	ment of existing irrigati Field capacity, NB-nar	on performance in Austr row bed and WB-wide b	ralia and Pakistan (SM- oed)	soil moisture, BD- bulk
Experimental	Australian sites		Pakistani sites		
measurement	Site A1 (Cambooya)	Site A2 (Dalby)	Site P1 (Mardan A)	Site P2 (Mardan B)	Site P3 (Mardan C)
Antecedent SM & BD No. of samples & positions Sampling depths Gravimetric SM & BD	8 – 4 @ head, 4 @ tail 0-10, 10-20,& 20-30 cm Undisturbed cores 5 cm diom 5 cm Innorth	4 - 2 @ head, 2 @ tail 0-10, 10-20,& 20-30 cm Undisturbed cores 5 cm diam.	4 - 2 @ head, 2 @ tail 0-15, 15-30 & 30-60 cm . Undisturbed cores 3 cm dia x	4 - 2 @ head, 2 @ tail 0-15, 15-30 & 30-60 cm Undisturbed cores 3 cm dia x	4 - 2 @ Head, 2 @ tail 0-15, 15-30 & 30-60 cm Undisturbed cores 3 cm dia x
Volumetric SM Method	Gravimetric SM x BD	Gravimetric SM x BD	ou cui toug tune Gravimetric SM x BD	Gravimetric SM x BD	ou cui toug uuce Gravimetric SM x BD
<u>SMD prior to irrigation</u> No. of samples & positions	4 – 2 @ head, 2 @ tail	4 – 2 @ head, 2 @ tail	4 - 2 @ head, 2 @ tail	4 - 2 @ head, 2 @ tail	4-2 @ head, 2 @ tail
Distance from furrow centre Profile depths	33, 6/, 100 cm 10 cm intervals to 100 cm	50 cm 10 cm intervals to 100 cm	44 cm 0-15, 15-30 & 30-60 cm	44 cm 0-15, 15-30 & 30-60 cm	44 cm 0-15, 15-30 & 30-60 cm
Method Calculation	Sentek capacitance probe SMD= FC -SM in 100 cm	capacitance probe SMD= FC -SM in 100 cm	Gravimetric (as above) SMD= FC -SM in 60 cm	Gravimetric (as above) SMD= FC -SM in 60 cm	Gravimetric (as above) SMD= FC -SM in 60 cm
Water advance in furrow					
No. of samples & positions Method	5 – @ 100 m intervals IRRIMATE TM Flow	5 – @ 90 m intervals IRRIMATE TM Flow advance	4 – @ 21 m interval Manually using measuring tap	5 – @ 18 m intervals eManually using measuring	5 – @ 18 m intervals Manually using measuring
No of furrows measured	advance meter 21	meter 20	and staff watch 20 NB & 16 WB	tape and staff watch 20 NB & 16 WB	tape and staff watch 20 NB & 16 WB
<u>Inflow rate</u>	IRRIMATE TM Siphon flow meter	IRRIMATE TM Siphon flow meter	Total inflow into 5 NB & 4 WB furrows by broad- crested weir (pipe) flume	Total inflow into 5 NB & 4 WB furrows by broad- crested weir (pipe) flume	Total inflow to 5 NB & 4 WB furrows by broad- crested weir (pipe) flume
Runoff	None- Furrow blocked at tail end	IRRIMATE TM Flume/ flow meter	None-Furrow blocked at tail end	None- Furrow blocked at tail end	None-Furrow blocked at tail end
Treatments Furrow spacing Bed width	200 cm ~150 cm	100 cm ~40 cm	NB (66 cm) & WB (132 cm) NB(29 cm) & WB (70 cm)	NB (66 cm) & WB (132 cm) NB(16 cm) & WB (90 cm)	NB (66 cm) & WB (132 cm) NB(18 cm) & WB (80 cm)
<u>Measured Irrigations</u> No. & sequence Dates	2 (1st & 2nd) Dec 26, 2009; Jan 29, 2010	2 (2nd & 3rd) Jan 7 & 27, 2010	2 (3rd & 4th) Jul 18 & 28, 2010	2 (1st & 2nd) Jul 19 & Aug 2, 2010	2 (5th & 6th) Jul 20 & 29, 2010

4.2.1 Data collection and instruments used

The site conditions and cultivation practices recorded were already discussed in Chapter 3. The IPARM model (Gillies & Smith 2005) was used to calculate the infiltration functions and the SIRMOD III (Walker 2003) model was used to evaluate the irrigation performance. Both these models are described in section 2.5.2.1 and their input data (Table 3.3) and collection procedure is described in section 3.6.3. Data sets for calibration of both models are given in Figure B1 & B2 (Appendix B). Specific field data collected to parameterise the IPARM and SIRMOD models and instrument used in both countries during this study are summarised below.

4.2.1.1 Australia

At site A1, a field topographic survey was conducted for determining the slope at the commencement of the trials. Weather data were recorded by an automatic weather station on the experimental site. The *SMD* was calculated as the difference of field capacity (43%) and antecedent soil moisture using soil moisture and bulk density data in surface 0-30 cm depth at 10 cm depth interval using the thermo-gravimetric method as discussed in section 3.6.1 and the four replications were in both the field head and tail reaches. Additional soil moisture data were collected prior to irrigation at 0-100 cm depth using automatic soil moisture meters (Sentek EnviroSCAN[®]), as discussed in section 2.5.1.2 and the method described in section 3.6.2, according to layout shown in Figure 3.8 at site A1. The measurements were replicated twice both at field head (50 m distance from distribution channel) and tail (350 m distance from distribution channel) to account for spatial variability. The capacitance probe accuracy was validated by the gravimetric soil moisture.

Two irrigations (26 December 2009 and 29 January 2010), were applied to a soybean crop at site A1. IRRIMATETM equipment (section 3.6.3.1) consisting of siphon flow meters were used to record Q in three furrows and flow advance monitoring sensors placed at ~100 m intervals along the furrow length were used to measure water advance in 24 furrows. Flow crossing in few furrow affected the water advance during first irrigation thus the furrow sizes were increased before the second irrigation and data of affected furrows discarded.

At site A2, the data collected include furrow slope, agronomic management (including land preparation, crops, inputs and machinery) and current irrigation management as discussed in section 3.4.2 and summarised in table 3.2. The *SMD* and irrigation data were obtained similar to site A1. However, the water advance sensors were placed 90 m apart in this case. Two irrigations (7 January 2010 and 27 January 2010) to cotton were measured at this site. The furrows were free draining and alternate furrow was irrigated. Runoff was measured by flumes with flow recording meters. Collection of two irrigations data was possible on both sites because no third irrigation was applied during the season. However, the replication of data during both irrigations on both sites was sufficient for evaluating statistical significance.

4.2.1.2 Pakistan

Two irrigations were measured at each of the three sites. The number of irrigation events monitored was limited due to time constraint but the number of sites and replication of data during each irrigation event at each site was sufficient for identifying statistical significance. Two bed furrow treatments were evaluated (1-NB: narrow bed at 66 cm furrow spacing and 2-WB: wide bed at 132 cm furrow spacing) at each site, as described in Chapter 3. Data from five furrows per set under NB plot and four furrows per set under WB plot were twice replicated at each site. The Q to a group of furrows was calculated from water head measurements over the crest level of broad crested weir type flume at 5 minute intervals. The average Q per set was divided by the number of furrows to calculate the single furrow Q. Similar Q in each furrow per set was obtained through parallel bunds by controlling flow depth near the head by adjusting bund crest height. Water advance data were measured at 21 m interval on site P1 and 18 m interval on site P2 and P3 manually.

The root zone (0-60 cm) *SMD* was determined according to procedure described in section 3.6.1 using field capacity of 23.5% of the local sandy clay loam (Hansen et al. 1979) as determined by Shafiq et al. (2002). The soil cores were collected at 0-15 cm, 15-30 cm and 30-60 cm depth intervals using 3 cm diameter x 60 cm long tube from the shoulders of WB and middle of NB with two replicates in each plot. Data of three sites were averaged for different treatments in majority of cases due to no significant differences between sites.

4.3 Results

4.3.1 Field conditions and soil moisture deficit before irrigation

The topographic survey on both Australian sites (A1 and A2) showed an average slope of ~0.002 m m⁻¹. The *SMD* was ~100 mm before both irrigations on site A1 and before the second irrigation on site A2. However, the average *SMD* was 76 mm before the first irrigation on site A2 (Figure 4.1). Weather data for site A1 are attached in Table A1 (Appendix A).



Figure 4.1: Soil moisture deficit *(SMD)* in the root zone (0-100cm depth) prior to two irrigation applications at both sites on Vertisol, Australia (vertical bars show SD)

The fields evaluated in all Pakistani sites were short (< 100 m) and field widths were also less than 100 m. The furrow slopes were ~0.002 m m⁻¹. Field observations of the soil profile up to 60 cm depth showed a fairly uniform soil structure and texture across the field. The average root zone *SMD* was ~60 mm/60 cm depth prior to both irrigations on all sites as given in Figure 4.2.



Figure 4.2: Average *SMD* at given depths on three sites, prior to two measured irrigation applications at Mardan, Pakistan (Vertical bars show SD)

4.3.2 Bed furrow dimensions

In Australia, the furrow spacing was 200 cm at site A1 with an initial furrow top width of 42 cm and depth of ~8 cm during the first irrigation. However, due to insufficient capacity for the applied Q during the first irrigation, the furrow dimensions were increased by ~26% before the second irrigation (Table 4.2). At site A2, the furrow spacing was 1 m with average furrow top width of ~61 cm (Table 4.2), and bed top width of ~39 cm. The wider furrow width avoided compaction of the furrow sides. However, cracks crossing the bed from wet to dry alternate furrow were obvious during the second irrigation, when *SMD* was high.

Table 4.2: Average furrow dimensions (TW = top width, MW = middle width, BW = bottom width and D = depth) recorded prior to two irrigations at two sites (A1 and A2) in Australia and three sites (P1. P2 and P3) in Pakistan (SD in brackets)

Bed &	Aust	tralia		Pakistan						
furrow	A1	A2]	P1	I	22]	P 3		
dimensions	200cm	100cm	NB	WB	NB	WB	NB	WB		
TW	48	61	37	62	50	43	48	52		
(cm)	(8)	(3)	(3)	(2)	(3)	(1)	(1)	(3)		
MW	27	43	25	52	35	34	31	33		
(cm)	(4)	(4)	(3)	(4)	(3)	(1)	(3)	(3)		
BW	13	24	19	38	20	15	14	18		
(cm)	(2)	(5)	(2)	(7)	(4)	(2)	(2)	(2)		
D	9	10	10	12	12	11	9	10		
(cm)	(1)	(1)	(1)	(1)	(2)	(1)	(1)	(1)		
L (<i>m</i>)	465	455	86	86	90	90	90	90		

In Pakistan, the furrow dimensions of both NB and WB were different (Table 4.2 and Figure 4.3). The furrow widths were smaller for NB than WB in majority of cases. Conversely, at site P2 the NB furrows were wider due to fresh renovation applied prior to measurements. However, the difference in furrow depth was not significant (P=0.05) between the NB and WB. For NB the furrow top width ranged from 37 to 50 cm, while for WB the furrow top width ranged from 43 to 62 cm. Thus, the NB furrow sides were subjected to more compaction by tractor and machinery tyres, as generally tractor and machinery widths range up to 45 cm, during the field operations compared with the WB furrows. Bed furrow sizes at all sites and cropping seasons are summarised in Appendix A: Table A4.



Figure 4.3: Average field measured furrow dimensions (recorded prior to two irrigations at all three sites for narrow bed (NB: 66cm furrow spacing) and wide bed (WB: 132cm furrow spacing) systems at Mardan, Pakistan (vertical bars show SD)

4.3.3 Inflow, water advance and time to cut-off.

In Australia, the field measured Q ranged from 1.91 L s⁻¹ to 1.98 L s⁻¹ with an average of 1.92 L s⁻¹ during the two irrigations at site A1 (Table 4.3). The total average inflow volume was 134 mm during the two irrigations with ~12% increased inflow volume during the first (146 mm) than the second irrigation (130 mm) (Table D1: Appendix D). At site A2, the average Q ranged from 2.5 L s⁻¹ to 2.6 L s⁻¹ with an average of 2.54 L s⁻¹ during the two irrigations. The total average inflow volume during the two irrigations with ~12% less inflow volume during the first (101 mm) than for the second irrigation (113 mm).

The average measured advance time to furrow tail end (Ta) at site A1 was 13% longer during the first (993 min) than during the second irrigation (862 min) (Figure 4.4a), indicating impact of seasonal changes on water advance along the furrow length, because impacts of *SMD* and *Q* were negligible due to their comparable values. At site A2, the average Ta during the first irrigation (387 minutes) was 34% shorter than for the second irrigation (584 minutes) (Figure 4.4b). Wider cracks due to larger *SMD* (100 mm) were observed during the second than for the first irrigation (Figure 4.5a), which reduced the advance rate and all furrows (wet and dry) were equally inundated (Figure 4.5b) till the end of the second irrigation. The advance times at all sites are summarised and attached in Table A5 (Appendix A).



Figure 4.4: Average field measured water advance curves of two irrigations at (a) sites A1, and (b) site A2 in Australia (vertical bars show SD)



Figure 4.5: Crop and field conditions before (a) irrigation 1 with no cracks; (b) irrigation 2 with flow crossing the bed through cracks at site A2, Australia

The average *Tco* at site A1, was 1078 minutes (Table 4.3) with 7.5% shorter *Tco* during the second (1057 min) than for the first irrigation (1143 min) (Table D1: Appendix D). The average *Tco* at site A2 was 646 minutes during the two irrigations with 15% longer *Tco* during the second than for the first irrigation (590 min).

In Pakistan, the measured Q ranged from 1.55 L s⁻¹ to 4.88 L s⁻¹ on all sites. The results showed a significantly (P=0.05) reduced Q on NB that ranged from 1.55 L s⁻¹ to 3.09 L s⁻¹ (2.3 L s⁻¹ average) than for the WB that ranged from 2.66 L s⁻¹ to 4.88 L s⁻¹ (3.5 L s⁻¹ average). The total inflow volume per single furrow of WB was 36% increased (5.40 m³ to 9.75 m³) than for the NB (2.9 m³ to 7.4 m³) during all measured irrigations. However, the total inflow volume per unit area was significantly (P=0.05) reduced (32%) on the WB (60 mm) than for the NB (Table 4.3).

Irrigation	Australian sites		Pakistani sites							
Performance	A1	A2	I	?1	I	22	Р	3	Ove	rall
I al alletel	200cm	100cm	NB	WB	NB	WB	NB	WB	NB	WB
Q	1.92	2.54	2.95	4.86	1.62	2.71	2.36	2.96	2.3a	3.5b
(L s-1)	(0.1)	(0.1)	(0.2)	(0.0)	(0.1)	(0.1)	(0.4)	(0.2)	(0.6)	(1.1)
Tco	1078	646	39	33	34	35	37	34	37a	34b
(min)	(40)	(51)	(2)	(2)	(3)	(1)	(1)	(1)	(3)	(2)
Inflow	134	108	121	83	56	47	88	50	88a	60b
(mm/irrigation)	(8)	(7)	(15)	(6)	(9)	(5)	(11)	(5)	(27)	(17)
DD	34	12	61	24	6	4	28	1	32a	8b
(mm/irrigation)	(8)	(6)	(4)	(2)	(1)	(1)	(2)	(0)	(24)	(6)
Ea	75	81	50	72	92	99	68	99	70a	90b
(%)	(4)	(6)	(6)	(4)	(8)	(1)	(9)	(1)	(18)	(9)
Er	99	97	100	99	83	77	99	83	94a	86b
(%)	(1)	(1)	(0)	(1)	(7)	(0)	(1)	(7)	(5)	(10)
DU	91	87	90	77	66	84	73	75	76a	79a
(%)	(3)	(3)	(6)	(2)	(7)	(8)	(1)	(9)	(13)	(10)

Table 4.3: Average irrigation performance of two irrigations measured on two sites (A1 and A2 with two bed sizes) in south east Queensland, Australia and three sites (P1, P2 and P3 with two bed sizes) in north west Pakistan (SD in brackets)

*Figure followed by same letter in rows are not significantly (P=0.05) different (All table abbreviations defined on page xxviii)

The average Q for the NB (2.95 L s⁻¹) at site P1 was ~39% less than for the WB (4.86 L s⁻¹) during both irrigations. The average total inflow volume per furrow for the NB (6.75 m³) was ~28% less than for the WB (9.35 m³) during both irrigations. But the total average inflow volume of the NB (121 mm) was ~46% larger than for the WB (83 mm) during both irrigations (Table 4.3). At site P2, the average Q was 1.62 L s⁻¹ on NB and 2.71 L s⁻¹ on WB. The average inflow volume per single furrow on NB (3.25 m³) was ~40% less than for the WB (5.45 m³). In contrast, the average inflow volume per unit area of NB (56 mm) was ~20% larger than for the WB (47 mm). At site P3, the average Q of NB (2.36 L s⁻¹) was ~20% less than for the WB (2.96 L s⁻¹), while the average inflow volume per single furrow of NB (5.15 m³) was ~13% less than for the WB (5.9 m³). However, the average inflow volume per unit area of NB (88 mm) was ~75% larger than WB (Table D2: Appendix D).

The average Ta on the NB treatment was 51% longer, similar and 14% longer than WB treatment during both irrigations at sites P1, P2 and P3, respectively. The data generally showed a significantly (P=0.05) longer (19%) Ta on NB than WB (Figure

4.6a). All trafficked furrows showed a significantly (P=0.05) faster (~21%) *Ta* than the non-trafficked furrows (Figure 4.6b and 4.7) of all sites.



Figure 4.6: Impact of (a) bed size and (b) wheel trafficking on average water advance along furrow for all three sites in Pakistan (Vertical bars show SD)



Figure 4.7: Field conditions of site P2 with (a) NB: Fast water advance on trafficked furrows and slow on non-trafficked alternate furrows; (b) WB: Uniform water advance on both trafficked furrows

The average *Tco* for the NB was ~18% longer, similar and ~9% longer than for the WB at site P1, P2, and P3 respectively during both irrigations (Table 4.3). Overall results indicated longer average *Tco* on NB than on WB with current management systems. Field observation showed that the farmer decision for *Tco* on WB was based on furrow filling and avoiding bed overtopping at the tail end of the field, which caused poor lateral infiltration at the head reaches. For the NB systems, the measured *Tco* exceeded the *Ta* with the existing farmer management. The water advance along the furrows showed linear response with the square root of time for both soils as provided in Figure A1 (Appendix A).

4.3.4 Infiltration

In Australia, at site A1, results showed a general tendency of reduced cumulative infiltration during the second than the first irrigation. Comparison of both irrigation results during the 900 minutes of infiltration opportunity time showed ~11% reduced cumulative infiltration during the second (132 mm) than the first irrigation (148 mm) (Figure 4.8). Similarly, the total cumulative infiltration at the time of water arrival to the tail end of the furrow was ~15% increased during the first (155 mm) than for the second irrigation (135 mm), indicating reduced soil porosity due to subsidence.



Figure 4.8: Comparison of SIRMOD simulated cumulative infiltration at given times during two irrigations at both sites (A1 and A2) on a Vertisol, in Australia

At site A2, the cumulative infiltration showed 25% increased infiltration during the second (100 mm) than the first irrigation (80 mm), when compared at 400 minutes of infiltration opportunity time (Figure 4.8). Overall data showed \sim 38% increased cumulative infiltrations during the second (110 mm) than the first irrigation (80 mm) at the time of water arrival to the furrow tail end during both irrigations, indicating the larger *SMD* increased the cumulative infiltration during the second irrigation.

The cumulative infiltration along the furrow length (Figure 4.9) indicated 27% to 40% and 34% to 42% less cumulative infiltration at the tail than at the head reaches on site A1 and A2, respectively. The cumulative infiltration at site A1 was 29% and 33% increased at the head and tail reaches, respectively, than site A2. At site A1, the *SMD* (100 mm) was fulfilled but with excessive *DD*, while at site A2, 11% (50 m)

and 33% (150 m) of the field length from field tail end received less than required irrigation application during the first and the second irrigation, respectively (Figure 4.9). The infiltration opportunity time at the tail reaches was 38% to 60% reduced at sites A1 and 58% to 78% at site A2 than at the head reaches.



Figure 4.9: Average cumulative infiltration and infiltration opportunity time along the furrow length during the two irrigations at (a) site A1 and (b) site A2 in Australia

In Pakistan, the volume infiltrated per unit furrow length was in the range of $0.04 \text{ m}^3 \text{ m}^{-1}$ to $0.15 \text{ m}^3 \text{ m}^{-1}$ on NB (Figure 4.10a) and $0.08 \text{ m}^3 \text{ m}^{-1}$ to $0.11 \text{ m}^3 \text{ m}^{-1}$ on WB (Figure 4.10b) during the 100 minutes infiltration opportunity time. Overall results showed 12% increased infiltrated volume per unit furrow length for WB than for the NB during the 100 minutes infiltration opportunity time.



Figure 4.10: Average volume infiltrated per unit furrow length on two bed sizes at three sites during two irrigations in a sandy clay loam, Pakistan (Vertical bars show SD)

The volume infiltrated per unit area ranged from 0.06 m³ m⁻² to 0.225 m³ m⁻² on NB (Figure 4.11a) and 0.06 m³ m⁻² to 0.11 m³ m⁻² on WB (Figure 4.11 b) during the 100 minutes infiltration opportunity time. The average volume infiltrated per unit area was ~35% less on WB than on NB during the 100 minutes infiltration opportunity time. Thus the infiltration properties varied from site to site.



Figure 4.11: Average volume infiltrated per unit area on two bed sizes at three sites in sandy clay loam, Pakistan (Vertical bars show SD)

The spatial distribution of average cumulative infiltration along the furrow length (Figure 4.12) clearly illustrated that 50% of the measured irrigation events were over-irrigated (e.g. site P1 both beds and site P3 NB) and ~50% were under-irrigated (e.g. site P2 both beds and site P3 WB) as shown by the reduced cumulative infiltration at head reaches than the target 60 mm *SMD*.



Figure 4.12: Spatial variation in average cumulative infiltration along the furrow length to replenish 60 mm soil moisture deficit (*SMD*) during two irrigations each at (a) site P1, (b) site P2 and (c) site P3 under two bed sizes in sandy clay loam, Pakistan.

4.3.5 Irrigation performance

In Australia, at site A1, the results showed *Ea* in the range of 68% to 77% with 75% average, *Er* above 99%, while *DU* ranged from 88% to 92% with 91% average (Table 4.3). Comparison of two irrigations showed improved *Ea*, *DU* and reduced inflow volume during the second irrigation than for the first irrigation (Table D1: Appendix D). The lower *Ea* caused excessive *DD* that ranged 30 mm/irrigation to 47 mm/irrigation as the furrows were blocked at tail end. The *DU* was increased by 4% during the second than the first irrigation (88%).

At site A2, the average *Ea* was 81% during the two irrigations (Table 4.3) with relatively reduced (74%) *Ea* during the first irrigation, due to excessive irrigation application with ~18% *DD* and ~8% tail drain runoff losses (Table D1: Appendix D). In contrast *Ea* during the second irrigation was increased (85%) resulting in reduced *DD* (6%) and tail drain runoff (9%) losses. The *Er* ranged above 95% while *DU* ranged above 81% during both irrigations.

In Pakistan, at site P1, *Ea* was the lowest at 50% on NB and 72% on WB followed by site P3 at 68% on NB and 99% on WB (Table 4.3). At site P2, *Ea* was the highest at 92% on NB and 99% on WB (Table 4.3). At site P1 excessive irrigation was applied, that resulted in increased *DD* (24 mm/irrigation to 61 mm/irrigation), while at site P2 the *SMD* was not fully met. Overall data showed significantly (P=0.05) lower *Ea* on NB that ranged from 46% to 97% (70% average) in comparison with the WB, that ranged from 69% to 100% (90% average) on all sites. At site P1 both bed sizes and at site P3 NB only, the lowest *Ea* was experienced with over-irrigation and large *DD* (up to 61 mm/irrigation). However, improved *Ea* (> 90%) for site P2 and P3 (WB only) reduced *DD* (< 6 mm/irrigation), but *SMD* was not met (Table D2: Appendix D).

Results showed higher *Er* at site P1 that ranged above 99% followed by site P3 at 91% and then site P2 at ~80%. The overall data showed a significantly (*P*=0.05) increased *Er* on NB (94%) than for the WB (86%), thus *SMD* was generally not fully met on the majority of WBs. A comparison of irrigation volume applied indicated a number of over-irrigated events (i.e. Er > 99%), especially at site P1 (both bed sizes)

and site P3 (NB only), in other words, the *SMD* was met, but with considerable losses below the root zone. While at site P2 (both bed sizes) and site P3 (WB only) Er ranged down to 77% and thus crop was mostly under-irrigated. Overall results showed comparable *DU* on WB (79%) and NB (76%) as given in Table 4.3.

The average simulated surface water storage fraction (ratio between surface storage of irrigation water in furrows to the total inflow) at the time of water arrival to the furrow tail end on site P2 was higher (30%) than site P1 (24%). The increased surface water collected at the tail end of site P2 caused dry bed middle at the head reaches. Comparison of results indicated increased surface water storage (Table 4.4) for larger furrows in comparison with smaller furrows at all sites indicating larger impact of furrow sizes on irrigation applications in small farms. For instance, at site P1 the surface storage fraction on the narrow furrow (Table 4.2) of NB was 20% (Table 4.4), which was 8% less than the wide furrow of WB with 28% surface storage at the same site. Similar trends were also identified at P2 (Table 4.4).

Table 4.4:Volume of surface water storage in furrows at the time of water arrival to
tail end on three sites with two bed sizes in sandy clay loam, Pakistan (SD in brackets)

Performance	P	21	P	2	P	3
Parameter	NB	WB	NB	WB	NB	WB
Inflow rate (L s ⁻¹)	2.95	4.86	1.62	2.71	2.36	2.96
Total inflow	6.250	6.875	2.400	4.200	4.075	4.550
(m^{3})	(1.06)	(0.18)	(0.28)	(0.57)	(0.32)	(0.64)
Surface storage	1.225	1.950	0.875	1.000	1.175	1.300
(m^3)	(0.04)	(0.07)	(0.04)	(0.28)	(0.18)	(0.00)
Surface storage (%)	20	28	36	24	29	29

4.4 Discussion

4.4.1 Sites conditions and existing management

The soil moisture deficit *(SMD)* identifies the required irrigation application, which plays a key role in the irrigation performance evaluation and scheduling (Luis S 1999). The *SMD* values of ~100 mm under lengthy furrows of Australian farms are generally considered suitable to ensure irrigation schedules with long gap periods due to higher cost associated with the irrigation and to make effective use of the inseason rainfall. In contrast, the weekly canal irrigation schedules in north west Pakistan and the potential pan evaporation values of ~8 mm day⁻¹ (Javaid & Khalid Usman 2009) during the summer season had made the 50 mm to 60 mm irrigation application very common. Our study showed the *SMD* values in the range of 76 to 100 mm in Australia and ~60 mm in north west Pakistan, which well represented the existing irrigation management and scheduling practices, thus contributed in the accurate benchmarking of the existing irrigation performance.

This study showed largely variable bed furrow dimensions spatially from farm to farm, which also indicated a temporal changing trend at different crop stages. For instance, the measured bed furrow dimensions significantly varied between both Australian farms and between two bed sizes on three Pakistani farms. The significantly larger furrow widths of WB in all Pakistani sites were due to the increased space between furrows, which easily accommodated the excavated soil compared with smaller space on NB. Bed furrow dimensions were also variable during the cropping season evidenced by the marginally deeper furrows in the early season (site P2) than for the late season (site P3). Variable bed furrow configurations of PRB have been shown to affect irrigation and crop performance (Hassan et al. 2005; Jin et al. 2007; Akbar et al. 2010), while the variable furrow dimensions of current study further increased its importance in irrigation management studies.

Furrow spacing and compaction by trafficking affected the water advance rate and inflow volume per unit area respectively. For instance, at site A1, the *Ta* during the first irrigation was 13% longer than for the second irrigation due to reduced soil infiltration capacity of relatively wider and compacted furrows that imparted greater

momentum to flowing water during the second than the first irrigation. According to Allen and Musick (1992) furrow compaction can enhance the water advance up to 45%. At site A2, the furrows were compacted and wide (~61 cm TW) but the furrow spacing was small (100 cm) with a bed top width of ~39 cm, which caused flow crossing between the wet and dry furrows during the second irrigation, when soil was cracked, thus making the water advance rate slower. The furrow spacing impact on total inflow volume can be evidenced by the 32% larger irrigation application to the NB than for the WB, which closely agrees with the findings of Hassan et al. (2005) and Akbar et al. (2009) under similar short furrows.

This study showed a close relationship of the Tco with Ta under actual field conditions, as the majority of farmers followed this rule. However, the Ta was affected by several factors. For instance the Ta was increased at low Q and high infiltration capacity soil, which agrees with the findings of Allen and Musick (1992), availability of larger cracks under dry soil conditions (e.g. site A2 irrigation 2 caused 24% larger infiltration than irrigation 1), which agrees with the findings of Bautista and Wallender (1993b) and furrow surface roughness due to abundance of weed (e.g. site P1) and manual weeding (e.g. site P3). Other reasons may be the deeper root penetration of cotton and maize crops as mentioned by Meek et al. (1992) and Mitchell et al. (1995), (e.g. site P1 and P3 that were at flowering stage), which seems to have made the soil more infiltrative. Other evidence was the 34% longer Ta caused 15% longer Tco during the second than for the first irrigation at site A2. Impact of wheel trafficking on water advance was significant (21%), which can largely affect the irrigation performance under longer furrows as mentioned by Allen and Musick (1997). Thus, a large number of factors affected Ta. Therefore, development of guidelines to relate *Tco* to *Ta* under variable field conditions seems to be instrumental in enhancing the irrigation performance.

The observed low DU and reduced lateral infiltration under the short furrow lengths in Pakistan is a serious problem. The local farmers generally divide the field into smaller segments (15 to 20 m long) with bunds, which control the downstream flow of water to increase the opportunity time in the field head reaches and to reduce the tail end overtopping. This study revealed low DU as a consequence of poor irrigation
management, reduced opportunity time, and short Tco due to the short furrow lengths. Consequently, the tail end was overtopped, while the head reaches received less irrigation depth, which is contrary to lengthy furrows (Trout 1990), thus the DU was negatively affected, and the poor lateral infiltration issue was exacerbated, at head reaches. Therefore, development of guidelines to improve DU can be beneficial in saving labour and in enhancing the lateral infiltration.

4.4.2 Renovation methods and irrigation performance interactions

This study showed variable renovation methods in actual practice. Renovation methods largely affect infiltration properties of soil (Akbar et al. 2010), which played a key role in the existing irrigation performance. For instance, the existence of loose soil conditions (during the first irrigation on site A1) due to deep cultivation prior to crop sowing made the furrow surface rough and permeable due to less compaction and trafficking, which enhanced the infiltration capacity of soil (9% increased after 900 minutes of infiltration opportunity time during the first than for the second irrigation), which conformed to the findings of Hulugalle et al. (2007). The loose soil conditions exacerbated the furrow filling with sediment. Consequently, a large area was inundated as a result of shallow furrow sizes, thus the water advance rate was retarded due to bed overtopping, and irrigation performance was negatively affected. On the other hand, shallow rotary hoeing (e.g. site P2) caused quicker Ta due to furrow sealing, which forced short *Tco*, and exacerbated the poor lateral infiltration issue. Poor lateral infiltration was also reported by Jin et al. (2007) and Akbar et al. (2007) in the sandy clay loam. Thus, appropriate renovation method seems vital for improving the soil infiltration properties, mitigating the poor lateral infiltration and for improving the irrigation performance.

The water advance along the furrows showed linear response with the square root of time (Philip 1969), which indicates that infiltration was dominated by capillary forces during the whole irrigation events for both soils. Thus the soil properties created by pore size distribution, clay type, soil structure and the stability of soil structure, a manifest of soil management, influenced irrigation performance. The practices that deliver water to the soil will also impact on irrigation performance.

4.4.3 Sites and irrigation performance interactions

This study revealed a poor irrigation performance in the majority of sites. However, the Ea identified in the range of 68% to 85% in Australia, was higher than reported in early studies (Raine & Bakker 1996; Smith et al. 2005) under sugarcane and cotton crops, which show improvement by using PRB. In Pakistan, Ea ranged from 45% to 99% with the majority of low Ea on NB and high Ea on WB. The major reasons were the current common over-irrigation practices in Australia and on NB (common bed size) in Pakistan. However, the high Ea values of WB were indicative of insufficient irrigation applications in sandy clay loam. The Er was above 96% in Australia due to excessive applications, while in Pakistan it was significantly reduced on WB than on NB. Current evaluations showed that the DU of all irrigation events in Australia were negatively affected by large cumulative infiltration at head reaches. In contrast, 67% of measured irrigation events in Pakistan (i.e. site P2 and P3) were affected by large cumulative infiltration at tail reaches than at head reaches. Further investigation revealed that DU was adversely affected in cases when cumulative infiltration was larger at the tail than in the head reaches. The low DU associated with high *Ea* for WB is consistent to Sepaskhah and Ghahraman (2004).

This study showed a higher potential for deep drainage due to irrigation excess that ranged 7 mm/irrigation to 47 mm/irrigation in Australia and up to 61 mm/irrigation in Pakistan, as the furrows were blocked at the tail end, which is consistent to the findings of Raine and Bakker (1996) and Smith et al. (2005). The recent increasing trend of land loss due to accelerated waterlogging and salinity problems in the Indus Basin (Yasin et al. 2002; Qureshi et al. 2008) can partly be attributed to the increased infield deep drainage losses under the most common NBs. Thus, deep drainage losses has a greater role in improving the current irrigation performance.

4.4.4 Irrigation management and performance interactions

This study indicated irrigation management as one of the major factors affecting the existing irrigation performance of PRB farming systems. The inflow volume, a manifest of the effectiveness of irrigation management for replenishing the soil moisture deficit with uniformity and efficiency (Khatri 2007; Smith et al. 2009), was variable at different sites for replenishing 76 mm to 100 mm *SMD* in Australia and

~60 mm in Pakistan, which showed that irrigation management is poor in both countries. The uninformed selection of *Tco*, based on either relative to water arrival to the furrow tail end, furrow filling, or any other farm manager preferences, negatively affected the irrigation performance. The farmer interest on both Australian sites was to replenish the *SMD* and to avoid dry field sections, however irrigation was applied for longer periods than required, which caused excessive deep drainage losses. Similarly, on site P1 in Pakistan fewer plots per turn were irrigated simultaneously, thus more Q per furrow was applied. However, at site P2 more plots were irrigated, thus less Q per furrow was applied.

The Q and Tco were revealed as perhaps the most important irrigation management factors. Inappropriate Q was also reported as the main reason of inefficiency by Raine and Bakker (1996) and Smith et al. (2005) in Australia and Kalwij (1997) in Pakistan. The selection of Q and Tco and their appropriate combination according to specific field conditions affected DU as mentioned by Raine and Bakker (1996). For instance, the higher Q with short Tco on WB at site P3 caused increased DU, when compared with NB at low Q with longer Tco. The low Q and prolonged Tco at site A2 negatively affected DU during the second more than for the first irrigation. Nevertheless, the current irrigation management was not capable of meeting the *SMD* for variable furrow spacing evidenced by 32% less irrigation application to WB (60 mm) than NB (88 mm). The reasons were attributed to ~9% larger Tco on NB than WB and ~52% larger Q on WB than NB under current management. Thus, development of decision support guidelines, especially for selection of Q and Tco, can be instrumental in improving the current irrigation performance.

Poor lateral infiltration under WB has been shown to adversely affect water productivity of many regions especially under the structurally unstable loamy soils with low lateral hydraulic conductivity (Hassan et al. 2005; Akbar et al. 2007; Jin et al. 2007). The current study revealed that soil moisture distribution was not uniform between the bed shoulder and middle on Vertisol and majority of WB in sandy clay loam. Beside the inappropriate soil management in both soils, one of the reasons was the reduced water application to WB than for the NB, thus the poor lateral infiltration issue was also exacerbated due to sub-optimal irrigation management.

4.4.5 Performance improvement options of existing permanent raised beds

The current soil management of PRB systems was dominated by annual or occasional deep ripping in Australia and seasonal shallow harrowing in Pakistan, which placed the sustainable crop production objective at risk. This study showed that poor soil management/renovation methods adversely affected the irrigation performance of PRB farming systems in both countries. Thus, cultivation practices/renovation of beds seem to be among the current major issues in both countries and therefore need a comprehensive evaluation for developing decision support guidelines according to the changed soil conditions.

This study showed that poor lateral infiltration in wide PRB farming systems is one of the main limiting factors in low infiltration soils and subsided beds constrained by the existing poor agronomic and irrigation management practices, which was also pointed out in several past studies (Lucy 1993; Akbar et al. 2007; Jin et al. 2007). The variable bed furrow configurations and resulting variable irrigation performance further complicate the lateral infiltration issue due to lack of understanding. Therefore, a comprehensive evaluation of lateral infiltration and its interaction with variable renovation methods and irrigation management needs to be undertaken for further refinement of understanding and for developing decision support procedures.

The existing irrigation performance is largely variable/poor in both countries and reasons are generally attributed to poor irrigation management and inappropriate soil management practices according to Gill (1994), Kalwij (1997), Raine et al. (1998), Smith et al. (2005) and Gillies (2008). Irrigation management strategies have been successfully tested by using simulation modelling in many past studies (Kalwij 1997; Raine et al. 1998; Smith et al. 2005) but have received only limited considerations under the actual field conditions of irrigated PRB in Australian farms in general and under Pakistani farms in particular. This study showed that current poor irrigation management in both countries and needs to be evaluated for developing decision support guidelines for the improved performance of existing PRB farming systems.

4.5 Conclusions

The existing soil and irrigation management practices of PRB are sub-optimal in both Australia and Pakistan, thus irrigation performance is poor. Several important conclusions emerged from this benchmarking study as listed below:

- The existing soil management of PRB practices involve intensives tillage, which is destructive to soil structure, thus negatively affecting PRB performance and placing the sustainable crop production at risk.
- The current irrigation performance of PRB is poor demonstrated by *Ea* down to 68% in Australia with up to 47 mm/irrigation deep drainage losses and down to 50% in Pakistan with up to 61 mm/irrigation deep drainage losses and the current irrigation management seems to be largely poor.
- Wide PRBs are more susceptible to poor lateral infiltration, especially under sandy clay loam, while renovation methods and irrigation management seems to have a greater role in lateral infiltration of wide PRBs.
- The water advance showed linear response with the square root of time, which shows that capillary forces dominated in infiltration. Thus, pore sizes and their distribution in the beds, a manifest of soil structure and soil management, either side of the furrow control irrigation performance.
- The SIRMOD evaluation was effective in benchmarking the existing irrigation performance, thus can be instrumental in mitigating the excessive deep drainage losses and in controlling the frequent under-irrigation practices on wide PRBs. However, the uniform soil moisture distribution assumption between the wetted furrows seems detrimental for wide PRBs.

Developing decision support guidelines for selection of appropriate renovation methods, irrigation management and field design of PRBs seems to be very beneficial for improving the current raised bed performance in both countries (Australia and Pakistan).

CHAPTER 5: Impact of PRB renovation on irrigation performance and productivity

5.1 Introduction

Renovation methods of PRB systems vary around the globe (Roth et al. 2005), and depend on traditional practices, farmer preferences, crop type, and available machinery. The effect of renovation methods on irrigation performance has generally not been studied in previous research. Hence, detailed studies on the management of soil structure and the effects on hydraulic properties and performance of PRB in an irrigation environment are sparse. The preliminary evaluation of irrigation performance reported in Chapter 4 suggested that improper renovation of beds is one of the main factors in the performance of existing PRB systems. Tillage has been shown to affect soil hydro-physical, chemical and biological properties (Smith et al. 1983; Hulugalle et al. 2004; Tullberg 2010). Thus, if PRB renovations using tillage affect soil properties then renovation would also be expected to influence irrigation performance, crop yield, and *WP*. This chapter describes field experiments to evaluate the comparative *WP* performance of different PRB renovation methods on an irrigated Vertisol. Chapters 6 and 7 report the investigation of lateral water movement into the beds.

5.2 Material and methods

5.2.1 Site description, treatments and experiment layout

The field trials for this experiment were conducted at the Australian field site A1 (as described in Chapter 3) used in the benchmarking study (Chapter 4) and commenced after the soybean crop. The experiments were conducted over four seasons (from 2010 winter to 2012 summer) with two summers and one winter crop, and with one winter fallow. The crops were consecutively 2010 wheat, 2011 hemp, and 2011 corn.

Three PRB renovation treatments were applied to the one season old PRB after the 2009 soybean season. The three treatments were:

<u>NT: No tillage:</u> The beds were not disturbed and furrows were cleaned with bed former. Only the soil in the sown rows was disturbed by the seeder during planting (Figure 1a).

<u>SC: Shallow cultivation:</u> The whole top of the bed cultivated to a depth of about 10-15 cm without relocating furrow positions using a six metre wide cultivator prior to furrow cleaning with bed former (Figures 5.1b and 5.2a),

<u>BP: Blade ploughing:</u> The furrows were reshaped and the soil in the bed opened by horizontal blades that were passed through the base of the bed at ~30 cm depth opening the soil without disturbing the roots of the previous crop and without causing any soil inversion using a 200 cm wide blade plough (Figures 5.1c and 5.2b).



Figure 5.1: Layout of three PRB renovation methods including (a) NT: no tillage, (b) SC: shallow cultivation and (c) BP: blade ploughing treatments



Figure 5.2: Establishment of two renovation treatments for 2010 wheat crop after 2009 soybean harvest at site A1, Marinya farm, Cambooya, Australia

The three treatments were thrice replicated using a randomised block design as shown in Figure 5.3. Each treatment consisted of three beds surrounded by a three bed wide buffer area to avoid interference of treatments. The two beds of buffer area were treated similar to respective adjacent treatments to reduce the influence of edge effects across the treatment furrows and beds.



Figure 5.3: Layout of bed renovation treatment plots at site A1, Australia

The various soil tillage treatments applied at the start of each season for each treatment are described in Table 5.1.

Cron	Voor		Renovation treatmen		
Crop	rear	NT	SC	BP	
		Furrow reshaped	Furrow reshaped	Furrows reshaped	
Wheat	2010	Soil in bed	Beds cultivated to	Beds blade ploughed	
		undisturbed	15 cm depth	at 30 cm depth	
	2011	Furrow reshaped	Furrow reshaped	Furrow reshaped	
Hemp		Soil in bed	Soil in bed	Soil in bed	
		undisturbed	undisturbed	undisturbed	
		Furrow reshaped	Furrow reshaped	Furrow reshaped	
Corn	2011	Soil in bed	Beds cultivated to	Beds blade ploughed	
		undisturbed	15 cm depth	at 30 cm depth	

 Table 5.1:
 Soil tillage applied to each treatment over three cropping seasons at site A1

<u>Note</u>: Blade ploughing for 2010 wheat used blades vertically inclined at an angle ~25 degrees to the direction of travel and backward rake angle of ~15 degrees. Blade ploughing for 2011 corn used blades with a vertical inclination of zero degrees to the direction of travel and a backward rake angle of 35 degrees.

5.2.2 Agronomic management practices

The first season treatments were established on one season old PRB and wheat was planted on 7 June 2010. Eight rows of wheat per bed on 20 cm row spacing were planted at a seed rate of 74 kg ha⁻¹ using a 6 m wide seed drill equipped with double disc openers (Figure 3.3b). The same quantities of inputs (seed, fertiliser, herbicides) were applied to all treatments. Nitrogen was applied as urea at a rate of 74 kg ha⁻¹ at sowing and again at 100 kg ha-1 on 10 July 2010. After planting, glyphosate isopropylamine was sprayed at prescribed rate on 9 June 2010 to reduce weeds. Only a single irrigation was applied during the cropping season (on 19 June 2010) due to the wet winter weather. The 2010 wheat crop was harvested as hay at ~5 cm above ground surface on 15 November, 2010.

Summer crop sowing in 2010-11 was delayed due to broad scale severe regional flooding. The hemp crop was planted late on 26 January 2011 without disturbing any treatment, due to wet soil conditions. Furrows of all treatments were cleaned (excavated) after sowing. Urea was applied at the rate of 74 kg ha⁻¹ at sowing and 100 kg ha⁻¹ on 16 February 2011. Four irrigations were applied during the season. The initial two irrigations were applied quickly at low *SMD* early in the season in response to poor crop establishment observed in the middle of the bed. The third and fourth irrigations were applied at high *SMD*. The 2011 hemp crop was harvested on 15 May 2011 leaving less than 5% crop residues in the field.

The third season corn crop was planted on 3 October 2011 after a winter fallow on freshly renovated beds. The BP treatment was applied with a newly designed bed former with two sharp horizontal blades for reducing soil disturbance. Round up weedicide was sprayed at the prescribed rate before sowing. Two rows of corn were planted on the bed shoulders at 100 cm row spacing and ~20 cm plant to plant spacing. Liquid nitrogen fertiliser was applied at prescribed commercial application rates at crop early stage. Urea fertiliser was applied at 74 kg ha⁻¹ before first irrigation and 100 kg ha⁻¹ before second irrigation. Two irrigations were applied, the first on 12 November 2011 and the second on 7 January 2012. The 2011corn crop was harvested on 15 February 2012 leaving ~5 cm crop residues above ground surface. All crops in all treatments were sown with the same double disc drill.

5.2.3 Sampling and instrumentation

Weather data including rainfall, temperature, humidity, and wind speed were recorded using a solar powered automatic weather station located adjacent to the experimental area during the three cropping seasons. The furrow dimensions data including TW, MW, BW and D (Figure 3.2c) were recorded at the head, middle and tail sections of each treatment at the time of sowing before irrigation, after irrigation and at harvest of all three seasons. Irrigation data including Q, water advance at multiple points along the furrow at 100 m intervals and *Tco* were recorded using IRRIMATETM tools including siphon flow meters, advance sensors (shown in Figure Figure 5.4 and described in section 3.6.3.1).



Figure 5.4: Overbank systems and IRRIMATETM meters for measuring inflow rate. Insert shows water advance sensors and soil moisture sensors layout during irrigation. Photo taken during 2010 wheat season at site A1, Cambooya, Australia

Volumetric soil moisture content was monitored using three capacitance probes (Sentek EnviroSCAN[®]) per treatment during all seasons from sowing to harvest. These sensors were installed at 50 m distance from head channel in replication 2 only (according to the layout shown in Figure 3.8 and described in section 3.6.2). During the 2010 wheat season the sensors were positioned to record at profile depths of 10, 20, 30, 40, 60, 80 and 100 cm as shown in Figure 3.8. During the 2011 hemp and 2011 corn seasons the sensors were positioned at 10, 20, 30, 40, 70 and 100 cm depths. This set up provided a complete record of soil moisture variation at 10 minute intervals across a half bed width and up to 100 cm depth during all three

cropping seasons. Gravimetric soil moisture and soil bulk density were measured in the 0-30 cm surface profile at 10 cm intervals using the thermo-gravimetric method described in section 3.6.1 (Chapter 3). Three replications were collected in each treatment at sowing, before irrigation, after irrigation and at harvest.

During the 2010 wheat season, the plant density (number of plants per metre row length), average plant height per row, and total plant biomass data were collected near maturity stage at harvest using a sample plot size of 2 m x 1 m. Three plant samples, one per each bed in a treatment, were collected from the head, middle, and tail sections of all treatments. The sample size spanned the whole or half bed width to account for possible growth variations across the bed while the 1 m perpendicular length of sample was adequate (Payero & Harris 2010) to cover spatial variability and to make the management possible with limited resources. These samples were packed in paper bags. The green weight of all samples was recorded and the samples were placed in an oven at 45° C. The oven dry weight of the samples was checked twice daily after 5 days of drying to ascertain any further weight loss. The final dry weight was recorded on the seventh day when no further weight loss was observed.

The plant height, plant density, and dry plant biomass data for the 2011 hemp crop were recorded at crop harvest from each treatment using sample plot size of (1 m x 1 m, spanning half bed width) with three replicates at head, middle and tail sections of each treatment plot. The above data for each one metre row length of the three rows per sample were also recorded separately to identify yield variations across the bed width. The samples were air dried for two weeks and the oven dry weight of samples was recorded.

The 2011 corn plot samples from a 1 m section of row were collected at harvest. Samples were taken at the head, middle, and tail sections of the treatment plots with three replicates per treatment. The number of plants and the three plants height within the one metre row were also recorded. After drying the ears, the grain was separated from the husk and then placed in an oven at 45^oC until fully dried. The dry weight of grain and dry plant biomass were recorded separately. Dry weight of 100 grains per sample, randomly selected, was also recorded.

5.2.4 Irrigation performance and water productivity

Irrigation management was based on controlling Q and Tco. Irrigation application was stopped when the furrows were filled with water and before the beds overtopped at the bottom end of the field. Irrigation was generally scheduled to occur at 65% depletion level of soil moisture in the 100 cm depth profile during the cropping season. However, due to water restrictions in the locality and demand for water in other parts of the farm, the irrigation scheduling plan was moderately adjusted on occasions. Irrigation management aimed to maximise WP with management adjusted based on expertise from previous irrigations results. In particular the inflow rate to the BP furrows was increased relative to the other treatments by increasing the head in the head channel. While field design limitations restricted adoption of optimum Q, Tco and total volume of irrigation applied (based on calculations of *SMD* prior to irrigation) were monitored and managed through observation.

The irrigation characteristics of four furrows were measured in each replicated treatment plot. The irrigation performance was determined using the procedure described in Chapter 3 and Chapter 4. The infiltration parameters were determined by adjusting both the f_0 and Manning's 'n' coefficient in the IPARM model to ensure that the observed water advance matched the simulated advance. The f_0 ranged from 0.0001 to 0.00005 during the 2010 wheat season and was 0 during the 2011 hemp and 2011 corn seasons. The irrigation performance indicators (*Ea*, *Er*, *DU* and total inflow volume applied) were evaluated using the SIRMOD (III) model. The furrows were blocked at the tail end thus deep drainage was the only source of loss, which was calculated according to equation 3.1 (Smith et al. 2005). Calibration of IPARM and SIRMOD for the three seasons is provided in Figure B3 (Appendix B).

The input water productivity (*WP*) was calculated as the ratio between the dry weight of crop yield (grain or straw) per unit area and total water applied during the cropping season. The total water applied was calculated by adding the seasonal rainfall (assumes all rainfall is effective) and total irrigation water applied during the season. Statistical analysis was conducted according to the procedure described in section 3.7. The experimental details of measurements during the three cropping seasons for three PRB renovation treatments are summarised in Table 5.2.

Table 5.2: Experimental detailsrenovation plot during three cro	of measurement of soil proper pping seasons at site A1, Australi	ties, irrigation, crop yield and ia (SM = soil moisture, BD = bul	water productivity per each PRB k density).
Measurements	Wheat 2010	Hemp 2011	Corn 2011
Crop sowing date	7 June 2010	26 January 2011	3 October 2011
<u>SM and BD</u> No of samples & position Sampling depths Grav. SM & BD method	3 – 1 each @ plot head, mid & tail 0-10, 10-20 & 20-30 cm Undisturbed cores 5 cm diameter. x	3 – 1 each @ plot head, mid & tail 0-10, 10-20 & 20-30 cm Undisturbed cores 5 cm diameter. x	3 – 1 each @ plot head, mid & tail 0-10, 10-20 & 20-30 cm Undisturbed cores 5 cm diameter. x
Volumetric SM method Sampling time	Gravimetric SM x BD Sowing, pre & post irri. & harvest	Gravimetric SM x BD Sowing, pre & post irri. & harvest	Gravimetric SM x BD Sowing, pre & post irri. & harvest
<i>SMD prior to irrigation</i> No of samples & position Distance from furrow centre Profile depth Equipment Method	1 – © 50 m distance from plot head 33, 67 and 100 cm 0-100cm Sentek EnviroSCAN [®] SMD=FC – Antecedent SM	 @ 50 m distance from plot head 33, 67 and 100 cm 0-100 cm Sentek EnviroSCAN[®] SMD=FC – Antecedent SM 	 @ 50 m distance from plot head 33, 67 and 100 cm 0-100cm Sentek EnviroSCAN[®] SMD=FC – Antecedent SM
<u>Irrigation data</u> No & position of furrow dimensions No & position of inflow rate No & positions of water advance No of furrows measured Time to cut-off reading No of irrigations measured	3-1 each @ plot head, mid & tail 1-siphon flow meter per treatment 5- @ ~100 m distance along furrow Four furrows per treatment plot One per treatment plot one	3-1 each @ plot head, mid & tail 1-siphon flow meter per treatment 5- @ ~100 m distance along furrov Four furrows per treatment plot One per treatment plot Four	3-1 each @ plot head, mid & tail 1 - siphon flow meter per treatment v 5 - @ ~100 m distance along furrow Four furrows per treatment plot One per treatment plot two
<u>Harvest data</u> No of samples & positions Sample size Data measured	9 – 3 each @ plot head, mid & tail 2 m x 1 m (span whole bed width) Tillers density, height and dry biomass	9 – 3each @ plot head, mid & tail 1 m x 1 m (span half bed width) Plant density, height and dry biomass	9 – 3 each @ plot head, mid & tail 1 m x 1 m (span half bed width) Plant density, height, dry biomass & 100 grain weight
Crop harvest date	15 Nov 2010	15 May 2011	15 Feb 2012

5.3 Results

5.3.1 Weather conditions

The winter (2010 wheat season) had an average temperature of 13^{0} C, average humidity of 72% and total rainfall of 196 mm. During the summer (2011 hemp and 2011 corn) seasons the average temperature was ~17⁰C and average humidity was ~69%. The total rainfall was 228 mm and 302 mm during the 2011 hemp and 2011 corn seasons, respectively. The majority of the individual rainfall events applied < 10 mm, but a maximum of 32 mm, 25 mm and 40 mm daily rainfall was recorded once during the 2010 wheat, 2011 hemp and 2011 corn seasons, respectively. The 2010 wheat harvest was followed by severe weather conditions, which delayed the 2011 hemp sowing. A total of 211 mm rainfall was recorded in the last half of December 2010, with 132 mm rain in a two day period (27 to 28 December, 2010). Severe flooding occurred in the surrounding areas and overland flow occurred in the experimental area. A monthly record of weather data is provided in Table A3 (Appendix A).

5.3.2 Observation of machinery operations

The available machinery consumed one extra machinery operation time and energy per bed for the SC and BP treatments compared with the NT treatment prior to furrow cleaning. The draft/energy required per operation was also higher for establishing the BP treatment (according to the farmer), especially during the 2010 wheat season due to inadequate machinery design and set up. Field coverage of the available single bed per pass machinery for the BP treatment was lower compared with the three beds per pass machinery for the SC treatment. Moreover, the operator was comfortable sowing the NT treatment with the available double disc, zero till seeder. In contrast, the loose soil on the SC and BP treatment beds caused hindrance during sowing operations. Thus, machinery design, set up and type of renovation method affected the machinery operation performance at this site.

The major hindrance faced was the frequent seed drill clogging and reduced steering control caused by the loose soil conditions when applying the SC and BP treatments. For the 2010 wheat BP treatment, the cutting edge sharpness and angle on the

available thick cutting blade was not appropriate to the soil conditions. This resulted in occasional pulling out and dragging of the previous soybean crop stubble affecting the bed and furrow shapes as well as the bed surface roughness. This issue was resolved for the 2011 corn BP treatment when a new bed former (adequately designed and set up) was used.

Another problem was the insufficient strength of the loose Vertisol soil to withstand the pressure inserted by the double discs, thus making it difficult for the operator to keep a uniform seeding depth and seed rate for the SC and BP treatments. Moreover, the closely spaced (20 cm) discs of zero till seeder did not provide sufficient clearance for the passage of loosely held stubble trash, especially on the SC treatment. This caused occasional accumulation at the front of the disc opener for both SC and BP treatments.

5.3.3 Effect of PRB renovation method on bed and soil properties

5.3.3.1 Furrow dimensions

The furrow dimensions varied between tillage treatments and seasons (Table 5.3). The largest furrows during the 2010 wheat and 2011 corn seasons were on the BP and NT treatments with the smallest on the SC treatment. The difference in furrow dimensions between the tillage treatments was negligible during the 2011 hemp season when no renovation occurred. Furrow top widths were 8 and 12 cm narrower for NT and SC respectively, compared with BP furrows during the 2010 wheat season. Furrows were also 2 cm and 3 cm shallower for NT and SC treatments compared with the BP treatment during the 2010 wheat season. During the 2011 corn season the furrows were wider on BP followed by NT and then SC treatment as given in Table 5.3.

Crong	Treatment	Average furrow configurations (cm)				$\mathbf{I}(\mathbf{m})$
Crops	Treatment	TW	MW	BW	D	L (III)
	NT	53	35	15	12	
Wheat 2010	IN I	(2)	(2)	(2)	(1)	
	SC	49	32	16	11	165
		(3)	(2)	(2)	(1)	405
	PD	61	39	19	14	
	Dr	(2)	(3)	(3)	(1)	
	NT	47	32	20	13	
	191	(5)	(2)	(5)	(1)	465
Homp 2011	SC	48	34	22	13	
Themp 2011		(6)	(3)	(3)	(1)	
	PD	49	35	21	13	
	Dr	(6)	(3)	(4)	(3)	
	NT	54	41	20	13	
	IN I	(3)	(4)	(4)	(1)	465
Corn 2011	SC BP	51	37	25	11	
		(5)	(4)	(5)	(1)	
		58	44	29	11	
		(4)	(4)	(5)	(1)	

Table 5.3: Average furrow dimensions (according to layouts shown in Figure 3.2c)under three PRB renovation treatments during each cropping seasons (SD in brackets)

The furrow dimensions varied during the 2010 wheat season after fresh renovation (Figure 5.5). In general, the furrow top width and depth decreased over time. In contrast, the furrow middle and bottom widths increased over time. This change continued into the following 2011 hemp season, which was not renovated. The furrow top width was reduced by ~11%, 2%, and 20% for the NT, SC, and BP treatments respectively from the 2010 wheat to 2011 hemp seasons. The full data record of furrow dimensions is provided in Table A4 (Appendix A).



Figure 5.5: In-season variations in furrow dimensions (TW = top width, MW = middle width, BW = bottom width and D = furrow depth) during 2010 wheat season (vertical bars show SD)

5.3.3.2 Bulk density

The average bulk density (gm cm⁻³) for the 2010 wheat season of one season old PRB over 0-30 cm depth was significantly (P=0.05) lower with BP when compared with NT (Figure 5.6). However, the bulk density of the SC treatment was not significantly (P=0.05) different from either the NT or the BP treatments.



Figure 5.6: Average seasonal bulk density of three renovation treatments during three cropping seasons at site A1, Australia (Vertical bars show SDs)

The average seasonal bulk density over 0-30 cm for the 2011 hemp season of BP was still significantly (P=0.05) lower than the NT and SC treatments (Figure 5.6) while this difference was not significant between NT and SC treatments. However, the bulk densities increased up to 6% for SC (~11% increase in 0-10 cm depth), followed by BP (4% increase) and the least with NT (3% increase) from wheat to hemp season.

The average seasonal bulk densities for the 2011 corn season over 0-20 cm depth were significantly (P=0.05) lower with BP than for the other two treatments (Figure 5.6). Bulk densities over 0-30 cm depth were similar for SC and BP and were 2% and 6% respectively, lower than NT. The seasonal increase in bulk density during the 2011 corn was more for SC (~8%) than the other two treatments (~5%).

The results showed a significantly (P=0.05) higher and similar bulk density in the 0-20 cm depth for the NT and SC treatments respectively compared with BP when renovation was freshly applied (i.e. wheat and corn seasons). However, the bulk density of 20-30 cm depth was similar between treatments during all seasons.

The soil moisture did not confound the treatment effect on bulk density because sampling was done when the soil moisture was not significantly (P=0.05) different between treatments. However, the bulk density tended to reduce at high moisture level and to increase at low moisture level during the cropping seasons irrespective of treatment differences Table A6 & Figure A2 (Appendix A).

5.3.3.3 Soil moisture deficit and infiltration properties

The *SMD* was ~100 mm prior to irrigation for all treatments during the 2010 wheat season (Figure 5.7). During the 2011 hemp season the soil was dry at the surface (0-10 cm) and wet at deeper depths in early crop season. The *SMD* was lower (~40 mm) prior to the initial two irrigations and higher (~100 mm) prior to the last two irrigations. The *SMD* in 100 cm depth prior to irrigation during the 2011 corn season was lower (60-74 mm) during the first than the second irrigation (106-134 mm).



Figure 5.7: Average soil moisture deficit in 100 cm depth prior to irrigations during the three seasons (vertical bars show SD)

The average cumulative infiltration depths were plotted against time for all treatments and crop seasons (Figure 5.8). Infiltration was significantly higher on freshly applied renovation treatments than on un-renovated beds. There were also larger differences between the renovation treatments when renovation was freshly applied. For instance, average total cumulative infiltration after fresh renovation (e.g. 2010 wheat and 2011 corn seasons) was significantly (P=0.05) higher for BP than for the other treatments. The BP treatment infiltration after 400 minutes was 23%

higher than the NT (89 mm). The beds were not renovated during the 2011 hemp season and the cumulative infiltration was significantly lower (48-55%) than during the other two crop seasons. The differences in furrow infiltration affected the advance rate of irrigation water in furrows during the three cropping seasons as shown graphically in Figure A3-A4 (Appendix A).



Figure 5.8: Average cumulative infiltration under (a) freshly applied renovation treatments during 2010 wheat and 2011 corn (average of one irrigation to wheat and two irrigations to corn) crop seasons and (b) un-renovated beds during 2011 hemp season (average of 4 irrigations) at given time at site A1 (Vertical bars show SD)

The lateral wetting front movement into the centre of the bed was largely influenced by the renovation methods during the 2010 wheat season. The wetting front took \sim 10, \sim 9 and \sim 6 hours to wet the bed middle to field capacity (FC) for NT, SC and BP treatments respectively (Figure 5.9). The majority of the bed width was wetted above field capacity within 10 hours in all three treatments. However, the wetting of the bed profile continued till irrigation cut-off time.



Figure 5.9: Volumetric soil moisture (100 cm depth) at given distances from furrow centre during irrigation to 2010 wheat crop at site A1, Australia (FC = field capacity)

During the 2011 hemp season, the un-renovated beds were settled while compaction and sealing of furrows were obvious and *SMD* was less during the initial two irrigations. Therefore, lateral infiltration was poor and the wetting front could not reach to the centre of the bed during the first two irrigations. The lateral infiltration was marginally increased during the subsequent third and fourth irrigations due to cracking but *SMD* in bed middle was still not fully met (Figure C1:Appendix C).

The first irrigation in the 2011 corn season was applied at a *SMD* of 60-70 mm prior to the first irrigation. Lateral infiltration was slow for the NT and SC treatments. However, the BP treatment indicated faster lateral wetting (~6 hrs) into the middle of the bed (Figure 5.10). The *SMD* was higher (~130 mm) before the second irrigation and the lateral infiltration was quicker due to the presence of soil cracks. More detailed analysis of the lateral infiltration is provided in Chapter 6 and 7.



Figure 5.10: Volumetric soil moisture (100 cm depth) at given distance from furrow centre during the first irrigation to 2011 corn at site A1, Australia

5.3.4 Water applied and profile water content

A single irrigation was applied twelve days after sowing (Figure 5.11a) during the first 2010 wheat season. The average irrigation volume applied was 103, 116 and 160 mm to the NT, SC, and BP treatments, respectively (Table 5.4). Four irrigations were applied (Figure 5.11b) during the 2011 hemp season. Total irrigations of 191 mm (29, 31, 68, and 63 mm) was applied to NT, 204 mm (28, 39, 70 and 67 mm) to SC and 212 mm (26, 42, 70 and 75 mm) to BP treatment. Two irrigations (Figure 5.11c) of 224 mm (100 and 124 mm) to NT, 244 mm (107 and 137mm) to SC and 248 mm (126, 122mm) to BP during the 2011 corn season.



Figure 5.11: Average daily volumetric water content in 100 cm profile, irrigation applied and rainfall during three cropping seasons under three PRB renovation treatments on a Vertisol. (Vertical bars show SD in irrigations applications between treatments)

Сгор	Treatment	Irrigation (mm)	Rainfall (mm)	TWA (mm)
	NT	103 (6)	196	299
Wheat 2010	SC	116 (3)	196	312
	BP	160 (9)	196	356
^a Hemp 2011	NT	191 (16)	228	419
	SC	204 (18)	228	432
	BP	212 (21)	228	440
Corn 2011	NT	224 (14)	302	526
	SC	244 (16)	302	546
	BP	248 (17)	302	550

 Table 5.4:
 Total water applied (TWA) during the three cropping seasons in three PRB renovation methods at site A1, Cambooya Australia.

^aSC and BP treatments were not freshly applied and were only furrow cleaned

The 2010 wheat was harvested early as hay, therefore it required less irrigation applications. Less irrigation was applied to the 2011 hemp due to large residual soil moisture at planting and low farm irrigation water availability at crop maturity. In 2010 wheat fresh renovation with BP resulted in 19% higher total water input (Table 5.4), SC 5% higher-similar to that for the non-renovated 2011 hemp; in the 2011 corn fresh renovation only resulted in about 5% higher total water input for both SC and BP-similar to that in the non renovated beds for the non-renovated 2011 hemp.

The volumetric soil moisture content in the beds was consistently higher for BP than for the NT and SC treatments during the experiments (Figure 5.11). The bed middle (67-100 cm from furrow centre) was relatively drier than the bed shoulder on both NT and SC treatments and vice versa for BP during 2010 wheat and 2011 corn cropping seasons. The average seasonal profile water content in 100 cm depth of the BP treatment was ~4% higher during all three cropping seasons when compared with NT. However, the average seasonal profile water content of SC treatment was ~4% less than the NT treatment during the first season but similar for the last two seasons.

5.3.5 Effect of PRB renovation on irrigation characteristics

5.3.5.1 Inflow

The data for two crops (2010 wheat and 2011 corn) were averaged because the results were not significantly different between both seasons. The inflow rate (Q) was slightly lower for NT and SC than for BP over all three cropping seasons, as practical adjustment to the different infiltration behaviour causing variations in the water advance rate (Table 5.5). There was also a trend toward longer *Tco* with cultivation. Thus, the irrigation volume applied to the BP and SC treatments during the wheat and corn seasons was significantly (P=0.05) larger than the NT treatment (109 mm) and similar during the hemp crop season.

Table 5.5:Irrigation performance for freshly applied PRB renovation treatmentsduring two cropping seasons (2010 wheat & 2011 corn) and on un-renovated PRB(2011 hemp) treatments at site A1, Australia (SD in brackets)

Irrigation	Wheat 2010 & Corn 2011			Hemp 2011				
Performance Parameter	NT	SC	BP	NT	SC	BP		
Q	1.87a	1.97b	2.21c	1.79a	1.79a	2.01b		
$(L s^{-1})$	(0.1)	(0.1)	(0.2)	(0.2)	(0.2)	(0.1)		
Тсо	904a	942a	963a	406a	438a	406a		
(min)	(90)	(99)	(135)	(124)	(126)	(145)		
Inflow	109a	120b	137c	47a	51a	53a		
(mm/irrigation)	(12)	(13)	(18)	(17)	(18)	(17)		
DD	13a	25b	49c	2a	3a	3a		
(mm/irrigation)	(4)	(9)	(11)	(1)	(1)	(1)		
Ea	87a	78b	66c	94a	92a	93a		
(%)	(10)	(15)	(16)	(11)	(11)	(10)		
Er	94a	97b	98b	65a	70b	71b		
(%)	(4)	(3)	(2)	(6)	(11)	(8)		
DU	83a	82a	80a	75a	73a	75a		
(%)	(7)	(8)	(7)	(20)	(18)	(15)		

*Figures followed by different letters in rows for different treatment in the same cropping seasons are significantly (P=0.05) different between treatments

5.3.5.2 Irrigation performance

The *Ea* was significantly (P=0.05) higher (9% and 21%) for NT compared with SC and BP, respectively, during the 2010 wheat and 2011 corn seasons (Table 5.5). The treatment difference in *Ea* was not significant (P=0.05) during the 2011 hemp season. The *DD* were significantly (P=0.05) higher on freshly applied BP (49 mm/irrigation) and SC (25 mm/irrigation) compared with NT (~13 mm/irrigation).

The *Er* was highest (> 94%) for the freshly renovated PRBs during the 2010 wheat and 2011 corn seasons (Table 5.5) but there was no significant difference between the treatments. Similarly, the treatment differences in *Er* were negligible during the 2011 hemp season but the *Er* was significantly (P=0.05) reduced (65-71%) during the 2011 hemp season. The full results are detailed in Table D3 & D4 (Appendix D).

There was no significant (P=0.05) difference between the treatments based on the combined average *DU* results for the 2010 wheat and the 2011 corn seasons (Table 5.5). However, comparing season to season results (Table D3: Appendix D) indicated a significantly (P=0.05) lower (~9%) *DU* for the SC and BP treatments compared with the NT treatment during the 2010 wheat season. The *DU* was also higher during the 2011 corn season than during the 2011 hemp season.

5.3.6 Dry biomass yield

The tillers density of the 2010 wheat was significantly (P=0.05) lower for BP than for NT and SC treatments (Figure 5.12a), while plant height was similar between treatments (Figure 5.13). The dry plant biomass was significantly (P=0.05) higher for NT (9.24 ton ha⁻¹) than for SC (8.71 ton ha⁻¹) and BP (6.88 ton ha⁻¹) treatments (Figure 5.12b & Table 5.6). Results showed reduced (~5%) dry plant biomass on the bed middle (row 3 and 4) for NT and increased biomass for SC (~9%) and BP (~19%) treatments compared with bed shoulder (row 1 and 2). Comparison of rows for different treatments showed significantly lower wheat tillers density and dry biomass for BP on all four rows than NT and SC and for SC on row 1 and 2 than NT. The correlation was positive when tillers density was regressed against dry biomass.



Figure 5.12: Tillers density (a) and dry biomass (b) for four rows (row 1 on bed shoulder and row 4 on bed middle) of 2010 wheat at site A1(vertical bars show SD).



Figure 5.13: Average plant height of three crops on three PRB renovation treatments at site A1, Australia (vertical bars show SD)

 Table 5.6:
 Average dry plant biomass for three crops and dry grain yield for 2011

 corn with three PRB renovation methods at site A1, Australia (SD in brackets)

Crop &	Biomass Yield (ton ha ⁻¹)					
Treatments	NT	SC	BP			
Wheat	9.24a	8.71b	6.88c			
2010	(075)	(0.9)	(0.92)			
Hamm 2011	8.88a	8.83a	8.48a			
Hemp 2011	(2.6)	(2.3)	(2.1)			
Com 2011	14.07a	14.59a	15.14a			
Corii 2011	(2.7)	(2.6)	(1.9)			
Corn 2011	8.94a	9.13a	9.24a			
grain	(1.7)	(2.3)	(1.9)			

*Figures followed by same letters in rows show no significant (P=0.05) difference between treatments

The 2011 hemp crop plant density and dry biomass was not significantly different (P=0.05) between the treatments (Figure 5.14a & b). But the plant density of row 3 was significantly (P=0.05) larger than rows 1 and 2 for SC and BP treatments. The average plant height across the bed width was significantly (P=0.05) lower for the NT than for the SC but BP was similar to both NT and SC (Figure 5.13). The plants were physically weaker in the bed middle of all treatments. The dry biomass of row 3 and 2 on the bed middle was significantly (P=0.05) reduced by up to 55%, ~35% and ~43% than row 1 on bed shoulder for the NT, SC and BP treatments, respectively. Row to row comparison between treatments for plant density and dry biomass indicated no significant difference between treatments.



Figure 5.14: Plant density (a) and dry plant biomass (b) for 2011 hemp three rows (row 1 on bed edge and row 3 on bed middle) across a half bed width for three tillage treatments at site A1, Australia (vertical bars show SD)

There was no significant (P=0.05) difference in either the dry biomass or dry grain yield for the renovation treatments during the 2011 corn season (Figure 5.15). The difference in grain weight, plant height or plant density between the treatments was not significant (Figure 5.15). However, 2011 corn performance on BP was comparable to NT and SC treatments, which was contrary to the 2010 wheat results.



Figure 5.15: Comparative performance of 2011 corn on three PRB renovation treatments at site A1, Australia (vertical bars show SD)

5.3.7 Water productivity

The biomass *WP* of NT and SC treatments were significantly (P=0.05) higher than the BP treatment during the 2010 wheat season (Table 5.7). The *WP* was 10% and 39% lower for the SC and BP respectively, compared with NT treatment during the 2010 wheat season. There was no significant (P=0.05) difference between treatments for the other two crop seasons (Table 5.7). However, the combined average WP for freshly applied treatments during the 2010 wheat and 2011 corn seasons indicated significantly (P=0.05) higher WP for NT compared with SC and BP.

Сгор	Treatment	Water applied (mm)	Dry biomass (ton ha ⁻¹)	<i>WP</i> (kg.ha ⁻¹ mm ⁻¹)
	NT	299	9.24a	31a
Wheat 2010	SC	312	8.71b	28b
	BP	356	6.88c	19c
	NT	419	8.87a	21a
Hemp 2011	SC	432	8.83a	20a
	BP	440	8.48a	19a
	NT	526	14.07a	27a
Corn 2011	SC	546	14.59a	27a
	BP	550	15.14a	28a

 Table 5.7: Dry biomass input water productivity of three crops under three PRB renovation methods at site A1, Australia.

*Figures followed by different letters in columns for each crop season show significant (P=0.05) differences between treatments within the season

The dry grain input water productivity of corn (Table 5.8) was comparable between treatments.

Table	5.8:	Dry	grain	input	water	productivity	of	2011	corn	under	three	PRB
renova	tion 1	metho	ods at s	ite A1,	Austral	lia.						

Treatment	Water applied (mm)	Dry grain yield (ton ha ⁻¹)	WP (kg.ha ⁻¹ mm ⁻¹)
NT	526	8.94a	17a
SC	546	9.13a	17a
BP	550	9.24a	17a

*Figures followed by different letters in columns for each crop season show significant (P=0.05) differences between treatments within the season

5.4 Discussion

5.4.1 Effect of PRB renovation methods on soil physical properties

Furrow dimensions

The furrow dimensions varied between tillage treatments despite the same bed former being used for furrow cleaning in all treatments. For instance, the furrows were largest for the freshly established BP treatment and smallest for the SC treatment during the 2010 wheat and 2011 corn seasons. BP furrow sizes were larger associated with the deeper (30 cm) cutting of the bed with horizontal blades. Another reason was the lifting of the soil in the bed area by the blade without inversion. This caused loosening of the bed soil, thus increasing bed height (Hamilton et al. 2005). The furrow dimensions of the shallow cultivated treatment were smaller as the loose soil in furrows provided less traction for deeper excavation, while the loose soil slid back from the bed shoulder to furrow bottom, thus reducing the furrow dimensions.

The furrow dimensions of SC and BP treatments were largely reduced due to bed settlement and refilling of the furrow bottom with loose soil from bed shoulders transported as sediment during irrigation and rainfall. Bed settlement may also have occurred due to the collapse of unstable macro pores, as soil strength against deformation and subsidence is negatively affected by tillage (Coughlan 1981; Dexter 1991; Cook et al. 1992). Furrow filling (Yang & Wander 1999; Six et al. 2000) caused a reduction in total furrow depth, thus elevating the furrow bottom width (BW) and middle width (MW) making these larger than the initial values. Therefore, PRB renovation methods affect furrow dimensions and need to be applied according to the soil properties and susceptibility to slumping, erosion, and subsidence during the cropping season.

Bulk density of the beds

The treatment effect on soil bulk density was not confounded by soil moisture because soil sampling was done when the difference between treatments was not significant. However, bulk density decreased generally with increase in soil moisture during a season irrespective of treatment differences presumably due to swelling as reported by McKenzie and McBratney (2001) and McKenzie et al. (2002). Average

bulk density during the 2010 wheat season was highest for the NT and lowest for the BP treatment, indicating higher soil porosity for the BP. Differences in bulk density due to tillage or removal of tillage were also reported by Cassel (1982) and McHugh et al. (2009). A maximum temporal increase of ~11% in bulk density of the surface profile (0-10 cm) and 8% increase in average seasonal bulk density for the SC treatment identified during two cropping seasons (2010 wheat and 2011 hemp) indicates that subsidence occurred in a short period of time as mentioned by Meki et al. (2012). The smaller seasonal increase in bulk density for the NT and BP treatments (~5%) suggests these treatments have a greater soil structural stability and resistance to deformation, which may favour improved irrigation performance and *WP* leading to sustainable crop production.

Infiltration

The soil infiltration properties were significantly influenced by the different renovation methods and varied throughout the season. This study found a 23% higher cumulative infiltration using BP treatment compared with NT. Similarly, cumulative infiltration was reduced by approximately 50% for all treatments between 2010 wheat and 2011 hemp seasons, indicating the transient nature of the infiltration improvement caused by improper machinery design and setup. The occurrence of heavy rainfalls before the 2011 hemp sowing may have affected the soil structure and porosity due to excessive wetting. This reduction in infiltration properties within two cropping seasons closely agrees with the findings of Dunn et al. (1998), who demonstrated similar changes in soil hydraulic conductivity and soil water retention properties and attributed them mainly to the collapse of larger temporary pores created due to tillage operations. Consequently, the seasonal reduction in soil infiltration properties negatively affected lateral infiltration and irrigation performance. Therefore, soil disturbance caused bed subsidence and sealing leading to poor infiltration and made it difficult to replenish the SMD, which negatively affected crop yield in the bed middle.

Wetting front penetration into the centre of wide beds is generally not considered a major issue for the Australian Vertisol with shrink-swell, cracking, and self-mulching qualities. Moreover, long furrows and irrigation times (Raine & Bakker

1996) generally overcome lateral infiltration problems in an Australian context. However, this study showed that under specific irrigation and crop conditions the reduced lateral infiltration due to bed subsidence (e.g. up to ~10 hours to wet the bed middle) may affect the PRB performance. For instance, the lower lateral infiltration to the centre of the un-renovated beds during the 2011 hemp season caused germination problems and reduced crop yield in the bed middle. This conform with the findings of Lucy (1993) and suggests that renovation, especially BP, may be helpful in improving the lateral infiltration under the subsided beds.

Seasonal profile water content and irrigation application

The seasonal profile water content differences between renovations treatments were largely influenced by the irrigation scheduling, soil infiltration, and retention properties. Irrigation scheduling is an important decision-making process (Wichelns & Oster 1990) which has consequences on irrigation performance. However, the limited control over infiltration when using surface irrigation systems makes it difficult to apply a target water volume, while the uncertain weather conditions, water availability restrictions, and management practices further confound the ability to apply an optimal schedule (Mailhol et al. 2011). The soil infiltration properties at the time of irrigation and the selection of Q and Tco (Shafique & Skogerboe 1983) to suit the changed soil properties largely impacted on the irrigation performance. Thus, the field limitations and changed soil conditions due to the different renovation treatments influenced irrigation applications and consequently the profile water content was also influenced for the different treatments.

The BP treatment was identified with larger water absorption and soil water storage capacity (~4%) during all three cropping seasons. The reason was higher lateral infiltration, which enlarged the volume of soil available to store water in the root zone across the bed width and greater porosity of loose soil, which agrees with several studies (Hamilton et al. 2005; Jin et al. 2007; Akbar et al. 2009). Therefore, blade ploughing seemed beneficial for conserving in-season rainfall, which can be enlarge the gap between the irrigation and can avoid in-season water stress. However, the long irrigation times and cracking during the 2010 wheat and 2011 corn seasons mitigated treatment impact on lateral infiltration.

The profile water content largely impacted on the irrigation performance of three PRB renovation methods because the heavy clay Vertisol was prone to affect irrigation performance due to its cracking properties (Raine et al. 1998; Gillies 2008), which were normally developed when *SMD* prior to irrigation was greater. For instance, the larger *SMD* prior to fourth irrigation of the 2011 hemp and second irrigation of the 2011 corn caused deeper and wider cracks across the bed, thus the water advance in the furrow was reduced. Consequently, *Ea* was also reduced and *Er* was increased. In contrast, the severe rainfall before the 2011 hemp season wetted the soil profile to deeper depths. However, the top soil surface turned dry till the 2011 hemp sowing with no cracks as the deep soil was still wet. Consequently the *Ea* was increased but *Er* was reduced during the initial two irrigations. This caused poor lateral infiltration and the two rows in the bed middle were poorly germinated initially, which conformed to the findings of Lucy (1993). Luckily the early season rainfall assisted in seed germination in the bed middle. Therefore, soil moisture at the time of irrigation largely affected the irrigation performance.

5.4.2 Treatment, irrigation management and performance interactions

Irrigation performance varied between treatments and crop seasons. The selected irrigation management criteria worked well in improving the irrigation performance and in controlling the excessive losses identified under farmer managed systems (Chapter 4). However, comparing the overall results showed a significantly different irrigation performance between the tillage treatments (as low as 61% *Ea* on BP and above 86% *Ea* on NT treatment), while field limitations in implementing the optimum *Q* and *Tco* seem to have largely impeded in achieving maximum irrigation performance. However, seasonal changes suppressed the treatment differences, evidenced by the negligible variations between irrigation performance factors during the four irrigations to the 2011 hemp crop. Hence, renovation methods largely matter in irrigation performance of PRB systems.

The potential to deep drainage due to irrigation excess were largely variable between the renovation methods. The higher infiltration capacity increased the deep drainage potential (up to 62 mm/irrigation) on freshly applied BP treatment, which agreed with the findings of Smith et al. (2005) and Raine and Bakker et al. (19996), compared with NT (up to 15 mm/irrigation). According to Raine and Bakker (1996) and Smith et al. (2005) deep drainage is one of the main reasons for poor irrigation performance on Australian farms. However, due to bed settlement and furrow compaction, *Ea* was improved but *Er* was significantly reduced during the subsequent irrigations to 2011 hemp. Thus, inadequate irrigation management for the changed soil conditions reduced the performance of freshly renovated PRBs.

Irrigation management played a key role in the irrigation performance of different tillage treatments. For instance, the improved Ea achieved during the 2011 hemp season was due to reduced Er, because the fast arrival of water to the tail end of blocked end furrows necessitated the need for shorter Tco, at applied Q, to avoid excessive tail end bed overtopping. In contrast, Ea of freshly applied BP treatment was reduced due to slow water advance caused by higher soil infiltration properties and low applied Q. Similarly, the uniformity of NT treatment during the first irrigation to 2010 wheat crop was higher due to the slight backing effect of blocked end furrows, which compensated the less infiltration opportunity time that normally occurs at the tail end during the 2011 hemp season negatively affected the DU. In other words, DU was negatively affected, when the difference of infiltrated depth between the head and the tail ends of the field was larger. In contrast, DU during the 2011 corn season was improved due to three seasons of furrow compaction. Thus, irrigation performance was influenced by the changed soil properties.

The root zone soil moisture largely influenced the irrigation performance. For instance, the *Ea* was reduced during the initial two irrigations to 2011 hemp crop, when the soil was wet and *SMD* was low compared with the last two irrigations, when *SMD* was high, which agreed with the findings of Sepaskhah and Ghahraman (2004). In contrast, a very high *SMD* can also negatively affect *Ea*, as evidenced by the lower *Ea* during the second compared with the first irrigation to the 2011 corn, which agreed with the findings of Bautista and Wallender (1993b). Thus, irrigation performance was largely affected by the increased porosity of renovated beds during the early period and by the *SMD* during the latter period after bed settlement. However, irrigation management did not lose its importance during both periods.

5.4.3 Treatment and crop yield interactions

The maximum 2010 wheat crop dry biomass was ~10% less compared with average dry biomass reported with 100 kg ha⁻¹ seed rate by (Khan et al. 2004). Similarly, the 26% higher dry biomass of wheat for NT conform with the findings of (Abu-Hamdeh 2003) and Sayre et al. (2005) who also identified greater yield on NT compared with conventional methods. However, equipment design and set up, its suitability for a particular operation, knowledge of soil properties, soil moisture during machinery operations and expertise in machinery operation for achieving the desired results are important considerations that can affect crop yields. For instance, the grain yield of 2011 corn was up to ~22 % less than the twenty years average reported by (Payero 2008) in the same locality. While, the grain and straw yield were increased by 3% and 8%, respectively on well managed BP treatment compared with NT, which was contrary to the 2010 wheat results due to poor wheat crop establishment.

The three renovation methods caused variable lateral infiltration, thus affecting crop yield, which agreed with the findings of Hamilton et al. (2005), Jin et al. (2007) and Akbar et al. (2009). For instance, the reduced lateral infiltration in NT treatment decreased crop yield in the bed middle by ~5% during 2010 wheat season and up to 55% during the 2011 hemp season (on two season old PRB). Impact of poor lateral infiltration can also be evidenced by the weaker plant height in the bed middle for the 2011 hemp crop. However, during the 2010 wheat season a single irrigation was applied and frequent rainfall during the season mitigated the negative impact of poor lateral infiltration on crop yield in the bed middle. Similarly, impact of treatments on 2011 corn season could not be established as the two rows were on the bed shoulder.

The irrigation performance may affect crop yield as analytical correlations between the crop yield and different irrigation performance parameters were established by Solomon (1984), Holzapfel et al. (1985), Juan et al. (1996) and Zhang et al. (1999), which seemed to have influenced the crop yield in the current study. It can be inferred that the general poor irrigation performance on BP and SC treatments have negatively contributed in their yield compared with the NT treatment. Thus, if achievable, improved irrigation performance through better irrigation management of BP treatment may also contribute to improved crop yield and *WP*.

5.4.4 Treatment and water productivity interaction

The *WP* varied for the renovation treatments and was affected by the irrigation and agronomic management practices, as reported by Playán and Mateos (2006). The results indicated increased *WP* for NT treatment followed by SC (10% less) and then BP (39% less) during the 2010 wheat season. However, seasonal changes due to bed settlement and slumping caused comparable *WP* among treatments during the 2011 hemp season. Despite poor crop performance was caused by bed settlement and subsidence in the bed middle of NT, the yield and *WP* of NT was comparable with the cultivated treatments during the 2011 hemp and the 2011 corn seasons.

The irrigation application and crop yield mainly contributed to the *WP*. The larger soil infiltration and deep drainage losses on BP treatment increased the total water applied, which negatively affected its *WP*. However, the larger soil water storage and increased lateral infiltration compensated for the losses by increasing crop yield in the bed middle. Large soil water absorption and retention with blade ploughing was also reported by Thomas et al. (1990) over seven years of grain sorghum planting in central Queensland, Australia. The yield loss of NT treatment in bed middle was compensated by the improved crop yield on the bed shoulder. Thus the *WP* was similar between treatments during the 2011 hemp and 2011 corn seasons.

This study showed similar *WP* between the three renovation treatments during the last two seasons (2011 Hemp and 2011 corn). Thus the NT treatment seems more beneficial because of its greater consistency with the principles of conservation agriculture and favourable long term impacts on soil structure and biology (Cook et al. 1992; McHugh et al. 2009). However, the reduced bulk density, enhanced lateral infiltration and greater soil moisture retention with less soil disturbance due to no surface inversion by BP (Hamilton et al. 2005) and gradually improved *WP* during the three cropping seasons indicate benefits in using BP compared with SC treatment for the controlled mechanical loosening of subsided PRB on a Vertisol.

5.4.5 PRB renovation options for improved performance on a Vertisol

The NT treatment significantly improved *Ea* while crop yield was not significantly reduced compared with SC or BP treatment. Thus, the NT treatment with no soil disturbance and reduced machinery requirements seems a preferable choice under normal field conditions in the irrigated Vertisol. The NT is also consistent with the principles of CA, which tends to restore the soil hydro-physical, chemical and biological properties through self-regeneration processes (McHugh et al. 2009).

The SC treatment caused greater soil disturbance, poor establishment, accelerated furrow filling and enhanced deep drainage losses. According to Coughlan (1981) the Vertisol is very sensitive to intensive tillage because it destroys the low volume of permanent pores as the shrinking and swelling generally close the unstable macro pores making the soil less permeable (Cook et al. 1992) and more susceptible to erosion and degradation (Smith et al. 1983). Thus SC is not preferable for Vertisol.

This study showed that renovating PRB with the BP method reduced irrigation performance under current management but improved soil physical properties (i.e. bulk density, lateral infiltration and soil water storage) due to non-inversion on the bed surface, thus leading to comparable crop yield and WP with the NT treatment in the last 2011 corn season. The closer consistency of BP with the principles of CA makes it a better choice compared with SC treatment for loosening the subsided beds if required. The BP can better manage ground cover because of no inversion (Hamilton et al. 2005) compared to SC treatment which destroy ground cover. It can also be inferred from these outcomes that BP can be beneficial for conserving more water from in-season rainfall, for reducing irrigation application frequency by enlarging irrigation application gaps and mitigating negative impacts of drought and water stress on crop yield, which is very common on Australian farms. However, BP needs careful selection of specialised equipment (right design and set up), correct soil moisture (soil moisture above or below plastic limit can cause smearing or soil disturbance), and operating expertise for improved performance. Additionally, the changed soil infiltration properties with BP require intensive irrigation management optimisation to avoid excessive application losses, while long term impacts of BP on soil structure and its optimal frequency of applications is not known on Vertisol.

5.5 Conclusions

The *Ea* of three freshly applied renovation methods on 2 m wide beds was significantly influenced by the current irrigation management due to the changed soil hydro-physical properties. However, productivity was not significantly affected except for the 2010 wheat season, suggesting that changes may take longer to significantly emerge in *WP* for this site. The key conclusions are:

- Renovation methods did affect bed bulk density, cumulative infiltration, and lateral wetting into the bed. However, the effect was largest immediately after renovation and decreased during the season. There was no difference between the treatments in the second season after renovation.
- While there was some difference between renovation method treatments on the rate of lateral wetting, the long irrigation times and cracking resulted in close to full wetting of beds under the majority of conditions observed.
- The *Ea* was significantly lower for the freshly applied SC and BP treatments during the 2010 wheat and 2011 corn seasons due to the higher soil infiltration rate, larger *Tco* and increased *DD* than with the NT treatment. However, the overall difference in *Er* and *DU* was not significant.
- There was no significant difference in irrigation performance between the treatments for the 2011 hemp crop, when the beds were not renovated.
- Except in the first 2010 wheat season, there was no significant difference in yield or WP between the NT, SC or BP treatments.
- First season BP (and possibly SC) yield may have been affected by poor seeding and establishment due to poor machinery performance.
- Reduced yield was observed in the bed middle for all treatments during the 2011 hemp season but full bed width yield was similar on all treatments.
- The increased infiltration of SC and BP treatments suggests that these treatments need to be irrigated more quickly using increased Q and reduced *Tco*. If achievable, the *Ea*, *Er* and *DU* of SC and BP should be able to increase substantially. Chapter 8 explores these possibilities.
CHAPTER 6: Field evaluation of lateral infiltration in permanent raised beds

6.1 Introduction

In most countries where water is either in short supply or is costly, wide beds are often preferred to narrow beds because smaller irrigation volumes normally are applied and land available to grow crop is increased (Jin et al. 2007). However, for such irrigation to be effective the soil in the beds needs to be highly sorptive to ensure water is drawn quickly to the centre of the bed. Poor lateral infiltration negatively affects WP of wide PRB systems (Lucy 1993; Akbar et al. 2007; Jin et al. 2007). The reasons for water not being absorbed quickly into the centre of beds are that the soil in the centre of the beds is compact or the soil structure is poor or unstable to rapid wetting. In consequence, farmers use a variety of renovation methods to restore the sorptive behaviour of beds in PRB farming systems. No field research has been found that is specifically focussed on this issue. The results of benchmarking studies in Australia and Pakistan (Chapter 4) and poor lateral infiltration revealed in the WP study in Australia (Chapter 5) have emphatically illustrated the need for research into renovation practices that embody all the principles of conservation agriculture and build and maintain a soil structure in beds that ensures they are permanently sorptive enough to ensure water reaches the centre of beds as quickly as possible.

Most furrow irrigation research and simulation models assume uniform penetration of water to the centre of the beds. Thus, a secondary need for research into managing the sorptive behaviour of PRB is to reinforce the legitimacy and applicability of this research and the models by ensuring practices exist that validate the assumption of uniform penetration of water to the centre of beds. Therefore, this study evaluated the effect of renovation methods on lateral infiltration into wide permanent raised beds. Three PRB renovation methods were evaluated on two soils in Australia and Pakistan. This chapter reports on the field evaluation of three PRB renovation methods while Chapter 7 evaluates irrigation strategies to improve lateral infiltration.

6.2 Material and methods

The experiments involved using a blocked furrow infiltrometer method (Walker & Skogerboe 1987) and measurements of soil moisture content across the beds to determine differences in lateral infiltration due to renovation methods on two soils. The experiments were conducted at site A1-Vertisol in Australia and site P4-sandy clay loam in Pakistan. The sites details are described in Chapter 3.

6.2.1 Site description, treatment and experimental layout

Two sets of measurements were obtained during the 2011 summer in a Vertisol planted with corn at site A1, Australia. Three renovation treatments with three replicates were applied in a randomised block design. The three renovation treatments were NT, SC and BP as described in Chapter 5 (section 5.2.1) and laid out as shown in Figure 5.3. The centre bed of each treatment was selected for measurement. A total of nine data collection points, one for each treatment, were selected at ~40 m distance from the head end of the experimental plot. The measurements of lateral infiltration were conducted immediately prior to the first and second irrigation of the 2011 corn crop (Chapter 5). The furrow dimensions of the three treatments at the time of the measurements are given in Table 6.1.

andy endy round (furrow unitensions explained in Figure 5.2c); (5D in brackets)						
Site/soil	Treatment	TW (cm)	MW (cm)	BW (cm)	D (cm)	
A1- Vertisol	NT	62 (3)	46 (3)	18 (3)	12(1)	
	SC	52 (4)	37 (4)	21 (3)	10(1)	
	BP	60 (3)	44 (2)	23 (2)	11(1)	
P4-Sandy clay loam	NT	56 (1)	41 (1)	22 (2)	13 (1)	
	SC	51 (1)	27 (2)	14(1)	13.5 (2)	
	BP	55 (2)	34 (1)	20 (3)	14(1)	

 Table 6.1: Average furrow dimensions during two irrigations on a Vertisol and a sandy clay loam (furrow dimensions explained in Figure 3.2c), (SD in brackets)

Two additional sets of measurements were obtained on a farmer's field at site P4 (north west Pakistan), on a sandy clay loam soil during the 2011 summer corn season. Four randomly selected replications of three treatments NT, SC, and BP were freshly applied to a two season old PRB. The main differences in renovation treatment between the sandy clay loam and Vertisol sites were the furrow spacing (132 cm rather than 200 cm) and the use of a rotavator (for 10-15 cm deep rotary

hoeing) instead of a tined cultivator for the establishment of the SC treatment. The data collection points were located at a 20 m distance from the field head.

The sandy clay loam at this site has a badly degraded and inherently weak structure as a consequence of excessive cultivation over many years. It was also marginally saline while analyses of soils in the area showed an exchangeable sodium percentage of 18% (Akbar and Hamilton, unpublished data), which will predispose them to dispersion and sealing when rapidly wet.

Two rows of corn crop were planted per bed on 25 June 2011 using a minimum till planter with 60 cm row spacing and 10 cm between plants in the rows. The measurements were obtained immediately prior to the first and second irrigations of the season. The average furrow dimensions at the time of measurements are shown in Table 6.1.

6.2.2 Bulk density and soil moisture measurements

Bulk density measurements were taken at two positions near each measuring point prior to wetting. Samples were taken in multiples of 10 cm to a depth of 30 cm according to the procedure described in section 3.6.1.

Volumetric soil moisture measurements were taken prior to and during the irrigation by a micro-gopher operated within PVC access tubes installed across the beds. Four access tubes (2 cm internal diameter) for soil moisture measurement were installed according to the layout shown in Figure 6.1. The measuring points were located at the furrow centre, 33 cm, 67 cm from the furrow centre and the bed middle on the Vertisol and at 22 cm, 44 cm and 66 cm distance from the furrow centre on the sandy clay loam. The holes for the access tubes were made with a soil auger slightly smaller than the outer diameter of the PVC access tube and slightly deeper (120 cm) than the maximum measuring depth of 100 cm. The bottom end of all access tubes were sealed with putting plug to ensure full water tightness. The access tubes were then installed by pushing and twisting using a nylon plug at the top of the tube. This procedure reamed the hole to the exact size of the PVC tube while the soil reamed was settled in the extra depth of the hole drilled. This procedure ensured a perfect contact between the access tube and the soil. To further reduce the possibility of vertical preferential flow along the walls of the access tube installed in the furrow centre, bentonite clay was applied around the tube (2 cm wide and 2 cm deep cross sectional ring). The access tubes were then tested for any physical deformation during the installation by placing the micro-gopher sensor head into the access tube and checking the uniformity of physical resistance whilst pushing it slowly from top to bottom through the entire length. After installation, the top of the access tubes were covered with a small plastic cup that ensured free air entry and protected from rain water entry.



Figure 6.1: Experimental set up showing location of four access tubes for micro-gopher across half bed width installed at nine locations (one/treatment) at site A1-Vertisol.

The micro-gopher was calibrated using the inbuilt volumetric function. The initial field capacity values were set to 43% for the Vertisol and 23.5% for sandy clay loam following the procedure described in the equipment manual. However, the equipment calibration was readjusted latter on for the given field capacity values by taking soil moisture readings at 10-20 cm depth after 2 days of full irrigation. A single point calibration was considered adequate because both soil profiles were uniform over the 100 cm depth being monitored. The micro-gopher readings were validated by matching with the calculated volumetric soil moisture obtained from a 0-30 cm profile in a nearby location using the gravimetric method (section 3.6.1). The gravimetrically determined soil moisture readings as given in Appendix C: Figure C2 for both soils.

6.2.3 Blocked furrow infiltrometer measurements

A one metre long section of the centre furrow in each treatment (Figure 5.3) was isolated at both ends by steel sheets (inserted to a depth of 10 cm into the furrow bottom and 5 cm into both adjacent bed shoulders). The experiment commenced with recording antecedent soil moisture in both soils in all tubes. Then the one metre long furrow section between metal covered with polythene sheet was filled with water to 9 cm depth. The water was released by gently sliding out the polythene sheet and the 9 cm depth was immediately checked and in case of any deficiency additional water was added quickly. The 9 cm depth was kept constant during the experiment and a log of time and volume of water added to each furrow was recorded for a maximum of 9 hours on both soils. The change in soil moisture of 100 cm profile at 10 cm depth intervals was monitored at 15 minute intervals during the first hour and at 30 minute intervals after one hour till the end of experiment.

6.2.4 Lateral infiltration and sorptivity

Lateral infiltration was determined by comparing the soil moisture at the three measured positions from the furrow centre to the bed centre. Cumulative infiltration for each measured position at each time interval was calculated as the increase over the antecedent volumetric soil moisture content for a 100 cm deep profile. Sorptivity was calculated as the slope of the linear regression line of best fit by plotting the sum of the increase in profile moisture content at all three lateral positions against the square root of time.

Sorptivity is not only a descriptor of early-time infiltration, but also of soil structure stability (Collis-George & Laryea 1972), because of its reliance on hydraulic conductivity and the moisture characteristic, both of which vary with moisture content and pore size distribution. Therefore, sorptivity evaluations were conducted to illustrate the effects of different PRB renovation methods on soil structure and the resulting variable lateral infiltration in two very different soil types.

ANOVA analysis was applied on selected data only to determine significant differences as some of the data violated the normality of distribution and equal variance assumption. Thus mean and standard deviations were generally provided.

6.3 Results

6.3.1 Site A1-Vertisol

The average bulk densities of the combined 0-30 cm layer were generally not significantly (P=0.05) different between irrigations for all treatments (Figure 6.2). Results showed significantly (P=0.05) lower bulk density of the combined (0-30 cm) layer on BP compared with NT. However the average bulk density of the combined 0-30 cm layer on SC was not significantly different (P=0.05) from either NT or BP. Treatment comparison showed significantly reduced bulk density for BP compared with NT while bulk density of SC was similar to both NT and BP at all measured individual depths. However, bulk density of the combined 0-20 cm layer was significantly (P=0.05) reduced for SC and BP in comparison with 20-30 cm layer.



Figure 6.2: Average bulk density (BD) before two irrigations for three PRB renovation treatments at site A1, Vertisol (vertical bars show SD)

Soil moisture variations in 100 cm profile at 10 cm depth intervals across the bed on all three treatments during both irrigations were evaluated by comparing the average antecedent and one, four and nine hours post soil moisture content. The antecedent soil moisture content of all treatments was not significantly (P=0.05) different from one to another preceding both the first and second irrigations (Figure 6.3 & 6.4). However, the antecedent soil moisture content of the first experiment was significantly higher compared with the second experiment. Thus, both irrigations were evaluated separately. These moisture content differences caused the rate of infiltration and the amount absorbed over 9 hours for the first and second irrigations to be very different, as was expected from the knowledge of sorptivity.



Figure 6.3: Impact of renovation method, wetting time and distance from furrow centre on average soil moisture in 100 cm profile depth during the first irrigation at site A1-Vertisol



Renovation methods

Figure 6.4: Impact of renovation method, wetting time and distance from furrow centre on average soil moisture in 100 cm profile depth during the second irrigation at site A1-Vertisol

Temporal variations in average profile soil moisture across the bed indicated no significant (P=0.05) difference between treatments after one hour during the first irrigation (Figure 6.3) but a rapid increase at the 33 cm for NT and SC and 67 cm for BP during the second irrigation (Figure 6.4). However, after four hours NT showed significantly (P=0.05) lower soil moisture across the bed than the other two treatments, which were similar. After nine hours of wetting during the first irrigation the average soil moisture at 33 cm was similar between all treatments. However, the average soil moisture at 67 cm and the bed middle was significantly (P=0.05) higher for BP and SC compared with NT. During the second irrigation the moisture content at the bed middle was lower for NT and SC than BP after nine hours of wetting (Figure 6.4).

Separate parallel plots for the four measured positions (furrow centre, 33 cm, 67 cm and bed centre) were plotted for all three treatments to illustrate the difference in vertical distribution of soil moisture across the bed (Figure 6.5 & 6.6). The results indicated greater soil moisture variations on the bed surface compared with the deeper depths for all treatments. The vertical soil moisture distribution at 67 cm was similar for all treatments after 9 hours of wetting (Figure 6.5a-c). However, the moisture content was higher in the BP bed centre than the SC or NT during the first irrigation (Figure 6.5d). The total change in soil moisture of 0-100 cm profile in the bed middle was similar during four hours of wetting (Figure 6.5d). The change in profile moisture content was two and three times higher for the SC (Figure 6.5d-2) and BP (Figure 6.5d-3) treatments respectively, compared with NT treatment (Figure 6.5d-1) after nine hours of wetting during the first irrigation.



Chapter 6: Field evaluation of lateral infiltration in permanent raised beds

Figure 6.5: Change in soil moisture content $(\Delta \theta)$ across a 200 cm wide bed at given distances from furrow centre and given times with three renovation treatments during the first irrigation at site A1-Vertisol

The larger infiltration (i.e. gap between antecedent and one hour post wetting water content curves) at 33 cm for NT and SC treatments and at 67 cm for BP treatment (Figures 6.6) illustrated increased lateral and vertical infiltration during the second than the first irrigation, as the water moved rapidly through cracks. However, the greater wetting front penetration in BP treatment is more obvious compared with the other two treatments evidenced by a 52% increase in soil moisture of BP treatment than the NT treatment after nine hours of wetting.



Figure 6.6: Change in soil moisture content $(\Delta \theta)$ across a 200 cm wide bed at given distances from furrow centre and given times with three renovation treatments during the second irrigation at site A1-Vertisol

The comparison of temporal variation in cumulative infiltration (Figure 6.7) for the first irrigation shows the highest values at 33 cm, followed by the furrow centre, 67 cm and the bed middle for all treatments. However, infiltration was similar at 67 cm and the furrow centre after 8 hours for NT and BP (Figure 6.7). The bed middle of SC and BP received 86% and 233% more cumulative infiltration respectively compared with NT (36 mm) after 9 hours during first irrigation. The cumulative infiltration rapidly increased across BP, while the bed middle for NT and SC (Figure 6.8) took longer to wet during the second irrigation.



Figure 6.7: Cumulative infiltration across a 200 cm wide bed with three renovation treatments during the first irrigation at site A1-Vertisol



Figure 6.8: Cumulative infiltration across a 200 cm wide bed with three renovation treatments during the second irrigation at site A1-Vertisol

The two dimensional soil moisture profiles (expressed as percent of field capacity) were graphed using the Krigging data interpolation function of software Surfer (version 8). Temporal variations in soil moisture (antecedent, one hour, four hour and nine hour) were plotted separately (Figure 6.9 & 6.10), which shows the substantially higher lateral infiltration for the BP compared with NT and SC (Table 6.2).

Irrigation	Treatment	80% FC	100% FC	120 % FC
	NT	90	75	46
1	SC	> 100	>100	45
	BP	> 100	>100	> 100
	NT	95	78	50
2	SC	> 100	90	77
	BP	> 100	>100	> 100

Table 6 2. Wetting front penetration (in cm from furrow centre) at the bed surface



Figure 6.9: Soil moisture distribution for three PRB renovation treatments across a 200 cm wide bed during the first irrigation at site A1-Vertisol



Figure 6.10: Soil moisture distribution for three PRB renovation treatments across a 200 cm wide bed during the second irrigation at site A1-Vertisol

6.3.2 Site P4-Sandy Clay Loam

The average bulk densities of the combined 0-30 cm layer were not significantly (P=0.05) different between irrigations for all treatments. However, the results showed significantly (P=0.05) lower bulk density for BP compared with NT (Figure 6.11) at all individual depths. The bulk density of the combined 0-20 cm layer on SC was significantly (P=0.05) lower than NT. However, the bulk density at 20-30 cm depth was similar between NT and SC. Overall results showed a similar average bulk density (0-30cm) between SC and the other two treatments, but the average bulk density of the combined (0-30cm) layer of BP was significantly (P=0.05) lower compared with NT.



Figure 6.11: Average bulk density (BD) before two irrigations for three PRB renovation treatments at site P4-sandy clay loam (vertical bars show SD)

Average antecedent soil moisture was similar across 132 cm beds for all three treatments during both irrigations (Figure 6.12). The antecedent soil moisture was slightly higher near the furrow compared with the bed middle. However, the differences in antecedent soil moisture between experiments and between treatments were not significant (P=0.05). Therefore, the data for both experiments were treated as a single population and analysed together.

The soil moisture at 22 cm after one hour and four hours of wetting was significantly (P=0.05) higher for SC and BP compared with NT (Figure 6.12). However, the difference in soil moisture between SC and BP was not significant (P=0.05) after one hour and four hours of wetting. All treatments soil moisture became significantly

(P=0.05) different after nine hours of wetting. BP had significantly (P=0.05) higher soil moisture at 44 cm compared with the other treatments after nine hours of wetting.



Figure 6.12: Impact of renovation method, wetting time, and distance from furrow centre on average soil moisture in 100 cm profile depth during two irrigations at site P4-sandy clay loam

There were considerable variations in infiltrated depth after nine hours of wetting, especially at 44 cm (Figure 6.13). The wetting front movement both vertically and laterally was considerably reduced for NT compared with SC and BP. The BP treatment considerably increased the soil moisture at 44 cm but none of the renovation treatments were successful in wetting the bed middle to field capacity during the nine hours of wetting.



Figure 6.13: Change in soil moisture content $(\Delta \theta)$ across a 132 cm wide bed at given distances from furrow centre and given times with three renovation treatments during two irrigations at site P4-sandy clay loam

The comparison of cumulative infiltration after nine hours of wetting (Figure 6.14) shows most infiltration occurred at 22 cm for all treatments. The least cumulative infiltration at a 22 cm distance was on NT (66 mm) compared with SC (23% higher) and BP (68% higher). Similarly, at 44 cm the average cumulative infiltration after nine hours of wetting was 36% and 190% higher for SC and BP respectively compared with NT (15 mm). There was no infiltration into the bed middle for any treatment.



Figure 6.14: Cumulative infiltration across a 132 cm wide bed with three renovation treatments during two irrigations at site P4-sandy clay loam

The two dimensional soil moisture profiles were graphed at four time intervals (antecedent, one hour, four hours and nine hours) of wetting. The soil moisture distribution below the furrow bottom was estimated by extrapolation of the measured values and by the difference between the soil moisture volume measured in the profile, and total volume of water added to the furrow. The lateral movement of the wetting front penetration on the bed surface after nine hours of infiltration is summarised in Table 6.3. The average results of soil moisture profiles during both experiments for all three treatments are shown in Figure 6.15.

 Table 6.3:
 Wetting front penetration (in cm from furrow centre) at the bed surface after 9 hrs of wetting for three renovation treatments during the two irrigation at site P4-sandy clay loam

Treatment	70% FC	90% FC	110 % FC
NT	60	50	38
SC	64	54	40
BP	> 66	57	48



Figure 6.15: Soil moisture distribution for three PRB renovation treatments across a 132 cm wide bed during two irrigations at site P4-sandy clay loam

6.3.3 Field sorptivity

Field sorptivity (S) on Vertisol was variable between treatments and soil types (Table 6.4 and Figure 6.16). The S for the BP treatment was 19% and 15% higher than the NT and SC treatments respectively, in Vertisol. For sandy clay loam the S for BP treatment was 71% and 16% higher than the NT and SC treatments respectively. Comparison of both soils showed 4 to 6 times higher field sorptivity in Vertisol compared with sandy clay loam.

Table 6.4: Field sorptivity (mm hr^{-0.5}) during 9 hrs wetting time for three PRB renovation treatments during the first irrigation at site A1-Vertisol and average of two irrigations at site P4-sandy clay loam (SD in brackets)

Site/Soil	NT	SC	BP
A1-	107	110	126
Vertisol	(3)	(9)	(4)
P4-sandy	17	25	29
clay loam	(3)	(5)	(6)



Figure 6.16: Effect of three PRB renovation methods on average field sorptivity for (a) site A1-Vertisol during first irrigation and (b) site P4-sandy clay loam during two irrigations

6.4 Discussion

6.4.1 Lateral infiltration on Vertisol

The significantly lower bulk density for BP compared with NT indicates that the renovation method has altered the soil structure in the bed. Thus, the differences in soil moisture distribution were most likely due to differences in soil hydraulic properties associated with the different renovation methods. As the antecedent soil moisture during each experiment was not significantly (P=0.05) different between treatments, this suggests that the results were not confounded by antecedent soil moisture effects even though the two irrigations had significantly (P=0.05) different antecedent soil moisture content.

This study found significantly lower soil moisture at the middle of the bed compared with the bed shoulder in the majority of cases. This confirms that the soil moisture distribution between the wetted furrow and the centre of a 200 cm wide PRB is not uniform. Thus, the assumption of uniform soil moisture distribution across the beds is not appropriate for irrigated wide PRB farming systems. Enhanced lateral infiltration appeared to be due to the presence of cracks (Lin et al. 1998) when the soil was dry during the second irrigation. This enhanced lateral infiltration may reduce the risk of water stress in the bed middle (Lucy 1993). However, deep drainage losses are also likely to be increased due to cracks.

The lower cumulative infiltration observed at the furrow centre of all treatments compared with the bed is consistent with the compacted permanent wheel tracks reducing vertical infiltration in the furrow area (Khalid & Smith 1976; Potter et al. 1995; Coutadeur et al. 2002). However, a lower vertical infiltration may also have been due to the relatively higher moisture content in the furrow (Turner & Sumner 1978) as occurred during the first irrigation.

Overall results showed greater lateral infiltration for BP followed by SC and then NT. The treatment effect was significantly different during both irrigations. This confirms that the renovation method of PRB can significantly alter the lateral infiltration on Vertisol. More interestingly, it has also confirmed that impacts of

renovation methods on lateral infiltration still existed under very dry soil conditions on a Vertisol in the presence of cracks. One of the reasons may have been the loose and porous soil conditions of the BP treatment. In any case, BP can be used as a tool to enhance lateral infiltration in wide PRBs on Vertisol soils.

BP had higher lateral infiltration but lower vertical infiltration compared with SC. One of the reasons was the soil tilth associated with SC that enhanced vertical infiltration by gravity due to the presence of temporary macro pores (Lipiec et al. 2006). However, the enhanced lateral infiltration for BP was likely due to higher lateral hydraulic conductivity, sorptivity and water storage capacity (Thomas et al. 1990; Eldridge & Robson 1997; Hamilton et al. 2005; Jin et al. 2007; Akbar et al. 2010). The objective of gentle, non-inverting, controlled mechanical loosening with BP was to retain roots and bio-pores, which seems likely to have increased the organic matter, stability of aggregates and retained a stable porous environment with enhanced sorptivity (Collis-George & Laryea 1972), thus lateral infiltration was improved. The uniform distribution of soil moisture across the bed for BP treatment indicates no smearing that could constrain water movement below the mechanically loosened 30 cm profile depth.

Variations in infiltration between irrigations were largely affected by the antecedent soil moisture levels, being drier prior to the second irrigation than the first. However, the drier profiles for the second irrigation do not account for all of the differences. If this was the sole reason, one would expect the magnitude of increase to be about the same for all treatments. It seems that the active growth of the corn crop and its roots along with the associated biological activity around the roots may have enhanced the soil aggregation and the stability of aggregates to wetting, as evidenced by the greater sorptivity. The larger horizontal infiltration under BP is consistent with a larger, stable porosity in this treatment. Greater root development has been documented for PRBs with 30 cm deep blade ploughing practices (Hamilton, D. pers. comm.). The resulting larger soil moisture storage under BP may also be helpful in extending the period between irrigations, making better use of in-season rainfall, and controlling deleterious effects on crops due to in-season water stress if the crop is not fully irrigated.

6.4.2 Lateral infiltration on sandy clay loam

The similar bulk densities for both irrigations indicated that there was no further degradation or subsidence over this time frame on the sandy clay loam at this site. However, the significantly lower bulk density (0-30cm) for BP indicated greater porosity compared with NT. Similarly, SC also showed significantly lower bulk density in the cultivated zone (10 cm) compared with NT. Thus, the three treatments significantly altered the soil structure and the resulting differences in lateral infiltration would be expected to be due to changes in hydraulic properties caused by the different soil management techniques.

The sandy clay loam had an extremely poor lateral and vertical infiltration. Although lateral infiltration in the three renovation treatments was significantly different, none of the treatments successfully wet the bed middle to field capacity during the nine hours of wetting. This is most likely due to the general poor soil structure at the site associated with a history of intensive cultivation with top soil inversion (Gill 1994; Kahlown et al. 1998; Akbar et al. 2007) and high soil salinity (Qureshi et al. 2008), which has made the soil susceptible to dispersion and sealing due to rapid wetting.

NT is not a common practice in north west Pakistan (Akbar et al. 2007) as it is normally associated with higher weed infestation, poor lateral infiltration and hard seed beds, which are difficult to manage manually. The current results showed NT resulted in lowest lateral infiltration. Therefore, NT could be expected to increase the risk of poor crop performance in the bed middle due to water stress under degraded soil conditions. Poor lateral infiltration was also reported by Jin et al. (2007) and Akbar et al. (2007), while Wang et al. (2012) did not recommend continuous NT on sandy clay loam.

SC is the most commonly used method of tillage in north west Pakistan due to the common perception that it helps in cleaning the weeds from the previous crop, improve soil infiltration capacity and also helps in making a soft seed bed for the convenient manual sowing of row crops (Akbar et al. 2007). However, this study found poor lateral infiltration under this treatment. It also seems likely that the continued use of SC may negatively affect soil structure and destroy the soil's natural

ability to ameliorate. Initially high vertical infiltration for SC may be due to the existence of temporary macro pores (Coquet et al. 2005; Bormann & Klaassen 2008). However, the macro pores would be expected to rapidly collapse or to clog due to soil dispersion (Smith et al. 1983). The resultant poor lateral infiltration has been found (Hassan et al. 2005) to negatively affect crop performance on the bed middle and crop yield.

The higher lateral infiltration observed for BP treatment was consistent with the findings of Jin et al. (2007) and Akbar et al. (2009). The higher lateral infiltration associated with BP was found in both irrigations. Although BP was also not successful in wetting the middle of the bed in the nine hours of wetting, the higher infiltration rates over the NT and SC raise the prospect of further improvement in combination with other irrigation and agronomic (CA) measures.

Increasing the infiltration opportunity time and/or furrow water head through optimised irrigation management practices may assist in wetting the middle of a 132 cm wide bed on sandy clay loam, which needs to be investigated and the following Chapter 7 explores this possibility. Alternatively, decreasing the bed width or leaving a larger gap in plant rows on the bed middle may also address poor lateral infiltration issues. However, the long term measures for such a soil with non-expansive mineralogy and degraded structure could involve structural improvement and stabilisation by organic matter and soil biology.

6.5 Conclusions

Lateral infiltration was increased by BP relative to both NT and SC on both the Vertisol and sandy clay loam. The higher lateral infiltration capacity and longer irrigation times used on Australian Vertisol (due to long fields) mean that wide beds (200 cm furrow spacing) are less susceptible to poor crop performance in the middle of the beds. In contrast, the poor lateral infiltration and short irrigation times used on smaller Pakistani farms make wide beds (132 cm furrow spacing) on sandy clay loams more susceptible to poor crop performance. The main conclusions from this study are:

- The soil moisture distribution across the beds was not uniform and varied under all three PRB renovation methods. Thus, the assumption of uniform soil moisture distribution across the beds is not appropriate for the irrigated wide bed farming systems.
- The three PRB renovation treatments significantly altered the lateral infiltration capacity on both soils. Thus, bed renovation can be used as a management tool to minimise yield loss of densely grown crops in the middle of wide beds.
- BP was more efficient than SC and NT in reducing the bulk density and increasing the sorptivity, lateral infiltration capacity and soil water storage on both soils.
- The increased sorptivity with BP treatment enhanced the legitimacy and validity of most surface irrigation simulation models that assume uniform distribution of soil moisture across the beds.

CHAPTER 7: Strategies to improve lateral infiltration in permanent raised beds using Hydrus 2D

7.1 Introduction

The previous field trials (Chapter 6) revealed large differences in infiltration between three PRB renovation methods due to soil structural and antecedent soil moisture differences. These differences in lateral infiltration on the Vertisol and sandy clay loam raised concerns about the appropriateness of current PRB irrigation management and field design. The common assumption of uniform sub-surface distribution of soil moisture across the beds was shown to be inappropriate for wide PRBs. In fact, without management aimed specifically at improving lateral infiltration, PRB farming will not reach its potential levels of irrigation efficiency and productivity. This lack of adequate wetting of the centre of PRB has already been shown to affect crop germination on Vertisol (Lucy 1993) and crop yield on sandy clay loam (Hassan et al. 2005) leading to low *WP*. Therefore, this study used a calibrated soil water simulation model to evaluate renovation and furrow water head (flow rate) strategies for improving lateral infiltration in wide PRBs.

The PRB renovation methods (described in Chapter 5) were simulated using Hydrus 2D (Šimunek et al. 1999). The lateral and vertical movement of the wetting front, cumulative infiltration, deep drainage losses (water penetrated beyond arbitrary root zone depths), and soil water storage were determined at a range of furrow water heads (flow rates). The outcomes were used to develop relationship between furrow inflow rate and bed width to improve irrigation management on both sites.

7.2 Material and methods

The Hydrus 2D model was calibrated on both soils (Appendix E) using the preprocessing procedures as outlined by Rassam et al. (2004) and Simunek et al. (2007). For the Vertisol, the field data measured during the first experiment (Chapter 6: 9 cm furrow water head) and during the first 2011 corn crop irrigation (Chapter 5: 6 cm furrow water head) were used. For the sandy clay loam, the average field data of both experiments (Chapter 6: 9 cm furrow water head) were used. The flow domain was designed to simulate the 200 cm wide PRB on Vertisol and 132 cm wide PRB on sandy clay loam. The depth of the flow domain was set at 100 cm below the bed surface to match the root zone depth for mass balance, deep drainage and soil water storage evaluations (Figure 7.1) on both soils. The calibrated model was run for 20 hrs on the Vertisol and for 40 hrs on the sandy clay loam. Six lateral observation nodes (L1 to L6) were located on the bed surface at 40, 60, 70, 80, 90 and 100 cm on the Vertisol (Figure 7.1) and at 36, 46, 51, 56, 61 and 66 cm distance from the furrow centre on the sandy clay loam (layout not shown). Ten observation nodes were fixed on the vertical boundary in the middle of bed (M1 to M10) and nine on the vertical boundary below the furrow centre (F1 to F9) at 10 cm depth intervals (Figure 7.1) for both soils. These observation nodes were used to evaluate temporal variations in soil moisture for the three renovation treatments with 9 cm furrow water head (as 9 cm represented large range of flow rates). Further analysis of lateral and vertical infiltration was undertaken using furrow water heads of 4 cm, 6 cm, 9 cm and full furrow depth (D in Table 6.1) to represent a range of furrow flow rates (Q). The Q versus furrow water head relationship was identified from field measured experimental data on both soils. Cumulative infiltration, deep drainage and volume stored were identified from the Hydrus output.



Figure 7.1: Locations of observation nodes on surface and vertical boundaries of flow domain for Vertisol

7.3 Results

7.3.1 Hydrus calibration

There was a strong correlation between measured and predicted soil moisture values on both soils and all bed renovation treatments as shown in Figure 7.2 and presented in detail in Appendix E. The errors identified were mainly due to a slight overprediction near the furrow and slight under-prediction in the bed middle. However, the overall results showed a good estimation of temporal variation in soil moisture distribution under three different PRB renovation methods on both soils confirming that the calibration of the model was appropriate.



Figure 7.2: Correlation between measured vs. predicted soil moisture during 9 hrs wetting for three renovation treatments with 9 cm furrow water head

7.3.2 Effect of renovation on infiltration with 9 cm furrow water head

7.3.2.1 *Lateral infiltration*

For Vertisol, there were substantial temporal and spatial differences in soil moisture on the top surface of the bed between treatments (Figure 7.3) identified from six observation nodes (location given in Figure 7.1). The field capacity front took ~17 hrs on NT, ~11 hrs on SC and ~6 hrs on BP treatments to reach the bed middle (node L6 in Figure 7.3) when 9 cm of furrow water head was applied.



Figure 7.3: Simulated soil moisture variation on the bed surface (shown by observation node number: Figure 7.1) for three PRB renovation treatments with 9 cm furrow water head applied on a Vertisol

For sandy clay loam there were substantial differences in lateral wetting between the renovation treatments (Figure 7.4). More than 40, ~32 and 20 hrs wetting time was required for the bed middle (node L6) to reach field capacity for the NT, SC and BP treatments respectively when 9 cm furrow water head was applied.



Figure 7.4: Simulated soil moisture variation on the bed surface at (L1 = 36, L2 = 46, L3 = 51, L4 = 56, L5 = 61 and L6 = 66 cm distance from furrow centre) for three PRB renovation treatments with 9 cm furrow water head applied on a sandy clay loam

7.3.2.2 Vertical infiltration

For Vertisol, vertical infiltration below the furrow base was quickest for SC and slowest for NT (Figure 7.5). It took ~8, ~5 and ~7 hrs for the wetting front to reach 100 cm depth below datum in the furrow centre (node F9) for NT, SC, and BP treatments respectively, with 9 cm furrow water head applied.



Figure 7.5: Simulated soil moisture variations below furrow (shown by observation node number: Figure 7.1) for three PRB renovation treatments with 9 cm furrow water head applied on a Vertisol

For Vertisol, infiltration to depths below the bed was slower than under the furrow (Figure 7.6). The lateral wetting front movement to the bed middle occurred faster on the bed surface than at deeper depths for BP (Figure 7.6c). It took ~17, ~13 and ~12 hrs for the wetting front to reach the 90 cm depth (M10) in the bed middle for NT, SC and BP treatments, respectively. The 0-30 cm depth filled much sooner with BP, while field capacity front took longer (8 to 16 hrs) on the other two treatments.



Figure 7.6: Simulated soil moisture variations on the vertical boundary under the bed middle (shown by observation node number: Figure 7.1) for three PRB renovation treatments with 9 cm furrow water head applied on a Vertisol

For sandy clay loam, vertical infiltration below the furrow centre also varied between treatments with infiltration occurring fastest on the BP treatment (Figure 7.7). The wetting front took ~35 hrs, ~30 hrs and ~27 hrs to reach 100 cm depth (node F9) on the NT, SC, and BP treatments, respectively.



Figure 7.7: Simulated soil moisture variations below furrow (shown by observation node number: Figure 7.1) for three PRB renovation treatments with 9 cm furrow water head applied on a sandy clay loam

Soil moisture on the vertical boundary beneath the bed middle was unchanged until ~35 hrs for NT, 26 hrs for SC and 17 hrs for BP, when a 9 cm furrow water head was applied (Figure 7.8). Water reached the 30 cm depth (node M3) before the bed surface (node M1) and 90 cm depth (node M10) in all treatments (Figure 7.8). Evaporation during the simulation period reduced soil moisture in the 10 cm depth (node M1). The wetting times of extreme boundary positions are given in Table 7.1.



Figure 7.8: Simulated soil moisture variations on the vertical boundary under the bed middle (shown by observation node number: Figure 7.1) for three PRB renovations with 9 cm furrow water head applied on a sandy clay loam

Table 7.1: Time required for the wetting front to reach the given positions on the twosoils with three PRB renovation methods at 9 cm furrow water head applied

	Wetting time (hrs)					
Position	Vertisol			Sandy clay loam		
	NT	SC	BP	NT	SC	BP
Bed middle at surface	17	11	6	> 40	32	20
Bed middle at 90 cm depth	17	13	12	^a 35	^a 26	^a 17
Furrow centre at 100 cm depth	8	5	7	35	30	27

^aWetting front arrival time to 90 cm depth in bed middle

7.3.3 Effect of furrow water head on soil moisture distribution

7.3.3.1 Lateral wetting on bed surface

For Vertisol, increasing the furrow water head decreased the time taken to wet the beds (Figure 7.9). For instance, to wet the bed middle to field capacity when the furrow was at full depth it required ~14, ~10 and ~5.5 hrs for NT, SC and BP, respectively. However, reducing the furrow water head to 4 cm increased the wetting time by 43% on NT, 35% on SC and on BP. Thus, the 200 cm wide beds would experience water stress in the bed middle at wetting times less than: ~20 hrs for NT, ~13.5 hrs for SC and ~7.5 hrs for BP if 4 cm furrow water head is applied.



Figure 7.9: Effect of furrow water head on the wetting time required to reach field capacity at different lateral distances from furrow centre on a Vertisol

For sandy clay loam, increasing the furrow water head decreased the time required to wet the bed middle (Figure 7.10). For instance, it took the field capacity front ~37 hrs, 24 hrs, and 15 hrs for NT, SC and BP respectively, to reach the bed middle when the furrow was full of water. However, decreasing the furrow water head to 4 cm resulted in the bed middle not being wet during 40 hrs for NT and SC. For BP, decreasing the furrow water head to 4 cm increased the wetting time to ~31 hours. This data suggests that during the maximum 7 hrs wetting time, normal local irrigation period, the maximum bed width should be < 90 cm for NT and SC and < 105 cm for BP. However, if the furrow water head is reduced to 4 cm, the maximum bed width should be < 55 cm for NT and SC and < 66 cm for BP to ensure soil moisture in the bed middle reaches field capacity in the available 7 hrs wetting time.



Figure 7.10: Effect of furrow water head on the wetting time required to reach field capacity at different lateral distances from furrow on a sandy clay loam

7.3.3.2 Vertical wetting below furrow centre

For Vertisol, vertical infiltration was fastest for SC and slowest for NT (Figure 7.11). When the furrow was full of water (Figure 7.11), it took ~8, ~6 and ~7 hrs in wetting the 100 cm profile depth for NT, SC, and BP, respectively. However, the wetting time increased by 65% for NT, 53% for SC and 40% for BP when the furrow water head was reduced to 4 cm.



Figure 7.11: Effect of furrow water head and wetting time on the depth of wetting beneath the furrow on a Vertisol

For sandy clay loam, the soil moisture movement below the furrow centre was faster for BP than for SC and NT. The wetting front took ~33 hrs, 26 hrs and 23 hrs to reach 100 cm depth for NT, SC and BP, respectively (Figure 7.12) when the furrow was full of water. Decreasing the furrow water head to 4 cm increased the wetting time by 21% on NT, 46% on SC and 62% on BP.



Figure 7.12: Effect of furrow water head and wetting time on the depth of wetting beneath the furrow on a sandy clay loam

7.3.3.3 *Cumulative infiltration*

BP had the highest cumulative infiltration at a short period (< 10 hrs) but SC had slightly higher cumulative infiltration after 12 hours (Figure 7.13). However, cumulative infiltration was lower for NT at all times. A comparison of cumulative infiltration after 17 hrs of wetting (wetting front arrival time to bed middle for NT) showed 33% and 25% higher infiltration for SC and BP, respectively, compared with NT (177 mm) at 9 cm furrow water head. Increasing furrow water head generally increased cumulative infiltration. For instance, cumulative infiltration after 20 hrs of wetting was increased by 58% for NT, 26% for SC and 22% for BP, when the furrow water head had a larger effect on cumulative infiltration for NT than the other treatments. The difference between 9 cm and furrow full head was negligible for SC and BP.



Figure 7.13: Effect of furrow water head on simulated cumulative infiltration for three PRB renovation methods on a Vertisol

For sandy clay loam, cumulative infiltration on NT was lower than SC and BP (Figure 7.14). After 20 hrs of wetting the cumulative infiltration was up to 24% greater for SC and 52% greater for BP compared with NT at all furrow water heads. Increasing furrow water head from 4 cm to furrow full water head increased cumulative infiltration by 85% for NT, 79% for SC and 49% for BP respectively, after 20 hrs of wetting.



Figure 7.14: Effect of furrow water head on simulated cumulative infiltration for three PRB renovation methods on a sandy clay loam

7.3.3.4 Soil water storage

For Vertisol, the volume of water stored in the flow domain was generally lower for NT than for SC and BP (Figure 7.15). For instance, the volume stored after 10 hrs of wetting was 40%, 18%, 13% and 6% higher for SC and 70%, 46%, 27% and 29% higher for BP compared with NT (73, 91, 106 and 116 mm), when the furrow water head was 4 cm, 6 cm, 9 cm and furrow full respectively. However, the difference between the treatments was smaller at longer wetting periods when deep drainage losses occurred in the SC and BP treatments. Increasing the furrow water head increased the rate at which the water was stored in the profile. For instance, increasing the furrow water head from 4 cm to 6 cm, 9 cm, furrow full head increased the volume of water stored after 10 hrs of wetting by: 25, 45, 59% for NT, 5, 18, 21% for SC and 7, 17, 21% for BP respectively. Comparison of soil water storage as to when the wetting front reaches the centre of the bed for the flow depths used ranged 100-150 mm for NT, 110-130 mm for SC and 80-100 mm for BP

treatments. The uniformly distributed soil moisture in BP across the bed shows greater potential for more efficient irrigation performance.



Figure 7.15: Effect of wetting time and furrow head on the water stored in 100 cm deep root zone across half bed width for three renovation treatments on a Vertisol

For sandy clay loam, the volume of water stored in the flow domain was generally lower for NT than for SC and BP (Figure 7.16). The volume stored after 20 hrs of wetting was up to 24% higher for SC and 49-52% higher for BP compared with NT at all furrow water heads. However, the difference between treatments was smaller at longer wetting periods when deep drainage losses occurred. For instance, after 40 hrs of wetting the water stored was 2-16% higher for SC and 17-42% higher for BP compared with NT. Increasing the furrow water head from 4 cm to full furrow depth increased the volume of water stored in the root zone by 45% for NT, 81% for SC and 47% for BP after 20 hrs of wetting.



Figure 7.16: Effect of wetting time and furrow head on the water stored in 100 cm deep root zone across half bed for three renovation treatments on a sandy clay loam

7.3.3.5 Deep drainage losses

For Vertisol, after 20 hrs of wetting, deep drainage losses were 263%, 155%, 143% and 100% higher for SC and 113%, 68%, 54% and 35% higher for BP compared with NT (24, 38, 54 and 69 mm) at 4 cm, 6 cm, 9 cm and full furrow depth (Figure 7.17)., respectively. Similarly, increasing the water head from 4 cm to the full furrow condition increased deep drainage losses by 188% for NT, 59% for SC and 82% for BP after 20 hrs of wetting. By the time the wetting front reached the bed middle the deep drainage losses ranged from 10-40 mm for NT, 22-40 mm for SC and < 2 mm for BP treatment.



Figure 7.17: Effect of wetting time and furrow head on deep drainage losses below 100 cm root zone depth (across half bed x-section) for three PRB renovation methods on a Vertisol

For sandy clay loam, deep drainage losses were incurred after 33 hrs wetting on NT, 26 hrs on SC and 25 hrs on BP when a full furrow water head was applied (Figure 7.18). Deep drainage losses after 40 hrs of full furrow head wetting were ~3 times and ~6 times higher for SC and BP, respectively compared with NT (10 mm).



Figure 7.18: Effect of wetting time and furrow head on deep drainage losses below 100 cm root zone depth (across half bed x-section) for three PRB renovation methods on a sandy clay loam
7.3.3.6 Soil moisture distribution for normal irrigation applications

For Vertisol, to apply 100 mm (a common irrigation application depth in Australia) with a 9 cm furrow water head took ~9 hrs, ~7 hrs and ~5 hrs for NT, SC, and BP treatments respectively. The soil moisture distribution at the time of 100 mm water application (Figure 7.19) shows lateral penetration of the field capacity wetting front has reached 70, 74 and 85 cm distance from furrow centre for the NT, SC, and BP treatments, respectively. The vertical penetration of the wetting front was at 93, 110, and 71 cm below the datum line in the furrow centre for NT, SC and BP treatments, respectively.



Figure 7.19: Simulated soil moisture distribution across the flow domain after 100 mm water application for three PRB renovation treatments with 9 cm furrow water head on a Vertisol

For sandy clay loam, to apply 60 mm (a common irrigation application in north west Pakistan) it took ~14 hrs, ~11 hrs and ~9 hrs for NT, SC and BP, respectively with full furrow water head (Figure 7.20). The wetting front reached 50 cm, 51 cm and 56 cm lateral distance on bed surface and ~61 cm, 62 cm, and 58 cm vertical distance below furrow centre with NT, SC and BP, respectively.



Figure 7.20: Simulated soil moisture distribution across the flow domain after 60 mm water application for three PRB renovation treatments with full furrow water head on a sandy clay loam

7.3.4 Implications of lateral infiltration for irrigation management

Inflow rate and furrow water heads were measured under field conditions as reported in Chapters 4 and 5. The data also revealed that beside inflow rate, the field measured water head varied along the furrow length presumably due to differences in furrow dimensions, furrow slope and antecedent soil moisture (Oyonarte et al. 2002). Water head in the furrow decreased up to 50% in the field tail section compared with the field head section (465 m long furrows in Vertisol) at low inflow rate (1 L s⁻¹) and up to 23% at high inflow rates (3 L s⁻¹). There was insufficient data to establish treatment specific relationships. Hence, the data was pooled to identify a single relationship (Figure 7.21). The furrow water head was found to vary from 4 cm to 10 cm when the inflow rate was changed from ~1 L s⁻¹ to ~5 L s⁻¹. This relationship indicated that flow rate can be increased with comparatively little increase in velocity and erosivity of flow, which reinforces the practical application of the aforementioned relationships. The inflow rate and the furrow head was subsequently used to develop charts to highlight the relationship between inflow rate and target wetting time for each soil.



Figure 7.21: Effect of inflow rate on furrow water head measured in the field head section during experiments on a Vertisol and a sandy clay loam

Using the above relationship with the earlier simulation results (Figure 7.9) revealed that the inflow rate affected the lateral infiltration on all renovation treatments. For instance, increasing the inflow rate from 1 to 5 L s⁻¹ reduced the time required to wet the bed middle (200 cm wide PRB) on a Vertisol by 30%, 26% and 27% for NT, SC and BP treatments, respectively (Figure 7.22). However, for a 66 cm wide PRB on the sandy clay loam the renovation treatment differences were negligible at high inflow rates (Figure 7.23). At low inflow rates (1 L s⁻¹) the wetting time was ~7 and 9 hrs on BP and SC treatments, respectively but 10 hrs for NT (Figure 7.23).



Figure 7.22: Effect of inflow rate on wetting time of bed middle (200 cm wide bed) to field capacity for three PRB renovation methods on a Vertisol



Figure 7.23: Effect of inflow rate on wetting time of bed middle (66 cm furrow spacing-NB) to reach field capacity for three PRB renovation treatments on a sandy clay loam

The maximum bed width was calculated based on lateral penetration of the wetting front for a seven hour wetting period (Figure 7.24). The maximum bed width whose middle could be wetted to field capacity when 5 L s⁻¹ is applied is 98 cm for BP compared with 83 cm for NT. However, the maximum bed width was ~55 cm for NT and ~67 cm for BP when 1 L s⁻¹ was applied. Similarly, the nominal bed width increased by 51%, 55% and 48% for NT, SC and BP treatments respectively, when inflow rate was increased from 1 L s⁻¹ to 5 L s⁻¹ (Figure 7.24). This figure may be useful to optimise irrigation management and bed width for a sandy clay loam soil under similar field conditions.



Figure 7.24: Effect of inflow rate on maximum bed width required to ensure wetting of bed middle to field capacity within 7 hrs of wetting time on a sandy clay loam

7.4 Discussion

7.4.1 Effect of renovation methods and furrow head on infiltration

The simulations confirmed the large impact of renovation methods on lateral infiltration into PRBs on both soils. The fast lateral infiltration shown for BP on both soils (Vertisol and sandy clay loam) suggests that this treatment should be beneficial at mitigating the negative impacts on crops in the bed middle caused by poor lateral infiltration (Jin et al. 2007). However, the maximum wetting time of 20 hrs identified for NT when a 4 cm furrow water head is applied may create difficulties on Vertisols with long furrows under Australian farm conditions. On sandy clay loam soils the wetting time required to wet the bed middle (> 40, > 40 and 32 hrs on NT, SC and BP, respectively) at low flow rates will be difficult to achieve if the furrows are short (as is common in Pakistan conditions). However, the fast lateral infiltration in BP treatment offers greater potential for improved irrigation performance within shorter irrigation application time.

There was more vertical infiltration in the SC and NT treatment on the Vertisol compared with the BP treatment (Figure 7.19). This resulted in larger deep drainage losses on both the NT and SC treatments. In contrast, the faster lateral infiltration in the surface layer of the BP treatment on both soils reduced the vertical infiltration beneath the furrow reducing deep drainage losses. Thus BP treatment may be used to improve irrigation efficiency by reducing excessive deep drainage losses.

While renovation methods did affect infiltration on the sandy clay loam soil the speed of wetting front movement in both directions (lateral & vertical) was up to 50% less than in the Vertisol. BP on the sandy clay loam was not successful in improving lateral infiltration to the extent that the bed middle could be wetted in less than 7 hrs. Therefore, alternative strategies, including increasing the furrow water head or decreasing the bed width, should be considered to ensure wetting to the middle of the bed.

Cumulative infiltration was indicative of soil infiltration capacity and was variable between treatments on both soils. The higher cumulative infiltration for BP on the Vertisol, at short periods, and for longer periods on the sandy clay loam, reflected the higher lateral infiltration capacity compared with other two treatments.

The volume of water stored in the root zone was indicative of soil porosity and water holding capacity, which was improved by BP treatment (Thomas et al. 1990; Eldridge & Robson 1997). The increased soil water storage on BP treatment suggested that this treatment may be helpful in reducing the frequency of irrigation application, improving effective use of in-season rainfall and avoiding economic crop damage from water stress in water limited/drought conditions.

Irrigation is generally aimed at applying approximately 100 mm on the Vertisol in Australia and ~60 mm on the sandy clay loam in Pakistan. An analysis of applying the required irrigation depth showed that the wetting front would still be short of reaching the middle of the bed in all treatments on both soils. Thus, wetting of the bed middle could only be achieved with deep drainage losses, which seems to be larger on SC and NT compared with BP on the Vertisol.

Furrow water head considerably affected lateral and vertical infiltration on all treatments. However, wetting front movement for NT on the Vertisol and for SC on the sandy clay loam was more sensitive to furrow water head changes. For instance, increasing furrow water head from 4 cm to full furrow reduced wetting time by up to 43% on the Vertisol and up to 113% on the sandy clay loam. Greater lateral infiltration with increased furrow water head was also demonstrated by Tabuada et al. (1995). Thus, increasing furrow water head can be used to improve lateral infiltration. However, this will also enhance the vertical infiltration, and can promote excessive deep drainage losses as evidenced by up to 188% increased deep drainage losses with increased furrow water head from 4 cm to full furrow on NT after 20 hrs of wetting on the Vertisol. However, increasing the furrow water head to full furrow on BP did not reduce the wetting time to below 7 hrs on the sandy clay loam. The relationships identified between inflow rate and furrow water head may be helpful in optimise irrigation management for improved lateral infiltration in both soils.

7.4.2 Implications for irrigation management

The shrink-swell nature of Vertisol and long Australian furrows make these systems less susceptible to poor lateral infiltration due to both soil crack development under dry soil conditions and often long irrigation cut-off times. However, at the furrow tail section due to less wetting time than at the head section or when dry soil overlies wet soil at planting time, absence of cracks (Dexter 1991) can negatively affect crop germination due to poor lateral infiltration (Lucy 1993). Poor lateral infiltration is a significant problem in the sandy clay loam soil and has been shown to affect crop performance (Hassan et al. 2005; Akbar et al. 2007; Jin et al. 2007). Therefore, alternative irrigation practices could be suggested. The most common alternative practice could be to fill the furrow and just overtop the wide beds at planting stage and as the season progresses the roots will enhance the infiltration. The second practice could be to use narrow beds with wider furrows in short fields so that a larger volume of water is held in the furrows for a longer period to ensure the wetting of the bed middle. The third option is to increase the space between the centre rows.

The hydrus simulation demonstrated large differences in infiltration with renovation methods and inflow rates on both soils. These figures may be helpful to optimise irrigation management and bed width design on similar site and field conditions. Although these guidelines are applicable to the specific field conditions encountered during the experiments, the general findings also have broader applications. The simulations demonstrated that increasing water head in the furrow and increasing wetting time can improve the lateral infiltration on wider beds. However, the longer term benefits of PRB and the renovation strategies could be expected to enhance soil structure and nutrients with reduced soil disturbance, crop residue retention, soil fauna, and biota (Smith et al. 1983; McHugh et al. 2009). These benefits seem vital for the sustainable improvement and crop production, especially for the degraded soil structure found on the sandy clay loam site. This suggests that the renovation methods need to be adopted judiciously, keeping in view their transient impacts on soil hydro-physical properties and production. However, the dominance of lateral infiltration in BP treatment can encourage farmers to combine CA practices for achieving improvement over time.

7.5 Conclusions

The variable renovation methods and furrow water head were shown to affect both lateral and vertical infiltration into PRBs on both the Vertisol and sandy clay loam. This suggests that both strategies can be used as management tools to improve the efficiency and effectiveness of irrigation. The specific conclusions are:

- Hydrus 2D, properly parameterised by and crossed checked against field measured data, adequately simulated the distribution of soil moisture in PRB systems and accounted for the variable soil management, thus was used to explore management options to improve irrigation practice and thereby provide insight to improve lateral infiltration.
- The BP treatment increased lateral infiltration and the SC treatment increased vertical infiltration compared with the NT treatment. Thus, different wetting times are required to replenish *SMD* depending on the renovation method.
- In the sandy clay loam, which was badly structurally degraded, the BP treatment showed sufficient improvement over the NT and SC treatments in the amount and dominance of lateral infiltration to encourage farmers to combine BP with CA practices to achieve further improvement over time.
- The wetting time of bed middle was 20, 14 and 8 hrs on the Vertisol and > 40,
 > 40 and ~32 hrs on the sandy clay loam when 4 cm furrow head (1 L s⁻¹) was applied to NT, SC and BP treatments, respectively. Hence, the required wetting time may be achievable under Australian conditions but is likely to be difficult to achieve on the sandy clay loam under Pakistani conditions.
- The minimum wetting time (15 hrs) required to wet the bed middle on the sandy clay loam with BP and full furrow water head is greater than the irrigation application period (~7 hrs) commonly available in Pakistan. This suggests that the bed width may need to be reduced. Another option could be to fill the furrow by just overtopping the bed at the crop early stage.

The figures developed in this work may be used as decision support tools to assist in improving irrigation management and for the selection of optimum bed width based on lateral infiltration.

CHAPTER 8: Managing irrigation and field design to improve performance

8.1 Introduction

The literature review outlined the significant gains in irrigation performance of furrow irrigation systems with optimised irrigation management and field design (Raine & Shannon 1996; Bakker et al. 1997; Dalton et al. 2001; Smith et al. 2005; Langat & Raine 2006; Smith et al. 2009). However, specific studies on this topic are rare for PRB farming systems. The benchmarking study (Chapter 4) revealed variable irrigation performance in existing PRB farming systems, which appeared to be largely inefficient in both Australian and Pakistani contexts. The main reasons for inefficiency seem to be the uninformed selection of Q, Tco and inappropriate furrow lengths in both countries. Field trials (Chapter 6) and simulation (Chapter 7) evaluating lateral infiltration showed non-uniform soil moisture distribution in wide PRBs. Prevailing irrigation management techniques assume uniform soil moisture distribution across the beds and do not generally account for lateral infiltration, which may limit crop performance in the middle of the bed.

Irrigation applications during three cropping seasons (Chapter 5) under three PRB renovation treatments were difficult to optimise both within and between seasons. For instance, field application losses were significantly larger on freshly applied BP during the 2010 wheat and 2011 corn seasons (Chapter 5) due to a larger infiltration capacity compared with NT. The high infiltration produced a slower water advance rate in the furrow, which may be managed with increased inflow rate (Q). Similarly, *SMD* was not fully met during the 2011 hemp crop due to insufficient water application. This was because the fast water advance in the compacted furrows at applied Q resulted in switching off the irrigation much sooner than other furrows to prevent bed overtopping at the furrow tail end. In this case, inadequate lateral infiltration reduced crop performance in the bed middle. Hence, there is a need to optimise irrigation management in response to the specific effects of bed renovation on infiltration.

It was therefore hypothesised that the existing irrigation performance of PRB farming systems in both countries (Chapter 4) was not optimal and optimised irrigation management and field design can improve the existing irrigation performance in both countries. It was also hypothesised that optimal irrigation management and field design can further improve the irrigation performance by reducing the irrigation application losses, replenishing *SMD* and improving lateral infiltration on the three PRB renovation treatments evaluated on a Vertisol (Chapter 5), with permissible trade-offs. Therefore, this chapter aimed to test these hypotheses and to develop guidelines for improving irrigation performance according to the specific field conditions of both countries as discussed in Chapter 3.

8.2 Material and methods

To achieve these objectives a desktop study was undertaken, where the derived infiltration and irrigation performance data, which were experimentally obtained in Chapters 4, 5, 6 & 7, were used to explore the scope of irrigation performance improvement and replenishment of *SMD* using optimised irrigation management and field design strategies. The strategies were selected on the basis of observed field conditions, relevance to prevailing practices and convenience of implementation.

Four strategies were evaluated that involved optimising: (1) *Tco* alone; (2) Q and *Tco* together; (3) furrow length; and (4) *Tco* relative to *Ta*. Each strategy has practical significance under actual field conditions. For instance, it may not always be possible to control Q and *Tco* together, but controlling *Tco* alone may still be beneficial. Similarly, optimising furrow length may not be feasible on most farms, especially the long Australian fields, due to cost, but could be feasible under some farm conditions, where infrastructure is currently under development or redevelopment. Additionally, exploring the optimum relationship between *Tco* and *Ta* under variable field conditions should be beneficial. Except for change to field length, the majority of the proposed strategies can be implemented with minimal cost to infrastructure, machinery or labour. Simulation of irrigation performance and calculation of performance indices and deep drainage losses were determined according to the procedure described by Smith et al. (2005). The statistical analysis was conducted using the procedure described in section 3.7.

8.2.1 Irrigation management and field design optimisation protocol

The SIRMOD III model, which was calibrated for the benchmarking of existing PRB systems in Chapter 4 and for evaluation of three PRB renovation methods in Chapter 5, was used to optimise the irrigation management of the benchmarked Australian and Pakistani irrigation events and those of the renovation treatments. The input parameters of Q, *Tco* and furrow length were adjusted in the model until the best simulated irrigation performance in terms of efficiency, uniformity and lateral infiltration was achieved for each measured furrow during irrigation events recorded. The lateral infiltration condition was fulfilled by matching simulated infiltration opportunity time (recession - advance time) along the furrow length with the field measured and simulated wetting time (Chapter 7) of the bed middle.

The irrigation efficiency outputs were based on certain criteria i.e. ensuring water arrived at the tail end of each furrow, $Er \ge 85\%$, maximising $Ea \ge 90\%$ and confirming lateral infiltration into the bed centre, especially for densely grown crops. In other words, the selected parameterisation and criteria ensured avoidance of dry field sections, fulfilment of SMD, minimised irrigation water losses, and maximised lateral infiltration. The efficiency settings were successfully achieved in the majority of the simulations. However, on a few occasions when $Er \ge 85\%$ was not possible under settled bed-furrows (e.g. 2011 hemp crop), then Ea > 90% was used to avoid losses and overtopping of the beds at the tail end of the field. Similarly, the lateral wetting to the bed centre was largely met at Er = -85% in the majority of cases on Vertisol, but in a few cases when this condition was not met then Tco was increased to ensure wetting to the middle of the bed. The lateral wetting times and distance recorded with the Sentek EnviroSCAN[®] during each cropping period (Chapters 4, 5 & 7) were used for identifying lateral infiltration. However, the lateral wetting to the bed middle condition was not met for a 132 cm wide bed on sandy clay loam due to prolonged wetting time and close cotton and corn crop rows to wetted furrow.

Optimisation of current PRB farming systems (Chapter 4) were analysed separately for Australia and Pakistan due to differences in soil properties, field designs and irrigation management. Similarly, the analyses of the three PRB treatments on the Vertisol (Chapter 5) are presented separately.

8.3 Results

8.3.1 Optimising irrigation management of PRB systems in Australia

The majority of the measured furrow irrigation events (Chapter 4) were overirrigated, thus irrigations were able to be successfully optimised by changes in management. The field measured and best simulated results with improved irrigation efficiencies and irrigation water saving were summarised in Table 8.1.

Table 8.1: Average irrigation performance parameters for strategies of optimising (i) time to cut-off (*Tco*) and (ii) inflow rate (*Q*) plus *Tco* compared with field measured farmer practice (FP) for two sites (site A1 = 465 m long furrows with blocked tail ends and site A2 = 455 m long furrows with open end furrows at tail end) evaluated during two irrigation events on a Vertisol (SD in brackets)

Irrigation	Si	ite A1-Soy	bean	Site A2-Cotton			
Performance	FD	Opt	imised	FD	Optimised		
parameter	rr -	Tco	Q & Tco	FP	Tco	Q & Tco	
Q	1.94a	1.94a	2.75b	2.54a	2.54a	3.13b	
$(L s^{-1})$	(0.1)	(0.1)	(0.1)	(0.1)	(0.1)	(0.2)	
Тсо	1100a	981b	503c	635a	473b	370c	
(min)	(61)	(72)	(38)	(64)	(78)	(40)	
Inflow	138a	127b	89c	107a	80b	77b	
(mm/irrigation)	(11)	(10)	(7)	(9)	(12)	(12)	
^{a}DD	37a	25b	2c	13a	2b	1b	
(mm/irrigation)	(3)	(2)	(0)	(0)	(0)	(0)	
Ea	73a	80b	98c	79a	97b	97b	
(%)	(6)	(6)	(3)	(5)	(2)	(3)	
Er	98a	95a	88b	97a	88b	85b	
(%)	(2)	(3)	(4)	(2)	(3)	(0)	
DU	90a	85b	90a	87a	77b	82c	
(%)	(3)	(3)	(3)	(2)	(2)	(4)	

^{*a}DD = Deep drainage losses per irrigation*</sup>

*Figures followed by the different letters in rows for each site are significantly (P=0.05) different between strategies

In the majority of cases, setting Er at ~85% was successful in avoiding dry field sections and in fulfilling lateral infiltration into the bed centre. However, in a few cases under densely grown crops (e.g. soybean and wheat) the simulated wetting time (difference of recession and advance time) was closely evaluated to ensure wetting of the bed middle in tail sections of the field. For instance, the lateral wetting time to the bed middle of ~300 minutes during soybean crop was not met in ~65 m length from the end of the furrow (Figure 8.1: dotted box) at Er = -85%, which was rectified by increased *Tco* to avoid water stress in centre rows.



Figure 8.1: Impact of optimizing *Tco* at Er = 85% on the distribution of infiltration opportunity time along the furrow length during two irrigations of 2009 soybean. The dotted box highlights the tail end area susceptible to poor lateral infiltration.

8.3.1.1 Optimising time to cut-off

Strategy 1 (*Tco* optimisation alone) suggested a significant reduction in *Tco* to improve performance (Table 8.1). At site A1, *Tco* optimisation produced an improvement (7%) in *Ea* with significant 8% irrigation water saving (32% reduction in deep drainage losses), however a reduction of ~3% in *Er* and ~5% in *DU* was observed for both irrigations compared with existing practice. At site A2, *Tco* optimisation resulted improved (~18%) *Ea* and ~25% irrigation water saving (85% reduced *DD*) with a reduction of 9% in *Er* and 10% in *DU* from existing practice.

8.3.1.2 *Optimising inflow rate and time to cut-off*

Strategy 2 (optimising Q and Tco together) produced a large improvement in irrigation performance (Table 8.1). At site A1, optimisation identified a significant (~25%) improvement in Ea and 36% irrigation water saving (95% reduction in deep drainage losses) with a reduction of ~10% in Er from farmer values. Similarly, at site A2, optimisation achieved a significant (~18%) improvement in Ea and 28% irrigation water saving (92% reduction in deep drainage losses) with a reduction of ~12% in Er and 5% in DU from farmer values. Optimised Tco over a range of Q was plotted on both sites (Figure 8.2), which may be used to support decision making under similar field conditions. For instance, comparisons of both sites indicated increased optimised Tco and Er on site A1 when compared with site A2 at given Q,

while increased Q negatively impacted on DU at site A1 when compared with site A2.



Figure 8.2: Example irrigation management tool for illustrating the optimised Tco vs. Q and resulting Ea, Er and DU evaluated for two irrigations at two sites

8.3.1.3 *Optimising furrow length*

Irrigation performance was greatly affected by furrow length (Figure 8.3). As furrow length increases, there is a need to increase Q to achieve the same Ea and DU. For instance on site A1, applying 2 L s⁻¹ to a blocked end furrow, achieved $Ea \sim 99\%$ for a 100 m furrow length but only ~60% for a 600 m furrow length (Figure 8.3a). In contrast on site A2, applying 5 L s⁻¹ to an open ended furrow achieved Ea = -90%when the furrow length was 600 m but only ~45% for a 200 m furrow (Figure 8.3b). The impact of furrow length on DU was greater on site A1 than at site A2. For instance, DU dropped to ~45% at 300 m furrow length with $Q = 5L \text{ s}^{-1}$, while DU was above 80% for all tested furrow lengths and Q on site A2. In contrast, the impact of furrow length on inflow volume was smaller on site A1 than on site A2. For instance, a maximum inflow volume of ~160 mm was applied to 600 m furrow length at $Q = 2 \text{ L s}^{-1}$ on site A1. In contrast, a maximum inflow volume of ~190 mm was required for 200 m furrow length at $Q = 5 \text{ L s}^{-1}$ on site A2. Figure 8.3 provides a simple chart to assist understanding of irrigation management and field design inputs under different field conditions on a Vertisol where there is a requirement to fully wet the middle of the beds according to site A1 with blocked furrows at tail ends and site A2 with open ends at tail ends.



Figure 8.3: Example irrigation management and field design tool illustrating the effect of furrow length and Q on *Tco*, inflow volume and *Tco* Ta^{-1} (solid lines) and *Ea* and *DU* (dotted line) for achieving $Er \ge 85\%$, water reaching furrow tail end on (a) site A1 with blocked end furrows and (b) site A2 with open ended furrows requiring wetting to bed middle on site A1 on Vertisol (vertical bars show SD)

8.3.1.4 *Optimising time to cut-off relative to water advance time*

The optimised *Tco* was evaluated relative to *Ta* at different furrow lengths. If optimisation suggests *Tco* $Ta^{-1} = 2$, then the irrigation needs to be switched off at twice the time that the water reached the furrow tail end. A comparison of *Tco* Ta^{-1}

showed larger values with shorter furrow length and larger Q as shown in Figure 8.3. The ratio converged to unity at longer furrow lengths, while the effect of Q on the ratio (*Tco Ta⁻¹*) was negligible on longer furrow lengths. Comparing the sites showed a larger *Tco Ta⁻¹* with higher Q and short furrow lengths on site A2 with open ended furrows than on site A1 with blocked end furrows.

8.3.2 Optimising irrigation management of PRB farming systems in Pakistan

Optimisation of Q, Tco and L were carried out to evaluate the irrigation performance on three sites in Pakistan. The bed middle wetting condition was not met due to the long wetting time requirement for sandy clay loam on 132 cm wide PRBs. However, the maize crop was less susceptible to water stress in the bed middle, as the crop rows were close to the wetted furrows. Thus, the optimisation was based on $Er \ge$ 85%, water reaching to the tail end, minimising water losses and avoiding excessive tail end water overtopping the bed, as the furrow ends were blocked at tail ends.

The results showed that the irrigation performance of existing PRBs in Pakistan (Chapter 4) can be improved with optimised irrigation management. However, the results were highly site specific. For instance, at site P1 the infiltration capacity of soil was high, thus increasing Q and reducing Tco (Figure 8.4) improved irrigation performance. In contrast, at site P2 the infiltration capacity of soil was low, thus using a lower Q and a longer Tco (Figure 8.5) improved the irrigation performance.



Figure 8.4: SIRMOD screen shots for showing the impact of irrigation management strategies on infiltration along a blocked end furrow (86 m) for a high infiltration (site P1-NB during first irrigation) sandy clay loam soil



Figure 8.5: SIRMOD screen shots for showing the impact of irrigation management strategies on infiltration along a blocked end furrow (90 m) for a low infiltration (site P2- NB during first irrigation) sandy clay loam soil

8.3.2.1 *Optimising time to cut-off*

Optimising the *Tco* at the existing inflow rate (Strategy 1) provided an average improvement of 11% in Ea, 3% in DU, 14% in irrigation water saving and a 44% reduction in deep drainage losses on the NB systems compared with farmer practice (Table 8.2). Optimising *Tco* for WB did not produce any significant benefits because optimisation suggested increased Tco, which was not possible due to tail end overtopping in majority of cases. However, the effect was highly variable between sites (Table D5: Appendix D). For instance, optimising Tco on NB systems at site P1 provided an average improvement of 9% in Ea and ~15% irrigation water saving, 31% reduction in deep drainage losses at the cost of 10% reduction in DU. For WB systems, Tco optimisation improved Ea by 27%, DU by 15% and irrigation water saving by 35% with up to 96% reduction in deep drainage losses. However, Er decreased by 10%. At site P2, the challenge was to replenish the SMD and to avoid tail end over topping. Hence, increasing Tco on NB increased Er by up to 5% but Ea was reduced by 5%. Similarly on WB, increasing Tco resulted in a 10% increase in Er but required a 13% increase in volume applied. For site P3, optimisation involved reducing Tco on NB which improved Ea by 30%, DU by 21% and irrigation water saving by 36% with up to 96% reduction in deep drainage losses but a 7% reduction in Er. In contrast, increasing Tco on WB increased Er by 4% and the applied volume by 8% reducing Ea by 2%. The average best achievable results for optimised Tco on individual sites are summarised in Table D5 (Appendix D).

Irrigation		Narrow b	oed	Wide Bed			
Performance parameter	FP	Opt	imised	FD	Optimised		
		Тсо	Q & Tco	ГГ	Тсо	Q & Tco	
Q	2.3a	2.3a	3.0a	3.5b	3.5b	1.5b	
(Ls^{-1})	(0.6)	(0.6)	(0)	(1.0)	(1.0)	(0)	
Тсо	37a	31a	24a	34b	34b	72b	
(min)	(3)	(7)	(7)	(2)	(2)	(8)	
Inflow	88a	74a	73a	60b	60b	55b	
(mm/irrigation)	(27)	(22)	(10)	(17)	(17)	(7)	
DD	32a	18a	13a	8b	8b	2b	
(mm/irrigation)	(14)	(12)	(5)	(6)	(6)	(1)	
Ea	70a	81a	82a	90b	90b	97b	
(%)	(18)	(17)	(11)	(9)	(9)	(2)	
Er	94a	93a	93a	86b	86b	89b	
(%)	(5)	(6)	(5)	(10)	(10)	(5)	
DU	76a	79a	80a	79a	79a	86b	
(%)	(13)	(14)	(12)	(10)	(10)	(9)	

Table 8.2: Average irrigation performance parameters for strategies of optimising (i) time to cut-off (*Tco*) and (ii) inflow rate (*Q*) plus *Tco* compared with field measured farmer practice (FP) for two bed sizes evaluated during two irrigation events (~90 m furrow length) on a sandy clay loam (SD in brackets)

* Figures followed by same letters in rows for a strategy are not significantly different (P=0.05) between the bed sizes

8.3.2.2 *Optimising inflow rate and time to cut-off*

Optimising the Q and Tco (strategy 2) achieved further improved irrigation performance and irrigation water saving for both NB and WB compared with farmer managed practices (Table 8.2). Comparisons for all three sites are summarised in Table D5 (Appendix D). The optimum Tco for a range of Q (1-5 L s⁻¹) is shown for both NB (Figure 8.6a) and WB (Figure 8.6b) and all three sites (Figure D1: Appendix D). These figures illustrate the impact of bed widths, site conditions, and management parameters (Q and Tco) on irrigation performance, which supported decision making under existing field conditions. For instance, at Q = 2 L s⁻¹ the NB would require ~40 minutes (Figure 8.6a) and WB ~55 minutes (8.6b) for achieving Ea = 80%, Er = 90% and DU = 80% on NB and Ea = 90%, Er = 85% and DU = 80%on WB.



Figure 8.6: Example irrigation management tool showing the effect of inflow rate on optimum time to cut-off for (a) narrow and (b) wide beds with ~90 m furrow length at three sites on a sandy clay loam (vertical bars show SD)

Optimising *Q* and *Tco* together were more beneficial than optimising *Tco* alone using the farmer managed inflow rate. For instance, optimising both Q and Tco improved Ea by 12% and 7%, DU by 4% and 7%, irrigation water saving by 17% and 8% and DD reduced by 60% and 75% on NB and WB systems, respectively (Table 8.2). However improvements in Ea (up to 30%), DU (up to 21%), and irrigation water saving (up to 38%) compared with farmer practice were achieved on individual sites (Table D5: Appendix D). Optimising both O and Tco was beneficial for the overirrigated events in saving water (Figure 8.7a) and under-irrigated events by fulfilling SMD with less additional water (Figure 8.7b) compared with Tco optimisation alone. For the over-irrigated events (Er > 85%, i.e. Site P1 both beds and site P3 NB only), applying the optimum Q and switching off the irrigation at the appropriate time, saved up to 32% on average and reduced deep drainage by up to 96% (Figure 8.7a). Half of the measured irrigation events were under-irrigated (Er < 85%), with 75% of the under-irrigated events identified in WB systems. These systems required a longer *Tco* at the existing *Q*. Thus, the irrigation efficiency was reduced when the *Tco* was increased (Figure 8.7b). However, by optimising both Q and Tco the Er was increased and irrigation performance improved with a negligible reduction in Ea (Figure 8.7b).

Chapter 8: Managing irrigation and field design to improve performance



Figure 8.7: Impact of irrigation management strategies on performance of (a) overirrigated events (NB & WB at site P1 and NB at site P3) and (b) under-irrigated events (NB & WB at site P2 and WB at site P3) in a sandy clay loam (vertical bars show SD)

8.3.2.3 *Optimising furrow length*

The interaction of furrow length (*L*), *Q* and *Tco* on irrigation performance is illustrated in Figure 8.8(a) for NB and Figure 8.8(b) for WB. A comparable irrigation performance (*Ea and DU*) is possible with different furrow length by optimising *Q* and *Tco* on both bed sizes. However, the optimisation is highly site specific (Figures D2 & D3: Appendix D).

Increasing furrow length generally reduces irrigation performance irrespective of bed size (Figure 8.8). However, site to site comparisons indicated that sub-optimal Q exacerbated the reduction on irrigation performance with longer furrow lengths. For instance, at site P3, *Ea* reduced to ~45% when Q was 1 L s⁻¹ and *L* was 300 m (Figure D2c: Appendix D), but when Q = 5 L s⁻¹, *Ea* was ~90% for the same furrow length. Similarly, *DU* and *Er* of under-irrigated site P2 (both bed sizes) was improved with a lower Q and longer furrow length (Figure D2b & D3b: Appendix D) which was contrary to results observed for sites P1 and P3.



Figure 8.8: Example field design and irrigation management tool illustrating the effect of furrow length and inflow rate on optimised *Tco* and *Tco* Ta^{-1} (solid lines) and *Ea* and *DU* (dotted lines) for (a) narrow beds (66 cm furrow spacing) and (b) wide beds (132 cm furrow spacing) in a sandy clay loam. Optimisation required $Er \ge 85\%$ and water reaching to the tail end)

8.3.2.4 Optimising time to cut-off relative to water advance time

The optimum *Tco* Ta^{-1} was affected by *Q*, *L*, bed sizes, and soil infiltration properties. The optimum ratio was below unity on NB (Figure 8.8a) and above unity on WB (Figure 8.8b) for all furrow lengths and inflow rates evaluated. Similarly, individual site comparisons indicated that the optimum ratio (*Tco* Ta^{-1}) was below unity for the over-irrigated sites (i.e. sites P1 NB & WB, site P3 NB) with high infiltrative soils and above unity for the under-irrigated sites (e.g. sites P2 NB & WB, site P3 WB) with low infiltrative soil (Figure D2 & D3: Appendix D). The optimum ratio tended to increase on short furrows and decrease on longer furrows. Thus, switching off irrigation at the time of flow arrival at the tail end of the field will cause over-irrigation on NB and under-irrigation on WB.

8.3.3 Optimising irrigation management in response to PRB renovation methods on the Vertisol in Australia

The irrigation management observed during the trials over three cropping seasons (2010 wheat, 2011 hemp and 2011 corn) and PRB renovation methods (NT, SC and BP) were not optimal (Chapter 5). Thus, simulations were conducted to identify improved irrigation management practice for each renovation method. These practices are reported to be helpful in controlling over or under-irrigation and to avoid dry bed middles under densely grown crops (e.g. wheat).

8.3.3.1 *Optimising time to cut-off*

Optimised *Tco* alone (strategy 1) showed variable results for the three renovation methods over the three cropping seasons (Table D6: Appendix D). For instance, the existing irrigation management during the first irrigation to 2010 wheat under freshly renovated PRB caused poor lateral infiltration at the tail end of the field in all treatments. Thus, *Tco* optimisation suggested increasing *Tco*, which significantly (P=0.05) reduced *Ea* and increased irrigation applications in the majority of treatments.

Fast water advance in settled bed furrows of un-renovated PRB during the 2011 hemp season caused under-irrigation (*Er* ranged 65-71%; Table D6: Appendix D). Thus *Tco* optimisation was aimed to replenish the *SMD*. However, evaluations showed no benefit from increasing *Tco* as the water overtopped the bed at the tail end of the field on all treatments.

The freshly renovated PRB during the 2011 corn season resulted in low *Ea* and high *Er*. Thus, *Tco* optimisation aimed to improve *Ea*, which suggested a reduction in *Tco*. However, this resulted in no significant (P=0.05) improvement in *Ea*.

8.3.3.2 *Optimising inflow rate and time to cut-off*

Optimising Q and Tco together (strategy 2) significantly (P=0.05) improved irrigation performance for the most sub-optimally managed and freshly applied BP treatment during the first (2010 wheat) and last (2011 corn) seasons (Table 8.3). Results indicated improved Ea (20%), DU (8%), irrigation water saving (24%) and reduced (71%) deep drainage losses on freshly applied BP with optimised Q and Tco. Optimising Q and Tco also significantly (P=0.05) improved Ea (7%), irrigation water saving (8%) and reduced deep drainage losses (44%) on freshly applied SC. However, there was no significant improvement in irrigation performance on the NT treatment (Table 8.3) as the Q and Tco applied in the field were similar to the optimised values.

Irrigation management optimisation required significantly (P=0.05) higher Q and significantly shorter Tco for freshly applied SC and BP during the wheat and corn crop seasons (Table 8.3). Optimisation also suggested increasing Q on NT but the resultant impact on Tco, Ea, Er, DU, inflow volume and DD were negligible.

 Table 8.3:
 Impact of three fresh PRB renovation methods and optimised (Q and Tco)
 on irrigation performance during two cropping seasons on Vertisol (SD in brackets)

Tuniastian	Wheat 2010 and corn 2011						
Parameter	Fie	eld measure	ed	Q & Tco optimised			
	NT	SC	BP	NT	SC	BP	
Q_{\perp}	1.87a	1.97a	2.21a	2.00b	3.25b	4.00b	
$(L s^{-1})$	(0.1)	(0.1)	(0.2)	(0.1)	(0.4)	(0.1)	
Тсо	904a	942a	963a	901a	540b	401b	
(min)	(90)	(99)	(135)	(104)	(110)	(59)	
Inflow	109a	120a	137a	109a	110b	104b	
(mm/irrigation)	(12)	(13)	(18)	(15)	(16)	(8)	
DD	13a	25a	49a	12a	14b	14c	
(mm/irrigation)	(4)	(9)	(11)	(5)	(4)	(5)	
Ea	87a	78a	66a	88a	85b	86b	
(%)	(10)	(15)	(16)	(10)	(9)	(15)	
Er	94a	97a	98a	95a	98a	97a	
(%)	(4)	(3)	(2)	(4)	(2)	(2)	
DU	83a	82a	80a	85a	83a	88b	
(%)	(7)	(8)	(7)	(6)	(5)	(5)	

*Figures followed by different letters for the same treatment in rows are significantly (P=0.05) different between strategies

Irrigation management optimisation was highly season and event specific. For instance, the treatment difference in irrigation performance was not significant (P=0.05) during the 2011 hemp season when all the treatments were under effective NT. However, the irrigation performance during the 2011 hemp season could have been improved with significantly (P=0.05) lower Q and increased Tco compared with field measured values (Table 8.4). This is contrary to the freshly applied

treatments (Table 8.3). In this case inflow volume and Er were significantly improved, while Ea, DU and deep drainage losses were unaffected. However, SMD was still not fully met due to bed over topping at the furrow tail end.

Table 8.4: Impact of one season old renovation methods and the optimised (Q and Tco) on irrigation performance across all experimental plots during the 2011 hemp season (SD in brackets)

Irrigation parameter	Field measured				Q & Tco optimised			
	NT	^a SC/NT	^a BP/NT	Mean	NT	^a SC/NT	^a BP/N7	^r Mean
Q	1.79a	1.79a	2.01a	1.86a	1.17b	1.18b	1.59b	1.32b
$(L s^{-1})$	(0.2)	(0.2)	(0.1)	(0.2)	(0.4)	(0.2)	(0.6)	(0.5)
Тсо	406a	438a	406a	417a	882b	834b	598b	764b
(min)	(124)	(126)	(145)	(133)	(177)	(218)	(202)	(192)
Inflow	47a	51a	53a	51a	66b	63b	61a	61b
(mm/irrigation)	(17)	(18)	(17)	(19)	(26)	(29)	(28)	(22)
DD	2a	3a	3a	4a	6b	5a	4a	7b
(mm/irrigation)	(1)	(1)	(1)	(2)	(2)	(3)	(3)	(3)
Ea	94a	92a	93a	93a	91a	92a	94a	92a
(%)	(11)	(11)	(10)	(11)	(9)	(8)	(4)	(7)
Er	65a	70a	71a	69a	78b	79b	80b	79b
(%)	(6)	(11)	(8)	(11)	(10)	(10)	(7)	(9)
DU	75a	73a	75a	74a	72a	75a	80a	76a
(%)	(20)	(18)	(15)	(18)	(16)	(16)	(10)	(15)

^aThe SC and BP were only furrow cleaned thus all treatments were effectively NT *Figures followed by different letters in rows on same treatment show significant (P=0.05) difference between field measured and optimised results

The optimised *Tco* for a range of Q (2 to 5 L s⁻¹) are shown (Figure 8.9) for the three freshly renovated PRBs during the wheat and corn crop seasons. The optimisation was based on confirming wetting of the bed middle throughout the furrow length, while accounting for variable furrow water head due to infiltration along the furrow (Chapter 7). This figure illustrates the impact of wetting time to the bed middle in irrigation management. The results show adverse effects on *Ea* and *DU* of increasing Q for NT and SC during the first wheat season (Figure 8.9a). In contrast, *Ea* and *DU* of BP were reduced when Q was reduced. During the 2011 corn season (Figure 8.9b) the BP treatment generally had a better irrigation performance than the other two treatments. However, the optimised irrigation performance was very low (e.g. *Ea* = 25-40% and *DU* = 30-80%) in all treatments during the 2011 corn season due to the long lateral wetting time. These figures highlight the interactions between renovation methods (i.e. lateral infiltration) on the irrigation performance for a Vertisol.



Figure 8.9: Impact of renovation methods and Q on optimal *Tco* and inflow volume (solid lines) and *Ea* and *DU* (dotted line) on (a) one season old PRBs (2010 wheat) (b) three season old PRBs (2011 corn) at site A1, Australia. Optimisation requires wetting of bed middle and accounts for furrow water head effect on lateral infiltration

8.3.3.3 *Optimising furrow length*

Furrow length greatly affects *Tco*, irrigation applications and efficiencies (Figure 8.10). For instance, *Ea* and *DU* was low in the 200 to 400 m furrow length range on all treatments, where wetting of the bed middle, especially in the head section of the field, was achieved only with large deep drainage losses. However, the optimal field length was a function of the flow rate applied. For instance, maximum *Ea* and *DU* was achieved at ~400, ~300 and ~200 m furrow length for NT, SC and BP, respectively at $Q = 2 \text{ L s}^{-1}$ (Figure 8.10). However, at $Q = 5 \text{ L s}^{-1}$, the maximum *Ea* and *DU* was identified at 1000, 800 and 600 m furrow lengths for NT, SC and BP, respectively.



Figure 8.10: Example field design and irrigation management tool for showing the effect of furrow length and Q on optimum Tco & Inflow volume (solid lines) and Ea & DU (dotted lines). Optimisation data for a single irrigation to 2010 wheat requiring lateral wetting of bed middle on a Vertisol at site A1, Australia

Improved irrigation performance during the hemp crop season could be achieved with increased furrow length, reduced Q and increased Tco (data not given). For instance, increasing furrow length to 600 m would increase Ea up to 4% in all treatments, Er would be increased by up to 23%, 13%, and 14%, while DU increased by 4%, 4% and 10% on NT, SC and BP, respectively.

The 2011 crop was grown on the bed shoulders. Thus, for the 2011 corn season simulations, the furrow length optimisation was conducted by requiring $Er \ge 85\%$, but without requiring field wetting to the bed middle. In this season, irrigation performance improved with lower Q applied to short furrows and higher Q applied to long furrows (Figure 8.11). Similarly, improved irrigation performance (*Ea* and *DU*) was achieved with higher Q on SC and BP treatments, while reduced Q in longer furrows adversely affected their irrigation performance. In contrast, irrigation performance was not affected at low Q (i.e. $Q = 2 \text{ L s}^{-1}$) on NT for furrow lengths.



Figure 8.11: Example field design and irrigation management tool for showing the effect of furrow length and Q on optimum *Tco* & Inflow volume (solid lines) and *Ea* & *DU* (dotted lines). Optimisation data for two irrigations on a 2011 corn crop requiring $Er \ge 85\%$ and water reaching to furrow tail ends at site A1, Australia

8.3.3.4 *Optimising time to cut-off relative to water advance time*

The *Tco* Ta^{-1} (Figure 8.12) showed higher values (4 to 5) on short furrows and smaller values (~1) on longer furrows. The ratio tended to be higher with NT compared with SC and BP. For instance *Tco* Ta^{-1} were ~4.5, ~4 and ~3.5 for NT, SC and BP treatments at $Q = 2 \text{ L s}^{-1}$ and L = 100 m, respectively. However, the ratio was unity at ~500 m on BP compared with ~600 m for SC and ~700 m for NT.



Figure 8.12: Effect of furrow length on optimum *Tco* Ta^{-1} (solid line) and *Ea* (dotted line) for freshly renovated PRBs (2010 wheat season) for different Q and requiring wetting of bed middle at site A1, Australia

8.4 Discussion

8.4.1 Irrigation management practices of PRB systems in Australia

The current irrigation management of Australian farms was not optimal, evidenced from current excessive deep drainage and tail drain runoff losses as discussed in Chapter 4. Therefore, the four irrigation management strategies were successful in improving the existing irrigation performance. The results of all four optimisation strategies have shown significant improvement in current irrigation performance, which conformed with the views of Playán and Mateos (2006) and also consistent with this study hypotheses.

The Q and Tco were identified as the key irrigation management parameters which significantly influenced the irrigation performance on both Australian farms. This study showed that up to 25% irrigation water savings could be achieved by adjusting Tco and that up to 36% irrigation water saving could be achieved by adjusting both Q and Tco. This agrees closely with the findings of Smith et al. (2005) despite the fact that the requirement to fulfil lateral infiltration to the bed middle reduced the irrigation water savings. The optimisation of Q and Tco together was more beneficial than Tco optimisation alone as demonstrated by reducing the applied water volume by up to 28% on site A1 and 3% on site A2 and by reducing deep drainage losses up to 85%. These improvements were achieved with an acceptable small trade-off in Er and DU.

Consistent with the findings of Sanchez et al. (2009) furrow length was found to greatly affect irrigation performance on both sites. The current study findings at both sites revealed that appropriate furrow length is one of the most important considerations and will determine the maximum achievable irrigation performance.

The relationship between Tco and Ta is often used as a convenient field tool for irrigation management. However, the current study has shown that $Tco Ta^{-1}$ was greatly affected by Q, furrow length, blocked end condition and field infiltration. For instance, the optimal ratio increased up to 5 times for short furrows (200 m) on site A1 (blocked ended furrows) and up to 7 times on site A2 (open ended furrows) with

larger Q (5 L s⁻¹). In longer furrows (L > 500 m) the ratio was found to converge to unity irrespective of Q at both sites. Thus, switching off irrigation at the time of water arrival to the furrow tail end largely resulted in under-irrigation on short furrows, while for long furrows it resulted in either over or under-irrigation depending on Q and soil infiltration properties. However, the relationships developed in graphical form may be helpful as decision support to identify optimised irrigation management.

8.4.2 Irrigation management practices of PRB systems in Pakistan

Irrigated raised beds are considered a relatively efficient irrigation system in Pakistan (Kahlown et al. 1998). These systems have the potential to save up to 60% irrigation water losses (Gill et al. 2005) compared with existing flat basin or bay irrigation systems if managed properly. However, this study has revealed that the irrigation management of existing irrigated raised bed farming systems in north west Pakistan is sub-optimal with low irrigation performance. The evaluations demonstrated the 50% *Ea* and up to 61 mm/irrigation deep drainage losses, which agree with results reported by World Bank (1997). Thus, there is potential to optimise irrigation management of PRB farming systems in Pakistan.

The small Pakistani fields have short Tco and the farmers are generally keen to apply the limited available canal water by personally monitoring the irrigation events. The current experiments were monitored carefully under farmer supervision, apparently leaving little scope for further optimisation. However, the SIRMOD evaluations indicated that the existing irrigation practices are not optimal due to a lack of decision support guidelines. In particular, further improvements are possible by selecting suitable field lengths, appropriate Q and irrigation switching off time.

Half of the measured irrigation events were over-irrigated with excessive tail end over topping and half were under-irrigated. The furrows are characterised by a quick recession time at the head section of the field leading to poor lateral infiltration (Akbar et al. 2007). All the four irrigation management strategies were shown to produce irrigation water saving in over-irrigated events and to adequately replenish *SMD* in under-irrigated events. Optimisation of Q and *Tco* together was effective at meeting the *SMD* and in improving DU along the furrow length, especially for WB. Hence, this strategy may be particularly beneficial for improving DU without dividing the field into smaller segments, a common practice in the locality, and thus may save the additional labour costs and time associated with shorter furrows. Furrow length optimisation showed a further improved irrigation performance.

Seasonal changes and field conditions played a key role in the irrigation performance (Gates & Clyma 1984) and effectiveness of irrigation management strategies. Similarly, irrigation performance was shown to be sensitive to management on short furrow lengths (Eldeiry et al. 2005). This may have made the current irrigation management optimisation highly sensitive to a range of contributing factors. The current results indicate that irrigation performance improvement strategies are highly site and event specific, subject to variations caused by infiltration capacity of the soil, weeds, cultural practices, and bed settlement during a cropping season. For instance, site P1 showed improved irrigation performance (irrigation water saving, Ea and DU) with increased Q due to higher infiltration capacity, which agree with the findings of Gillies et al. (2010). However, site P2 and P3 showed improved irrigation performance with reduced Q due to a lower surface roughness and low infiltration. Thus, the irrigation management optimisation strategies evaluated were able to account for seasonal and spatial (site to site) variations. This suggests that these processes used in this study may be used to develop guidelines for other field conditions.

According to Sanchez et al. (2009) optimum furrow length (strategy 3) is an important factor for achieving improved irrigation performance. This study showed that optimising furrow length improved irrigation performance with reduced furrow length, increased Q and reduced Tco at all sites. The relationships identified and plotted graphically provided guidelines for the optimum irrigation management and field design under the existing field conditions. Maximised irrigation water saving and irrigation performance with optimised Q, Tco and L strategies were identified as very beneficial, which can be very important for the water deficient Indus Basin in improving the existing poor WP.

Relating *Tco* to *Ta* for improved performance required better understanding of the irrigation and field conditions. For instance, optimisation required increased *Tco* up to two fold of *Ta*, on low infiltration soils, when furrow lengths were shorter (50 m) and Q were larger (3 L s⁻¹). In contrast, irrigation performance improved for high infiltration soil when *Tco* was equal to *Ta* at longer furrows and low Q. Similarly, NB required switching off irrigation before the water reached the furrow tail end. In contrast switching off irrigation at the time of water arrival to the furrow tail end was identified as not adequate to replenish *SMD* on WB.

8.4.3 Irrigation management practices in response to renovation methods

Irrigation applications of three PRB renovation treatments during three cropping seasons (Chapter 6) were strictly monitored during the experiments. The Tco was based on fulfilling *SMD* by applying the required volume of irrigation water, without overtopping the beds at the tail ends of furrows. The Q per furrow selection was restricted by the available Q from the source, water head in distribution channel, farmers' commitments for fulfilment of water demand in nearby vegetable farms, lack of decision support tools and high cost involved with prolonged irrigation times. Thus, optimum irrigation management was difficult to implement under existing field conditions. Therefore, impacts of four irrigation management issues experienced during field trials that negatively affected irrigation performance.

During the 2010 wheat season, optimised Tco alone (strategy 1) significantly reduced Ea with significant increases in inflow volume and deep drainage losses to ensure wetting of the bed middle. In contrast, optimised Q and Tco together significantly improved Ea, DU, irrigation water saving and reduced deep drainage losses while wetting of the bed middle was confirmed. Moreover, strategy 2 was more effective improving the existing poor irrigation performance of freshly applied BP treatment with increased Q and reduced Tco compared with the other two treatments.

During the 2011 hemp season, the early two irrigations were applied under low *SMD* to avoid poor crop germination in the bed middle, when the top soil was dry and the deep soil below 10 cm depth was moist. Due to quicker water advance in relatively

wet and compacted furrows the irrigation was switched off earlier to avoid excessive tail end overtopping, as the furrows were blocked at the tail end. Thus, the wetting front did not penetrate to the bed middle. Evaluations suggested no further increase in Tco at existing Q due to fast water advance. Therefore, overall results of Tco optimisation did not produce any benefit during the 2011 hemp season. However, strategy 2 improved Er and increased lateral infiltration with reduced Q and increased Tco.

During the 2011 corn season, the three seasons' old PRB had compacted furrows, which increased water advance rate in furrows but renovations were freshly applied and cracks were developed due to dry antecedent soil moisture, especially before the second irrigation, which improved soil infiltration properties in the bed area. Therefore, *Tco* optimisation did not produce any benefits in improving irrigation performance, except inflow volume was significantly reduced for BP treatment. However, strategy 2 significantly improved irrigation performance on SC and BP with up to 17% irrigation water saving and up to ~48% reduced deep drainage losses.

Furrow length optimisation suggested further improved performance during all three cropping seasons. However, optimum furrow lengths varied for the different seasons and treatments. For instance, the results suggested decreased furrow length during the 2010 wheat season when the beds and furrows were relatively loose and porous due to fresh renovation and recent realignment. Similarly, the results indicated the largest improvement on BP treatment with reduced furrow length during the 2010 wheat crop season. However, due to seasonal bed settlement and furrow compaction with machinery traffic, sealing, and sedimentation, the furrow length optimisation suggested increased furrow length in the subsequent 2011 hemp and 2011 corn seasons for achieving improved irrigation performance.

The current study revealed significant variation in water advance rate in furrows among irrigations and seasons caused by variable furrow compaction, bed settlement, antecedent soil moisture, furrow length, Q and miscellaneous field conditions. Thus, deciding *Tco* on the basis of *Ta* may be affected by *L*, *Q* and PRB renovation methods. The graphical illustrations may supported decision making.

8.5 Conclusions

This study showed that optimised irrigation management and field design can significantly improve the irrigation performance of existing farmer managed irrigated PRB systems and that comparable irrigation performance among the three freshly applied renovation methods is possible. The main conclusions are:

- Irrigation performance of existing PRB farming systems in Australia and Pakistan were poor leading to over or under-irrigation. However, significant improvements in irrigation performance are achievable through irrigation management.
- Optimising Q and Tco together was more beneficial than optimisation of Tco alone. Optimisation of Q and Tco saved up to 38% water and reduced deep drainage losses by up to 96%.
- Shorter furrow lengths generally had high irrigation performance on high infiltration soils while longer furrows were beneficial on compacted furrows. However, the irrigation performance of short furrows was sensitive to Q and Tco, thus requiring more refined and intensive irrigation management to achieve improved performance.
- Relating *Tco* to *Ta* required an understanding of field infiltration characteristics, *Q* and furrow length. Using a *Tco Ta*⁻¹ ratio of 1 caused underirrigation, especially in short furrows and wide beds. The graphical illustration may be helpful in improving understanding for the identification of improved irrigation management practices.
- Renovation methods involving tillage affected irrigation performance due to changed soil infiltration properties. Renovated PRB, especially BP, require monitoring and evaluation to identify the appropriate irrigation management practices to improve irrigation performance.
- The optimised irrigation management practices (e.g. *Q* and Tco) can improve irrigation performance, which can normally be implemented without incurring significant costs to infrastructure, machinery, or labour.

CHAPTER 9: General discussion and recommendations

9.1 Introduction

The permanent raised bed (PRB) farming system has evolved over multiple attempts to implement various aspects of CA. Control traffic farming (CTF) has been shown to successfully improve soil hydro-physical properties (Tullberg et al. 2007; McHugh et al. 2009). The transformed farming system in the form of PRB has been adopted in some developed countries (e.g. Australia and USA) but is still in its infancy in many developing countries (e.g. China, India and Pakistan). Lack of adequate guidelines and decision support aids for agronomic and irrigation management, poor lateral infiltration in wide spaced furrows and resulting poor *WP* are some of the main impediments to the successful implementation of PRBs in many parts of the world.

The majority of previous studies evaluating PRB have focussed on comparison with traditional flat basin, border irrigation, or random traffic systems. Measurements are often taken using point evaluation methods over small areas, which are subject to large variations. Similarly, the majority of previous work evaluating the impact of PRB farming systems on soil hydro-physical properties has been conducted under rain-fed conditions. Therefore, the effects of different agronomic and irrigation management factors on the *WP* of irrigated PRBs are largely unknown.

This research evaluated the potential to improve the *WP* of irrigated PRBs by focussing on four key areas:

- Benchmarking the irrigation performance of PRB farming systems;
- Evaluating the effect of PRB renovation methods on irrigation performance and productivity;
- Evaluating the effect of PRB renovation methods on lateral infiltration into the beds; and
- Identifying strategies to improve lateral infiltration and irrigation performance of PRBs with optimised irrigation management and field design.

9.2 Major outcomes and key findings

9.2.1 Benchmarking irrigation performance of PRB farming systems

This study showed that the interaction of agronomic, soil and irrigation variables make it difficult for farmers to identify optimal irrigation management practices. The uninformed selection of Q and Tco resulted in application losses (*Ea* down to 68% in Australia and 50% in Pakistan) and low *DU*, which conforms with the findings of Schwankl et al. (2000) and Mailhol et al. (2005). Soil infiltration, Q, *SMD* prior to irrigation, bed furrow configurations and furrow compaction affected the water advance rate along the furrow as mentioned by Raine and Bakker (1996) and Bakker et al. (1997). Thus, farmer decision making was affected by multiple field factors and irrigation performance was negatively affected.

Different renovation and tillage techniques were identified as being used in commercial practice on both soils. Practices ranged from no tillage to either annual or occasional deep ripping with the aim of loosening the compacted soil and to control pests and diseases in Australia. Differences in PRB renovation methods in Australia were also reported by Bell et al. (2003), Hulugalle and Daniells (2005), and Garside (2005). In Pakistan, seasonal shallow cultivation or rotary hoeing is the common practice (Akbar et al. 2007). However, the majority of these renovation methods cause excessive soil disturbance, which reduces soil infiltration, soil resistance to wetting and increase soil erosion. Degraded soil structure and accelerated erosion, which was attributed to current intensive cultivation methods, was observed on both soils and is consistent with reports in past studies (Smith et al. 1983; Dexter 1991; Cook et al. 1992; Ishaq et al. 2001; Akbar et al. 2007).

The water advance in the furrows showed a linear response with the square root of time in both soils. This indicates that capillary forces were dominant in the soil infiltration (Philip 1969). Hence it can be inferred that the soil properties of pore size distribution, clay type, and soil structural stability to wetting are influencing irrigation performance. Certainly the practices that deliver the water to the soil will also impact on irrigation performance. So, any particular irrigation practices of properties of properties of properties of properties of properties of performance.

rate and duration of application and length of furrow over which free water must pass to wet the entire length of a field. Both Australian and Pakistani irrigation practices were shown to be sub-optimal and would probably respond favourably to both improved soil management and irrigation practices.

9.2.2 Impact of renovations on irrigation performance and productivity

This study highlighted the large impact of machinery design, operation control and renovation methods on PRB performance. Generally, PRBs renovated with excessive tillage required extra machinery operating time, created fatigue during sowing, had lower crop establishment rates, and soil disturbance. According to Canakci et al. (2005) and Hussain et al. (2010) land preparation and tillage consume large amounts of total energy, which can also affect farm income. All these factors favoured the NT treatment than SC and BP on Vertisol at this site under normal field conditions.

The tillage/renovation methods were shown to significantly impact on soil hydrophysical properties which is in close agreement with other studies (Lipiec et al. 2006; Verhulst et al. 2011). The significantly reduced (~6%) seasonal bulk density and increased cumulative infiltration (23%) of BP and the smaller seasonal increase (~5%) for NT and BP indicate greater soil structural stability compared with SC, which conform to the findings of Hussain et al. (1998), Jin et al. (2007) and Wang et al. (2012). However, the 3 fold reduction in lateral infiltration during the three cropping seasons for the NT indicate bed renovation may be helpful to avoid negative impacts of poor lateral infiltration. Thus different renovation techniques can alter soil hydro-physical properties and can improve *WP* of PRB farming system.

The different renovation methods altered profile water storage evidenced by increased (up to 4%) seasonal profile water content on BP compared with NT and SC. This was also demonstrated by the consistently dry bed middles on NT and SC treatments due to poor lateral infiltration, which effectively reduced the volume of soil storing water in the active root zone. Dry bed middles under NT and SC treatments were also reported by Jin et al. (2007) on a sandy clay loam and Lucy (1993) on a Vertisol. The lack of inversion of the bed surface by BP improved the soil structural stability during wetting and reduced evaporation losses. According to
(Hamilton et al. 2005) the undisturbed crop roots under BP work as reinforcing rods, which enhance soil strength and aggregation. The enhanced water storage in BP can be beneficial in reducing irrigation frequency and improving capture and use of inseason rainfall (Thomas et al. 1990; Eldridge & Robson 1997).

The furrow sizes and shapes have been shown to affect irrigation performance due to changes in wetted perimeter (Raine et al. 1998). The large furrow sizes (up to 17%) of BP than NT were due to greater porosity and deeper furrow excavation. However, the greater seasonal reduction (up to 47%) in furrow sizes for the cultivated treatments indicates the improvements in soil hydro-physical properties may be temporary due to adverse seasonal changes. This reflected increased soil resistance to deformation by wetting under NT compared with BP and SC treatment. Therefore, mechanical loosening need to be applied judiciously keeping in view their negative impacts on soil structure stability and adverse seasonal changes.

Impact of renovation methods on crop yield and WP was not significant. However, the significantly higher 2010 wheat yield and WP on NT during the first season was attributed to poor seeder performance on renovated beds evidenced the significantly lower tillers density for BP than NT and SC. The double disc planter performance on renovated beds was poor because it was designed and set up to operate normally in firm zero till soils as has been the case on the farm for 40 years. During the 2011 hemp season, the excessive pre-sowing rainfall caused bed subsidence, sealing and resulted in poor lateral infiltration as the beds were not freshly renovated. Thus, the 2011 hemp yield was reduced (~55%) in the middle of the beds compared with the bed shoulder. Decreased yield due to poor lateral infiltration was also reported by Lucy (1993), Jin et al. (2007) and Akbar et al. (2007). The comparable crop yield and WP of fresh BP compared with NT during the 2011 corn season was indicative of a good rooting environment, larger soil water storage and reduced soil disturbance resulting from newly designed equipment, with enhanced performance (Hamilton et al. 2005). As with structural stability, the similar yield and WP of mechanically loosened PRBs (SC and BP) and NT suggests that NT and BP may be preferable on a Vertisol but the benefits may take longer than three seasons to significantly emerge.

This study confirmed that BP reduced bulk density, improved lateral infiltration, improved profile water storage, and gradually increased *WP* on a Vertisol. One of the reasons for the gradual improvement on BP compared with other treatments was the fact that the BP treatment was not most appropriately imposed until the third season. Additionally, BP was shown to be sensitive to the selection of equipment, soil moisture (may be destructive to soil structure above or below plastic limit and can promote smearing) and operating expertise for its successful implementation. It can be concluded that due to the fragile and sensitive nature of Vertisol to soil disturbance (Coughlan 1981; Smith et al. 1983; Dexter 1991; Cook et al. 1992), the NT may be the preferred option at this site. However, keeping in view the aforementioned benefits in soil hydro-physical properties, the BP treatment can be more beneficial compared with SC. However, long term impacts of BP on soil structure needs to be further investigated.

9.2.3 Effect of renovation on lateral infiltration

Poor lateral infiltration is generally considered as one of the major limiting factors for achieving high *WP* on wide PRBs (Lucy 1993; Hassan et al. 2005; Jin et al. 2007). However, only few studies have investigated lateral infiltration in wide PRBs. This study revealed a significantly different lateral infiltration capacity among the three PRB renovation treatments on both soils. Other factors including *SMD* and soil type were also identified to have major contribution in lateral infiltration.

The study revealed increased lateral infiltration on BP, increased vertical infiltration on SC and the least cumulative infiltration (both lateral and vertical) on NT treatments. The freshly applied BP increased lateral infiltration and soil moisture storage capacity. Increased infiltration and soil moisture storage for BP were also reported by Thomas et al. (1990), Eldridge and Eldridge and Robson (1997), Hamilton et al. (2005) and Jin et al. (2007). Its impact was far greater on the Vertisol with its inherently greater level of aggregation and stability of structure than on the highly unstable soil structure on the sandy clay loam (Alfisol) soil, which produced the same patterns of responses but with a much smaller magnitude and significance. Thus, the BP did not wet the middle of 132 cm wide beds within (~7 hrs) irrigation time. The significantly different soil moisture content identified between the shoulder and middle of the bed, on both soils, confirmed the non-uniform nature of soil moisture distribution on wide PRB farming systems (Tabuada et al. 1995) and raised concerns regarding the adequacy of field design and irrigation management practices that assume uniform soil moisture distribution across the beds. The shrink swell nature of Vertisol and the long irrigation times under Australian conditions, make the 200 cm wide PRBs less vulnerable to poor lateral infiltration compared with the 132 cm wide PRB on rigid and degraded sandy clay loam soil.

9.2.4 Strategies to improve lateral infiltration and irrigation performance

When the Hydrus 2D (Šimunek et al. 1999) model was appropriately parameterised by field measured data for both soil types, the lateral infiltration process of these soil types with the same treatments as imposed in the field experiments in Australia showed just how this BP treatment and the soil structure it produces is capable of very large improvements in the rate and distribution of water absorption two dimensionally, but dominantly laterally to the bed centre. The contrasting extent of these improvements between the two soils indicates the potential improvement in irrigation performance and productivity for Pakistan sandy clay loam would be likely to be proportionately greater once the soil structure had built up and become stable. The enhancement of lateral infiltration rate indicated that soils so treated could be watered faster (with larger Q), completely to the middle of the beds with less water and less deep drainage loss.

This study showed differences in lateral infiltration due to renovation methods and water head in furrows. The maximum wetting time of bed middle to reach field capacity (~17, ~11 and ~6 hours on NT, SC and BP, respectively) at 9 cm furrow water head seems manageable on the Vertisol under Australian field conditions. The increased lateral infiltration with increased furrow water head was consistent with the findings of Tabuada et al. (1995). Thus both strategies (renovation method and furrow water head) can affect lateral infiltration and may be helpful as management tools on a Vertisol.

Lateral infiltration was very poor in sandy clay loam evidenced by the 37, 24 and 15 hours predicted minimum wetting time of the bed middle on NT, SC and BP respectively, at full furrow water head, which seems difficult to manage under Pakistani field conditions. Poor lateral infiltration was also reported on sandy clay loam in many past studies (Hassan et al. 2005; Akbar et al. 2007; Jin et al. 2007). The minimum time (~15 hrs) required to wet the bed middle, identified with BP at full furrow water head, is likely to be difficult to achieve using the existing bed width (132 cm) and degraded soil conditions. However, this research has evaluated decision support tools, which may be helpful in selecting the optimum bed width, inflow rate and infiltration opportunity time for different bed renovation methods.

The SIRMOD III model, appropriately parameterised with the field data from Australia and Pakistan, showed how each of the soil type and management induced soil conditions would affect irrigation performance. The four optimisation strategies were: (1) *Tco* optimisation, (2) *Q* and *Tco* optimisation, (3) furrow length optimisation and (4) Relating optimum *Tco* to *Ta*. Evaluation of the strategies for PRBs on a Vertisol (Australia) highlighted the potential for significant improvement in irrigation water saving (36%), efficiency, and uniformity. Optimising furrow length further improved irrigation performance. Increasing field length generally reduced irrigation performance. These findings agreed with findings of Raine and Bakker (1996), Smith et al. (2005) and Gillies et al. (2010).

The sandy clay loam (Pakistan) with short furrows (< 100 m) generally showed improved irrigation performance with each of the four irrigation management and field design strategies, which conformed to the findings of Kalwij (1997). The Q and *Tco* optimisation produced greater (~38%) irrigation water saving and up to 93% reduction in deep drainage potential. Poor DU is a significant problem and the current local method of improving DU by putting bunds in the furrow is labour intensive. Thus improving DU by optimised Q and *Tco* can save labour. *Tco* Ta^{-1} was sensitive to Q, soil infiltration properties and furrow configurations (Kalwij 1997). The analysis suggested the (*Tco* Ta^{-1}) ratio should be above unity for WB and below unity for NB and high infiltration soils.

Renovation of a long furrow Vertisol reduced irrigation performance unless Q was increased to match the increased soil infiltration. In all cases the BP treatment showed the greatest improvements (up to 33% *Ea* and up to 35% irrigation water saving) but also, because of its responsiveness was the most sensitive to changed conditions. Importantly, the BP treatment also showed itself to be more flexible in its furrow length, its capacity to absorb higher Q rates and maintain $Tco Ta^{-1} = 1$ over a longer length than the soils treated by SC or NT. Interestingly, the compacted furrows required reduced Q and increased Tco for improved performance. Hence, SC and BP treatments benefitted more from irrigation management optimisation than NT as the farmer managed irrigation matched the infiltration of the NT treatment.

Irrigation practice, especially on BP treated soil, would require awareness of the antecedent moisture content in particular. But offsetting this is the fact that the increased lateral infiltration behaviour allows the simply observed and applied management practice of equating Tco with time when water advance in the furrow reaches the tail end. Because of the rapidity of lateral infiltration this management guide is needed, otherwise too much water infiltrates and deep drainage increase.

The four strategies evaluated have practical significance and have the potential to reduce deep drainage losses, enhance infiltration opportunity time, and improve uniformity. Importantly, these strategies offer an opportunity of PRB performance improvement which can be implemented without incurring significant increases in infrastructure, machinery, or labour costs.

9.3 Summary

The water productivity of PRB farming systems may be constrained by sub-optimal irrigation management, excessively wide beds, and renovation methods involving intensive tillage. Consequently, excessive water losses, soil structural degradation, and poor lateral infiltration are frequently observed. This study has shown that improved agronomic management, specifically renovation of PRB consistent with the principles of conservation agriculture, combined with optimised irrigation management, has great potential to improve irrigation and crop performance that can lead to improved *WP* of existing PRB systems.

Renovating PRBs involving inversion can reduce soil structural stability and can make the bed vulnerable to slumping. If renovation is applied without modifying Q and *Tco* then irrigation performance will also be adversely affected. Hence, NT would be the preferred choice for managing beds on a Vertisol under normal field conditions. However, if poor lateral infiltration is observed and mechanical loosening is required then BP renovation is more beneficial compared with SC. This is because BP improves lateral infiltration, enhances profile water content, and maintains soil structural stability. However, BP requires better machinery operating control, specialised equipment, and the correct soil moisture during renovation to avoid soil smearing and clodiness. Similarly, BP requires improved irrigation management, optimised Q and *Tco* for the changed soil infiltration properties.

Optimised irrigation management and appropriate bed and field design has the potential to improve the *WP* of PRB farming systems. The procedures developed for understanding the soil, water and agronomic interactions may be beneficial in making the PRB farming system a more attractive option in both countries. This will be especially important in developing countries, where the adoption of PRB systems is still in its infancy and water and food security are pressing issues.

All these outcomes confirmed that the hypotheses and objectives of this study have been successfully satisfied to a greater or lesser extent. However, further extensive work is needed to assess the level of probable *WP* improvement.

9.4 Practical implications of this research

This research involved evaluation of irrigation performance, PRB renovation methods/cultivation methods, soil hydro-physical properties, lateral infiltration, and crop performance for existing and variably renovated PRB farming systems. Evaluations were conducted in the two contrasting environmental conditions of Australia and Pakistan. This research has elucidated the plausible consequences of prevailing irrigation management practices, which generally assume uniform soil moisture distribution across the beds. The following are the implications arising from the research:

- PRB have not been widely adopted yet in developing countries and is still in transition in many developed countries. This research identified inappropriate agronomic and irrigation management as the major areas, which can enhance the current *WP* and thus may assist in stimulating the adoption of PRB farming systems worldwide.
- The *WP* of PRBs under commercial practices has been shown to be generally poor and that improvement is possible at no significant cost. Thus well managed PRBs can be helpful for sustainable crop production to feed the growing population and to conserve the natural resource for the future.
- This work has identified temporal and spatial variations in soil hydro-physical properties of PRB, which may be helpful in improving understanding and refining decision making for improved irrigation and soil management of PRB under a wide range of environmental conditions.
- The soil management involve intensive tillage in both Australia and Pakistan, which have deleterious consequences on soil hydro-physical characteristics. This research showed that loosening of compacted soil is possible with controlled mechanical loosening without excessive soil disturbance, which may lead to increased soil porosity, lateral infiltration and profile water storage. Thus the general practice of diseases, pests, weeds control and soil loosening with excessive tillage in both countries can be replaced with improved soil management practices.
- Poor lateral infiltration is perceived to be a barrier in PRB adoption in developing countries. The outcomes of this study are of particular benefit in supporting decision making about the selection of the renovation methods, irrigation management, and bed furrow sizes for improved lateral infiltration.
- The measurement and modelling procedures demonstrated for the development of management guidelines are robust and have broad implications.
- This research compared PRB performance under contrasting environmental and socio economic conditions. This has made the scope of these outcomes comparable, representative, beneficial, and applicable for PRB systems under wide ranging farming conditions.

9.5 Recommendations for further research

The outcomes of this study have confirmed that the water productivity of PRB farming systems can be increased if managed properly. The irrigation performance of PRBs may be improved using appropriate irrigation management, field design optimisation, and renovation methods matched to specific soil and field conditions. However, opportunities for further research include:

- Development of a simplified routine irrigation performance evaluation and management optimisation protocol that can account for lateral infiltration into the bed with different furrow spacing.
- Field evaluation of the optimisation techniques on different PRB farming systems and raised beds across a broad range of environmental conditions (e.g. throughout the Indus Basin).
- Testing the proposed strategies of improving lateral infiltration under variable soil salinity levels of PRB systems on both soils.
- Evaluating the renovation methods on a wider range of soils (including soil affected by salinity and sodicity).
- Investigating the longer term impacts and optimal frequency of applying the different PRB renovation techniques, especially the blade ploughing method, under different environmental conditions.
- Evaluating the energy requirement and practicalities of the different PRB renovation techniques under a wider range of field, environmental and socioeconomic conditions (including comparison with traditional cultivation).
- Evaluating the long term impact of renovation methods on the *WP* of both soils, but particularly the sandy clay loam in Pakistan. This will need the consistent application of conservation agriculture and blade ploughing practices for 4 to 5 years before the improvement in soil structure is sufficient to demonstrate the limit to potential improvements achievable to both *WP* and irrigation performance

CHAPTER 10: Conclusions

The literature review identified two key areas of research (i.e. agronomic and irrigation management) for improving the *WP* of PRB farming systems. The benchmarking study successfully quantified the performance of existing PRB farming systems on a Vertisol (Australia) and a sandy clay loam (Pakistan) under usual farmer management. The results of this benchmarking study guided the research to focus on bed renovation, the effect on lateral infiltration, and the potential to improve irrigation management.

This study assessed three PRB renovation methods (no tillage, shallow cultivation, and blade ploughing) for changes in: soil hydro-physical properties (bulk density, lateral infiltration, and profile water content), bed furrow configurations, crop yield, and *WP* over three cropping seasons on the Vertisol. The effects of (1) renovation method and (2) furrow water head on lateral infiltration into both soils were evaluated using the Hydrus 2D model. Four strategies for improving irrigation performance including optimising (1) *Tco* alone (2) *Q* and *Tco* together (3) furrow length and (4) optimum *Tco* relative to *Ta*, were assessed using SIRMOD. The four strategies were also used to explore the potential irrigation performance benefits for the three PRB renovation methods. The outcomes of these studies have shown that:

1. The irrigation and soil management of existing PRB farming systems in Australia and Pakistan is sub-optimal leading to poor irrigation performance. This was demonstrated by:

- The uninformed selection of sub-optimal inflow rate and time to cut-off.
- Application efficiencies as low as 68% in Australia and 50% in Pakistan.
- Excessive irrigation application in both countries with deep drainage potential up to 61 mm/irrigation, while wide beds in Pakistan were under-irrigated.
- Renovation methods involving intensive tillage are destructive to soil structural stability and reduced irrigation performance if the irrigation management was not altered to match the changed infiltration characteristics.

2. PRB renovation methods (e.g. no tillage with furrow cleaning, shallow cultivation and blade ploughing) significantly altered soil hydro-physical properties, irrigation performance, crop yield and thereby has the potential to improve the water productivity on a Vertisol. The three seasons results demonstrated that:

- No tillage enhanced *Ea* and productivity, thus for this site, soil and operating conditions are the preferred option for sustainable production.
- Shallow cultivation was destructive to soil structure, accelerated furrow filling, and enhanced deep drainage losses, thus is not preferable.
- Seasonal bed subsidence reduced irrigation performance and lateral infiltration, thereby reduced crop performance in the bed middle (e.g. ~55% in 2011 hemp crop with no tillage) when compared with bed shoulder.
- Blade ploughing reduced soil bulk density (~6%), and enhanced profile water storage, lateral infiltration and productivity, compared with shallow cultivation but was sensitive to *Q* & *Tco* for high irrigation performance.

3. Lateral infiltration was significantly affected by renovation methods and furrow water head. Thus, these may be used to avoid yield loss due to dry bed middle caused by low infiltration soils or bed subsidence. The outcomes demonstrated that:

- Blade ploughing enhanced lateral infiltration compared with shallow cultivation and no tillage treatment on both sandy clay loam and Vertisol.
- Increased furrow water head increased lateral infiltration on both soils.
- Bed middle of 200 cm wide PRB on Vertisol under normal Australian field condition is less susceptible to poor crop performance due to fast lateral infiltration, cracking and prolonged irrigation times on long furrows.
- The 132 cm wide PRB on sandy clay loam required a minimum of 15 hours to wet the bed middle under blade ploughing treatment at full furrow head. Where the beds are renovated with blade ploughing the bed width would need to be reduced to 105 cm to permit full wetting in the 7 hrs evaluated with normal irrigation. However, if no tillage or shallow cultivation were used then the beds would need to be < 90 cm because of the reduced lateral infiltration.</p>

4. Irrigation management and field design strategies improved irrigation performance of PRB systems on both soils. These strategies may be used to improve water productivity. This was evidenced by:

- Optimising *Q* and *Tco* together saved up to 38% of water, which was up to 23% more than optimising *Tco* alone for the farmer managed PRB systems.
- Furrow length optimisation further improved irrigation performance.
- Optimising *Tco* relative to *Ta* improved understanding of the complex interaction between variable field conditions.

5. Several examples of decision support tools were illustrated in this research for the specific soils and sites evaluated. While the applicability of these decision support tools is not universal, they do demonstrate the principles of performance variations and their interactions with changed soil and field conditions, which are applicable to a wider range of environmental conditions.

This study demonstrated the effectiveness of improved soil management and optimised irrigation management in improving irrigation performance and *WP* of PRB farming systems. This work has shown no tillage with furrow cleaning as the most efficient and productive PRB renovation method on a Vertisol, which is also consistent with the principles of conservation agriculture. However, seasonal subsidence, compaction, and low infiltration soils can reduce crop performance, in the middle of a no tillage bed. In these conditions, controlled mechanical loosening with blade ploughing can be beneficial for enhancing lateral infiltration, profile water storage and soil stability compared with shallow cultivation. However, the long term impacts on soil properties and optimal frequency of blade ploughing application needs to be investigated. Improving irrigation management and field design has the potential to significantly improve irrigation performance of PRB farming systems. Importantly, these strategies may be implemented with little infrastructure, labour, or machinery costs and offer an opportunity of improved *WP* on a sustainable basis.

References

- ABS 2010, Water use on Australian Farms, 2009-10, Australian Bureau of Statistics.
- Abu-Hamdeh, N 2003, 'Effect of weed control and tillage system on net returns from bean and barley production in Jordan', *Canadian Biosystems Engineering*, vol. 45, pp. 23-8.
- Adem, H & Tisdall, J 1984, 'Management of tillage and crop residues for doublecropping in fragile soils of south-eastern Australia', *Soil and Tillage Research*, vol. 4, no. 6, pp. 577-89.
- Adem, H, Tisdall, J & Willoughby, P 1984, 'Tillage management changes sizedistribution of aggregates and macro-structure of soils used for irrigated rowcrops', *Soil and Tillage Research*, vol. 4, no. 6, pp. 561-76.
- Ahmad, M, M., A, Shabbir, MB, Yasin, MJ & Amin, R 1986, 'Soil Taxonomy and Agrotechnology Transfer, Second volume: Field Excursion', In: Proceedings of XII International Forum, SMS Services (ed.), Soil survey of Pakistan, USA, p. 315.
- Akbar, G, Hamilton, G & Hussain, Z 2009, 'Permanent Raised Bed Cropping System Improves Water Use Efficiencies of Wheat and Maize Crops, Mardan/Pakistan Experience', *Pakistan Journal of Water Resources*, vol. 13, no. 1, pp. 17-24.
- Akbar, G, Hamilton, G & Raine, S 2010, 'Permanent raised beds configurations and renovation methods affect crop performance', In: 19th World Congress of Soil Science, Soil Solutions for a Changing World, Australian Society of Soil Science Brisbane, Australia, pp. 171-4.
- Akbar, G, Hamilton, G, Hussain, Z & Yasin, M 2007, 'Problems and Potentials of permanent raised bed cropping systems in Pakistan', *Pakistan Journal of Water Resources*, vol. 11, no. 1, pp. 11-21.
- Akbar, G, Yasin, M, Ahmad, S, Khan, Z & Ashraf, M 2001, 'Hydro-chemical performance of a dugwell for skimming shallow fresh groundwater', *Sarhad Journal of Agriculture (Pakistan)*, vol. 17, no. 3, pp. 395-400.
- Alazba, A 1999, 'Simulating furrow irrigation with different inflow patterns', *Journal* of irrigation and drainage engineering, vol. 125, no. 1, pp. 12-8.
- Alberts, J & Kalwij, IM 1999, *Disseminating the bed-and-furrow irrigation method* for cotton cultivation in Bahadarwah Minor, IWMI, Paksiatn Lahore.
- Allen, R & Musick, J 1992, 'Furrow traffic and ripping for control of irrigation intake', *Applied Engineering in Agriculture*, vol. 8, no. 2, pp. 243-8.
- Allen, R & Musick, J 1997, 'Furrow irrigation infiltration with multiple traffic and increased axle mass', *Applied Engineering in Agriculture*, vol. 13, no. 1, pp. 49-53.
- Allen, RG, Pereira, LS, Raes, D & Smith, M 1998, 'Crop evapotranspiration-Guidelines for computing crop water requirements-FAO Irrigation and drainage paper 56', *FAO*, *Rome*, vol. 300, p. 6541.

- Alva, A & Fares, A 1998, 'A new technique for continuous monitoring of soil moisture content to improve citrus irrigation', In: *Florida State Horticultural Society*, pp. 113-6.
- Arya, LM & Paris, JF 1981, 'A physicoempirical model to predict the soil moisture characteristic from particle-size distribution and bulk density data', *Soil Science Society of America Journal*, vol. 45, no. 6, pp. 1023-30.
- Azooz, R & Arshad, M 1996, 'Soil infiltration and hydraulic conductivity under long-term no-tillage and conventional tillage systems', *Canadian Journal of Soil Science*, vol. 76, no. 2, pp. 143-52.
- Baker, S 1979, 'Pore size distribution A factor to be considered in infiltration studies?', *Journal of Hydrology*, vol. 41, no. 3-4, pp. 279-90.
- Bakker, D, Raine, S & Robertson, M 1997, 'A preliminary investigation of alternate furrow irrigation for sugarcane production', In: *Australian Society of Sugar Cane Technologist*, pp. 302-9.
- Bakker, D, Hamilton, G, Houlbrooke, D & Spann, C 2005, 'The effect of raised beds on soil structure, waterlogging, and productivity on duplex soils in Western Australia', *Australian Journal of Soil Research*, vol. 43, no. 5, pp. 575-85.
- Bakker, D, Hamilton, G, Hetherington, R & Spann, C 2010, 'Productivity of waterlogged and salt-affected land in a Mediterranean climate using bed-furrow systems', *Field Crops Research*, vol. 117, no. 1, pp. 24-37.
- Bakker, DM, Plunkett, G & Sherrard, J 2006, 'Application efficiencies and furrow infiltration functions of irrigations in sugar cane in the Ord River Irrigation Area of North Western Australia and the scope for improvement', *Agricultural Water Management*, vol. 83, pp. 162-72.
- Barral, JP 1994, 'Development of a water-course model to assess the canal water supply at the farm level', ENGREF, Montpellier, Lahore, Pakistan.
- Baumhardt, R & Lascano, R 1996, 'Rain infiltration as affected by wheat residue amount and distribution in ridged tillage', *Soil Science Society of America Journal*, vol. 60, no. 6, pp. 1908-13.
- Baumhardt, R, Lascano, R & Evett, S 2000, 'Soil material, temperature, and salinity effects on calibration of multisensor capacitance probes', *Soil Science Society of America Journal*, vol. 64, no. 6, pp. 1940-6.
- Bautista, E & Wallender, W 1985, 'Spatial variability of infiltration in furrows', *Transactions of the ASAE*, vol. 28, no. 6, pp. 1851-5.
- Bautista, E & Wallender, W 1993a, 'Optimal management strategies for cutback furrow irrigation', *Journal of irrigation and drainage engineering*, vol. 119, no. 6, pp. 1099-114.
- Bautista, E & Wallender, W 1993b, 'Identification of furrow intake parameters from advance times and rates', *Journal of irrigation and drainage engineering*, vol. 119, no. 2, pp. 295-311.
- Beddow, JM, Pardey, PG & Alston, JM 2009, 'The Shifting Global Patterns of Agricultural Productivity Choices', *The magazine of food, farm and resources issues*, vol. 24, no. 4.

- Beecher, H, Thompson, J, Dunn, B & Mathews, S 2005, 'Successful permanent raised beds in the irrigated farming systems of the Murrumbidgee/Murray Valleys of NSW, Australia', In: Evaluation and performance of permanent raised bed cropping systems in Asia, Australia and Mexico, C Roth, et al. (eds.), ACIAR Proceeding No: 121, Grifith, NSW, Australia, pp. 129-43.
- Beecher, H, Dunn, B, Thompson, J, Humphreys, E, Mathews, S & Timsina, J 2006, 'Effect of raised beds, irrigation and nitrogen management on growth, water use and yield of rice in south-eastern Australia', *Australian Journal of Experimental Agriculture*, vol. 46, no. 10, pp. 1363-72.
- Bell, M, Halpin, N, Garside, A, Moody, P, Stirling, G & Robotham, B 2003, 'Evaluating combinations of fallow management, controlled traffic and tillage options in prototype sugarcane farming systems at Bundaberg', In: Australian Society of Sugar Cane Technologists, p. 26.
- Berkout, N, Yasmeen, F, Maqsood, R & Kalwij, I 1997, Farmers' use of basin, furrow and bed-and-furrow irrigation systems and the possibilities for traditional farmers to adopt the bed-and furrow irrigation method, IWMI, Lahore, Pakistan.
- Bondurant, JA 1957, 'Developing a furrow infiltrometer', *Agricultural Engineering*, vol. 38, pp. 602-4.
- Bormann, H & Klaassen, K 2008, 'Seasonal and land use dependent variability of soil hydraulic and soil hydrological properties of two northern German soils', *Geoderma*, vol. 145, no. 3-4, pp. 295-302.
- Borrell, A & Garside, A 2005, 'Early work on permanent raised beds in tropical and subtropical Australia focusing on the development of a rice-based cropping system', In: *Evaluation and performance of permanent raised bed cropping systems in Asia, Australia and Mexico*, C Roth, et al. (eds.), ACIAR Proceeding No: 121, Grifith, NSW, Australia, pp. 120-8.
- Borrell, A, Fukai, S & Garside, A 1991, *Alternative methods of irrigation for rice production in tropical Australia*, International Rice Research Newsletter.
- Borrell, A, Garside, A, Fukai, S & Reid, D 1999, 'Grain quality of flooded rice is affected by season, nitrogen rate, and plant type', *Australian journal of agricultural research*, vol. 50, pp. 1399-408.
- Bos, MG 1985, *Summary of ICID definitions of irrigation efficiency*, International institute for land reclamation and improvement, Wageningen, Netherland.
- Boulal, H & Gómez-Macpherson, H 2010, 'Dynamics of soil organic carbon in an innovative irrigated permanent bed system on sloping land in southern Spain', *Agriculture, Ecosystems & Environment*, vol. 139, no. 1, pp. 284-92.
- Boulal, H, Gómez-Macpherson, H & Villalobos, FJ 2012, 'Permanent bed planting in irrigated Mediterranean conditions: short-term effects on soil quality, crop yield and water use efficiency', *Field Crops Research*, vol. 130, pp. 120-7.
- Bouwer, H 1986, 'Intake rate: Cylinder infiltrometer ', in A Klute (ed.), *Methods of Soil Analysis, Part 1: Physical and Mineralogical Methods*, American Society of Agronomy and Soil Science Society of America, USA, vol. 9, pp. 825-44.

- Bruinsma, J 2003, World agriculture: towards 2015/2030: an FAO perspective, 1844070077, United Nations Food and Agricultural Organisation (FAO), London, UK.
- Brussaard, L & Van Faassen, H 1994, 'Effects of compaction on soil biota and soil biological processes', in *Soil Compaction in Crop Production*, Elsevier Science, Amstredam, pp. 215–35.
- Burrough, P 1993, 'Soil variability: a late 20th century view', *Soils and fertilizers*, vol. 56, no. 5, pp. 529-62.
- Burt, C, Appling, D & Norton, L 1995, 'The surface irrigation manual: a comprehensive guide to design and operation of surface irrigation systems', Waterman Industries, Exeter.
- Canakci, M, Topakci, M, Akinci, I & Ozmerzi, A 2005, 'Energy use pattern of some field crops and vegetable production: Case study for Antalya Region, Turkey', *Energy Conversion and Management*, vol. 46, no. 4, pp. 655-66.
- Carsel, RF & Parrish, RS 1988, 'Developing joint probability distributions of soil water retention characteristics', *Water Resources Research*, vol. 24, no. 5, pp. 755-69.
- Cassel, D 1982, 'Tillage effects on soil bulk density and mechanical impedance', In: *Predicting tillage effects on soil physical properties and processes*, PW Unger, et al. (eds.), American Society of Agronomy, pp. 55-67.
- Cavero, J, PlayÃ_in, E, Zapata, N & Faci, JM 2001, 'Simulation of maize grain yield variability within a surface-irrigated field', *Agronomy journal*, vol. 93, no. 4, pp. 773-82.
- Chan, K 1989, 'Effect of tillage on aggregate strength and aggregation of vertisols', *Soil and Tillage Research*, vol. 13, no. 2, pp. 163-75.
- Childs, JL, Wallender, WW & Hopmans, JW 1993, 'Spatial and seasonal variation of furrow infiltration', *Journal of irrigation and drainage engineering*, vol. 119, p. 74.
- Christiansen, J, Bishop, A, Kiefer Jr, F & Fok, YS 1966, 'Evaluation of intake rate constants as related to advance of water in surface irrigation', *Transaction of ASAE*, vol. 9, no. 5, pp. 671-4.
- Clemmens, A & Strelkoff, T 1999, *SRFR*, U.S. Water Conservation Laboratory, Agricultural Research Services, A surface irrigation simulation model.
- Collis-George, N & Laryea, K 1972, 'An examination of the wet aggregate analysis, the moisture characteristic, and infiltration-percolation methods of determining the stability of soil aggregates', *Soil Research*, vol. 10, no. 1, pp. 15-24.
- Connolly, R, Freebairn, D, Bell, M & Thomas, G 2001, 'Effects of rundown in soil hydraulic condition on crop productivity in south-eastern Queensland-a simulation study', *Soil Research*, vol. 39, no. 5, pp. 1111-29.
- Connolly, R, Bell, M, Huth, N, Freebairn, D & Thomas, G 2002, 'Simulating infiltration and the water balance in cropping systems with APSIM-SWIM', *Soil Research*, vol. 40, no. 2, pp. 221-42.

- Cook, FJ, Knight, JH, Humphreys, E, Tisdall, J, McHugh, AD & Hamilton, G 2008, Modelling water and solute processes and scenarios for optimization of permanent raised bed systems in China, India, Pakistan and Indonesia, Australian Centre for International Agricultural Research (ACIAR), Canberra ACT 2601, Australia GPO Box 1571.
- Cook, GD, So, HB & Dalal, RC 1992, 'Structural degradation of two Vertisols under continuous cultivation', *Soil and Tillage Research*, vol. 24, no. 1, pp. 47-64.
- Cooper, J 1999, 'A grower survey of rotations used in the New South Wales cotton industry', *Australian Journal of Experimental Agriculture*, vol. 39, no. 6, pp. 743-55.
- Coquet, Y, Vachier, P & Labat, C 2005, 'Vertical variation of near-saturated hydraulic conductivity in three soil profiles', *Geoderma*, vol. 126, no. 3-4, pp. 181-91.
- Coughlan, K 1981, 'The structure of Vertosols', In: *The properties and utilisation of cracking clay soils*, JW McGarity, et al. (eds.), University of New England, Armidale, New South Wales, Australia, pp. 87-96.
- Coutadeur, C, Coquet, Y & Roger Estrade, J 2002, 'Variation of hydraulic conductivity in a tilled soil', *European Journal of Soil Science*, vol. 53, no. 4, pp. 619-28.
- Czarnes, S, Hallett, P, Bengough, A & Young, I 2000, 'Root and microbial derived mucilages affect soil structure and water transport', *European Journal of Soil Science*, vol. 51, no. 3, pp. 435-43.
- Dalton, P, Raine, SR & Broadfoot, K 2001, Best management practices for maximizing whole farm irrigation efficiency in the Australian Cotton industry, National Centre for Engineering in Agriculture Report 179707/2, USQ, Toowoomba.
- David, R 2005, *Australian synthetic daily class A pan evaporation*, Queensland government, Department of Natural Resources and Mines.
- Dexter, A 1991, 'Amelioration of soil by natural processes', *Soil and Tillage Research*, vol. 20, no. 1, pp. 87-100.
- Dunn, G, Ahuja, L, Fiedler, F, Garrison, A & Benjamin, J 1998, 'Changes in soil water retention curves due to tillage and natural reconsolidation', *Soil Science Society of America Journal*, vol. 62, no. 5, pp. 1228-33.
- Dwyer, L, Ma, B, Stewart, D, Hayhoe, H, Balchin, D, Culley, J & McGovern, M 1996, 'Root mass distribution under conventional and conservation tillage', *Canadian Journal of Soil Science*, vol. 76, no. 1, pp. 23-8.
- Edgerton, MD 2009, 'Increasing crop productivity to meet global needs for feed, food, and fuel', *Plant Physiology*, vol. 149, no. 1, pp. 7-13.
- Edwards, J, Wood, C, Thurlow, D & Ruf, M 1992, 'Tillage and crop rotation effects on fertility status of a Hapludult soil', *Soil Science Society of America journal* (USA), vol. 56, pp. 1577-82.

- Eldeiry, A, Garcia, L, El-Zaher, ASA & Kiwan, MES 2005, 'Furrow irrigation system design for clay soils in arid regions', *Applied Engineering in Agriculture*, vol. 21, no. 3, pp. 411-20.
- Eldridge, D & Robson, A 1997, 'Bladeploughing and exclosure influence soil properties in a semi-arid Australian woodland', *Journal of Range Management*, vol. 50, no. 2, pp. 191-8.
- Elliott, R & Walker, W 1982, 'Field evaluation of furrow infiltration and advance functions', *ASAE Transaction*, vol. 25, no. 2, pp. 396–400.
- Esfandiari, M & Maheshwari, B 1997, 'Application of the optimization method for estimating infiltration characteristics in furrow irrigation and its comparison with other methods', *Agricultural Water Management*, vol. 34, no. 2, pp. 169-85.
- Fares, A & Alva, A 2000, 'Soil water components based on capacitance probes in a sandy soil', *Soil Science Society of America Journal*, vol. 64, no. 1, pp. 311-8.
- Findeling, A, Ruy, S & Scopel, E 2003, 'Modeling the effects of a partial residue mulch on runoff using a physically based approach', *Journal of Hydrology*, vol. 275, no. 1-2, pp. 49-66.
- Fischer, R, Sayre, K & Monasterio, IO 2005, 'The effect of raised bed planting on irrigated wheat yield as influenced by variety and row spacing', In: *Evaluation and performance of permanent raised bed cropping systems in Asia, Australia and Mexico*, C Roth, et al. (eds.), ACIAR Proceedings No. 121, Grifith, NSW, Australia, pp. 1-11.
- Flury, M, Flühler, H, Jury, WA & Leuenberger, J 1994, 'Susceptibility of soils to preferential flow of water: A field study', *Water Resources Research*, vol. 30, no. 7, pp. 1945-54.
- Gardner, P, Angus, J, Pitson, G & Wong, P 1998, 'A comparison of six methods to control take-all in wheat', *Australian journal of agricultural research*, vol. 49, no. 8, pp. 1225-40.
- Garside, A, Borrell, A, Ockerby, S, Dowling, A, McPhee, J & Braunack, M 1992, 'An irrigated rice based cropping system for tropical Australia', In: 6th Australian Agronomy Conference, University of New England, Armadale, pp. 259–61.
- Garside, AL 2005, 'The potential for permanent raised beds in sugarcane cropping systems in Australia', In: Evaluation and performance of permanent raised bed cropping systems in Asia, Australia and Mexico, C Roth, et al. (eds.), ACIAR Proceeding No. 121, Griffith, NSW, Australia, pp. 154-61.
- Gates, TK & Clyma, W 1984, 'Designing Furrow-Irrigation Systems for Improved Seasonal Performance', *ASAE Paper*, vol. 80, no. 2517, p. 21.
- Gharbi, A, Daghari, H & Cherif, K 1993, 'Effect of flow fluctuations on free draining, sloping furrow and border irrigation systems', *Agricultural Water Management*, vol. 24, no. 4, pp. 299-319.

- Gill, MA 1994, 'On-Farm Water Management: A historical overview', In: Water and Community: An Assessment of On-Farm Water Management Programme, Sustainable Development Policy Institute (SDPI), Islamabad, pp. 24-39.
- Gill, MA, Ahmad, M & Awan, NA 2005, 'Evaluation and performance of growing crops on permanent raised beds in Pakistan's Punjab', In: Evaluation and performance of permanent raised bed cropping systems in Asia, Australia and Mexico, C Roth, et al. (eds.), ACIAR Proceeding No. 121, Griffith, NSW, Australia, pp. 66-71.
- Gilland, B 2006, 'Population, nutrition and agriculture', *Population & Environment*, vol. 28, no. 1, pp. 1-16.
- Gillies, M 2008, 'Managing the effect of infiltration variability on the performance of surface irrigation', PhD thesis, University of Southern Queensland (USQ) Toowoomba, Queensland.
- Gillies, M & Smith, RJ 2005, 'Infiltration parameters from surface irrigation advance and run-off data', *Irrigation Science*, vol. 24, no. 1, pp. 25-35.
- Gillies, M, Smith, R & Raine, S 2008, 'Measurement and management of furrow irrigation at the field scale', In: *Irrigation Australia Conference, Share the Water, Share the Benefits,* Melbourne, Australia.
- Gillies, MH, Smith, RJ, Williamson, B & Shanahan, M 2010, 'Improving performance of bay irrigation through higher flow rates', In: Australian Irrigation Conference and Exibition 2010: One Water Many Futures, Irrigation Australia Ltd., Sydney, Australia.
- Govaerts, B, Sayre, KD, Lichter, K, Dendooven, L & Deckers, J 2007, 'Influence of permanent raised bed planting and residue management on physical and chemical soil quality in rain fed maize/wheat systems', *Plant and soil*, vol. 291, no. 1-2, pp. 39-54.
- Govaerts, B, Sayre, K, Ceballos-Ramirez, J, Luna-Guido, M, Limon-Ortega, A, Deckers, J & Dendooven, L 2006, 'Conventionally tilled and permanent raised beds with different crop residue management: effects on soil C and N dynamics', *Plant and soil*, vol. 280, no. 1-2, pp. 143-55.
- Govaerts, B, Mezzalama, M, Sayre, KD, Crossa, J, Lichter, K, Troch, V, Vanherck, K, De Corte, P & Deckers, J 2008, 'Long-term consequences of tillage, residue management, and crop rotation on selected soil micro-flora groups in the subtropical highlands', *Applied Soil Ecology*, vol. 38, no. 3, pp. 197-210.
- Gupta, SC & Larson, W 1979, 'Estimating soil water retention characteristics from particle size distribution, organic matter percent, and bulk density', *Water Resources Research*, vol. 15, no. 6, pp. 1633-5.
- Halvorson, AD, Wienhold, BJ & Black, AL 2002, 'Tillage, nitrogen, and cropping system effects on soil carbon sequestration', *Soil Science Society of America Journal*, vol. 66, no. 3, pp. 906-12.
- Hameed, A & Solangi, AK 1993, 'Water management strategies for area with poor drainage or shallow water table conditions', In: Irrigation System Management Research Symposium, Lahore Pakistan.

- Hamilton, G, Bakker, D, Houlbrooke, D & Span, C 2005, *Raised Bed Farming in Western Australia*, Department of Agriculture, Western Australia, (Bulletin 4646).
- Hansen, VE, Israelson, OA & Stingham, G 1979, *Irrigation Principles and Practice*, John Wiley & Sons, Inc., New York.
- Hanson, BR, Prichard, TL & Schulbach, H 1993, 'Estimating furrow infiltration', *Agricultural Water Management*, vol. 24, no. 4, pp. 281-98.
- Hao, C & Yali, Y 2012, 'Effect of Wheel Traffic Induced Soil Compaction on Soil Water Infiltration', *Journal of Agricultural Mechanization Research*, vol. 2, p. 034.
- Harris, PS, Biggs, AJW & Coutts, AJ 1999, Field Manual, in Central Darling Downs Land Management Manual. Department of Natural Resources, Queensland. DNRQ990102.
- Hassan, I, Hussain, Z & Akbar, G 2005, 'Effect of permanent raised beds on water productivity for irrigated maize-wheat cropping system', In: Evaluation and performance of permanent raised bed cropping systems in Asia, Australia and Mexico, C Roth, et al. (eds.), ACIAR Proceeding No. 121, Griffith, NSW, Australia, pp. 59-65.
- Hillel, D 1982, Introduction to soil physics, vol. 364, Academic Press New York.
- Hillel, D 2004, *Introduction to environmental soil physics*, Elsevier Academic Press, Amsterdam, Netherlands.
- Hobbs, PR 2007, 'Conservation agriculture: what is it and why is it important for future sustainable food production?', *Journal of Agricultural Science-Cambridge*, vol. 145, no. 2, p. 127.
- Hobbs, PR, Sayre, K & Gupta, R 2008, 'The role of conservation agriculture in sustainable agriculture', *Philosophical Transactions of the Royal Society B: Biological Sciences*, vol. 363, no. 1491, pp. 543-55.
- Holzapfel, E, Jara, J, Zuniga, C, Marino, M, Paredes, J & Billib, M 2004, 'Infiltration parameters for furrow irrigation', *Agricultural Water Management*, vol. 68, no. 1, pp. 19-32.
- Holzapfel, EA, Marino, MA & Chavez-Morales, J 1985, 'Performance irrigation parameters and their relationship to surface-irrigation design variables and yield', *Agricultural Water Management*, vol. 10, no. 2, pp. 159-74.
- House, ML, Powers, WL, Eisenhauer, DE, Marx, DB & Fekersillassie, D 2001, 'Spatial analysis of machine-wheel traffic effects on soil physical properties', *Soil Science Society of America Journal*, vol. 65, no. 5, pp. 1376-84.
- Hulme, PJ 1987, 'An evaluation of seedbed preperation methods for growing irrigated cotton in grey clays', UNE, Armidale, NSW.
- Hulugalle, N, Nehl, D & Weaver, T 2004, 'Soil properties, and cotton growth, yield and fibre quality in three cotton-based cropping systems', *Soil and Tillage Research*, vol. 75, no. 2, pp. 131-41.

- Hulugalle, NR & Daniells, IG 2005, 'Permanent beds in Australia cotton production systems', In: Evaluation and performance of permanent raised bed cropping systems in Asia, Australia and Mexico, C Roth, et al. (eds.), ACIAR Proceedings No. 121, Griffith, NSW, Australia, pp. 162-72.
- Hulugalle, NR, Weaver, TB & Finlay, LA 2010, 'Soil water storage and drainage under cotton-based cropping systems in a furrow-irrigated Vertisol', *Agricultural Water Management*, vol. 97, no. 10, pp. 1703-10.
- Hulugalle, NR, McCorkell, BE, Weaver, TB, Finlay, LA & Gleeson, J 2007, 'Soil properties in furrows of an irrigated Vertisol sown with continuous cotton (Gossypium hirsutum L.)', *Soil and Tillage Research*, vol. 97, no. 2, pp. 162-71.
- HulugalleA, NR, WeaverA, TB & FinlayA, LA 2010, 'Drainage under permanent beds in a furrow-irrigated Vertisol', In: Proceedings of the 19th World Congress of Soil Science: Soil solutions for a changing world, International Union of Soil Sciences (IUSS), Brisbane, Australia, pp. 21-4.
- Hunsaker, D, Clemmens, A & Fangmeier, D 1999, 'Cultural and irrigation management effects on infiltration, soil roughness, and advance in furrowed level basins', *ASAE Transaction*, vol. 42, no. 6, pp. 1753-62.
- Hussain, I, Olson, K & Siemens, J 1998, 'Long-term tillage effects on physical properties of eroded soil', *Soil science*, vol. 163, no. 12, pp. 970-81.
- Hussain, Z, Khan, MA & Irfan, M 2010, 'Water energy and economic analysis of wheat production under raised bed and conventional irrigation systems: A case study from a semi-arid area of Pakistan', *Soil and Tillage Research*, vol. 109, no. 2, pp. 61-7.
- Ishaq, M, Hassan, A, Saeed, M, Ibrahim, M & Lal, R 2001, 'Subsoil compaction effects on crops in Punjab, Pakistan: I. Soil physical properties and crop yield', *Soil and Tillage Research*, vol. 59, no. 1-2, pp. 57-65.
- Iwata, S, Tabuchi, T & Warkentin, BP 1995, Soil-water interactions: mechanisms and applications, Dekker, New York.
- Izadi, B & Wallender, W 1985, 'Furrow hydraulic characteristics and infiltration', *Transactions of ASAE*, vol. 28, no. 6, pp. 1901-8.
- Jat, M, Gathala, MK, Singh, K, Ladha, J, Gupta, RK, Singh, S, Sharma, S, Saharawat, Y & Tetarwal, J 2008, 'Experiences with permanent beds in the rice-wheat systems of the western Indo-Gangetic Plain', in ACIAR PROCEEDINGS: proceedings of theACIAR PROCEEDINGS ACIAR; 1998, p. 98.
- Javaid, AT & Khalid Usman 2009, 'Regulated deficit irrigation scheduling of maize crop', *Sarhad Journal of Agriculture (Pakistan)*, vol. 25, no. 3, pp. 441-50.
- Jaynes, D & Hunsaker, D 1989, 'Spatial and temporal variability of water content and infiltration on a flood irrigated field', *ASAE Transaction*, vol. 32, no. 4, pp. 1229-38.
- Jensen, M, Rangeley, W & Dieleman, P 1990, 'Irrigation trends in world agriculture', *Agronomy (USA)*, vol. 30, pp. 31-67.

- Jin, H, Hongwen, L, Kuhn, N, Xuemin, Z & Wenying, L 2007, 'Soil loosening on permanent raised-beds in arid northwest China', *Soil and Tillage Research*, vol. 97, no. 2, pp. 172-83.
- Jobling, GA & Turner, AK 1973, 'Physical Model Study of Border-Strip Irrigation', Journal of the Irrigation and Drainage Division, vol. 99, no. 4, pp. 493-510.
- Johnson, NC, Copeland, PJ, Crookston, RK & Pfleger, F 1992, 'Mycorrhizae: possible explanation for yield decline with continuous corn and soybean', *Agronomy journal*, vol. 84, pp. 387-90.
- Jones, H 1990, 'Plant water relations and implications for irrigation scheduling', *Acta Horticulturae*, vol. 278, pp. 67-76.
- Jones, HG 2004, 'Irrigation scheduling: advantages and pitfalls of plant-based methods', *Journal of Experimental Botany*, vol. 55, no. 407, pp. 2427-36.
- Juan, JA, Tarjuelo, J, Valiente, M & Garcia, P 1996, 'Model for optimal cropping patterns within the farm based on crop water production functions and irrigation uniformity I: Development of a decision model', *Agricultural Water Management*, vol. 31, no. 1-2, pp. 115-43.
- Kahlown, M, Shafique, M & Iqbal, M 1998, *Improved irrigation methods for efficient use of irrigation water under different water-table depths*, Mona Reclamation Experimental Project, WAPDA, Sargodha, Pakistan.
- Kalwij, IM 1997, Assessing the field irrigation performance and alternative management options for basin surface irrigation systems through hydrodynamic modelling, IWMI, Lahore, Pakistan.
- Khalid, M & Smith, JL 1976, 'Control of furrow infiltration by compaction', *American Society of Agricultural Engineers*, no. 77, p. 160.
- Khan, HU, Sadozai, GU, Khan, EA & Zaidi, N 2004, 'Growth behaviour of wheat as affected by various seed rates and herbicides', *Pak. J. Weed Sci. Res*, vol. 10, no. 3/4, pp. 129-32.
- Khatri, KL 2007, 'Toward real-time control of surface irrigation', PhD thesis, University of Southern Queensland (USQ), Toowoomba, Queensland.
- Kibblewhite, M, Ritz, K & Swift, M 2008, 'Soil health in agricultural systems', *Philosophical Transactions of the Royal Society B: Biological Sciences*, vol. 363, no. 1492, p. 685.
- Koskinen, WC & McWhorter, CG 1986, 'Weed control in conservation tillage', *Journal of soil and water conservation*, vol. 41, no. 6, pp. 365-70.
- Lal, R 1985, 'A soil suitability guide for different tillage systems in the tropics', *Soil and Tillage Research*, vol. 5, no. 2, pp. 179-96.
- Lal, R 2004, Principles of soil physics, Marcel Dekker, Inc, New York and London.
- Langat, PK & Raine, SR 2006, 'Using simulation modelling to improve the design and management of furrow irrigation in small-holder plots', In: 10th Southern and Eastern Africa Rainwater Network Conference, Southern and Eastern African Rainwater Network, Mombasa, Kenya.

- Langdale, G, Blevins, R, Karlen, D, McCool, D, Nearing, M, Skidmore, E, Thomas, A, Tyler, D & Williams, J 1991, 'Cover crop effects on soil erosion by wind and water', *Cover crops for clean water. Soil and Water Conserv. Soc., Ankeny, IA*, pp. 15-22.
- Levy, G & Torrento, J 1995, 'Clay dispersion and macroaggregate stability as affected by exchangeable potassium and sodium', *Soil science*, vol. 160, no. 5, pp. 352-8.
- Li, Y, Tullberg, J & Freebairn, D 2001, 'Traffic and residue cover effects on infiltration', *Soil Research*, vol. 39, no. 2, pp. 239-47.
- Limon-Ortega, A, Govaerts, B, Deckers, J & Sayre, KD 2006, 'Soil aggregate and microbial biomass in a permanent bed wheat-maize planting system after 12 years', *Field Crops Research*, vol. 97, no. 2–3, pp. 302-9.
- Lin, H, McLNNES, K, Wilding, L & Hallmark, C 1998, 'Macroporosity and initial moisture effects on infiltration rates in vertisols and vertic intergrades', *Soil science*, vol. 163, no. 1, p. 2.
- Lipiec, J, KuÅ, J, SÅ,owiÅ,,ska-Jurkiewicz, A & Nosalewicz, A 2006, 'Soil porosity and water infiltration as influenced by tillage methods', *Soil and Tillage Research*, vol. 89, no. 2, pp. 210-20.
- Lucy, MJ 1993, Permanent wide beds a controlled traffic system for irrigated vertisols, Department of Primary Industries, Queensland.
- Luis S, P 1999, 'Higher performance through combined improvements in irrigation methods and scheduling: a discussion', *Agricultural Water Management*, vol. 40, no. 23, pp. 153-69.
- Ma, Z, Boa, X, Shu, Q, Liu, S, Sun, J & Yang, R 2006, 'Studies on yield and benefit of conservation tillage for wheat-corn intercropping', In: *China-Canada Conservation Agriculture Forum on Conservation Agriculture Technology and Application*, Beijing, PRC, pp. 139-51.
- Ma, ZM & Wang, FH 2005, 'Studies on yield and benefit of conservation tillage for wheat-corn intercropping', In: Evaluation and performance of permanent raised bed cropping systems in Asia, Australia and Mexico, C Roth, et al. (eds.), ACIAR Proceeding No. 121, Griffith NSW, Australia.
- Mailhol, J, Ruelle, P & Popova, Z 2005, 'Simulation of furrow irrigation practices (SOFIP): a field-scale modelling of water management and crop yield for furrow irrigation', *Irrigation Science*, vol. 24, no. 1, pp. 37-48.
- Mailhol, JC, Ruelle, P, Walser, S, Schütze, N & Dejean, C 2011, 'Analysis of AET and yield predictions under surface and buried drip irrigation systems using the Crop Model PILOTE and Hydrus-2D', Agricultural Water Management, vol. 98, no. 6, pp. 1033-44.
- Malhi, SS, Lemke, R, Wang, Z & Chhabra, BS 2006, 'Tillage, nitrogen and crop residue effects on crop yield, nutrient uptake, soil quality, and greenhouse gas emissions', *Soil and Tillage Research*, vol. 90, no. 1-2, pp. 171-83.

- Mark, WS, Timothy, RG & James, CAI 2008, 'Tillage effects on soil hydraulic properties in space and time: State of the science', *Soil & Tillage Research*, vol. 99, pp. 4-48.
- Marshall, TJ & Holmes, JW 1988, *Soil Physics*, Second edn, Cambridge University Press, New York.
- Martin, D, Stegman, E & Fereres, E 1990, 'Irrigation scheduling principles', In: Management of Farm Irrigation Systems, G Hoffman, et al. (eds.), American Society of Agricultural Engineers, St. Joseph, MI, pp. 155-203.
- McBeath, JH & McBeath, J 2010, 'Plant diseases, pests and food security', in *Environmental Change and Food Security in China*, Springer, pp. 117-56.
- McBride, RG 2004, 'Relationships Between Soil Properties and Yield Variability and the Potential for Establishing Management Zones for Site-Specific Management in North Carolina', PhD thesis, North Carolina State University.
- McClymont, D & Smith, R 1996, 'Infiltration parameters from optimisation on furrow irrigation advance data', *Irrigation Science*, vol. 17, no. 1, pp. 15-22.
- McFarlane, D, Wheaton, G, Negus, T & Wallace, J 1992, 'Effects of waterlogging on crop and pasture production in the upper Great Southern Region of Western Australia', *Australian Journal of Soil Research*, vol. 32, pp. 377-87.
- McGarry, D 1989, 'The effect of cultivation at two different soil water contents on the structure and fabric of a vertisol', *Journal of Soil Science*, vol. 40, pp. 199-207.
- McGarry, D 1993, 'Degradation of soil structure', in GM Tainsh & WC Boughton (eds), *Land Degradation Processes in Australia*, Longman Cheshire, Melbourne, pp. 271-305.
- McGarry, D 1995, 'The optimisation of soil structure for cotton production', In: *Challenging the future. World Cotton Research Conference*, GA Constable & NW Forrester (eds.), Brisbane, Queensland, pp. 169–76.
- McGarry, D 2003, 'Tillage and soil compaction', *Conservation agriculture:* environment, farmers experiences, innovations, socio-economy, policy, p. 307.
- McHugh, A, Tullberg, J & Freebairn, D 2009, 'Controlled traffic farming restores soil structure', *Soil and Tillage Research*, vol. 104, no. 1, pp. 164-72.
- McKenzie, D & McBratney, A 2001, 'Cotton root growth in a compacted Vertosol (Grey Vertosol). I. Prediction using strength measurements and'limiting water ranges'', *Soil Research*, vol. 39, no. 5, pp. 1157-68.
- McKenzie, DC 1998, SOILpak for cotton growers, 3rd edition, NSW Agriculture, Orange, NSW, Australia.
- McKenzie, DC & Chan, KY 1990, 'Structural improvement of compacted Vertosols', In: *Soil Compaction workshop*, DPI, Toowoomba, Queensland, pp. 36-40.
- McKenzie, N, Henderson, B & McDonald, W 2002, *Monitoring soil change:* principles and practices for Australian conditions, CSIRO Land & Water.

- McLaughlin, MJ, Tiller, K, Naidu, R & Stevens, D 1996, 'Review: the behaviour and environmental impact of contaminants in fertilizers', *Soil Research*, vol. 34, no. 1, pp. 1-54.
- Mead, R, Ayars, J & Liu, J 1995a, *Evaluation of the Sentek EnviroSCAN RT5 capacitance probe: Laboratory calibration and field analysis*, Water Management Research Laboratory ARS-USDA, Fresno, Calif.
- Mead, R, Ayars, J & Liu, J 1995b, 'Evaluating the influence of soil texture, bulk density and soil water salinity on a capacitance probe calibration', *ASAE Paper*, pp. 95-3264.
- Meek, B, Rechel, E, Carter, L, DeTar, W & Urie, A 1992, 'Infiltration rate of a sandy loam soil: effects of traffic, tillage, and plant roots', *Soil Science Society of America Journal*, vol. 56, no. 3, pp. 908-13.
- Meki, MN, Snider, JL, Kiniry, JR, Raper, RL & Rocateli, AC 2012, 'Energy sorghum biomass harvest thresholds and tillage effects on soil organic carbon and bulk density', *Industrial Crops and Products*, vol. 43, pp. 172-82.
- Merriam, JL & Keller, J 1978, *Farm irrigation system evaluation: a guide for management*, Agricultural and irrigation engineering department, Utah State University, Logan.
- Michael, AM 1999, *Irrigation: Theory and practice*, Vikas Publishing House Pvt. Ltd Jangapura, New Delhi, India.
- Mitchell, A, Ellsworth, T & Meek, B 1995, 'Effect of root systems on preferential flow in swelling soil', *Communications in Soil Science & Plant Analysis*, vol. 26, no. 15-16, pp. 2655-66.
- Molden, D 1997, Accounting for water use and productivity, International Water Management Institute (IWMI), Colombo, Sri Lanka.
- Morgan, CLS & Lowery, JM 2003, 'Estimating plant-available water across a field with an inverse yield model', *Soil Science Society of America Journal*, vol. 67, no. 2, pp. 620-9.
- Mungomery, V & Byth, D 1976, 'Genetic variability for zinc deficiency symptoms in soybeans', In: Australian Oilseeds and Protein Grains Research Conference pp. 16-8.
- Nickel, SE, Crookston, RK & Russelle, MP 1995, 'Root growth and distribution are affected by corn-soybean cropping sequence', *Agronomy journal*, vol. 87, no. 5, pp. 895-902.
- Ockerby, S, Adkins, S & Garside, A 1999, 'The uptake and use of nitrogen by paddy rice in fallow, cereal, and legume cropping systems', *Australian journal of agricultural research*, vol. 50, pp. 945-52.
- Oliveira, MRG, Serralheiro, RP, Reis, MPZ & Santos, FL 1998, 'Maize Root System Response to Furrow Irrigation in a Mediterranean Brown Soil: Root Growth Related to Water Distribution', *Journal of Agricultural Engineering Research*, vol. 71, no. 1, pp. 13-7.
- Oster, J, Hoffman, G & Robinson, F 1984, 'Management alternatives: crop, water, and soil', *California Agriculture*, vol. 38, no. 10, pp. 29-32.

- Oyonarte, N & Mateos, L 2003, 'Accounting for soil variability in the evaluation of furrow irrigation', *ASAE Transactions*, vol. 46, no. 1, pp. 85-94.
- Oyonarte, N, Mateos, L & Palomo, M 2002, 'Infiltration variability in furrow irrigation', *Journal of irrigation and drainage engineering*, vol. 128, no. 1, pp. 26-33.
- Pagliai, M, Vignozzi, N & Pellegrini, S 2004, 'Soil structure and the effect of management practices', *Soil and Tillage Research*, vol. 79, no. 2, pp. 131-43.
- Panettieri, M, Carmona, I, Melero, S, Madejón, E & Gómez-Macpherson, H 2013, 'Effect of permanent bed planting combined with controlled traffic on soil chemical and biochemical properties in irrigated semi-arid Mediterranean conditions', *CATENA*.
- PARC 1982, *Consumptive Use of Water for Crops in Pakistan*, Land and Water Resources Department, Natural Resources Division, Pakistan Agricultural Research Council, Islamabad.
- Payero, J 2008, 'Optimisng maize yield under irrigation, http://www.grdc.com.au/research-and-development/GRDC-Updatepapers/2008/09/optimising-maize-yield-under-irrigation', pp. http://www.grdc.com.au/research-and-development/GRDC-Updatepapers/2008/09/optimising-maize-yield-under-irrigation.
- Payero, J & Harris, G 2010, *Maximising prfitability with limited water in coton farming systems*, Final report to Cotton Catchment Communities CRC Project No. 1.02.09.
- Paz, J, Batchelor, W, Tylka, G & Hartzler, R 2001, 'A modeling approach to quantify the effects of spatial soybean yield limiting factors', ASAE Transaction, vol. 44, no. 5, pp. 1329-34.
- Peries, R & Gill, JS 2011, 'The synergy of raised beds, controlled traffic, minimum tillage and stubble retention deliver higher water use efficiency in South West Victoria, Australia', In: World Congress on Conservation Agriculture, Brisbane.
- Philip, JR 1957a, 'The theory of infiltration: 4. Sorptivity and algebraic infiltration equations', *Soil science*, vol. 84, no. 3, pp. 257-64.
- Philip, JR 1969, 'Theory of infiltration', Advances in Hdroscience, vol. 5, pp. 215-96.
- Phillips, RE, Thomas, GW, Blevins, RL, Frye, WW & Phillips, SH 1980, 'No-tillage agriculture', *Science*, vol. 208, pp. 1108-13.
- Pinheiro, E, Pereira, M & Anjos, L 2004, 'Aggregate distribution and soil organic matter under different tillage systems for vegetable crops in a Red Latosol from Brazil', *Soil and Tillage Research*, vol. 77, no. 1, pp. 79-84.
- Playán, E & Mateos, L 2006, 'Modernization and optimization of irrigation systems to increase water productivity', *Agricultural Water Management*, vol. 80, no. 1-3, pp. 100-16.
- Postel, SL 1993, 'Water and agriculture ', In: *Water in Crisis: A Guide to the World's Fresh Water Resources*, PH Gleick (ed.), Oxford University Press, Oxford, pp. 56-66.

- Potter, K, Torbert, H & Morrison, J 1995, 'Tillage and residue effects on infiltration and sediment losses on vertisols', *Transactions of the ASAE-American Society* of Agricultural Engineers, vol. 38, no. 5, pp. 1413-44.
- Purcell, B & Associates 1999, *Determining a framework, terms and definitions for water use efficiency in irrigation*, Land and Water Resources Research and Development Corporation, Narrabri, NSW.
- Qureshi, A, McCornick, P, Qadir, M & Aslam, Z 2008, 'Managing salinity and waterlogging in the Indus Basin of Pakistan', *Agricultural Water Management*, vol. 95, no. 1, pp. 1-10.
- Qureshi, R & Barrett-Lennard, E 1998, *Saline agriculture for irrigated land in Pakistan: a handbook*, Australian Centre for International Agricultural Research, Monograph MN 050, Canberra.
- Rahimi, A 2011, 'Evaluation of soil infiltration in furrow irrigation', *Australian Journal of Basic and Applied Sciences*, vol. 5, no. 11, pp. 1542-5.
- Raine, S & Bakker, D 1996, 'Increased furrow irrigation efficiency through better design and management of cane fields', In: Australian Society of Sugar Cane Technologists, Mackay pp. 114-24.
- Raine, S & Shannon, E 1996, 'Improving the efficiency and profitability of furrow irrigation for sugarcane production', In: *Research Towards Efficient and Sustainable Production*, W JR, et al. (eds.), CSIRO Division of Tropical Crops and Pastures, Brisbane, pp. 211-2.
- Raine, S & Foley, J 2002, 'Comparing application systems for cotton irrigation-what are the pros and cons', In: 11th Australian Cotton Conference, Australian Cotton Growers Research Association Wee Waa, NSW and Brisbane Queensland, Australia, pp. 575-83.
- Raine, S & Smith, R 2007, Simulation modelling for surface irrigation evaluation: Training in using SIRMOD and IPARM, National Centre for Engineering in Agriculture (NCEA), Toowoomba, Queensland.
- Raine, S, McClymont, D & Smith, R 1998, 'The effect of variable infiltration on design and management guidelines for surface irrigation', In: Australian Society of Sugar Cane Technologists, pp. 293-301.
- Rassam, D, Simunek, J & Genuchten, M, Th, van. 2004, *Modelling variably* saturated flow with Hydrus -2D, ND Consult, Brisbane, Australia.
- Reddell, D 1981, 'Modified rate of advance method for an automatic furrow irrigation system', *ASAE Paper*.
- Reddell, D & Latimer, E 1987, 'Field evaluation of an advance rate feedback irrigation system', In: Irrigation Systems for the 21st Century, ASCE, Portland, Oregon (USA).
- Renault, D & Wallender, W 1992, 'ALIVE (Advance Linear Velocity): Surface irrigation rate balance theory', *Journal of irrigation and drainage engineering*, vol. 118, p. 138.
- Rezanezhad, F, Quinton, W, Price, J, Elliot, T, Elrick, D & Shook, K 2010, 'Influence of pore size and geometry on peat unsaturated hydraulic

conductivity computed from 3D computed tomography image analysis', *Hydrological processes*, vol. 24, no. 21, pp. 2983-94.

- Rhoton, F 2000, 'Influence of time on soil response to no-till practices', *Soil Science Society of American Journal*, vol. 64, pp. 700-9.
- Richards, L 1931, 'Capillary conduction of liquids through porous mediums', in *Physics*, vol. 5, pp. 318–33.
- Roth, C, Fischer, R, Piggin, C & Meyer, W 2005, 'Evaluation and performance of permanent raised bed systems in Asia, Mexico and Australia: A synopsis', In: *Evaluation and performance of permanent raised bed cropping systems in Asia, Australia and Mexico*, C Roth, et al. (eds.), ACIAR Proceeding No. 121, Griffith, NSW, Australia, pp. 200-8.
- Samrah, AK, Pillai McGarry, U & McGarry, D 1996, 'Repair of the structure of a compacted Vertosol via wet/dry cycles', *Soil and Tillage Research*, vol. 38, no. 1-2, pp. 17-33.
- Sanchez, C, Zerihun, D & Farrell-Poe, K 2009, 'Management guidelines for efficient irrigation of vegetables using closed-end level furrows', Agricultural Water Management, vol. 96, no. 1, pp. 43-52.
- Sayre, K & Hobbs, P 2004, 'The raised-bed system of cultivation for irrigated production conditions', In: *Sustainable agriculture and the international rice-wheat system*, Marcel Dekker, Inc., New York (USA), pp. 337-55.
- Sayre, K, Limon, A & Govaerts, B 2005, 'Experiences with permanent bed planting systems CIMMYT, Mexico', In: Evaluation and performance of permanent raised bed cropping systems in Asia, Australia and Mexico, C Roth, et al. (eds.), ACIAR Proceeding No. 121, Griffith, NSW, Australia, pp. 12-25.
- Schoenfi sch, M 1999, *MACHINE-pak A machinery manual for the cotton industry*, Cotton Research & Development Corporation, Narrabri, NSW, and University of Southern Queensland, Queensland, Australia.
- Schwankl, LJ & Wallender, WW 1988, 'Zero inertia furrow modeling with variable infiltration and hydraulic characteristics', ASAE Transaction, vol. 31, no. 5, pp. 1470-5.
- Schwankl, LJ, Raghuwanshi, NS & Wallender, WW 2000, 'Furrow irrigation performance under spatially varying conditions', *Journal of irrigation and drainage engineering*, vol. 126, pp. 355-61.
- Segeren, AG & Trout, TJ 1991, 'Hydraulic resistance of soil surface seals in irrigated furrows', *Soil Science Society of America Journal*, vol. 55, pp. 640-6.
- Sepaskhah, A & Ghahraman, B 2004, 'The effects of irrigation efficiency and uniformity coefficient on relative yield and profit for deficit irrigation', *Biosystems Engineering*, vol. 87, no. 4, pp. 495-507.
- Shafiq, M, Hassan, I & Hussain, Z 2002, 'Influence of Irrigation methods on the productivity of summer maize under saline/sodic environment', Asian Journal of Plant Science, vol. 6, pp. 678-80.

- Shafique, M & Skogerboe, G 1983, 'Impact of seasonal infiltration function variation on furrow irrigation performance', In: Advances in Infiltration, American Society of Agricultural Engineers, St Joseph MI, pp. 292-301.
- Shepard, J, Wallender, W & Hopmans, J 1993, 'One-point method for estimating furrow infiltration', *ASAE Transactions*, vol. 36, no. 2, pp. 395-404.
- Shipitalo, M, Dick, W & Edwards, W 2000, 'Conservation tillage and macropore factors that affect water movement and the fate of chemicals', *Soil and Tillage Research*, vol. 53, no. 3-4, pp. 167-83.
- Shull, H 1961, 'A by-pass infiltrometer', ASAE Transaction, vol. 4, no. 1, pp. 15-7.
- Simunek, J, Sejna, M & Genuchten, MT 2007, HYDRUS (2D/3D): Software package for simulating the two-and three-dimensional movement of water, heat, and multiple solutes in variably-saturated media, Available at www. pc-progress. cz, PC-Progress, Prague, Czech Republic.
- Šimunek, J, Van Genuchten, MT & Šejna, M 1999, The HYDRUS-2D software package for simulating the two-dimensional movement of water, heat, and multiple solutes in variably-saturated media: Version 2.0, International Ground Water Modelling Centre, US Salinity Lab, Colorado School of Mines, Golden, Colorado.
- Singer, MJ & Munns, DN 1999, *Soils: An introduction, 4th Edition* Prentice Hall: Upper Saddle River.
- Six, JE, Elliott, ET & Paustian, ET 2000, 'Soil structure and soil organic matter: II. A normalized stability index and the effect of mineralogy', *Soil Science Society* of America Journal, vol. 64, pp. 1042-9.
- Smith, G, Yule, D & Coughlan, K 1983, 'Soil physical factors in crop production on Vertosols in Qld. Australia ', In: *International Workshop on soils-research in the tropics,* Townsville.
- Smith, K, Jackson, D & Pepper, T 2001, 'Nutrient losses by surface run-off following the application of organic manures to arable land. 1. Nitrogen', *Environmental Pollution*, vol. 112, no. 1, pp. 41-51.
- Smith, R, Raine, S & Minkevich, J 2005, 'Irrigation application efficiency and deep drainage potential under surface irrigated cotton', Agricultural Water Management, vol. 71, no. 2, pp. 117-30.
- Smith, R, Gillies, M, Shanahan, M, Campbell, B & Williamson, B 2009, 'Evaluating the performance of bay irrigation in the GMID', In: *Irrigation Australia Ltd*, Swan Hill, Victoria, Australia, pp. 1-12.
- Smith, RJ 1990, 'The distribution of water from rigid and lay-flat gated pipe', In: Australian Society of Sugar Cane Technologists, Townville.
- Soane, B 1976, Studies on some soil physical properties in relation to cultivations and traffic, Technical bulletin-Ministry of Agriculture, Fisheries and Food, London.
- Solomon, K 1984, 'Yield related interpretations of irrigation uniformity and efficiency measures', *Irrigation Science*, vol. 5, no. 3, pp. 161-72.

- Soon, Y & Arshad, M 1996, 'Effects of cropping systems on nitrogen, phosphorus and potassium forms and soil organic carbon in a Gray Luvisol', *Biology and fertility of soils*, vol. 22, no. 1, pp. 184-90.
- Starr, JL & Paltineanu, IC 1998, 'Real-time soil water dynamics over large areas using multisensor capacitance probes and monitoring system', *Soil and Tillage Research*, vol. 47, no. 1-2, pp. 43-9.
- Tabuada, MA, Rego, ZJC, Vachaud, G & Pereira, LS 1995, 'Two-dimensional infiltration under furrow irrigation: modelling, its validation and applications', *Agricultural Water Management*, vol. 27, no. 2, pp. 105-23.
- Tanji, KK 1990, Agricultural salinity assessment and management, American Society of Civil Engineers, Irrigation and Drainage Division. Water Quality Technical Committee.
- Tarboton, K & Wallender, W 1988, 'Field: wide furrow infiltration variability', *American Society of Agricultural Engineers*, vol. 32, no. 3, pp. 913-8.
- Thomas, GA, Standley, J, Webb, AA, Blight, GW & Hunter, HM 1990, 'Tillage and crop residue management affect vertisol properties and grain sorghum growth over seven years in the semi-arid sub-tropics. 1. Crop residue and soil water during fallow periods', *Soil and Tillage Research*, vol. 17, no. 34, pp. 181-97.
- Thompson, JA 1995, 'Irrigated cropping on permanent beds in southern New South Wales', In: *National Controlled Traffic Conference*, Rockhampton Queensland, pp. 21-4.
- Thompson, JA & North, S 1994, *Raised beds reduce winter waterlogging*, Department of Primary Industries, Australia.
- Tilman, D, Cassman, KG, Matson, PA, Naylor, R & Polasky, S 2002, 'Agricultural sustainability and intensive production practices', *Nature*, vol. 418, no. 6898, pp. 671-7.
- Tilman, D, Fargione, J, Wolff, B, D'Antonio, C, Dobson, A, Howarth, R, Schindler, D, Schlesinger, WH, Simberloff, D & Swackhamer, D 2001, 'Forecasting Agriculturally Driven Global Environmental Change', *Science*, vol. 292, no. 5515, pp. 281-4.
- Tim, N, House, D & House, N 2008, Land & water management plan, Marinya Farm, 1831 Karara Road, Cambooya, Control traffic farming, CTF solutions.
- Tisdall, J, Cockroft, B & Uren, N 1978, 'The stability of soil aggregates as affected by organic materials, microbial activity and physical disruption', *Soil Research*, vol. 16, no. 1, pp. 9-17.
- Tisdall, JM 1994, 'Possible role of soil microorganisms in aggregation in soils.', *Plant and soil*, vol. 159, pp. 115-21.
- Topp, GC 2003, 'State of the art of measuring soil water content', *Hydrological processes*, vol. 17, no. 14, pp. 2993-6.
- Tricker, AS 1978, 'The infiltration cylinder: Some comments on its use', *Journal of Hydrology*, vol. 36, no. 3-4, pp. 383-91.

- Trout, T 1990, 'Furrow inflow and infiltration variability impacts on irrigation management', *Transaction of ASAE*, vol. 33, no. 4, pp. 1171-8.
- Trout, TJ 1992, 'Flow velocity and wetted perimeter effects on furrow infiltration', *Transaction of ASAE*, vol. 35, pp. 855-63.
- Trout, TJ & Kemper, W 1983, 'Factors which affect furrow intake rates', In: *National Conference on Advances in Infiltration,* ASAE, pp. 302-12.
- Trout, TJ & Mackey, BE 1988a, 'Furrow inflow and infiltration variability', *ASAE Transactions*, vol. 31, no. 2, pp. 531-7.
- Tullberg, J 2010, 'Tillage, traffic and sustainability--A challenge for ISTRO', *Soil and Tillage Research*, vol. 111, pp. 26-32.
- Tullberg, J & Murray, S 1988, 'Controlled traffic in sub-tropical grain production', In: 11th Conference: International Society of Soil and Tillage Research Organisation Germany, pp. 323–7.
- Tullberg, J, Yule, D & McGarry, D 2007, 'Controlled traffic farming-From research to adoption in Australia', *Soil and Tillage Research*, vol. 97, no. 2, pp. 272-81.
- Turner, D & Sumner, M 1978, 'The influence of initial soil moisture content on field measured infiltration rates', *Water SA*, vol. 4, no. 1, pp. 18-24.
- Ullah, MK, Habib, Z & Muhammad, S 2001, Spatial distribution of reference and potential evapotranspiration across the Indus Basin Irrigation Systems, IWMI, Lahore, Pakistan.
- UN 2005, Intrenational Decade for Action 'Water for Life' 2005-2015, United Nations Department of Economic and Social Affairs.
- UN 2009, Water in a changing world. The United Nations World Water Development Report 3, UNESCO, UK, London.
- UN 2010, World Population Prospects, the 2010 Revision, United Nations, Department of Economic and Social Affairs.
- Unger, PW 2006, Soil and water conservation handbook: policies, practices, conditions, and terms, Haworth Food & Agricultural Products Press, NY.
- Utomo, W & Dexter, A 1982, 'Changes in soil aggregate water stability induced by wetting and drying cycles in non-saturated soil', *Journal of Soil Science*, vol. 33, no. 4, pp. 623-37.
- Van Es, HM, Cassel, DK & Daniels, RB 1991, 'Infiltration variability and correlation with surface soil properties for an eroded Hapludult', *Soil Science Society of American Journal*, vol. 55, no. 2, pp. 486-92.
- Verhulst, N, Carrillo-GarcÃa, A, Moeller, C, Trethowan, R, Sayre, KD & Govaerts, B 2011, 'Conservation agriculture for wheat-based cropping systems under gravity irrigation: increasing resilience through improved soil quality', *Plant* and soil, vol. 340, no. 1, pp. 467-79.
- Vörösmarty, C, Green, P, Salisbury, J & Lammers, R 2000, 'Global water resources: vulnerability from climate change and population growth', *Science*, vol. 289, no. 5477, pp. 284-8.

- Walker, JP, Willgoose, GR & Kalma, JD 2004, 'In situ measurement of soil moisture: a comparison of techniques', *Journal of Hydrology*, vol. 293, no. 1-4, pp. 85-99.
- Walker, WR 1989, *Guidelines for designing and evaluating surface irrigation systems*, 45, FAO, Irrigation and Drainage.
- Walker, WR 2003, *Surface irrigation simulation, evaluation and design*, Utah State University, Logan, Utah.
- Walker, WR & Skogerboe, GV 1987, Surface irrigation: theory and practice, Prentice-Hall, Englewood Cliffs.
- Wallace, J 2000, 'Increasing agricultural water use efficiency to meet future food production', Agriculture, Ecosystems & Environment, vol. 82, no. 1-3, pp. 105-19.
- Wang, X, Wu, H, Dai, K, Zhang, D, Feng, Z, Zhao, Q, Wu, X, Jin, K, Cai, D & Oenema, O 2012, 'Tillage and crop residue effects on rainfed wheat and maize production in northern China', *Field Crops Research*, vol. 132, pp. 106-16.
- Werth, JA, Preston, C, Roberts, GN & Taylor, IN 2006, 'Weed management practices in glyphosate-tolerant and conventional cotton fields in Australia', *Animal Production Science*, vol. 46, no. 9, pp. 1177-83.
- White, SC 2007, 'Partial rootzone drying and deficit irrigation in cotton for use under large mobile irrigation machines', PhD thesis, University of Southern Queensland, Toowoomba.
- Wichelns, D & Oster, J 1990, 'Potential economic returns to improved irrigation infiltration uniformity', Agricultural Water Management, vol. 18, no. 3, pp. 253-66.
- Wightman, B, Peries, R, Bluett, C & Johnston, T 2005, 'Permanent rasied bed cropping in southern Australia: practical guidelines for implementation', In: *Evaluation of performance of permanent rasied bed cropping systems in Asia, Australia and Mexico*, CH Roth, et al. (eds.), ACIAR Proceeding No. 121, Griffith, NSW, Australia, pp. 173-90.
- Wilding, L & Drees, L 1983, 'Spatial variability and pedology', *Developments in Soil Science*, vol. 11, pp. 83-116.
- World Bank 1997, *Staff appraisal report, Pakistan National Drainage Project*, Report No. 1531U, World Bank, Washington DC.
- Yang, XM & Wander, MM 1999, 'Tillage effects on soil organic carbon distribution and storage in a silt loam soil in Illinois', *Soil and Tillage Research*, vol. 52, no. 1-2, pp. 1-9.
- Yasin, M, Ahmad, MM, Hussain, Z, Khan, Z & Akbar, G 2002, 'Issues of farmers in saline and waterlogged areas', *Journal of Drainage and Water Management*, vol. 6, no. 1, pp. 9-18.
- Zhang, H, Wang, X, You, M & Liu, C 1999, 'Water-yield relations and water-use efficiency of winter wheat in the North China Plain', *Irrigation Science*, vol. 19, no. 1, pp. 37-45.

Appendix A: Specific field measured data

 Table A 1: Soil properties at given depths at site A1, Australia (Soil analysis conducted by Topoclimate services Pty Ltd)

Soil Element	0-20cm	21-100cm
рН	6.7-7.4	7.3-8.5
Na (meq/100gm)	0.48-4.38	1.13-16.6
Ca (meq/100gm)	13.7-26.0	12.4-21.8
Mg (meq/100gm)	16.8-22.3	15.4-26.6
K (meq/100gm)	0.71-1.86	025-0.40
Organic Carbon (%)	1.0-1.8-	0.6-1.4
Phosphorus (mg/kg)	6.4-17.6	5.716.6
Sulphur (mg/kg)	5.3-21.2	8.5-17.0
Soil type	Irving-Black Vertisol	Irving Black Vertisol

Table A 2: Soil properties at	given depths on	two bed sizes	measured at t	he harvest of
wheat 2008-09 crop at site P1,	Pakistan.			

	Bed size and profile depth (cm)									
Soil Elements	Wide bed (132 cm) middle			Wide	Wide bed (132 cm) Edge			Narrow Bed (66 cm)		
	010	1120	2130	010	1120	2130	010	1120	2130	
EC(µS)	413.7	331.3	322.9	349.8	326.9	307.1	372.4	331.1	324.1	
рН	8.1	8.1	8.0	8.0	8.1	7.9	8.1	8.1	7.9	
Na(mg/l)	10.6	11.3	13.5	10.2	10.9	11.0	10.1	10.5	11.6	
Ca(mg/l)	55.0	50.9	48.6	51.4	48.0	51.6	50.6	48.4	50.2	
Mg(mg/l)	12.2	11.7	10.8	12.2	11.1	10.9	12.3	11.4	11.1	
SAR	2.7	2.1	2.5	1.8	2.1	2.0	1.8	1.9	2.1	
FeSO4(mg/l)	17.2	17.3	17.1	17.3	17.1	17.3	17.3	17.0	17.1	
Meq of FeSO4 Meg of	8.6	8.7	8.5	8.6	8.6	8.7	8.6	8.5	8.6	
K2Cr2O7	10.0	10.0	10.0	10.0	10.0	10.0	10.0	10.0	10.0	
Difference	1.4	1.4	1.5	1.4	1.4	1.3	1.4	1.5	1.4	
O.M (%)	1.0	0.9	1.0	0.9	1.0	0.9	0.9	1.0	1.0	
N2(mg/l)	0.0	0.0	0.1	0.0	0.0	0.0	0.0	0.1	0.0	
Saturation	46	44	44	45	43	42	44	42	41	

Month	Ave. Temj (°c	perature	Ave. Humidity	Ave. Wind Speed	Total Rainfall	
	Min.	Max.	(%)	(m/s)	(mm)	
Dec. 2009	18 (2)	30 (2)	67 (8)	4(1)	27	
Jan. 2010	17 (3)	32 (3)	60 (10)	4(1)	29	
Feb. 2010	18 (2)	29 (3)	68 (6)	4(1)	35	
Mar. 2010	16 (3)	27 (2)	71 (7)	4(1)	81	
Apr. 2010	12 (3)	26 (2)	68 (6)	3 (1)	14	
May 2010	5 (5)	22 (3)	66 (12)	3 (1)	9	
June 2010	3 (4)	18 (2)	71 (7)	3(1)	3	
July 2010	4 (5)	18(3)	72 (7)	3(1)	26	
Aug. 2010	3 (5)	17 (3)	70 (8)	3(1)	59	
Sept. 2010	9 (5)	21 (3)	75 (7)	2(1)	47	
Oct. 2010	10 (4)	23 (4)	74 (6)	2(1)	63	
Nov. 2010	14 (4)	25 (4)	69 (6)	2(1)	48	
Dec. 2010	16 (3)	26 (2)	75 (9)	3(1)	260	
Jan. 2011	17 (2)	29 (4)	70 (8)	3(1)	94	
Feb. 201	17 (3)	30 (3)	68 (5)	3(1)	23	
Mar. 2011	15 (4)	27 (3)	71 (8)	3(1)	86	
Apr. 2011	10 (2)	25 (3)	73 (5)	3(1)	87	
May 2011	3 (2)	21 (3)	75 (7)	2(1)	48	
June 2011	1 (4)	19 (2)	72 (11)	2 (2)	0	
July 2011	1 (5)	17 (2)	60 (6)	3 (1)	4	
Aug. 2011	2 (3)	20 (2)	63 (7)	3 (1)	20	
Sep. 2011	5 (3)	23 (4)	65 (7)	3 (1)	15	
Oct. 2011	8 (7)	24 (3)	70 (7)	3 (1)	82	
Nov. 2011	10 (4)	29 (3)	64 (8)	3 (1)	37	
Dec. 2011	12 (3)	27 (3)	66 (7)	3 (1)	83	
Jan. 2012	14 (6)	29 (4)	66 (12)	3 (1)	49	
Feb. 2012	15 (7)	30 (3)	72 (6)	3 (1)	52	

 Table A3:
 Average monthly weather data at site A1 during four cropping seasons (SD in brackets)

top width, hi vi – midule width, i	Irrig.	Furrow dimensions (cm)					
Site/Crop/Treatment	No	Spacing	TW	MW	BW	D	
Site A1/Soybean/Existing	1	200 cm	41.7	23.7	11.0	8.3	
Site A1/Soybean/Existing	2	200 cm	53.3	30.3	14.3	10.0	
Site A2/Cotton/Existing	1	100 cm	62	42	20.5	8.75	
Site A2/Cotton/Existing	2	100 cm	60.0	43.7	27.3	10.2	
Site P1/Cotton/NB-Existing	1	66 cm	36.7	24.6	18.2	10	
Site P1/Cotton/NB-Existing	2	66 cm	37	25.7	18.8	9.74	
Site P1/Cotton/WB-Existing	1	132cm	62.1	52.3	39.4	11.9	
Site P1/Cotton/WB-Existing	2	132cm	62	50.8	37.4	11.3	
Site P2/Maize/NB-Existing	1	66 cm	50.1	35.7	22.4	12.4	
Site P2/Maize/NB-Existing	2	66 cm	50.6	33.8	18.0	11.8	
Site P2/Maize/WB-Existing	1	132cm	42.8	34.8	15.8	11.5	
Site P2/Maize/WB-Existing	2	132cm	42.4	32.8	15.1	11.1	
Site P3/Maize/NB-Existing	1	66 cm	47.4	30.6	14.8	9.2	
Site P3/Maize/NB-Existing	2	66 cm	48.1	31.2	14.1	9.3	
Site P3/Maize/WB-Existing	1	132 cm	51.5	33.4	17.9	9.7	
Site P3/Maize/WB-Existing	2	132 cm	51.6	32.9	17.6	9.5	
Site A1/Wheat/No till	1	200 cm	53.4	35.1	15.3	12.4	
Site A1/Wheat/Shallow cultivation	1	200 cm	48.8	32.0	15.6	10.6	
Site A1/Wheat/Blade ploughing	1	200 cm	60.6	38.6	18.6	14.3	
Site A1/Hemp/No till	1	200 cm	44.8	33.0	21.3	12.6	
Site A1/Hemp/Shallow cultivation	1	200 cm	45.7	34.7	23.2	13.3	
Site A1/Hemp/Blade ploughing	1	200 cm	45.7	34.8	22.7	12.8	
Site A1/Hemp/No till	2	200 cm	44.8	33.0	21.3	12.6	
Site A1/Hemp/Shallow cultivation	2	200 cm	45.7	34.7	23.2	13.3	
Site A1/Hemp/Blade ploughing	2	200 cm	45.7	34.8	22.7	12.8	
Site A1/Hemp/No till	3	200 cm	52.5	30.5	14.5	13.3	
Site A1/Hemp/Shallow cultivation	3	200 cm	51.7	32.7	18.3	12.5	
Site A1/Hemp/Blade ploughing	3	200 cm	57.0	35.0	16.5	14.8	
Site A1/Hemp/No till	4	200 cm	53	31	14.5	13.0	
Site A1/Hemp/Shallow cultivation	4	200 cm	52	33	18.0	12.0	
Site A1/Hemp/Blade ploughing	4	200 cm	57.0	35.0	16.5	14.5	
Site A1/Corn/No till	1	200 cm	57.0	43.5	17.5	13.0	
Site A1/Corn/Shallow cultivation	1	200 cm	54.5	39.3	23.5	11.8	
Site A1/Corn/Blade ploughing	1	200 cm	59.8	46.8	26.8	11.8	
Site A1/Corn/No till	2	200 cm	52.7	39.5	21.3	12.4	
Site A1/Corn/Shallow cultivation	2	200 cm	48.3	35.0	26.5	11.1	
Site A1/Corn/Blade ploughing	2	200 cm	56.3	41.5	30.5	11.2	

Table A4: Average bed furrow dimensions measured before irrigations during the benchmarking study in both soils and three renovation experiments in Australia (TW = top width, MW = middle width, BW = Bottom width and D = furrow depth)

Country, Site	Irrigation	Statistical Average advance time (min)						
no. & bed		Statistical	Distance along furrow					
width	INO.	parameters	100m	200m	300m	400m	465m	
ustralia								
	1.4	Mean	54.7	160.7	415.0	696.7	993.3	
	1St Invigation	n	21	21	21	21	21	
Site A1	Infigation	SD	5.0	23.7	31.2	50.3	51.3	
200cm	and	Mean	32.4	119.3	248.6	423.1	861.8	
	2110 irrigation	п	24	24	24	24	24	
	Infigation	SD	4.3	9.6	15.2	14.2	17.4	
_			90m	180m	270m	360m	450m	
	1st	Mean	11.3	64.2	131.7	234.7	386.7	
Site A2	Ist	n	22	22	22	22	22	
100cm -	Ingation	SD	2.3	5.5	9.9	9.5	13.6	
Tootin	2nd	Mean	69.9	168.6	299.6	454.9	584	
	irrigation	n	20	20	20	20	20	
	Intigation	SD	10.0	16.0	8.4	7.1	6.7	
akistan				Dist	ance alon	g furrow		
ampun			21m	42m	64m	86m		
	1ct	Mean	4.0	11.8	23.2	38.2		
	ISI Irrigation	n	10	10	10	10		
Site P1 NR -	Ingation	SD	0.7	0.8	1.5	1.5		
Shellind	2nd	Mean	5.0	12.0	21.4	33.8		
	irrigation	n	10	10	10	10		
	Intigation	SD	0.7	2.4	2.4	2.9		
	1st	Mean	3.8	10.5	17.8	23.5		
	Irrigation	n	8	8	8	8		
Site P1 WR -		SD	0.5	1.7	2.1	2.1		
	2nd	Mean	4.5	10.8	17.5	24.5		
	211u irrigation	n	8	8	8	8		
	Intigation	SD	1.0	2.1	2.4	2.4		
akistan				Dista	ance alon	g furrow		
amstan			18m	36m	54m	72m	90m	
	1et	Mean	4.2	9.2	14.2	20.2	24.8	
	151 Irrigation	n	10	10	10	10	10	
Site P2 NR -	Ingation	SD	1.1	1.8	2.2	3.0	1.7	
SHE I 2 IND	and	Mean	4.4	9.6	15.6	21.8	26.0	
	2AQ irrigation	n	10	10	10	10	10	
		SD	0.9	1.3	3.0	4.1	0.8	
	1 of	Mean	3.5	7.8	12.5	18.0	23.8	
	15t Irrigation	n	8	8	8	8	8	
Site P2 WR -	IIIgation	SD	0.6	1.7	3.1	4.8	6.2	
	2nd	Mean	3.8	9.0	14.5	22.0	29.5	
	211U irrigation	n	8	8	8	8	8	
	IIIgaului	SD	0.5	0.8	1.0	1.4	1.3	
	1et	Mean	5.2	11.0	17.4	24.0	31.2	
	131 Irrigation	n	10	10	10	10	10	
Site P3 NB -		SD	1.1	1.9	2.7	3.5	3.9	
	Ind	Mean	4.6	9.6	15.4	21.4	27.8	
	∠11U irrigation	n	10	10	10	10	10	
	IIIgaului	SD	0.5	0.5	0.9	0.9	1.1	
	1et	Mean	3.8	8.8	14.3	20.8	27.8	
	181 Irrigation	n	8	8	8	8	8	
Site D3 WR -	IIIgation	SD	0.5	1.3	1.7	2.2	3.8	
		Mean	3.5	8.0	13.3	18.8	24.5	

n SD

8 0.6

8 1.2

8 1.5

8 1.5

2nd

irrigation

Appendix A: Specific field measured data

8

1.3

Table A6: Temporal variation in average bulk density (BD in gm cm⁻³) and soil moisture (SM in %) of 0-30 cm profile at 10 cm interval for three PRB renovation treatments during three cropping seasons in a Vertisol, Australia (SD in brackets)

Cron	Data	Depth	No tillage		Shallow cultivation		Blade ploughing	
Сгор	Date	(cm)	BD	\mathbf{SM}	BD	SM	BD	SM
	13/06/2010	010	1.01	40(2)	0.95	41	0.96	42
	13/00/2010	0 10	(0.02)	10(2)	(0.06)	(3)	(0.04)	(2)
		1020	1.05	44	1.06	46	1.03	48
		10 20	(0.03)	(3)	(0.04)	(3)	(0.03)	(1)
		2030	1.10	47	1.11	52	1.06	51
		20 30	(0.02)	(2)	(0.02)	(2)	(0.03)	(2)
	30/06/2010	010	1.00	47	0.96	50	0.96	44
	20100/2010	0 10	(0.01)	(4)	(0.04)	(2)	(0.05)	(3)
		1020	1.05	52	1.03	54	1.02	53
			(0.02)	(2)	(0.05)	(1)	(0.03)	(5)
		2030	1.09	56	1.09	54	1.05	57
Wheat		20 00	(0.03)	(1)	(0.02)	(2)	(0.04)	(2)
2010	4/10/2011	010	1.06	45	0.98	44	0.96	47
			(0.03)	(4)	(0.03)	(2)	(0.03)	(5)
		1020	1.09	53	1.07	48	1.03	54
			(0.02)	(4)	(0.06)	(3)	(0.02)	(4)
		2030		55	1.14	53	1.08	57
			(0.04)	(4)	(0.02)	(3)	(0.01)	(4)
	15/11/2011	010	1.03	46	0.97	39	0.96	51
			(0.05)	(3)	(0.06)	(/)	(0.05)	(2)
		1020	1.10	49	1.03	4/	1.04	54
			(0.03)	(4)	(0.03)	(3)	(0.00)	(2)
		2030	(0.08)	(3)	1.09	(8)	(0.02)	(3)
			1.04	(3)	1.03	45	0.02)	46
	26/01/2011	010	(0.03)	(3)	(0.05)	(3)	(0.04)	(3)
			1.08	52	1.06	51	1.06	52
		1020	(0.04)	(2)	(0.03)	(4)	(0.04)	(3)
			1 12	55	1 12	54	1 10	53
		2030	(0.01)	(3)	(0.02)	(4)	(0.01)	(3)
			1.05	42	1.05	47	0.99	46
	4/02/2011	010	(0.04)	(4)	(0.05)	(5)	(0.01)	(2)
		10.00	1.09	54	1.09	56	1.07	53
		1020	(0.05)	(4)	(0.02)	(2)	(0.05)	(1)
		20 20	1.12	56	1.11	57	1.10	56
		2030	(0.03)	(4)	(0.01)	(2)	(0.03)	(2)
	16/02/2011	0 10	1.05	49	1.07	52	0.99	50
Hemp	16/02/2011	010	(0.06)	(2)	(0.04)	(2)	(0.09)	(1)
2011		10 20	1.09	55	1.10	57	1.08	56
		1020	(0.05)	(3)	(0.03)	(2)	(0.03)	(2)
		20 30	1.12	59	1.13	59	1.11	57
		2030	(0.04)	(3)	(0.03)	(2)	(0.02)	(1)
	25/03/2011	010	1.07	40	1.07	42	1.01	45
	25/05/2011	010	(0.05)	(2)	(0.04)	(1)	(0.05)	(1)
		1020	1.12	45	1.10	45	1.09	46
		10 20	(0.05)	(3)	(0.03)	(2)	(0.03)	(2)
		2030	1.13	46	1.14	50	1.13	50
		00	(0.03)	(3)	(0.03)	(2)	(0.02)	(1)
	2/04/2011	010	1.09	49	1.08	49	1.01	49
		0 10	(0.01)	(2)	(0.04)	(5)	(0.03)	(2)
		1020	1.11	52	1.11	50	1.10	52
	-		(0.01)	(3)	(0.05)	(3)	(0.03)	(3)
	_							
------	------------	-------	--------	-----	--------	------	--------	-----
		2030	1.15	54	1.14	51	1.12	56
		20 50	(0.05)	(2)	(0.04)	(3)	(0.01)	(2)
	22/04/2011	010	1.10	38	1.09	39	1.01	43
	22/04/2011	0 10	(0.04)	(3)	(0.04)	(4)	(0.02)	(1)
		1020	1.12	44	1.12	43	1.10	47
		1020	(0.03)	(1)	(0.03)	(2)	(0.04)	(3)
		20 30	1.16	48	1.14	46	1.12	52
		2050	(0.03)	(2)	(0.03)	(2)	(0.01)	(3)
	14/05/2011	0 10	1.11	48	1.10	53	0.98	51
	14/03/2011	010	(0.02)	(3)	(0.01)	(2)	(0.02)	(1)
		10 20	1.11	52	1.13	54	1.11	56
		1020	(0.04)	(2)	(0.02)	(1)	(0.04)	(3)
		20 20	1.16	54	1.16	56	1.13	59
		2050	(0.03)	(3)	(0.02)	(1)	(0.05)	(3)
	10/10/2011	0 10	1.05	36	0.97	36	0.94	40
	10/10/2011	010	(0.10)	(2)	(0.04)	(3))	(0.04)	(2)
		10 20	1.07	39	1.02	40	1.00	43
		1020	(0.04)	(1)	(0.02)	(3)	(0.04)	(2)
		20 20	1.11	42	1.09	44	1.08	45
		2050	(0.10)	(1)	(0.04)	(2)	(0.02)	(2)
	12/11/2011	0 10	1.08	33	1.01	36	0.96	37
	13/11/2011	010	(0.04)	(2)	(0.02)	(2)	(0.05)	(2)
		10 20	1.11	35	1.09	38	1.05	42
		1020	(0.01)	(4)	(0.03)	(2)	(0.03)	(2)
		20 20	1.17	39	1.12	41	1.11	45
		2050	(0.02)	(5)	(0.01)	(3)	(0.01)	(1)
	11/12/2011	0 10	1.06	33	1.01	36	0.94	38
	11/12/2011	010	(0.04)	(3)	(0.02)	(3)	(0.03)	(2)
Corn		10 20	1.14	36	1.10	39	1.07	43
2011		1020	(0.02)	(2)	(0.02)	(4)	(0.02)	(4)
		20 20	1.16	39	1.17	42	1.14	46
		2050	(0.02)	(2)	(0.02)	(4)	(0.02)	(3)
	5/01/2012	0 10	1.07	14	1.08	15	1.03	17
	5/01/2012	010	(0.02)	(1)	(0.05)	(2)	(0.03)	(3)
		10 20	1.12	17	1.13	17	1.07	20
		1020	(0.01)	(1)	(0.02)	(3)	(0.05)	(3)
		20 30	1.17	19	1.19	20	1.13	21
		2030	(0.01)	(2)	(0.03)	(4)	(0.04)	(3)
	15/02/2011	0 10	1.08	48	1.06	51	1.02	51
	13/02/2011	010	(0.02)	(2)	(0.03)	(1)	(0.04)	(2)
		10.20	1.13	51	1.10	52	1.05	52
		1020	(0.02)	(1)	(0.01)	(1)	(0.01)	(2)
		20 20	1.18	53	1.16	54	1.10	55
		2050	(0.03)	(2)	(0.03)	(1)	(0.03)	(1)
						-		



Figure A1: Average water advance distance along furrow vs. square root of time during two irrigations (I1 = irrigation 1 and I2 = irrigation 2) for (a) site A1 and A2 on a Vertisol, Australia and (b) site P1, P2 and P3 on a sandy clay loam, Pakistan



Figure A2: Temporal variations in bulk density and soil moisture for 0-30 cm profile of three PRB renovation treatments during two cropping seasons (2010 wheat and 2011 hemp) seasons at site A1, Australia (vertical bars show SD)



Figure A3: Impact of three fresh renovation methods and irrigation management on water advance rate along furrow during single irrigation to wheat crop (a) and two irrigations to corn crop (b) at site A1, Australia



Figure A4: Impact of *SMD* (low during irrigation 1 & 2 and high during irrigation 3 & 4) on water advance rate along furrow during four irrigations to 2011 hemp crop under three PRB renovation treatments (one season old) at site A1, Australia

Appendix B: IPARM and SIRMOD calibration

Comparison of field measured advance time during benchmarking study (Chapter 4) with IPARM and SIRMOD simulated advance times to tail end was analysed using trend line analysis. The results showed a strong correlation (R2 > 0.99) between the two parameters at both sites (site A1 and site A2) in Australia (Figure B1). This confirmed the validity of the evaluations using IPARM and SIRMOD simulation techniques under the field conditions of both sites in Australia.



Figure B1: Measured vs. simulated water advance times along furrow length during two irrigations at (a) site A1 and (b) site A2 in Australia

The trend line analysis was also indicative of a very strong correlation ($R^2 > 0.98$) between the fields measured and SIRMOD simulated advance times (Figure B2) in Pakistan, which confirmed the validity of IRPARM and SIRMOD evaluation under the north west Pakistan field conditions.



Figure B2: Measured vs. simulated water advance times along furrow length at three sites under both (NB and WB) bed sizes in Mardan, Pakistan



Figure B3: Measured vs. simulated water advance times along furrow length of thee fresh PRB renovation treatments during three cropping seasons at site A1, Australia

Appendix C: EnviroSCAN and Micro-Gopher calibration validation



Figure C1: Volumetric soil moisture in 0-30 cm at 10 cm depth interval measured gravimetrically vs. volumetirc soil moisture measure with calibrated Sentek enviroSCAN across bed during three cropping seasons



Figure C2: Volumetric soil moisture in 0-30 cm at 10 cm depth interval measured gravimetrically vs. volumetric soil moisture measured with calibrated Micro-Gopher (measured before and after irrigation during 2011 corn crop experiments) across bed under (a) Vertisol, Australia and (b) sandy clay loam, Pakistan.

Appendix D: Irrigation performance and optimisation for

all sites and seasons

Table D1: Existing irrigation performance of raised beds and deep drainage losses (DD) during two irrigations on two sites in south east Queensland, Australia (SD in brackets)

		0	Tco	SI	RMOD	simula	tion	DD
Site	Event	$(I s^{-1})$	100 (min)	Ea	Er	DU	Inflow	(mm)
		(LS)	(mm)	(%)	(%)	(%)	(mm)	(1111)
	Innia 1	1.98	1143	68	100	88	146	47
A 1	Irrig.1	(0.11)	(15)	(3)	(0)	(4)	(5)	(2)
Al Junia (Innia 2	1.91	1057	77	99	92	130	30
	IITIg.2	(0.03)	(7)	(1)	(0)	(2)	(3)	(1)
	Innia 1	2.58	590	74	98	88	101	18
Irri	1111g.1	(0.03)	(0)	(1)	(0)	(1)	(2)	(0)
AZ	Irria 2	2.50	680	85	96	86	113	7
	Irrig.2	(0.02)	(42)	(3)	(1)	(4)	(4)	(0)

Table D2: Existing irrigation performance of raised beds and deep drainage losses(DD) during two irrigations on three sites with two bed sizes in north west Pakistan (SD in brackets)

Sito/			0	Tco -	SIRN	/IOD sim	ulated r	esults	ממ																
Dod	Crop	Event	$(I s^{-1})$	(min)	Ea	Er	DU	Inflow	(mm)																
Deu			(L.S)	(mm)	(%)	(%)	(%)	(mm)	(mm)																
		Innia 1	2.00	40	45	99	88	120	72																
P1 /	Cotton	Img.1	5.09	40	(1)	(1)	(2)	152	15																
NB	Cotton	Irria 2	2.81	37	55	99	94	110	50																
		nng.2	2.01	57	(1)	(1)	(5)	110	50																
		Irria 1	188	31	75	99	75	70	20																
P1 /	Cotton	iiig.i	4.00	51	(2)	(1)	(7)	1)	20																
WB	Cotton	Irrig 2	4 83	34	69	99	77	87	27																
		1115.2	4.05	54	(1)	(1)	(2)	07	21																
		Irrig 1	1 55	32	97	78	71	52	2																
P2/	Maize	1115.1	1.55	52	(3)	(4)	(4)	52	2																
NB	Maize	Irrig 2	1 68	36	86	88	61	62	9																
		1119.2	1.00		(2)	(1)	(4)	02	,																
		Irrig.1	2.76	34	99	77	79	46	0																
P2/	Maize			0.	(1)	(1)	(5)		Ū																
WB	1. Luize	Irrig.2	2.66	35	99	77	90	46	0																
				00	(1)	(1)	(6)		Ũ																
		Irrig.1	2.09	38	74	99	74	80	21																
P3/	Maize		,	20	(2)	(1)	(7)	00																	
NB	warze	Irrig.2	2.63	36	62	100	72	96	36																
		0			(I)	(0)	(5)																		
D2 /	P3/ Maize	Irrig.1	3.08	35	98	88	80	54	1																
P5/		0			(2)	(2)	(6) 70																		
wв	-	Irrig.2 2.85	2.85	33	100	78	70	47	0																
	Irrig.2		nng.2	ning.2	nng.2	nng.2	nng.2	nrig.2	Irrig.2	Irrig.2	Irrig.2	Irrig.2	Irrig.2	Irrig.2 2.85		(0)	(2)	(5)							

	1	T	0	Tee	SI	RMOI	D simul	lation	מת
Treatment	Crop	IIII.	\mathcal{Q}		Ea	Er	DU	Inflow	
	-	INO.	(LS)	(min)	(%)	(%)	(%)	(mm)	(mm)
	Wheat	1	1.90	843	90	93	82	103	10
	2010	1	(0.1)	(70)	(2)	(4)	(8)	(6)	(2)
		1	1.95	243	99	76	88	31	0
		1	(0.1)	(40)	(1)	(18)	(8)	(7)	(0)
		2	1.49	338	77	62	46	32	7
	Hemp	Z	(0.1)	(13)	(11)	(7)	(17)	(3)	(2)
No Tillago	2011	3	1.97	508	99	64	87	65	1
nu image		3	(0.1)	(7)	(1)	(5)	(4)	(5)	(0)
		4	1.75	534	100	60	81	61	0
		4	(0.0)	(18)	(0)	(3)	(5)	(4)	(0)
		1	1.8	857	74	98	79	99	26
	Corn	1	(0.1)	(17)	(2)	(1)	(3)	(3)	(2)
	2011	2	1.90	1013	98	91	89	124	2
		2	(0.1)	(26)	(2)	(3)	(2)	(7)	(0)
	Wheat	1	1.9	957	81	94	76	116	22
	2010	1	(0.03)	(32)	(2)	(2)	(4)	(5)	(3)
		1	1.66	288	94	72	75	31	6
		1	(0.1)	(62)	(5)	(17)	(11)	(7)	(1)
		2	1.65	353	77	72	52	37	9
	Hemp 2011	2	(0.02)	(24)	(14)	(13)	(23)	(2)	(2)
Shallow		3	2.01	538	98	68	79	70	1
Cultivation			5	(0.03)	(13)	(2)	(1)	(5)	(1)
		4	1.83	573	99	67	85	67	1
		-	(0.2)	(10)	(1)	(5)	(5)	(5)	(0)
		1	2.00	828	59	99	78	107	44
	Corn	1	(0.1)	(16)	(2)	(1)	(4)	(5)	(3)
	2011	2	2.03	1040	95	97	91	137	7
		2	(0.1)	(73)	(2)	(1)	(1)	(3)	(1)
	Wheat	1	2.12	1167	61	97	73	160	62
	2010	1	(0.1)	(25)	(3)	(1)	(5)	(9)	(4)
		1	2.04	200	95	63	74	26	1
		1	(0.05)	(4)	(3)	(3)	(7)	(1)	(0)
		2	1.85	337	79	79	58	40	8
	Hemp	2	(0.08)	(28)	(12)	(7)	(16)	(3)	(3)
Blade	2011	3	2.03	543	97	69	77	71	2
Ploughing		5	(0.07)	(7)	(4)	(4)	(13)	(2)	(1)
		Δ	2.13	543	99	74	89	75	1
		Ŧ	(0.11)	(36)	(1)	(7)	(4)	(7)	(0)
		1	2.3	857	47	99	80	126	67
	Corn	1	(0.23)	(69)	(2)	(1)	(3)	(3)	(5)
	2011	n	2.20	867	85	98	85	122	18
		2	(0.2)	(65)	(4)	(1)	(7)	(4)	(1)

Table D3: Average irrigation performance for all irrigation events applied on threePRB renovation methods during three cropping seasons on a Vertisol at site A1,Australia (SD in brackets)

		0	Taa	SI	RMOD) simul	ation	DD
Crop	Treatment	$(I \sigma^{-1})$	1co	Ea	Er	DU	Inflow	
		(LS)	(mm)	(%)	(%)	(%)	(mm)	(mm)
	NT	1.90a	843a	90a	93a	82a	103a	10a
	191	(0.1)	(55)	(2)	(4)	(8)	(6)	(1)
Wheat	SC	1.9a	957b	81b	94a	76b	116b	22b
2010	be	(0.03)	(32)	(2)	(2)	(4)	(5)	(1)
	RD	2.12b	1167c	61c	97b	73b	160c	62c
	DI	(0.1)	(25)	(3)	(1)	(5)	(9)	(4)
	NT	1.79a	406a	94a	65a	75a	47a	2a
	191	(0.2)	(62)	(5)	(11)	(20)	(16)	(1)
Hemp	SC	1.79a	438a	92a	70ab	73a	51a	3a
2011	be	(0.2)	(47)	(6)	(9)	(11)	(18)	(1)
	RP	2.01b	406a	93a	71b	75a	53a	3a
	DI	(0.1)	(58)	(7)	(8)	(15)	(21)	(1)
	NT	1.85a	935a	86a	94a	84a	112a	15a
	111	(0.1)	(83)	(12)	(5)	(6)	(14)	(6)
Corn	SC	2.02b	934a	77ab	98b	85a	122b	26a
2011	BC	(0.1)	(95)	(19)	(1)	(7)	(16)	(9)
	RP	2.25c	862b	66b	99b	83a	124b	42b
	BĽ	(0.2)	(65)	(8)	(1)	(6)	(13)	(10)

Table D4: Average seasonal irrigation performance for three PRB renovation methods during three cropping seasons on a Vertisol at site A1, Australia (SD in brackets)

*Figures followed by same letters in columns for each cropping season are not significantly (P=0.05) between treatments

Table D5:	Effect	of two	irrigation	management	optimisation	strategies	on current
irrigation]	perform	ance of	three sites	with two bed	l sizes during	two irriga	tions under
sandy clay	loam du	iring be	nchmarkin	g study in Ma	rdan, Pakista	n (SD in br	ackets)

Site	Bed Size	Strategy ^a	Q (L.s ⁻¹)	Tco (min)	Ea (%)	Er (%)	DU (%)	Inflow (mm)	DD (mm)
		0	3.0	39a	50a	100a	90a	121a	610
			(0.1)	(2)	(6)	(0)	(6)	(11)	01a
	NR	1	3.0	32b	59b	99ab	80b	103b	42h
	ΠD		(0.2)	(4)	(10)	(1)	(5)	(17)	420
		2	5.0	14c	80c	98b	85c	75c	15c
P1			(0.0)	(2)	(10)	(2)	(4)	(10)	150
•••		0	4.9	33a	72a	99a	77a	83a	23a
			(0.0)	(2)	(2)	(1)	(5)	(4)	204
	WB	1	4.9	21b	99b	89b	92b	54b	1b
			(0.0)	(1)	(1)	(2)	(5)	(3)	10
		2	4.9	21b	99b	89b	92b	54b	1b
			(0.0)	(1)	(1)	(2)	(5)	(3)	10
		0	1.6	34a	90a	83a	66a	56a	6a
			(0.1)	(2)	(7)	(8)	(11)	(6)	
	NB	1	1.6	38b	85a	88b	63a	62b	9a
			(0.1)	(3)	(4)	(4)	(10)	(3)	
		2	1.0	58c	90a	89b	72a	58ab	6a
P2			(0.0)	(9)	(5)	(4)	(14)	(6)	
		0	2.7	35a	99a	77a	84a	47a	0a
		1	(0.1)	(I)	(I)	(I)	(13)	(4)	
	WB	I	2.7	38b	98a	87b	82a	53b	1a
		•	(0.1)	(I)	(2)	(1)	(9)	(3)	
		2	1.5	68c	99a	87b	86a	526	1a
		0	(0.0)	(1)	(1)	(1)	(10)	(4)	
		0	2.4	$\frac{3}{a}$	68a	99a	/3a	88a	28a
		1	(0.3)	(1)	(4)	(I)	(0) 0.41-	(8) 5 (1-	
	NB	I	2.4	24D	98D	920	94D	560	1b
		2	(0.4)	(3)	(2)	(2)	(3)	(5) 5 (1-	
		2	2.4	24D	98D	920	94D	560	1b
P3		0	(0.4)	(2)	(2)	(2)	(3)	(4)	
		0	5.0	54a	99a	85a	/5a	50a	1a
		1	(0.1)	(1)	(1)	(4) 07h	(10)	(4) 5 4 h	
	WB	1	3.0	30a	9/a	8/D	/5a	54D	2a
		2	(0.2)	(2) 40h	(2)	(2) 061-	(10)	(2) 521-	
		Z	2.5	42D	98a	80D	/ða	55D	1a
			(0.0)	(1)	(2)	(2)	(11)	(2)	

*Figures followed by different letters in columns for each bed size on each site show significantly (p=0.05) different values between strategies

^a0: Farmer managed, 1: Tco optimised and 2: Q and Tco optimised

Table D6:	Effect o	f two	irrigati	ion m	anagement	optimisation	n strate	gies o	n current
irrigation	performa	nce of	three	PRB	renovation	treatments	during	three	cropping
seasons on	Vertisol a	at site A	A1, Aus	tralia	(SD in brac	kets)			

Crop	Treatment	^a Strategy	$\mathcal{Q}_{(L.s^{-1})}$	Tco (min)	Ea (%)	Er (%)	DU (%)	Total Inflow (<i>mm</i>)	^b DD (mm)
		0	1.90(0.1)	843a(55)	90a(2)	93a(2)	82a(7)	103a(6)	10a
	No tillage	1	1.90(0.1)	965b(<i>30</i>)	84b(<i>3</i>)	99b(1)	93b(3)	118b(5)	19b
01		2	2.00(0.0)	853a(60)	89a(5)	98b(1)	93b(5)	110a(8)	12a
201	Shallow	0	1.90(0.0)	957a(32)	81a(2)	94a(2)	76a(4)	116a(3)	22a
eat	Cultivation	1	1.90(0.0)	1125b(18)	73b(2)	100b(0)	90b(2)	137b(4)	37b
Vhe		2	3.00(0.0)	598c(16)	86c(1)	99b(1)	82c(4)	116a(2)	16c
-	Blade		2.12(0.1)	1167a(24)	61a(3)	97a(1)	73a(5)	160a(9)	62a
	Ploughing	1	2.12(0.1)	1245b(<i>31</i>)	59a(3)	100b(0)	85b(2)	170b(8)	70b
	riougining		4.00(0.0)	403c(28)	94b(5)	97a(2)	89b(<i>3</i>)	104c(7)	6c
		0	1.81(0.1)	415a(19)	94a(2)	65a(8)	75a(10)	191a(<i>16</i>)	3a
	No tillage	1	1.81(0.1)	415a(19)	94a(2)	65a(8)	75a(10)	189a(16)	3a
-		2	1.00(0.0)	941b(67)	91a(4)	78b(7)	72a(8)	240b(18)	5a
201	Shallow	0	1.79(0.1)	438a(47)	92a(6)	70a(9)	73a(11)	204a(18)	4a
du	Cultivation	1	1.79(0.1)	438a(47)	92a(6)	70a(9)	73a(11)	206a(18)	4a
Hen		2	1.13(0.1)	810b(60)	91a(5)	79b(8)	75a(10)	246b(29)	6a
H	Blada	0	2.00(0.1)	406a(58)	93a(6)	71a(8)	72a(9)	212a(21)	4a
	Ploughing	1	2.00(0.1)	406a(48)	93a(6)	71a(8)	72a(9)	212a(21)	4a
	Trong	2	1.50(0.1)	797b(85)	94a(4)	80b(7)	80a(8)	243b(27)	4a
		0	1.85(0.1)	935a(55)	86a(12)	94a(5)	84a(6)	224a(14)	16a
	No tillage	1	1.85(0.1)	913a(65)	87a(11)	94a(4)	84a(6)	218a(17)	14a
-		2	1.85(0.1)	913a(65)	87a(11)	94a(4)	84a(6)	218a(17)	14a
201	Shallow	0	2.02(0.1)	934a(60)	77a(12)	98a(1)	85a(3)	244a(16)	28a
Corn 2(Cultivation	1	2.02(0.1)	898a(65)	79a(11)	97a(1)	84a(3)	33ab(16)	24a
		2	3.5(0.0)	483b(70)	84b(8)	95a(1)	83a(3)	208b(22)	17b
-	Blade	0	2.25(0.2)	862a(47)	66a(8)	98a(2)	83a(6)	248a(7)	42a
	Ploughing	1	2.25(0.2)	833a(49)	68a(8)	98a(2)	81a(5)	240b(6)	38b
	Ploughing	2	4.00(0.0)	399b(<i>33</i>)	79b(5)	99a(1)	86a(6)	206c(10)	22c

*Figures followed by different letters in columns for each treatment in each crop season are significantly (P=0.05) different between strategies

^a0: Existing management

1: Tco optimised

2: Tco and Q optimised ^bDD = Deep drainage losses in mm/irrigation



Figure D1: Example irrigation management tool showing the effect of inflow rate on optimum time to cut-off on irrigation performance (*Ea* and *DU*) by confirming $Er \ge 85\%$ for two bed sizes (NB & WB) with 86-90 m furrow length at three sites on a sandy clay loam (vertical bars show SD)



Figure D2: Example field design and irrigation management tool showing the effect of furrow length and inflow rate on optimum *Tco*, inflow volume and *Tco Ta⁻¹* (solid lines) and; *Ea and DU* (dotted lines) for <u>narrow beds</u> (66 cm furrow spacing), by confirming $Er \ge 85\%$ and water reaching tail end with 60 mm *SMD*, at given three sites in a sandy clay loam, at Mardan, Pakistan (Vertical bars show SD)



Figure D3: Example field design and irrigation management tool showing the effect of furrow length and inflow rate on optimum *Tco*, inflow volume and *Tco* Ta^{-1} (solid lines) and *Ea* and *DU* (dotted lines) for <u>wide beds</u> (132 cm furrow spacing) by confirming *Er* \geq 85% and water reaching to tail end with 60 mm *SMD*, at given three sites in a sandy clay loam, at Mardan, Pakistan (vertical bars show SD)

Appendix E: Hydrus 2D calibration

E1 Introduction

The Hydrus 2D (Šimunek et al. 1999) model was calibrated according to the field measured data of the first experiment (Chapter 6) and during the first irrigation to the 2011 corn crop on a Vertisol. For sandy clay loam the average field measured data of both experiments (Chapter 6) was used for calibrating the model. This study was aimed to calibrate the Hydrus 2D model for evaluating different strategies (Chapter 7) for improving the lateral infiltration on both soils.

E2 Material and methods

The Hydrus 2D model was calibrated according to the pre-processing procedure outlined in (Rassam et al. 2004). The model was set to simulate water flow only in the main processes. A general geometry type in vertical plane with cm as the length unit was selected. The soil profile was selected with single (layer) material for NT treatment and two materials (heterogeneity) for SC and BP treatments. Time units were set to hours with a final time of 9 hours. A single number of time variable boundary records was selected. All available print options were enabled and the number of print times was set to 9 in both soils for obtaining hourly details. The maximum number of iterations was selected at 30 and all other default values of iteration criteria were not changed. Initial conditions were set to water content instead of pressure head. The soil hydraulic model of van Genuchten Maulem with no hysteresis options, were selected. The air entry value of -2 cm was selected on Vertisol to account for the shrink swell characteristics. The soil hydraulic parameters for the analytical functions were initially selected from the Hydrus inbuilt database, which were subsequently moderately adjusted as given in Table E1 according to field measured values by taking guidance from previous studies (Carsel & Parrish 1988; Connolly et al. 2001; Connolly et al. 2002; Cook et al. 2008) and by attaining a good fit between the measured and simulated values. In the time variable boundary conditions precipitation was set to zero while evaporation and transpiration was set to 0.004 cm hr^{-1} and rGWL was set to zero.

Site/Soil	Treatment	Material	θr	θs	α	п	Ks	Ι
A1/ Vertisol	NT	1	0.1	0.55	0.0098	1.41	2.2	0.5
	SC	1	0.1	0.56	0.0103	1.41	3.0	0.5
		2	0.1	0.55	0.0104	1.41	2.9	0.5
	BP	1	0.1	0.56	0.005	1.41	2.7	0.5
		2	0.1	0.55	0.007	1.41	2.1	0.5
	NT	1	0.1	0.34	0.032	1.48	0.9	0.5
P4/	SC	1	0.1	0.35	0.026	1.48	1.2	0.5
Sandy clay loam	30	2	0.1	0.34	0.028	1.48	1.18	0.5
	BP	1	0.1	0.35	0.019	1.48	1.25	0.5
		2	0.1	0.34	0.021	1.48	1.21	0.5

Table E1: Soil hydraulic parameters for the analytical functions of Van Genuchten equation after calibrating the Hydrus 2D model according to the measured field conditions (Units for time (t) = hr and length (l) = cm)

* $\theta r = residual moisture content (l^3.l^{-3})$

 $\theta s = saturated moisture content (l^3.l^{-3})$

 α = inverse of air entry value (or bubbling pressure) (1.1⁻¹)

n= *pore size distribution index*

Ks=saturated hydraulic conductivity (l.t⁻¹)

I= *pore connectivity parameter*

The soil profile geometry was set according to the layout of bed furrow configurations measured in the field on both soils as listed in Table E2 and drawn in Figure E1. The simulated soil profile was initially set to 200 cm deep x 100 cm wide on Vertisol and 200 m deep x 66 cm wide on sandy clay loam soils. The flow domain was capable of simulating one half bed cross sections at both sites. A fine mesh density in boundary of the wetted furrow (to enhance precision) and coarse density away from the furrow (to reduce simulation time) was introduced (Figure E1).

Table E2: Field measured furrow dimensions (TW = top width, BW = bottom width and D = depth of furrow) during experiments conducted on Vertisol and sandy clay loam

Site/soil	Treatment	*TW (cm)	BW (cm)	D (cm)
	NT	62	18	12
A1/Vertisol	SC	52	21	10
	BP	60	23	11
D4/Sondy	NT	56	22	13
r 4/5alluy	SC	51	14	13.5
clay loam	BP	55	20	14



Figure E1: Flow domain x-sections for Hydrus 2D calibration on two soils. The notch at the top right side is a furrow with positions marked for measured furrow dimensions

The flow domain was divided into two layers (materials) in SC and BP treatments. The two layers represented the variable soil hydraulic parameters and soil structure caused by different renovation methods. The depth of material number 1 in the bed surface profile was 15 cm for SC treatment (Figure E2 (a)) and 30cm for blade plough renovation treatment (Figure E2 (b)). However, the NT treatment consisted of a single material (not shown).



Figure E2: Material distribution in (a) shallow cultivation, and (b) blade plough renovation treatments in both soils

The flow domain was set according to the antecedent soil moisture levels measured in the field on both soils (Table E3) immediately prior to the experiments. Soil moisture of 100-200 cm depth range (not measured) was assumed to be equal to antecedent soil moisture of 100 depths on both soils. A no flux boundary was set on the vertical side boundaries of the flow domain to stop lateral water movement into and out of the soil profile, which was consistent with the assumption of soil-water potential symmetry across the boundaries. A constant pressure boundary was applied on the furrow surface to equalise a 9 cm water head on the furrow (to match the furrow water head during trials on both soils as given in Chapter 6) and 6 cm (to match irrigation 1 to corn on Vertisol only). An atmospheric boundary was placed along the remaining surface boundary exposed to atmosphere to enable interactions between the soil and the atmosphere. A free drainage boundary was set at the base of the flow domain (Figure E3). The evapo-transpiration was estimated according to local climatic conditions by getting guidance from David (2005), PARC (1982) and Ullah et al. (2001) according to both site conditions. The crop was in the establishment stage at both sites during the experiments thus root water extraction was negligible while evaporation and transpiration were each assumed to be equal to 0.004 cm hr⁻¹ according to weather conditions during the experimental period.

Soil and	Vol	Volumetric soil moisture (%) at given depth (cm)									SMD
Treatment	10	20	30	40	50	60	70	80	90	100	(<i>mm</i>)
Vertisol											
NT	27	28	29	32	36	37	38	44	44	48	74
SC	27	31	32	33	35	39	41	45	46	48	63
BP	28	32	34	34	36	38	40	42	45	47	60
Sandy clay loam											
NT	14	14	14	14	13	13	13	13	15	16	59
SC	13	14	14	13	14	14	14	14	15	16	59
BP	14	14	14	14	14	13	14	14	16	17	58

 Table E3:
 Antecedent soil moisture and soil moisture deficit (SMD) in 100 cm root

 zone depth in Vertisol and 60 cm depth in sandy clay loam



Figure E3: Outline of flow domain with applied boundary conditions

Calibration of the model was undertaken by first comparing the predicted and half of the total volume applied to the furrow during field trials on both soils (Chapter 6). At the same time the predicted and measured lateral wetting front infiltration were visually compared on the bed surface. The soil hydraulic parameters were moderately adjusted to get a good fit between the measured and predicted soil moisture profiles. After getting a good fit, the measured and simulated soil moisture at positions (where probes were located) and three time intervals (1 hr, 4 hrs and 9 hrs) were compared by utilising the Hydrus utility of values along cross sections. To calibrate the model at 6 cm water head in the furrow (recorded during first irrigation to the corn crop) on Vertisol, the constant pressure boundary was modified to 6 cm water head and the results were compared with the measured soil moisture data during the first irrigation to corn crop, following the same procedure as given for the 9 cm furrow water head. Coefficient of determination R^2 in percent was determined by plotting the measured and predicted values. The measured and predicted cumulative infiltration at all measured positions were also compared, to identify the model prediction capability with depth and lateral distance.

E3 Results

For Vertisol with 6 cm furrow water head, the coefficient of determination (R^2) between measured and predicted soil moisture values across the bed was higher at shallow depths (0-50 cm) compared with deeper depths (60-100 cm). The model slightly over-predicted the soil moisture on NT treatment and slightly under-predicted on SC and BP treatments (Table E4). However, the overall results were satisfactory.

Treatment	Time	All depths at all distances	Distance from furrow (cm)			Depths at all distances (cm)		
			33	67	100	0-50	60-100	
NT	1 hr	94	89	94	99	85	81	
	4 hrs	92	82	97	98	97	70	
	9 hrs	94	79	72	96	97	57	
SC	1 hr	96	88	96	98	97	95	
	4 hrs	97	93	97	98	99	75	
	9 hrs	84	27	53	96	99	91	
BP	1 hr	96	95	95	97	97	90	
	4 hrs	97	93	88	98	98	83	
	9 hrs	85	83	55	76	89	72	

Table E4:Relationship between measured and predicted soil moisture at given times
and depths across the bed on a Vertisol at 6 cm furrow water head (R^2 values in %)

For Vertisol with 9 cm furrow water head, the coefficient of determination was initially higher when most of the bed profile was dry. The good match between initial conditions shows the model calibration was adequate. Overall results showed a good match between measured and predicted values. However, the coefficient of determination was identified to be very sensitive to individual values when the range between the minimum and maximum values was low, which was frequently noted on positions where the bed profile was saturated. For instance, the observed difference between measured and predicted values for NT at 9 hours was apparently not great but the poor match (Table E5) shown (i.e. $R^2 = 51$) was due to its sensitivity to the smaller range between the minimum and the maximum values.

Treatment	Time	All depths at all	Dist	ance : (from f cm)	Depths at all distances (cm)		
		distances	0	33	67	100	0-50	60-100
NT	1 hr	95	96	94	98	97	97	82
	4 hrs	89	95	95	87	99	91	81
	9 hrs	95	85	97	51	95	98	80
SC	1 hr	97	93	94	99	98	99	78
	4 hrs	99	94	97	98	98	82	92
	9 hrs	97	70	75	79	96	99	95
BP	1 hr	98	97	98	99	97	98	94
	4 hrs	99	98	97	97	98	99	93
	9 hrs	95	89	98	96	88	96	92

Table E5:Relationship between measured and predicted soil moisture at given timesand depths on Vertisol at 9 cm furrow water head (\mathbb{R}^2 values in %)

For sandy clay loam soil with a 9 cm furrow water head, the coefficient of determination was poor when the range between the minimum and the maximum values was smaller, which was generally found in dry profile e.g. bed middle and in deeper depths. The coefficient of determination was stronger in the wetted profile near the furrow (i.e. 22 cm position) compared with dry profile in the bed middle. However, the measured and predicted values were well matched with an increase in wetting time as the wetting front moved towards the bed middle. For instance, the value of R^2 at 44 cm distance from furrow centre was 52%, 55% and 45% at 4 hours, which was increased to 86%, 91% and 99% after 9 hours of wetting on NT, SC and BP treatments respectively (Table E6).

Treatment	Time	All depthsDistance fromat allfurrow (cm)			Depths at all distances (cm)		
		distances	22	44	66	0-50	60-100
NT	1 hr	76	89	52	38	89	67
	4 hrs	91	96	52	38	86	67
	9 hrs	96	99	86	38	98	61
SC	1 hr	90	97	55	57	97	66
	4 hrs	95	99	55	57	98	66
	9 hrs	97	99	91	57	98	64
BP	1 hr	89	97	22	53	94	66
	4 hrs	96	99	45	53	98	66
	9 hrs	98	99	99	53	99	68

Table E6: Relationship between measured and predicted soil moisture at given times and depths across bed on sandy clay loam soil at 9 cm furrow water head (\mathbf{R}^2 in %)

Comparison of measured and predicted cumulative infiltration values on Vertisol across the bed (Figure E4) indicated a marginal over-prediction near the furrow to: 33 cm distance on NT, 67 cm distance on SC and 33 cm distance on BP treatment. The over prediction was higher in the furrow, which may be due to the models' inability to account for furrow compaction. However, the model slightly under-predicted the cumulative infiltration in the middle of the bed.



Figure E4: Comparison of measured (solid line) vs. predicted (dotted line) cumulative infiltration at four given distances from furrow centre for three PRB renovation methods at 9 cm furrow water head on a Vertisol

For sandy clay loam, the model slightly over-predicted the cumulative infiltration at 22 cm and under-predicted at 44 cm for NT and SC treatments. However, the overall match between the measured and predicted values was satisfactory (Figure E5).



Figure E5: Comparison of measured (solid line) vs. predicted (dotted line) cumulative infiltration at three given distances from furrow centre for three PRB renovation methods at 9 cm furrow water head on a sandy clay loam

E4 Discussion

The Hydrus 2D model well predicted soil moisture movement for three PRB renovation methods on a Vertisol. Most errors in simulation were due to slight overprediction in the top surface layers and slight under-prediction in the bottom layers of the profile when water head in the furrows was 9 cm. Similarly, the slight overprediction on NT and under-prediction on the other two treatments, at 6 cm furrow water head, seems to be due to errors in fixing initial conditions. The slight overprediction near the furrow and under-prediction in the bed middle seems to be due to the models' limitation to precisely predict the changed soil hydraulic properties.

For sandy clay loam, the stronger correlation between the measured and predicted soil moisture near the furrow compared with the bed middle was due to large differences in antecedent soil moisture below the furrow. However, the correlation gradually improved where the wetting front reached, which is good for this study.

The strong correlation between the measured and simulated results, especially with no tendency of increasing deviation at longer times and shallow depths, as indicated on both soils, suggests that the prediction capability of the model will not reduce with time and will tend to further improve. This assumption is particularly beneficial for this study, as most of the simulation times are intended to be longer than the times tested in this calibration study. The increased preciseness of simulation at shallow depths is also beneficial because the Hydrus simulation is intended to evaluate the lateral infiltration in the surface profile of beds.

E5 Conclusions

The Hydrus 2D appropriately parameterised by field measured data can adequately simulates the sub-surface distribution of soil moisture in PRB farming systems and can account for the variable soil management/renovation methods. Therefore, the calibrated model can be used to identify the lateral and vertical infiltration, irrigation applications, deep drainage losses, and soil water storage capacity for the three PRB renovation methods under the given environmental conditions.